

BOARD OF TRUSTEES MEETING AGENDA

September 12, 2023 at 7:00 PM

Ridgeway Community Center - Room 101/102 208 Jarvis Street, Ridgeway, WI 53582

CALL TO ORDER AND ROLL CALL

CONFIRMATION OF OPEN MEETING

PLEDGE OF ALLEGIANCE

PUBLIC COMMENT

ANNOUNCEMENTS, DEPARTMENT REPORTS, AND CORRESPONDENCE

- 1. Announcements
- 2. Department Reports
- <u>3.</u> Correspondence

CONSENT AGENDA

- Minutes to be Adopted: Regular Board of Trustees Meeting 08/08/2023; Finance Committee Meeting Minutes 08/04/2023 and 08/11/2023
- 5. ACH Payments and General Fund Disbursements
- 6. Adoption of Agenda

ITEMS FOR CONSIDERATION AND ACTION

7. 705 Main Street - Phelan Properties - CDI Grant Application

Incentives and Available Funding (FY23) \$8,250,000 CDI Grant - \$7,500,000 The maximum award generally does not exceed \$250,000 unless the request for funds is for a project that, due to the size and scope of the investment, clearly justifies an award beyond normal parameters. No more than one grant per fiscal year shall be located within the boundary of a municipality.

- 8. 2023 Village Park Improvements
- 9. Village Green Improvements Julene Garner
- <u>10.</u> Bike Rack Ridgeway Farmer's Market Iowa County Tourism Grant
- 11. Faherty, INC Contract Renewal
- 12. Resolution in Support of Iowa County Application for Grant Funds for County Hwy HHH
- 13. Sludge Transport and Interfacility Cooperation
- 14. Stihl Chainsaw Purchase for \$639.99

- 15. Impact Wrench and Grease Gun Kit Public Works
- <u>16.</u> Shop Parking Lot Repair
- <u>17.</u> Wastewater Training Braden Losby
- <u>18.</u> Inflow and Infiltration Update and Study Quote from Delta3
- 19. Utility Clerk/Deputy Clerk
- 20. WEDC Connect Communities Membership
- 21. Ehler's Financial Consulting Hourly Rates and Agreements

ADJOURNMENT



Village of Ridgeway Income Survey

2 messages

Jamie Marcue <MarcueJ@delta3eng.biz> To: Hailey Roessler <clerk@ridgewaywi.gov> Cc: Bart Nies <NiesB@delta3eng.biz> Thu, Sep 7, 2023 at 3:52 PM

Item 1.

Good Afternoon, Hailey,

Just wanted to inform you that I have sent out 2nd mailings to 26 Ridgeway residents in the Target Area this afternoon.

Next week and the week after I'll keep taking a look at how things are going, but so far I am only at 19.23% LMI for the Target Area and only at 17.53% LMI for the Community-wide. Of the 283 income survey requests we've sent out, we've only had 120 completed income surveys returned.

Just wanted to bring you up-to-date on where we are with tallying the results. We will probably have to continue with the 2^{nd} mailing for the remaining 137 residents (rest of the community who has still not yet responded) and I will continue to work on it next week (with help from other members of our office). Thanks and have a great day!

If you have any questions or need further information, please feel free to contact me.

Jamie Marcue



Grant Administrator

875 South Chestnut Street	898 Jackson Street
Platteville, WI 53818	Dubuque, IA 52001
608-348-5355 x 1000	563-542-9005 x 1000

I

marcuej@delta3eng.biz



CONFIDENTIALITY NOTICE: This message, including any attachments, is for the sole use of the intended recipient(s) and magnetic contain confidential and privileged information. Any unauthorized review, use, disclosure, or distribution is prohibited. If you are the intended recipient, please contact the sender by reply email and destroy all copies of the original message. Thank you and have a great day.

Hailey Roessler <clerk@ridgewaywi.gov> To: Jamie Marcue <MarcueJ@delta3eng.biz> Cc: Bart Nies <NiesB@delta3eng.biz> Thu, Sep 7, 2023 at 4:39 PM

Thank you for the update. [Quoted text hidden]

Hailey E. Roessler

www.ridgewaywi.gov

Office Phone: 608-924-5881

Cell Phone: 608-574-1797





We Need Your Help!

The Village of Ridgeway is conducting a survey to determine our eligibility in applying for state and federal grants to improve the lives of residents in our community.

This funding could help improve the water/sewer system, repair streets, and provide recreational services. Street Improvement Projects completed in 2018 and 2021 were partially funded by state and federal grants. This grant funding made the projects possible!

You will receive a letter asking you to complete the survey in the next few days. If we do not receive enough responses, someone may come to your home to follow up. The Village of Ridgeway needs accurate and up-todate information about the income of families in order to apply for these grant funds. Your responses are crucial in determining eligibility for funding. We ask that you please complete the survey when you receive it. Without your cooperation, this effort cannot succeed. **All information will be held strictly confidential.**

Frequently Asked Questions and Answers

Q: How long will this take?

A: The survey will take only 5 minutes of your time.

Q: What is the purpose of the survey?

A: The Village of Ridgeway needs accurate and up-todate information about the income of families in the area to determine its eligibility to receive state funding for important city projects. One example is the Community Development Block Grant (CDBG), which can be used for many kinds of activities that would improve the lives of city residents, including infrastructure improvements, economic development, and affordable housing.

Q: Who should complete the survey?

A: One survey should be completed for each family in a housing unit ("family" means anyone related by blood, marriage, or adoption). Roommates should fill out separate surveys. Additional families living in the same housing unit should fill out separate surveys. If you have an accessory rental unit, those residents should fill out a separate survey. We provide three IDs for each address, but if you need more, just call our toll-free number and we will share additional IDs.

Q: How will the data be used?

A: The data will only be used to submit an application for grant funds from state or federal sources. The data from all families will be combined to produce a median family income for the city (how many families have incomes above and below a certain level).

Q: How can I be sure that my information will be kept confidential?

A: The data will only be shared with the researchers and select village officials. Your name and the information you provide will never be directly connected. We will keep all the data secure, and it will only be used in the application process. All data will be destroyed following the grant application period for which the data is being collected.

If you have any questions, please call (608) 924-5881



iNuestra Comunidad Necesita su ayuda!

Nuestra comunidad está llevando a cabo una encuesta que nos ayudará a solicitar subvenciones estatales y federales para mejorar la vida de los residentes en nuestra comunidad.

Esta financiación nos ayudaría, por ejemplo, a mejorar el sistema de alcantarillado, reparar las calles y proporcionar servicios recreativos. También se puede emplear para conceder incentivos a pequeñas empresas y para ofrecer viviendas asequibles.

Una empresa de investigación, Delta3, nos está ayudando a realizar esta encuesta. Es posible que usted reciba una carta pidiéndole que complete la encuesta en los próximos días y que alguien acuda a su casa para darle seguimiento. Necesitamos su cooperación.

Nuestra comunidad necesita información precisa y actualizada sobre los ingresos de las familias. Sus respuestas son cruciales para determinar si cumplimos los requisitos para recibir financiación. Le pedimos por favor que complete la encuesta cuando la reciba. Sin su cooperación este esfuerzo no puede tener éxito. Toda la información será estrictamente confidencial.

Preguntas y respuestas frecuentes

P: ¿Cuánto tiempo me tomará esto?

R: La encuesta solo le tomará 5 minutos de su tiempo.

P: ¿Cuál es el propósito de la encuesta?

R: La ciudad necesita información precisa y actualizada sobre los ingresos de las familias de la zona para determinar si la ciudad cumple los requisitos para recibir fondos estatales para proyectos locales importantes. Un ejemplo es la Subvención en Bloque para el Desarrollo Comunitario (CDBG), que puede utilizarse para muchos tipos diferentes de actividades que mejorarían la vida de los residentes de la comunidad, incluyendo mejoras en las infraestructuras, desarrollo económico y vivienda asequible.

P: ¿Quién debe completar la encuesta?

R: Se debe completar una encuesta por cada familia que viva en una unidad de vivienda. "Familia" significa cualquier persona relacionada por consanguinidad, matrimonio o adopción. Los compañeros de cuarto deben completar encuestas separadas. Las familias adicionales que vivan en la misma unidad de vivienda deben completar una encuesta cada una. Si usted posee una unidad de alquiler, los residentes en dicha vivienda deben completar también una encuesta distinta a la suya. Si usted recibe una carta de invitación a completar la encuesta, le proporcionaremos tres IDs para cada dirección pero, si necesita más, simplemente llame a nuestro número gratuito y le facilitaremos IDs adicionales.

P: ¿Cómo se utilizarán los datos?

R: Los datos solo se utilizarán para presentar una solicitud de fondos ante fuentes estatales o federales. Los datos de todas las familias se combinarán para calcular el promedio de ingresos familiares de la ciudad (cuántas familias tienen ingresos por encima y por debajo de un determinado nivel).

P: ¿Cómo puedo estar seguro de que mi información será confidencial?

R: Los datos solo se compartirán con los investigadores y con funcionarios selectos de la ciudad. Su nombre y la información que proporcione nunca estarán directamente relacionados. Mantendremos todos los datos seguros y solo serán utilizados durante el proceso de solicitud. Todos los datos serán destruidos después del período de solicitud de subvenciones para el que se están recopilando.

Si tiene alguna pregunta, puede llamar al (608) 924-5881 Item 1.

07/24/2023 - 07/30/2023 [7 days]

PD011 [GORHAM, MICHAEL]

Employee ID	PD011	DEPT(G2)	PD	Pay Policy	601
Pay Type	1	Last Name	GORHAM	First Name	MICHAEL

Time Card

Time C	aru						
	Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hrs
	07/24/2023 Mon	205 [POP]	07:00PM*	11:00PM*	4.00	4.0000000	
	07/25/2023 Tue	205 [POP]	01:30PM*	12:30AM*	11.00	11.0000000	
	07/26/2023 Wed	205 [POP]	01:30PM*	07:00PM*		5.5000000	
		205 [POP]	08:00PM*	12:30AM*	10.00	4.5000000	
	07/27/2023 Thu	205 [POP]	05:30PM*	01:00AM*	7.50	7.5000000	
	07/28/2023 Fri	205 [POP]	11:30AM*	01:15PM*	1.75	1.7500000	
	07/29/2023 Sat	205 [POP]	08:00PM*	01:00AM*	5.00	5.000000	

Summary - PD011 [GORHAM, MICHAEL]

					11 12 14 S R		Accrual		Salar and the
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]									97.00
3 [SICK]									139.00
6 [FH]									8.00
205 [POP]	1[UNUSED]	39.25		39.25					
TOTALS		39.25		39.25					244.00
	0								

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT X Employee Signature

Supervisor Signature

X

VILLAGE OF RIDGEWAY Report Date: 08/07/2023 Report Time: 8:19:59 AM

07/31/2023 - 08/06/2023 [7 days]

Employee ID	PD011	DEPT(G2)	PD	Pay	Policy 601	
Рау Туре		Last Name	GORHAM	First		
ard						
Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hr
07/31/2023 Mon	205 [POP]	08:00PM*	12:30AM*	4.50	4.5000000	
08/01/2023 Tue	205 [POP]	07:00PM*	12:30AM*	5.50	5.5000000	
08/02/2023 Wed	205 [POP]	09:00AM*	03:00PM*	6.00	6.0000000	
08/03/2023 Thu	205 [POP]	12:00PM*	01:00PM*		1.0000000	
	205 [POP]	07:30PM*	01:30AM*	7.00	6.0000000	
08/04/2023 Fri	205 [POP]	2:00PM*	05:00PM*	3.00	3.0000000	
08/05/2023 Sat	205 [POP]	08:00PM*	02:30AM*	6.50	6.5000000	
08/06/2023 Sun	205 [POP]	01:00PM*	05:00PM*		4.0000000	
00/00/2020 0011	200 [. 0.]					

Summary - PD011 [GORHAM, MICHAEL]

			and the second second	State of the second	Accrual					
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available	
2 [VACA]									97.00	
					139.00	4.00			143.00	
3 [SICK]					100.00				8.00	
6 [FH]									0.00	
205 [POP]	1[UNUSED]	36.50		36.50						
208 [PADJ]	1[UNUSED]	-3.75		-3.75						
TOTALS		32.75		32.75	139.00	4.00			248.00	

BOVE INFORMATION TO BE CORRECT I CERTIFY Х Employee Signature

208 [PADJ]

Supervisor Signature

X_

0.25

-3.7500000

07/24/2023 - 07/30/2023 [7 days]

Employee ID	AD002	DEPT(G2) AD		Pay P	olicy 500		
	Pay Type 3		Last Name ROESSLER		ame HAILEY	AILEY	
ard							
Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hr	
07/24/2023 Mon		08:02AM	12:05PM		4.000000		
0172 112020 1111	501 [TRW]	12:44PM	04:25PM	7.75	3.7500000		
07/25/2023 Tue		08:07AM	12:13PM		4.2500000		
07723/2020 100	501 [TRW]	01:09PM	04:41PM	7.75	3.5000000		
07/26/2023 Wed		08:09AM	12:12PM		4.0000000		
07720/2023 Wet	501 [TRW]	01:04PM	04:52PM	7.75	3.7500000		
07/07/0002 Thu		Vacatio	NA NA	8.00	8.0000000		
07/27/2023 Thu	504 [TKV]	Juluni					

SICK

09:50AM

503 [TRS]

501 [TRW]

5.00

0.50

5.0000000

0.5000000

Summary - AD002 [ROESSLER, HAILEY]

07/28/2023 Fri

07/30/2023 Sun

			the second se		Contract of the second	DAL STREET		Accrual		
Paycode		N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]	Construction and the					57.25		8.00		49.25
3 [SICK]						5.00		5.00		
6 [FH]										8.00
7 [BREV]										
501 [TRW]	1[UNUSED]		23.75		23.75					
503 [TRS]	1[UNUSED]		5.00		5.00					
504 [TRV]	1[UNUSED]		8.00		8.00					
TOTALS			36.75		36.75	62.25		13.00		57.2

10:09AM

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

paper х Supervisor Signature **Employee Signature** Gen Corresp. Phone (Email; HR; AP/AR; SAL; Legal Notices WEDC/CDI Grant; Util Lakefees/ACH General; Water Moter Reg (Trans Upgrade; CDBG Income Survey Dog Park Lighting 12023 Park Improvements; Connectr/ Facilitics Maint; Payroll Page 1

07/31/2023 - 08/06/2023 [7 days]

Employee ID	AD002	DEPT(G2) AD			Pay Policy	500		
Pay Type		Last Nan	ROESSLER		First Name HAILEY			
ard								
Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hrs		
07/31/2023 Mon	501 [TRW]	08:01AM	12:30PM		4.5000000			
	501 [TRW]	01:21PM	04:30PM	7.75	3.2500000			
08/01/2023 Tue	501 [TRW]	08:11AM	12:38PM		4.5000000			
	501 [TRW]	01:11PM	04:36PM	7.75	3.2500000			
08/02/2023 Wed	501 [TRW]	08:06AM	12:22PM		4.2500000			
	501 [TRW]	12:41PM	01:34PM		0.7500000	=		
	501 [TRW]	01:56PM	05:00PM*	8.00	3.0000000			
08/03/2023 Thu	501 [TRW]	08:02AM*	12:18PM		4.2500000			
	501 [TRW]	01:07PM	04:39PM	8.00	3.7500000			
08/04/2023 Fri	501 [TRW]	08:01AM	12:19PM		4.2500000			
	501 [TRW]	12:36PM	05:00PM*	8.75	4.2500000	0.250000		

	ADUUZ [ROESSLER, HAILE			Composition (Section	Accrual					
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available	
2 [VACA]									49.25	
						4.00			4.00	
3 [SICK]									8.00	
6 [FH]									0.00	
7 [BREV]										
501 [TRW]	1[UNUSED]	40.00	0.25	40.25						
TOTALS		40.00	0.25	40.25		4.00			61.2	

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

Employee Signature

Supervisor Signature

X

GenCorresp [Email Phone | AP | AR | HR Fin Comm Mtg 814/23 Prep / Post/Atlend/Minutes 2023 Budget Prep 2023 Budget Maint. Board Mtg Prep (Agenda (Does | Memo | Resolutions

07/24/2023 - 07/30/2023 [7 days]

07/24/2023 Mon STT [CW] 07.00AM 02:59PM 07/25/2023 Tue 511 [CW] 08:00AM* 02:59PM 07/26/2023 Wed 511 [CW] 07:59AM 02:28PM 07/27/2023 Thu 511 [CW] 07:57AM 03:02PM	First Name MAGGIE Reg Hrs 01 7.50 7.5000000 7.00 7.0000000 6.50 6.5000000 7.00 7.0000000 4.00 4.0000000	T Hrs
Date Paycode IN OUT Daily Total 07/24/2023 Mon 511 [CW] 07:58AM 03:33PM 03:33PM 07/25/2023 Tue 511 [CW] 08:00AM* 02:59PM 02:59PM 07/26/2023 Wed 511 [CW] 07:57AM 02:28PM 02:28PM 07/27/2023 Thu 511 [CW] 07:57AM 03:02PM 07:07/28/2023 Fri 07/28/2023 Fri 511 [CW] 08:01AM 12:01PM 04	7.50 7.5000000 7.00 7.0000000 6.50 6.5000000 7.00 7.0000000	T Hrs
Date Flaycod OT/24/2023 Mon S11 [CW] O7:58AM O3:33PM 07/25/2023 Tue S11 [CW] 08:00AM* 02:59PM O7:59AM 02:28PM 07/26/2023 Wed S11 [CW] 07:59AM 02:28PM O7:07:27/2023 Thu S11 [CW] 07:57AM 03:02PM 07/28/2023 Fri S11 [CW] 07:57AM 03:02PM O7:20PM O7:20PM 07/28/2023 Fri S11 [CW] 08:01AM 12:01PM O7:20PM O7:20PM	7.50 7.5000000 7.00 7.0000000 6.50 6.5000000 7.00 7.0000000	T Hrs
07/24/2023 Mon 511 [CW] 07:58AM 03:33PM 07/25/2023 Tue 511 [CW] 08:00AM* 02:59PM 07/26/2023 Wed 511 [CW] 07:59AM 02:28PM 07/27/2023 Thu 511 [CW] 07:57AM 03:02PM 07/28/2023 Fri 511 [CW] 07:57AM 03:02PM 07/28/2023 Fri 511 [CW] 08:01AM 12:01PM	7.00 7.0000000 6.50 6.5000000 7.00 7.0000000	
07/25/2023 Tue STT [CW] 00.000M 02.001 M 07/26/2023 Wed 511 [CW] 07:59AM 02:28PM 07/27/2023 Thu 511 [CW] 07:57AM 03:02PM 07/28/2023 Fri 511 [CW] 08:01AM 12:01PM	6.50 6.5000000 7.00 7.000000	
07/26/2023 Wed S11 [CW] 07.334 W 02.201 W 07/27/2023 Thu 511 [CW] 07:57AM 03:02PM 07/28/2023 Fri 511 [CW] 08:01AM 12:01PM	7.00 7.000000	
07/27/2023 Tri 511 [CW] 07.07.01 12:01PM		
ummary - AD003 [JOHNSON, MAGGIE]		
Pavcode N/A Reg Hrs OT1 - OT-2 Total Hrs Prior Bal Adjust	Accrual	
	Used Earned Avail	43.50
		76.75
[SICK] [FH]		8.00
11 [CW] 1[UNUSED] 32.00 32.00		
TOTALS 32.00 32.00	12	28.25
Employee Signature Supervisor S	ignature	6
General: Utility payments, library/open gym	communication	ns,
Employee Signature General: Utility payments, library/open gym dog license		
Monday: Meter Change acts, Public Works & Sa	fety minutes	
Tues day: Cooling station post, 10 day disconnects, 1 prep For Procisin' Ruptiles & avoesance amph	bians show	jeni
Nednesdy: Upduted trimble, Rockin Reptills Shaw		
Thursday: Collips processed utility billing		
Friday: Utility billing, posted park & rec agenda	L	
Fritang. Uning mining, point a ree augeries		

Item 2.

11

VILLAGE OF RIDGEWAY Report Date: 08/07/2023 Report Time: 8:19:59 AM

AD003 [JOHNSON, MAGGIE]

Employee Timecard - Hourly Distribution Report

07/31/2023 - 08/06/2023 [7 days]

1	Employee ID AD003			DEPT(G2)	DC				Policy 5			
	Pay Type 3			Last Name	JOHNSON		First Name MAGGIE					
ime Card												
	Date	Paycode		IN	OUT		Daily Total		R	eg Hrs	OT Hr	
07/	31/2023 Mon	511 [CW]	07	:56AM	03:31PM	Λ		7.50	7.5	5000000		
	/01/2023 Tue	511 [CW]	07	7:57AM	03:00PM	Λ			7.0	000000		
007	0112020	511 [CW]	06	6:00PM	08:17PN	Λ		9.25	2.2	2500000		
08/	02/2023 Wed	511 [CW]	07	7:56AM	11:09AN	л			3.2	2500000		
00/	02/2020 1104	511 [CW]	12	2:08PM	02:29PN	л		5.50	2.3	2500000		
08	/03/2023 Thu	511 [CW]	0	7:55AM	11:00AN	N			3.0	0000000		
	105/2025 1110	511 [CW]	1	1:57AM	03:01PM	N		6.00	3.	0000000		
30	3/04/2023 Fri	511 [CW]	0	7:56AM	11:59AM	M		4.00	4.	0000000		
						- (1.5. S.		- Carolina				
summary - A	AD003 [JOHNSON,	MAGGILJ						Accr		Famad	Available	
Paycode	N//	Ą	Reg Hrs	OT1 - OT-2	2 Total Hrs	Prior B	al Adjust	Use	be	Earned	43.5	
[VACA]											80.7	
[SICK]						76	4.0				8.0	
6 [FH]											0.0	
511 [CW]	1[UNUSED]		32.25	5	32.25							
TOTALS			32.25	5	32.25	76	.75 4.0	00			132.2	
Mate	r=5	Sewer	= 4	Ĺ	Sepi	cle	ハ(< =	- 2	13	12:	5	
I CERTIFY T	HE ABOVE INFOR	MATION TO BE	CORRECT	r			1	-	\cap			
4.0	Λ						11	4	VI	1 00-		
x TW	NAL					X	TP	ST	7	for		
Employee	Signature						Supervisor	Signatu	re			
Geni	ral: ()ti	lity po	ymen	175,1	ibran	1/OP	en gyn	n cc	mm	nunica	tions.	
00.00		0		· U · O -1	Cost	Fal	Do 11	t loo	~ .	0.0.0.00		
Mando	uj : Poste Pan	LE EVEC.	me	ting	Prep	ea	pierre	e rec	_· C	igen	ju,	
Tuesc	lay: Park libra	A rec. 1	meeti Ina da	ng p	rep, Ht	allor	Veen,	libre	ing	p nun	Ibers	
	libra	AND JUPP	in the	11011	un c		ind					
1	solay: par											
n.	V	10	10	1. 11	11.	1	illoop (01 . 0	1 -	ind	A 1 0-	

Thursday: Battle of the bats, Halloween, Village carendar updates Friday: Chaned library, Meter Change acts, Halloween

07/24/2023 - 07/30/2023 [7 days]

	Employee ID	PW003			DEPT(G2)	M			Pay Policy	401	
	Рау Туре	1		a and they	Last Name	OHNSON			First Name	HARRY	
ime Card	NS LANGE	anterna Seri				14. 12 mil					
	Date		Paycode		IN	OUT		Daily Total		Reg Hrs	OT Hrs
07/	24/2023 Mon		401 [STW]	(08:14AM	12:47PI	M	4	.50	4.5000000	
	25/2023 Tue		101 [PAW]		08:06AM	12:23PI	M	4	.50	4.5000000	
	26/2023 Wed		401 [STW]		08:10AM	10:36A	M	2	2.25	2.2500000	
	07/27/2023 Thu		401 [STW]		08:08AM	12:34P	м	4	4.25	4.2500000	
	07/28/2023 Fri		611 [FMW]		08:09AM		М		4.50	4.5000000	
Summary - F	PW003 [JO	HNSON, H	ARRY]						Accrual	-	1
Paycode	and the second	N/A		Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
101 [PAW]	1[UNUSED			4.50		4.50					
401 [STW]	1[UNUSED	-		11.00		11.00					
611 [FMW]				4.50		4.50					
TOTALS				20.00		20.00					

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

Employee Signature

Supervisor Signature

M ð 7, 6 00 Page 4

Item 2.

13

VILLAGE OF RIDGEWAY Report Date: 08/07/2023 Report Time: 8:19:59 AM

Employee Timecard - Hourly Distribution Report

ltem 2.

07/31/2023 - 08/06/2023 [7 days]

	Employee ID	PW003			DEPT(G2)	FM			Pay Policy	401	
	Pay Type	1			Last Name	JOHNSON			First Name	HARRY	
ime Card						1					
	Date		Paycode		IN	OUT		Daily Total		Reg Hrs	OT Hrs
07/	31/2023 Mon		401 [STW]		08:10AM	01:18P	M	5.	.00 5	5.0000000	
	01/2023 Tue		401 [STW]		08:19AM	01:10P	м	5	.00	5.0000000	
			401 [STW]	Proville	08:14AM	01:57P	м	5	.75	5.7500000	
	08/02/2023 Wed		401 [STW]		08:20AM	01:28P	м	5	.25	5.2500000	
	08/03/2023 Thu 08/04/2023 Fri		101 [PAW]		08:18AM		M	4	.25	4.2500000	
Summary - F	0012 200/00	HNSON H	ARRYI						1.279		
summary - r		1143014, 11	alerer 1		1				Accrual		
Paycode		N/A		Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
01 [PAW]	1[UNUSED		FU	4.25		4.25					
101 [STW]	1[UNUSED	1-	2500	21.00		21.00					
TOTALS				25.25		25.25					

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

Employee Signature

Supervisor Signature

lind ary 2 Bal 0 imetalent Pel U ng de Lields Mawed Part Hrom for To. 4 , Mario (den Lane 60 dig -Acer re en

Page 5

07/24/2023 - 07/30/2023 [7 days]

E	mployee ID	PW005			DEPT(G2)	PW			Pay Policy	300	
	Pay Type	3			Last Name	PETERSON			First Name	DALE	
ime Card						-					
	Date		Paycode		IN	OUT		Daily Total	1	Reg Hrs	OT Hrs
07/2	4/2023 Mon		301 [SEW]	(06:56AM	11:53AI	М		5	0000000	
Cost Control			601 [WAW]		12:20PM	03:28PI	м	8	.00 3	0000000	
07/2	5/2023 Tue		601 [WAW]		06:56AM	12:25P	M		5	5000000	
			301 [SEW]		12:54PM	03:28P	M	8	.00 2	.5000000	
07/2	6/2023 Wed		301 [SEW]		06:57AM	02:36P	м	7	.50 7	.5000000	
	7/2023 Thu		304 [SEV]					8	.00 8	.0000000	
0.0.65	28/2023 Fri		604 [WAV]					8	.00 8	.0000000	
Summary - P				200000000				estato linearo	Carl and the	an a	
Summary - P		rendon, e			<u> </u>	1		1.000	Accrual		Section and
Paycode		N/A		Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]							68.00		16.00		52.0
B [SICK]											4.0
6 [FH]											8.0
301 [SEW]	1[UNUSED	1		15.00		15.00					
304 [SEV]	1[UNUSED	1		8.00		8.00					
601 [WAW]	1[UNUSED			8.50		8.50					
604 [WAV]	1[UNUSED			8.00		8.00					
	-	-								States in the second	

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

X_

Employee Signature

Supervisor Signature

Dale Peterson Weekly Work Log 7-24-2023

Monday 7-24-2023:

Daily Labs at WWTP 2 Hours

3 Meter install appointments 2.5 Hours, Cleaned and flushed Sewer main in alley between Weaver and Jarvis and flushed out 3 water main valve boxes 3.5 Hours

Tuesday 7-25-2023:

Daily Labs at WWTP 2 Hours

4 Meter install appointments 3 Hours, Troubleshot Blower VFD issue 2 Hours, Cleaned clarifiers 1 Hour

Wednesday 7-26-2023:

Daily Labs at WWTP 2 Hours

Cleaned clarifiers and digesters at WWTP 4 Hours, Continued removing scrap from Village properties 1.5 Hours.

Thursday 7-27-2023:

Vacation 8 Hours

Friday 7-28-2023:

Vacation 8 Hours

07/31/2023 - 08/06/2023 [7 days]

Employe	ID	PW005		DEPT(G2)	PW				Policy 300	
Pay T	/pe	3		Last Name	PETERSON			First Name	DALE	
ne Card					1			1		
Date		Payo	code	IN	OUT		Daily Total		Reg Hrs	OT Hrs
07/31/2023	Mon	604 [\						4	.0000000	
		304 [SEV]				8.	00 4	.0000000	
08/01/2023	Tue	301 [5		06:58AM*	11:55A	M*		5	0000000	
00/0 //2020		601 [V		12:26PM	03:28P	M	8.	00 3	3.0000000	
08/02/2023	Ned	BOUTEN -	SEW]	06:55AM	12:25F	M		5	5.5000000	
00/02/2020		601 [\		12:58PM	03:26F	M	8.	00 2	2.5000000	
08/03/2023	Thu		WAW]	06:54AM	11:22A	M		4	1.2500000	
00/00/2020	ma		WAW]	11:54AM	03:27P	M*	7.	75 3	3.5000000	
08/04/2023	Fri		WAW]	06:51AM	04:13F	PM	9.	50 9	9.5000000	
Divisor	DET		Contraction of the second		NOTE OF LAND			SAL STINES	1000	
ummary - PW005	PEI	ERSON, DALEJ					-	Accrual		
Paycode		N/A	Reg H	Irs OT1 - OT-	2 Total Hrs	Prior Bal	Adjust	Used	Earned	Available
	-					52.00		8.00		44.00

10.50

4.00

22.75

4.00

41.25

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

1[UNUSED]

1[UNUSED]

1[UNUSED]

1[UNUSED]

Employee Signature

2 [VACA]

3 [SICK]

6 [FH]

301 [SEW]

304 [SEV]

601 [WAW]

604 [WAV]

TOTALS

X_

8.00

Supervisor Signature

4.00

4.00

4.00

56.00

10.50

4.00

22.75

4.00

41.25

8.00

8.00

60.00

Dale Peterson Weekly Work Log 7-31-2023

Monday 7-31-2023:

Vacation 8 Hours

Tuesday 8-1-2023:

Daily Labs at WWTP 2 Hours

Cleaned clarifiers 4 Hours, Traveled to Ritchies for mower pricing 1 Hour, started going over budget details 1 Hour

Wednesday 8-2-2023:

Daily Labs at WWTP 2 Hours

Cleaned skimmers and clarifiers 2 Hours, Traveled to Meullers for mower pricing 1 Hour, continued budget details 3 Hours..

Thursday: 8-3-2023:

Daily Labs at WWTP 2 Hours

6 Water meter install appointments 6 Hours

Friday 8-4-2023:

Daily Labs at WWTP 2 Hours

Cleaned and scrubbed both clarifiers and skimmed the resulting scum 3 Hours, 1 meter appointment .5 Hour, Budget Meeting 3 Hours.

07/24/2023 - 07/30/2023 [7 days]

Employee ID	SP003	DEP	T(G2) ST		Pay Policy	400		
Pay Type		Last	Name LOSBY		First Name BRADEN			
ard								
Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hrs		
07/24/2023 Mon	401 [STW]	06:55AM	12:01PM		5.000000			
	401 [STW]	12:36PM	03:33PM	8.00	3.0000000			
07/25/2023 Tue	601 [WAW]	06:56AM	11:45AM		4.7500000			
0112012020 100	601 [WAW]	12:19PM	03:31PM	8.00	3.2500000			
07/26/2023 Wed	401 [STW]	06:55AM	12:50PM		5.7500000			
01/20/2023 Wed	401 [STW]	01:20PM	03:38PM	8.25	2.5000000			
07/27/2023 Thu	401 [STW]	07:00AM	03:23PM	8.50	8.5000000			
0/12/12023 1110						1 75000		

04:00PM

12:24PM*

11:12AM*

Total Hrs

2.00

33.75

10.00

45.75

OT1 - OT-2

2.00

1.75

2.00

5.75

06:56AM

10:24AM

09:12AM

32.00

8.00

40.00

Reg Hrs

401 [STW]

601 [WAW]

301 [SEW]

TOTALS

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

N/A

X

07/28/2023 Fri

07/29/2023 Sat

07/30/2023 Sun

Paycode

2 [VACA]

3 [SICK]

301 [SEW]

401 [STW]

601 [WAW]

6 [FH]

Summary - SP003 [LOSBY, BRADEN]

1[UNUSED]

1[UNUSED]

1[UNUSED]

x Jacob

Employee Signature

Supervisor Signature

9.00

2.00

2.00

Adjust

Prior Bal

7.2500000

Accrual

Used

Item 2.

1.7500000

2.0000000

2.0000000

Available

34.50

20.00

8.00

62.50

Earned

Weekly Work Log July 24 – July 30 2023 Braden Losby

Monday July 24 2023:

Daily rounds. - 2 hours

Meter replacements. - 2 hours

Weed control at fire department. - 2 hours

Flushed out manholes on the ally between Weaver and Jarvis. - 2 hours

Tuesday July 25 2023

Daily rounds. - 2 hours

Meter replacements. - 2 hours

Mowing. - 2 hours

Working on quotes for budget and getting parts for street sweeper. - 2 hours

Wednesday July 26 2023:

Daily rounds. - 2 hours

Fire call. - 2 hours

Working with marshal on the speed sign. - 2 hours

Working on scrapping metal. - 2 hours

Thursday July 27 2023:

Daily rounds. - 2 hours

Working on cdl permit. - 1 hours

Meter reads. - 2 hours

Fire call. - 3 hours

Friday July 28 2023:

Daily rounds. - 2 hours

Weed control on main street. - 2 hours

Washing the f-550. - 1 hour

Power washing and garbage at park shelter. - 2.5 hours

Moving ice machine. - 1 hour

Saturday July 29 2023:

Weekend rounds. - 2 hours

Sunday July 30 2023:

Weekend rounds. - 2 hours

07/31/2023 - 08/06/2023 [7 days]

Name BRADEN s OT Hrs 00 00
00
00
00
00
00
00
00
000
000
000
1.750000
2.000000

	SP003 [LOSBY, BRADEN]						Accrual		
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]									34.50
					20.00	4.00			24.00
3 [SICK]									8.00
6 [FH]									
101 [PAW]	1[UNUSED]	1.50		1.50					
301 [SEW]	1[UNUSED]		2.00	2.00					
401 [STW]	1[UNUSED]	38.50	1.75	40.25					
601 [WAW]	1[UNUSED]		2.00	2.00					
TOTALS		40.00	5.75	45.75	20.00	4.00			66.5

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

1h X

Employee Signature

X

Supervisor Signature

Weekly Work Log July 31 – August 6 2023 Braden Losby

Monday July 31 2023: Daily rounds. - 2 hours Mowing around fire hydrant by the colleens devision with the brush hog. - 2 hours Weed eating and spraying. - 4 hours Tuesday August 1 2023: Daily rounds. - 2 hours Farm and fleet run and doing my cdl written test at the dmv. - 3 hours Fixing both welcome home signs. - 1 hour Going to richeis implement to get a quote for a zero turn for budget. 2 hours Park and rec meeting. - 1.5 hours Wednesday August 2 2023: Daily rounds. - 2 hours Going to mullers implement to get quotes for zero turn mowers for budget. - 2 hours Mowing at sewer plant. - 4 hours Thursday August 3 2023: Daily rounds. - 2 hours Meter replacements. - 5 hours Friday August 4 2023: Daily rounds. - 2 hours Watering trees and plants. - 2 hours Meter replacement. - 0.5 hours Changing garbage. - 1.5 hours Finance committee meeting. - 3.5 hours Saturday August 5 2023: Weekend rounds. - 2 hours Sunday August 6 2023: Weekend rounds. - 2 hours

PD011 [GORHAM, MICHAEL]

Employee Timecard - Hourly Distribution Report

08/21/2023 - 08/27/2023 [7 days]

Employee ID PD011 DEPT(G2) PD Pay Policy 601 Pay Type 1 Last Name GORHAM First Name MICHAEL **Time Card** Date Paycode IN OUT Daily Total Reg Hrs OT Hrs 08/21/2023 Mon 205 [POP] 07:00PM* 02:30AM* 7.50 7.5000000 08/23/2023 Wed 205 [POP] 02:00PM* 01:00AM* 11.00 11.0000000 08/24/2023 Thu 205 [POP] 11:00AM* 11:15PM* 12.25 12.2500000 08/25/2023 Fri 206 [POG] 06:00PM* 01:30AM* 7.50 7.5000000 08/26/2023 Sat 205 [POP] 02:30PM* 05:00PM* 2.5000000 205 [POP] 11:00PM* 12:30AM* 4.00 1.5000000 Summary - PD011 [GORHAM, MICHAEL] Accrual Paycode N/A Reg Hrs OT1 - OT-2 Total Hrs Prior Bal Adjust Used Earned Available 2 [VACA] 97.00 3 [SICK] 143.00 6 [FH] 205 [POP] 1[UNUSED] 34.75 34.75 206 [POG] 1[UNUSED] 7.50 7.50 TOTALS 34.75 7.50 42.25 240.00

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

х

Employee Signature

Supervisor Signature

X

7.5 Grant OT Hours

TOTALS

Employee Timecard - Hourly Distribution Report

08/28/2023 - 09/03/2023 [7 days]

Emp	oloyee ID	PD011			DEPT(G2)	PD			Pay Policy	601	
F	Pay Type	1			Last Name	GORHAM			First Name	First Name MICHAEL	
Time Card											
Da	ite		Paycode		IN	OUT	[Daily Total		Reg Hrs	OT Hrs
08/28/20	23 Mon		205 [POP]	0	8:00PM*	02:30AM	*	6	.50 6	5.5000000	
08/29/20	23 Tue		205 [POP]	1	0:00PM*	12:30AM	*	2	.50 2	2.5000000	
08/30/20	23 Wed		205 [POP]	0	2:00PM*	12:00AM	*	10.	.00 10	0.0000000	
08/31/20	23 Thu		205 [POP]	1	2:00PM*	10:00PM	*	10	.00 10	0.0000000	
09/01/20	023 Fri		205 [POP]	0	1:30PM*	06:15PM			4	.7500000	
			205 [POP]	1	1:30PM*	01:30AM	*	6	.75 2	.0000000	
09/03/20)23 Sun		208 [PADJ]								
Summary - PD01	11 [GOR	HAM, MI	CHAEL]								
									Accrual		
Paycode		N/A		Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]											97.00
3 [SICK]							143.00	4.00			147.00
6 [FH]											
205 [POP] 1[U	INUSED]			35.75		35.75					
208 [PADJ] 1[U	JNUSED]										

35.75

I CERTIF BOVE INFORMATION TO BE CORRECT THE X Employee Signature

x___

143.00

35.75

Supervisor Signature

4.00

244.00

08/21/2023 - 08/27/2023 [7 days]

Emp	loyee ID PD012		DE	PT(G2) PD			Pay Policy		201	
Р	ay Type 3		Last	Name CH	HARLSON COLLINS				ELIZABETH	
Time Card										
-	Date	Paycode		IN	OUT	•3	Daily Total		Reg Hrs	OT Hrs
80	08/24/2023 Thu 205 [I		DP] 10:0		03:53P	M*	6.00		6.0000000	
Summary - I	PD012 [CHARLSON	COLLINS, ELIZA	BETH]							
Devende							A	crual		
Paycode	N/A		Reg Hrs	OT1 - OT	-2 Total Hrs	Prior Bal	Adjust l	Ised	Earned	Available
205 [POP]	1[UNUSED]		6.00		6.00					
TOTALS			6.00		6.00					

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

Employee Signature

X Supervisor Signature

0

08/28/2023 - 09/03/2023 [7 days]

Emplo	yee ID	PD012		DE	PT(G2) PD			Pay Policy		201	
Pay	у Туре	3		Last Name CHA		ARLSON COLLINS		First Name		ELIZABETH	
ime Card											
	Date Paycode			IN	OUT		Daily Total		Reg Hrs	OT Hrs	
08/:	08/31/2023 Thu 201 [F		201 [POW]	1	1:00AM*	07:00P	M*	8.00		8.0000000	
Summary - P	D012	[CHARLSON	COLLINS, ELI	ZABETH]							
								A	ccrua	al	
Paycode		N/A		Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust l	Jsed	Earned	Available
201 [POW]	1[UNL	JSED]		8.00		8.00					

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X_____

Employee Signature

X Supervisor Signature

ltem 2.



Hours

1 message

Lizzi Charlson <lizzicharlson@gmail.com> To: Hailey Roessler <clerk@ridgewaywi.gov>

Wed, Sep 6, 2023 at 4:52 PM

Item 2.

Hello,

It was 1145-7p last Thursday. I won't be in tomorrow as the Marshal is gone.

Lizzi Charlson Collins

Employee Timecard - Hourly Distribution Report 08/21/2023 - 08/27/2023 [7 days]

AD002 [ROESSLER, HAILEY] Employee ID AD002 DEPT(G2) AD Pay Policy 500 HAILEY Pay Type 3 Last Name ROESSLER First Name **Time Card** OT Hrs OUT Date Paycode IN **Daily Total Reg Hrs** 08/21/2023 Mon 08:05AM 12:19PM 4.2500000 501 [TRW] 04:32PM 7.50 3.2500000 501 [TRW] 01:10PM 08/22/2023 Tue 501 [TRW] 08:01AM 11:14AM 3.2500000 501 [TRW] 11:59AM 04:29PM 7.75 4.5000000 3.7500000 08/23/2023 Wed 501 [TRW] 08:09AM 11:59AM 501 [TRW] 01.24PM 04:29PM 6.75 3.0000000 4.5000000 501 [TRW] 08:05AM 12:30PM* 08/24/2023 Thu 501 [TRW] 01:21PM 04:30PM 7.75 3.2500000 08/25/2023 Fri 501 [TRW] 08:01AM 12:02PM 4.00 4.0000000 08/26/2023 Sat 501 [TRW] 12:00PM* 12:30PM* 0.5000000 0.50 Summary - AD002 [ROESSLER, HAILEY] Accrual Paycode N/A Reg Hrs OT1 - OT-2 **Total Hrs** Prior Bal Adjust Used Earned Available

TOTALS		34.25	34.25	47.25
501 [TRW]	1[UNUSED]	34.25	34.25	
7 [BREV]				
6 [FH]				8.00
3 [SICK]				4.00
2 [VACA]				35.25

Gin Admin = 30.25 MIG ton Admin = (**BOVE INFORMATION TO BE CORRECT** I CERTIN

Employee Signature

Supervisor Signature

X

Gen Corresp./Phone/Email/In-person. AP/AR; HR/Timeshoot Work Comp/Insurance; Transition/Term/ Recruitment; WSLR fload Gatification; FMP/Budget; 2023 Park Improvement Proj. & ACH; Payroll; Grant Research; BikeRack Mono; Liquor Gionsing; Water Motor Schod. BikeRack Mono; Liquor Gionsing; Water Motor Schod. MRSINFF/Mpo; Sell 2 Portable AC Units; Late Fer MRSINFF/Mpo; Sell 2 Portable AC Units; Late Fer

ltem 2.

Employee Timecard - Hourly Distribution Report 08/28/2023 - 09/03/2023 [7 days]

 AD002 [ROESSLER, HAILEY]

 Employee ID
 AD002
 DEPT(G2)
 AD

 Pay Type
 3
 Last Name
 ROESSLER

Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hr
08/28/2023 Mon	503 [TRS]				1.0000000	
	501 [TRW]	07:58AM	08:41AM		0.7500000	
	501 [TRW]	09:54AM	12:33PM		2.5000000	
	501 [TRW]	12:58PM	04:45PM*	8.00	3.7500000	
08/29/2023 Tue	501 [TRW]	07:59AM	12:09PM		4.2500000	
	501 [TRW]	12:48PM	04:30PM	8.00	3.7500000	
08/30/2023 Wed	501 [TRW]	07:57AM	12:30PM*		4.5000000	
	501 [TRW]	01:00PM*	04:30PM*	8.00	3.5000000	
08/31/2023 Thu	501 [TRW]	08:00AM	12:03PM		4.0000000	
	501 [TRW]	12:52PM	04:33PM	7.75	3.7500000	
09/01/2023 Fri	501 [TRW]	08:02AM	12:02PM	4.00	4.0000000	

Summary - AD002 [ROESSLER, HAILEY]

							Accrual		
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]									35.25
3 [SICK]					4.00	4.00	1.00		7.00
6 [FH]									8.00
7 [BREV]									
501 [TRW]	1[UNUSED]	34.75		34.75					
503 [TRS]	1[UNUSED]	1.00		1.00					
TOTALS		35.75		35.75	4.00	4.00	1.00		50.25

chAd 7 Sewer Admin = 7. WaterAdmin dun=

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

Employee Signature

Supervisor Signature

Gen Corresp. / Email / Phone / In Person: HR/Timesheets / Nor Klomp Insurance; Volunteer (cord.; Utillty Billing / Mater Updates/ Account Updates / Scheduling / Messagging / Seasonal Cons. Hurrod off for Sept.; Bank Reconcisi ations / AP / Pik WEDC/ Bus Corresp.; RCC Scheduling (Coord.; Post Park + Rec/ Draft / Prep; Well Pump History; CIP 30

Item 2.

Pay Policy

First Name

500

HAILEY

08/21/2023 - 08/27/2023 [7 days]

Employee ID AD003			DEPT(G2) DC				Pay Policy 550				
	Pay Type 3			Last Name	JOHNSON		Fi	rst Name MAGGIE			
ime Card											
	Date	Paycode		IN	OUT		Daily Total	Reg Hrs	OT Hrs		
08	3/21/2023 Mon	511 [CW]	0	7:59AM	03:41PM	Λ	7.75	7.7500000			
30	8/22/2023 Tue	511 [CW]	0.	7:55AM	03:00PM	Л	7.00	7.0000000			
08	3/23/2023 Wed	511 [CW]	0	7:56AM	11:58AN	Л		4.0000000			
		511 [CW]	0	1:21PM	02:32PM	Л	5.25	1.2500000			
08	8/24/2023 Thu	511 [CW]	0	7:56AM	03:14PN	Л	7.25	7.2500000			
0	08/25/2023 Fri	511 [CW]	0	7:59AM	12:01PM	л	4.00	4.0000000			
Summary -	AD003 [JOHNSON	, MAGGIE]					Δ	ccrual			
Paycode	N	/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	10000	Used Earned	Available		
2 [VACA]									39.50		
B [SICK]									77.25		
6 [FH]									8.00		
511 [CW]	1[UNUSED]		31.25		31.25				_		
TOTALS			31.25		31.25				124.7		
	PA - C	5.25	Wo	tu	= (4 1	Depa	Uerk=	25		
Sev		0	v v								

Supervisor Signature

Employee Signature

Last Day was 8/25/2023

08/21/2023 - 08/27/2023 [7 days]

Рау Туре	1									
				Last Name	JOHNSON			First Name	a HARRY	
Date		Paycode		IN	OUT		Daily Total		Reg Hrs	OT Hrs
1/2023 Mon		401 [STW]	()8:13AM	02:09P	M		6.00	3.0000000	
2/2023 Tue		101 [PAW]	()8:13AM	01:01P	M		4.75	4.7500000	
3/2023 Wed		401 [STW]	(08:06AM	11:56A	M		4.00	4.0000000	
24/2023 Thu		611 [FMW]	(08:15AM	12:10P	M		4.00	4.0000000	
25/2023 Fri		401 [STW]	(08:11AM	01:07P	M		4.75	4.7500000	
W003 [JOH	INSON, H	ARRY]							_	
								Accrual		
	N/A		Reg Hrs	OT1 - OT-2	Total Hrs	Prior Ba	al Adjust	Used	Earned	Available
1[UNUSED]	1:	5.5	4.75		4.75					
1[UNUSED]		4	14.75		14.75					
1[UNUSED]		4	4.00		4.00					
		l	23.50		23.50					
	21/2023 Mon 22/2023 Tue 33/2023 Wed 24/2023 Thu 25/2023 Fri W003 [JOH 1[UNUSED] 1[UNUSED]	21/2023 Mon 22/2023 Tue 33/2023 Wed 24/2023 Thu 25/2023 Fri W003 [JOHNSON, H/ N/A	21/2023 Mon 401 [STW] 22/2023 Tue 101 [PAW] 32/2023 Wed 401 [STW] 24/2023 Thu 611 [FMW] 25/2023 Fri 401 [STW] W003 [JOHNSON, HARRY] N/A 1[UNUSED] 15,5 1[UNUSED] 44	1/2023 Mon 401 [STW] 0 22/2023 Tue 101 [PAW] 0 3/2023 Wed 401 [STW] 0 24/2023 Thu 611 [FMW] 0 25/2023 Fri 401 [STW] 0 W003 [JOHNSON, HARRY] 0 N/A Reg Hrs 1[UNUSED] 15, 5 -4.75 1[UNUSED] 44 44.75 1[UNUSED] 44 44.00	1/2023 Mon 401 [STW] 08:13AM 22/2023 Tue 101 [PAW] 08:13AM 32/2023 Wed 401 [STW] 08:06AM 24/2023 Thu 611 [FMW] 08:15AM 25/2023 Fri 401 [STW] 08:15AM 25/2023 Fri 401 [STW] 08:11AM W003 [JOHNSON, HARRY] N/A Reg Hrs 0T1 - 0T-2 1[UNUSED] 15.5 -4.75 1[UNUSED] 44 44.75 1[UNUSED] 40 4.00	N/A Reg Hrs OT1 - OT-2 Total Hrs 1[UNUSED] 15.5 4.75 4.75 14.75	N/A Reg Hrs OT1 - OT-2 Total Hrs Prior B: 1[UNUSED] 15.5 -4.75 4.75 14.75	N/A Reg Hrs OT1 - OT-2 Total Hrs Prior Bal Adjust 1[UNUSED] 15,5 -4.75 14.75 14.75 14.75	1/2023 Mon 401 [STW] 08:13AM 02:09PM 6.00 6 22/2023 Tue 101 [PAW] 08:13AM 01:01PM 4.75 4 3/2023 Wed 401 [STW] 08:06AM 11:56AM 4.00 4 24/2023 Thu 611 [FMW] 08:15AM 12:10PM 4.00 4 24/2023 Thu 611 [FMW] 08:15AM 12:10PM 4.00 4 25/2023 Fri 401 [STW] 08:11AM 01:07PM 4.75 4 W003 [JOHNSON, HARRY] 0T1 - OT-2 Total Hrs Prior Bal Adjust Used 1[UNUSED] 15.5 4.75 4.75 4.75 4 4 1[UNUSED] 4.00 4.00 4.00 4.00 4 4 4 4	1/2023 Mon 401 [STW] 08:13AM 02:09PM 6.00 6.000000 22/2023 Tue 101 [PAW] 08:13AM 01:01PM 4.75 4.750000 32/2023 Wed 401 [STW] 08:06AM 11:56AM 4.00 4.000000 24/2023 Thu 611 [FMW] 08:15AM 12:10PM 4.00 4.000000 24/2023 Thu 611 [FMW] 08:15AM 12:10PM 4.00 4.0000000 25/2023 Fri 401 [STW] 08:11AM 01:07PM 4.75 4.7500000 W003 [JOHNSON, HARRY] Accrual 1[UNUSED] 15.5 -4.75 4.75 4.75 1[UNUSED] 15.5 -4.75 4.75 4.75 4.75 1[UNUSED] 14.75 14.75 14.75 4.00 4.00 4.00

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

1

Employee Signature

110

10

X Supervisor Signature

Q.

de



08/28/2023 - 09/03/2023 [7 days]

Er	mployee ID	vee ID PW003		DEPT(G2) FM			Pay Policy 401				
	Pay Type	1		Last Name	JOHNSON		First Name		HARRY		
Time Card		1	1		1						
	Date	Paycoo	de	IN	OUT		Daily Total		Reg Hrs	OT Hr	
08/28	/2023 Mon	401 [ST	W] (08:18AM	01:20P	M	5.	00 5	0000000		
08/29	/2023 Tue	101 [PA	.w]	08:17AM	11:20A	M	3.	00 3	0000000		
08/30	/2023 Wed	401 [ST	w]	08:14AM	12:42P	M	4.	50 4	.5000000		
08/31	/2023 Thu	401 [ST	w]	08:21AM	01:33P	M	5.3	25 5	.2500000		
09/0	1/2023 Fri	611 [FN	IW]	08:12AM	11:18A	M	3.	00 3	0000000		
Summary - PW	/003 [JOH	INSON, HARRY]									
								Accrual			
Paycode		N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available	
	[UNUSED]	5.25	3.00		3.00						
	[UNUSED]	12.5	14.75		14.75						
	[UNUSED]	3.0	3.00		3.00						
TOTALS			20.75	-	20.75						

I CERTIFY THE ABOVE, INFORMATION TO BE CORRECT

Village Shop Laun

Moded

in

Mowed Acchaum, at

Maced westend of

a lon

caned Board oom

131 Spray a both Bal

Inlon X

Employee Signature

Man

Tues

5/28

205-11

, Keik færking løt

elle

08/21/2023 - 08/27/2023 [7 days]

Employee ID	PW005	DEPT(G2)	PW	Pay Po	licy 300	
Pay Type	3	Last Name	PETERSON	First Na	me DALE	
ard				1		
Date	Paycode	IN IN	OUT	Daily Total	Reg Hrs	OT Hrs
08/21/2023 Mon	601 [WAV	V] 06:56AM	02:05PM*	7.00	7.0000000	
08/22/2023 Tue	604 [WAN	/]			2.0000000	
	301 [SEV	V] 06:32AM	11:00AM*		4.5000000	
	304 [SEV	ŋ		8.00	1.5000000	
08/23/2023 Wed	301 [SEV	V] 06:27AM	11:02AM	4.50	4.5000000	
08/24/2023 Thu	601 [WAV	V] 06:25AM	12:01PM		5.5000000	
	301 [SEV	/] 12:25PM	03:27PM	8.50	3.0000000	
08/25/2023 Fri	601 [WAV	V] 06:57AM	01:55PM	7.00	7.0000000	

Summary - PW005 [PETERSON, DALE]

							Accrual		
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]					44.00		3.50		40.50
3 [SICK]									
6 [FH]									8.00
301 [SEW]	1[UNUSED]	12.00		12.00					
304 [SEV]	1[UNUSED]	1.50		1.50					
601 [WAW]	1[UNUSED]	19.50		19.50					
604 [WAV]	1[UNUSED]	2.00		2.00					
TOTALS		35.00		35.00	44.00		3.50		48.50

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

Employee Signature

X_____

Supervisor Signature

Dale Peterson Weekly Work Log 8-21-2023

Monday 8-21-2023:

Daily Labs at WWTP 2 Hours

Meter install 1 Hour, Repaired Chlorine Leak at Well 2, 2 Hours, cleaned the clarifiers and skimmers 1 Hour, Dropped chain saws off for repair 1 Hour.

Tuesday 8-22-2023:

****08/23/2023 08:13AM HROESSLER****

Dale Peterson

Tue, Aug 22, 10:39?AM (21 hours ago)

My HVAC guy is supposed to be onsite at 12, so I am going to take a half day at leave at 11.

Daily Labs at WWTP 2 Hours

Prepped water meters for the weeks schedule 2 Hours, Used 4 Hours of Vacation for HVAC issues at home.

Wednesday 8-23-2023:

Daily Labs at WWTP 2 Hours

Cleaned Clarifiers and Influent screen and trough 2 Hours

Thursday: 8-24-2023:

Daily Labs at WWTP 2 Hours

2 Meter installs 1.5 Hours, Repaired chlorine leak at Well 2 1 Hour, Cleaned clarifiers and Influent screen and trough 2 Hours, prepped letter to Monticello for taking back sludge 1.5 Hours

Friday 8-25-2023:

Daily Labs at WWTP 2 Hours

1 Meter install appointment and 1 checkup for a leak 1.5 Hours,

Vacuumed the foam off of the digesters and aeration basins 2.5 Hours, cleaned the clarifiers 1 Hour

Item 2.

8.00

52.50

08/28/2023 - 09/03/2023 [7 days]

	Employee ID PW005			DEPT(G2) PW			Pay Policy		300	
	Pay Type 3			Last Name	PETERSON		First Name DAL		DALE	
me Card										
	Date	Paycode		IN	OUT		Daily Total		Reg Hrs	OT Hr
30	3/28/2023 Mon	301 [SEW]		06:55AM	12:43F	M		5	.7500000	
		301 [SEW]		01:16PM	03:25F	M	8	3.00 2	.2500000	
08	8/29/2023 Tue	601 [WAW]		07:00AM	11:54A	M		5	0000000	
		301 [SEW]		12:23PM	03:30F	M	8	3.00 3	0000000	
30	3/30/2023 Wed	301 [SEW]		06:56AM	11:38A	M		4	.7500000	
		601 [WAW]		12:15PM	02:57F	M	7	7.50 2	.7500000	
08	8/31/2023 Thu	301 [SEW]		06:57AM	12:19F	M		5	.2500000	
		601 [WAW]		12:45PM	03:28F	M	8	3.00 2	.7500000	
0	09/01/2023 Fri	301 [SEW]		06:56AM	01:56F	M	7	7.00 7	.0000000	
ummary -	PW005 [PETERSC	N, DALE]								
								Accrual		
Paycode	N	I/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
[VACA]										40.50
[SICK]							4.00			4.00

TOTALS		38.50	38.50	4.00
601 [WAW]	1[UNUSED]	10.50	10.50	
301 [SEW]	1[UNUSED]	28.00	28.00	

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

X

Employee Signature

6 [FH]

Supervisor Signature
Dale Peterson Weekly Work Log 8-28-2023

Monday 8-28-2023:

Daily Labs at WWTP 2 Hours

Turned off water service for leak repair at Malcolm Stack Foundation 1 Hour, Pumped sludge at WWTP 1 Hour, Drafted letter to Monticello for sludge handling 1 Hour, Cleaned influent trough and screen 2 Hours, general WWTP maintenance 1 Hour.

Tuesday 8-29-2023:

Daily Labs at WWTP 2 Hours

3 Meter installs 2.5 Hours, Meter reads 3.5 Hours

Wednesday 8-30-2023:

Daily Labs at WWTP 2 Hours

2 Meter installs 1.5 Hours, Filled several potholes around the Village 2.5 Hours, General office duties 2 Hours.

Thursday: 8-31-2023:

Daily Labs at WWTP 2 Hours

1 Meter install 1 Hour, Cleaning concrete and drains at the ballpark 4 Hours, Pumped sludge 1 Hour,

Friday 9-1-2023:

Daily Labs at WWTP 2 Hours

Cleaned up remaining scrap steel etc 3 Hours, Cleaned and scrubbed the clarifiers 2 Hours

VILLAGE OF RIDGEWAY Report Date: 08/28/2023 Report Time: 8:17:12 AM

Employee Timecard - Hourly Distribution Report

08/21/2023 - 08/27/2023 [7 days]

SP003 [LOSBY, BRADEN]

Employee	ID SP003	DEPT(G2)	ST	Pay Policy	400
Pay Ty		Last Name	LOSBY	First Name	BRADEN
Fay Iy	pe 5				

Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hrs
08/21/2023 Mon	401 [STW]	06:55AM	11:17AM		4.2500000	
08/21/2023 1001	401 [STW]	11:55AM	03:33PM	7.75	3.5000000	
08/22/2023 Tue	401 [STW]	06:56AM	12:06PM		5.0000000	
00/22/2020 100	301 [SEW]	12:25PM	03:26PM	8.00	3.0000000	
08/23/2023 Wed	401 [STW]	06:55AM	12:28PM		5.5000000	
00/20/2020 1100	601 [WAW]	12:57PM	03:31PM	8.00	2.5000000	
08/24/2023 Thu	401 [STW]	06:24AM	12:03PM		5.5000000	
00/24/2020 1114	401 [STW]	12:37PM	03:30PM*	8.50	3.0000000	
08/25/2023 Fri	401 [STW]	06:56AM	11:12AM	4.25	4.2500000	
08/26/2023 Sat	601 [WAW]	10:08AM	12:08PM*	2.00	2.000000	
08/27/2023 Sun	301 [SEW]	08:49AM	10:49AM*	2.00	1.5000000	0.5000000

Summary - SP003 [LOSBY, BRADEN] Accrual Earned Used Prior Bal Adjust Total Hrs OT1 - OT-2 N/A Reg Hrs Paycode 2 [VACA] 3 [SICK] 6 [FH] 5.00 4.50 0.50 301 [SEW] 1[UNUSED] 31.00 31.00 401 [STW] 1[UNUSED] 4.50 4.50 1[UNUSED] 601 [WAW] 0.50 40.50 40.00 TOTALS

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

Employee Signature

Supervisor Signature

Available

34.50

24.00

8.00

66.50

Weekly Work Log August 21 – August 27 2023 Braden Losby

Monday August 21 2023:

Daily rounds. – 2 hours

Working on chemical issue at well 2. - 2 hours

Meter replacement. - 1 hour

Working on quotes for street projects and getting a diggers ticket for sign installation. - 1 hour

Pressure washing and trying to unclog drains. - 2 hours

Tuesday August 22 2023:

Daily rounds. - 2 hours

Watering trees and gardens. - 3 hours

Getting supplies at menards for new signs. - 2 hours

Trying to figure out issue with the dump bed on the f-550. - 1 hour

Wednesday August 23 2023:

Daily rounds. - 2 hours

Water meter replacement. - 1 hour

Mowing. - 2 hours

Street sweeping. - 3 hours

Thursday August 24 2023:

Daily rounds. - 2 hours

Meter replacements. - 1 hour

Going to dmv for cdl written test. - 2 hours

Working on furnace and cleaning at sewer plant. - 3 hours

Friday August 25 2023:

Daily rounds. - 2 hours

Meter replacement and trying to find curb box. - 1 hour

Watering trees and gardens. - 2 hours

Saturday August 26 2023:

Weekend rounds. - 2 hours

Sunday August 27 2023:

Weekend rounds. - 2 hours

08/28/2023 - 09/03/2023 [7 days]

SP003 [LOSBY, BRADEN]

Time Card

Employee ID	SP003	DEPT(G2)	ST	Pay Policy	400
Pay Type	3	Last Name	LOSBY	First Name	BRADEN

OUT

03:38PM

11.00444

Daily Total

3.50

Date	Paycode	IN
08/28/2023 Mon	401 [STW]	12:16PM
08/29/2023 Tue	401 [STW]	06:55AM
	401 [STW]	12:05PM
08/30/2023 Wed	401 [STW]	06:52AM
	401 [STW]	12:16PM

08/29/2023 Tue	401 [STW]	06:55AM	11:28AM		4.5000000		
	401 [STW]	12:05PM	03:30PM*	8.00	3.5000000		
08/30/2023 Wed	401 [STW]	06:52AM	11:39AM		4.7500000		
	401 [STW]	12:16PM	03:30PM		3.2500000		
	201 [POW]	09:48PM	11:48PM*	10.00	2.0000000		
08/31/2023 Thu	401 [STW]	06:50AM	12:12PM		5.2500000		
	401 [STW]	12:45PM	03:30PM	8.00	2.7500000		
09/01/2023 Fri	101 [PAW]	06:55AM	02:20PM*	7.25	7.2500000	39 	
09/02/2023 Sat	301 [SEW]	10:26AM	12:26PM*	2.00	2.0000000		
09/03/2023 Sun	601 [WAW]	09:33AM	11:33AM*	2.00	1.2500000	0.7500000	

Summary - SP003 [LOSBY, BRADEN]

							Accrual		
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]									34.50
3 [SICK]					24.00	4.00			28.00
6 [FH]									8.00
101 [PAW]	1[UNUSED]	7.25		7.25					
201 [POW]	1[UNUSED]	2.00		2.00					
301 [SEW]	1[UNUSED]	2.00		2.00					
401 [STW]	1[UNUSED]	27.50		27.50					
601 [WAW]	1[UNUSED]	1.25	0.75	2.00					
TOTALS		40.00	0.75	40.75	24.00	4.00			70.50

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

Employee Signature

Supervisor Signature

Item 2.

OT Hrs

Reg Hrs

3.5000000

Weekly Work Log August 28 – September 3 2023

Braden Losby

Monday August 28 2023:

Street sweeping. - 2 hours

Washing f-550. - 1 hour

Tuesday August 29 2023:

Daily rounds. - 2 hours

Installing signs at park. - 3 hours

Watering trees and gardens. - 2 hours

Meter reading. - 1 hour

Wednesday August 30 2023:

Daily rounds. - 2 hours

Meter replacements. - 1 hour

Cold patching. - 2 hours

Changing garbage. - 3 hours

Thursday August 31 2023:

Daily rounds. - 2 hours

Meter replacement. - 1 hour

Working on metal dumpster. - 1 hour

Power washing at park. - 4 hours

Friday September 1 2023:

Daily rounds. - 2 hours

Working on filling scrap metal dumpster. - 3 hours

Mowing at sewer plant. - 2 hours

Saturday September 2 2023:

Weekend rounds. - 2 hours

Sunday September 3 2023:

Weekend rounds. - 2 hours

TOTALS

Employee Timecard - Hourly Distribution Report 08/07/2023 - 08/13/2023 [7 days]

	Employee ID	PD011		1.	DEPT(G2)	PD			Pay Policy	601	
	Pay Type	1			Last Name	GORHAM			First Name		
Time Card											
	Date	Pa	aycode		IN	OUT		aily Total		Reg Hrs	OT UN
08/	07/2023 Mon	205	5 [POP]	12:	00AM*	03:00AM*		any rotar		0000000	OT Hrs
		205	[POP]	08:	00PM*	01:00AM*		8.0		0000000	
08/	08/2023 Tue	205	[POP]	01:	00PM*	10:00PM*		9.0	9.	0000000	
08/	10/2023 Thu	205	5 [POP]	12:	30PM*	03:30PM*				0000000	
		205	[POP]	08:	00PM*	01:00AM*		8.0	5.	0000000	
08	/11/2023 Fri	205	[POP]	02:	00PM*	12:00AM*		10.0		.0000000	
08/	13/2023 Sun	205	[POP]	07:	45PM*	12:15AM*		4.5	50 4.	5000000	
Summary - I	PD011 [GOF	RHAM, MICHAE	L]			The second					
Dennel				(Classical)					Accrual		
Paycode		N/A		Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
											97.00
											143.00
6 [FH]	100000000										8.00
205 [POP]	1[UNUSED]			39.50		39.50					

39.50

39.50

I CERTIFY THE OME INFORMATION TO BE CORRECT X Employee Signature

Supervisor Signature

X

Item 2.

248.00

08/14/2023 - 08/20/2023 [7 days]

Employ	yee ID	PD011		DEPT(G2)	PD			Pay Policy	601	
Pay	Туре	1		Last Name	GORHAM		-	First Name		
ime Card										
Date		Paycode	IN	u	OUT	Daily Tot	al	Reg Hrs		T Hrs
08/14/2023 N	/Ion	205 [POP]	02:30	PM* 1	0:00PM*		7.50	7.5000000		1110
08/15/2023 T	Гue	207 [POFH]						8.0000000		
		205 [POP]	02:30	PM* 0	4:00PM*		9.50	1.5000000		
08/16/2023 W	Ved	205 [POP]	10:00	AM* 1	2:00PM*			2.0000000		
		205 [POP]	06:00	PM* 1	1:30PM*		7.50	5.5000000		
08/17/2023 T	Thu	206 [POG]	04:00	PM* 1	1:59PM*		8.00		8.0	000000
08/18/2023	Fri	205 [POP]	12:01	AM* 0	1:00AM*		1.00	1.000000	0.0	
08/19/2023 5	Sat	205 [POP]	07:00	PM* 0	2:00AM*		7.00	7.0000000		
08/20/2023 S	Sun	208 [PADJ]								
mmary - PD011	IGOR	RHAM, MICHAEL]							1	
	LOON	NICHAEL]	-	1						
Paycode		N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Accrual	Fornad	Augilati
VACA]			grino	01.2	rotarris	T IIVI Dal	Adjust	Used	Earned	Available
SICK]										97.0
										143.0

TOTALS		32.50	8.00	40.50	8.00	8.00	240.00
208 [PADJ]	1[UNUSED]						
207 [POFH]	1[UNUSED]	8.00		8.00			
206 [POG]	1[UNUSED]		8.00	8.00			
205 [POP]	1[UNUSED]	24.50		24.50			
6 [FH]					8.00	8.00	
e forend							143.00

INFORMATION TO BE CORRECT I CERTIFY BO х

Employee Signature

X

Signature 3/17/2023 Supervisor Signature Grant Thursday 4p.m. to Michnight BHOURS

08/07/2023 - 08/13/2023 [7 days]

AD002 [ROESSLER, HAILEY]

Employee ID AD002	DEPT(G2) AD	Pay Policy 500
Pay Type 3	Last Name ROESSLER	First Name HAILEY

Time Card

Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hrs
08/07/2023 Mon	501 [TRW]	07:57AM	11:33AM		3.5000000	
	501 [TRW]	11:59AM	04:39PM	8.25	4.7500000	
08/08/2023 Tue	501 [TRW]	08:04AM	01:46PM		5.7500000	
	501 [TRW]	02:29PM	05:04PM		2.5000000	
	501 [TRW]	06:38PM	09:01PM	10.50	2.2500000	
08/09/2023 Wed	501 [TRW]	08:13AM	12:36PM		4.2500000	
	501 [TRW]	01:29PM	04:30PM	7.25	3.0000000	
08/10/2023 Thu	504 [TRV]	Vacation	\mathbf{c}	8.00	8.0000000	
08/11/2023 Fri	504 [TRV]	Vacation	m	6.00	6.0000000	

Summary - AD002 [ROESSLER, HAILEY]

Designed			OT1 - OT-2		Accrual					
Paycode	N/A	Reg Hrs		Total Hrs	Prior Bal	Adjust	Used	Earned	Available	
2 [VACA]					49.25		14.00)	35.25	
3 [SICK]									4.00	
6 [FH]									8.00	
7 [BREV]										
501 [TRW]	1[UNUSED]	26.00		26.00						
504 [TRV]	1[UNUSED]	14.00		14.00						
TOTALS		40.00		40.00	49.25		14.00		47.25	

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

Employee Signature

Supervisor Signature

X

Gen Corresp / Phone / Email / Office; AP/AR / Print Checks; HR; Payroll BOT Mtg Prep/Publish / Post/Attend /Minutes / Wrap-Up Met W/ Larry Bierke/ Iowa County Development; CDBG pricome Survey; Liquor Licensing; TID/Budget

Page 1

Employee Timecard - Hourly Distribution Report 08/14/2023 - 08/20/2023 [7 days]

AD002 [ROESSLER, HAILEY]

Employee ID AD002	DEPT(G2) AD	Pay Policy 500
Pay Type 3	Last Name ROESSLER	First Name HAILEY

Time Card

Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hr
08/14/2023 Mon	501 [TRW]	08:04AM	12:00PM		4.0000000	
	501 [TRW]	12:44PM	04:29PM	7.75	3.7500000	
08/15/2023 Tue	501 [TRW]	08:10AM	12:04PM		3.7500000	
	501 [TRW]	12:45PM	04:30PM	7.50	3.7500000	
08/16/2023 Wed	501 [TRW]	08:06AM	12:01PM		4.0000000	
	501 [TRW]	12:58PM	04:27PM	7.50	3.5000000	
08/17/2023 Thu	501 [TRW]	08:04AM	12:34PM		4.5000000	
	501 [TRW]	01:31PM	04:30PM	7.50	3.0000000	
08/18/2023 Fri	501 [TRW]	08:02AM	12:03PM	4.00	4.0000000	

Summary - AD002 [ROESSLER, HAILEY]

					Accrual					
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available	
2 [VACA]									35.25	
3 [SICK]									4.00	
6 [FH]					*:				8.00	
7 [BREV]									0.00	
501 [TRW]	1[UNUSED]	34.25		34.25						
TOTALS		34.25		34.25					47.25	

Water=5 Sewer= 2

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

Supervisor Signature Gen Corresp Phone/Email Office; HR/AP/AR; Submit ACH Util Payments to FSB; DNRI NWCF/Parkitecture /FOTMRST correspondence / planning / Bid Docs for Park Proj.j Development (TID' Research; Ordinance Revision Corresp.; Dog Park Safety Lights; Bike Rack Research; MHTClasse; FOE Eredits; WRS Research Prep; Payroll Prende; Utility Motor Work Ordens (Corresp. 7 It I research (memo drafting;

08/07/2023 - 08/13/2023 [7 days]

AD003 [JOHNSON, MAGGIE]

Employee ID	AD003	DEPT(G2) DC	Pay Policy	550
Pay Type	3	Last Name JOHNSON	First Name	

Time Card

Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hr
08/07/2023 Mon	513 [CS]				3.5000000	
	511 [CW]	07:57AM	12:00PM	7.50	4.0000000	
08/08/2023 Tue	511 [CW]	07:56AM	03:01PM	7.00	7.0000000	
08/09/2023 Wed	511 [CW]	07:55AM	02:30PM	6.50	6.5000000	
08/10/2023 Thu	511 [CW]	07:55AM	03:00PM*	7.00	7.0000000	
08/11/2023 Fri	514 [CV]			4.00	4.0000000	

Summary - AD003 [JOHNSON, MAGGIE]

Danada						Section des	Accrual		
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]					43.50		4.00		39.50
3 [SICK]					80.75		3.50		77.25
6 [FH]									8.00
511 [CW]	1[UNUSED]	24.50		24.50					0.00
513 [CS]	1[UNUSED]	3.50		3.50					
514 [CV]	1[UNUSED]	4.00		4.00					
TOTALS		32.00		32.00	124.25		7.50		124.75

Water = 4 Sewer =

Dep Clerk= 25

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

Employee Signature

Supervisor Signature

Redealers Utility payments, library/open gym communications, General: burn permits

Monday: Halloween Theodory: Free Somedule, office haves communications Wednesday: I day disconnects, library Mursday: Chaned library, account more in/act, helped w/ summer camp deposit Friday: Off Page 2

08/14/2023 - 08/20/2023 [7 days]

	Employee ID A	D003		DEPT(G2)	DC			Pay Policy	550	
	Pay Type 3			Last Name	JOHNSON			First Name	MAGGIE	
ne Card										
	Date	Paycode		IN	OUT		Daily Total		Reg Hrs	OT Hrs
08.	/14/2023 Mon	511 [CW]	0	7:57AM	03:31P	и		7.50	7.5000000	011112
08	/15/2023 Tue	511 [CW]	0	7:57AM	03:01PM	м	1	7.00	7.0000000	
	16/2023 Wed	511 [CW]	0	7:58AM	02:29PM	M		6.50	6.5000000	
	/17/2023 Thu	511 [CW]		7:58AM	03:08PM				7.2500000	
	3/18/2023 Fri	511 [CW]	0	7:56AM	12:02PM	M		4.00	4.0000000	
mmary - A	D003 [JOHN	SON, MAGGIE]	1			1		Accerval		
Paycode		N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Accrual Used	Earned	Available
ACA]										39.50
іск] н]										77.25
⊓] [CW]	1[UNUSED]		32.25		32.25					8.00
ALS			32.25		32.25					404.74
			02.20		52.25					124.75
Vat	er = 1	f Sew	en=	4	Gen	Depi	lerk	= 2	14.2	-5
RTIFY TI	HE ABOVE IN	FORMATION TO BE	CORRECT			9	1			
	. /						NO	n		
-111	ntf	1				x	Jal 1	ler		
nployee	Signature					Sur	ervisor Si	anature		
2PM1	val: 1)-	tility Payl mmunity (nents,	libr	anylo	spen !	3YM	commu	nicatio	ms,
		mminity (enter	rento	150'	1- (1.			· ·
10.00	/ /		201110							
	U	third in a	done -	tras						
nonday	f: Tree	trimming	dccr -	tags,						
ionday	f: Tree	trimming	dccr -	tags,		<i>C</i> 2		1.1		
ionday	f: Tree	trimming	dccr -	tags,		y as	nto	fall s	Check	le,
nonday	f: Tree	trimming	dccr -	tags,		y Ces	nto	fall 5	Checiu	le,
iesolag	f: Tree : Par Fall	trimming K & rec. ac news lette	door - gench,	tags, Comr	munit	y les	nto	fall s	Checlu	le,
iesolay lednes	f: Tree : Par Fall day: B	trimming K & rec. ac news lette OCK G Th	door - yench, 1 u wee	tags, Comr LK, fo	m Unit	usle++	er			
iesolay lednes	f: Tree : Par Fall day: B	trimming K & rec. ac news lette OCK G Th	door - yench, 1 u wee	tags, Comr LK, fo	m Unit	usle++	er			
iesolag lednes nursda	f: Tree : Par Fall day: B g: Me	trimming K & rec. ac news lette OCK GF Th ter Change	door - yench, 1 u wee	tags, Comr LK, fo	m Unit	usle++	er			
nonduy Tesolag Iednes hursda	f: Tree : Par Fall day: B g: Me	trimming K & rec. ac news lette OCK G Th	door - yench, 1 u wee	tags, Comr LK, fo	m Unit	usle++	er			

1

1

Employee Timecard - Hourly Distribution Report 08/07/2023 - 08/13/2023 [7 days]

Employee ID	PW003		DEPT(G2)	FM		Pay Polic	y 401	
Рау Туре	1		Last Name	JOHNSON		First Name	e HARRY	
me Card		THE REAL						
Date	Payco	le	IN	OUT	Daily Total		Reg Hrs	OT Hrs
08/07/2023 Mon	401 [ST	W]	08:14AM	12:55PM	4.	.75	4.7500000	
08/08/2023 Tue	611 [FN	W]	08:18AM	01:07PM	4.	.75	4.7500000	
08/09/2023 Wed	401 [ST	W]	08:17AM	12:21PM	4.	.00	4.0000000	
08/10/2023 Thu	401 [ST	W]	08:16AM	12:48PM	4.	.50	4.5000000	
08/11/2023 Fri	401 -[ST	W] Fac	07:57AM	11:39AM	3.	.75	3.7500000	
ummary - PW003 [JOH	INSON, HARRY]							m 71
Paycode	N/A	Reg	Hrs OT1 - OT-2	Total Hrs Prio	r Bal Adjust	Accrual Used	Earned	Available
1 [STW] 1[UNUSED]			17.00	17.00			Lantea	, tranabio
11 [FMW] 1[UNUSED]			4.75	4.75				
OTALS				The second second				
Streets=		7	Fac flo EECT	21.75 unt =	8,5	P	0.	
Streets =		7	Fac Mo		x day	R	e	
Streets =		7	Fac Mo		8,5 x Jay Supervisor Sign Access	Anture Are:	e mpa	mber
Streets =		7	Fac Mo		x day	Anture Are.	e mpa	mber
Streets =		7	Fac Mo		x day	Anture fro.	e mpa hcc	mber
Streets = CERTIFY THE ABOVE Hard Employee Signature in 8/7 Mi Hard, Vil Hard, Vil Hard, Vil Hard, Vil	INFORMATION TO E Anson Ewed we Walled w Gestation	Terrer ster ster sky reed	Fac plant	illage, n lanter. hec, co	x day Supervisor Sign Access in Access in Access in Access in Access in Access	fre. fr H Mbe	e mpa hcc v fo	mber md,
Streets =	INFORMATION TO E Anson Ewed we Walled w Gestation	Terrer ster ster sky reed	Fac plant	illage, n lanter. hec, co	x day Supervisor Sign Access in Access in Access in Access in Access in Access	fre. fr H Mbe	e mpa hic s fo at k	mber md, cc

201

of fait, we

out gardage

08/14/2023 - 08/20/2023 [7 days]

	Employee ID	PW003			DEPT(G2)	FM			Pay Polic	y 401	
	Pay Type	1			Last Name	JOHNSON			First Nam	HARRY	
Time Card											
	Date		Paycode		IN	OUT		Daily Total		Reg Hrs	OT Hrs
30	/15/2023 Tue		101 [PAW]		08:09AM	01:12P	M		5.00	5.0000000	
08	/16/2023 Wed		401 [STW]		08:20AM	02:35P	M		6.25	6.2500000	
08	/17/2023 Thu		401 [STW]		08:19AM	01:22P	м		5.00	5.0000000	
0	8/18/2023 Fri		401 [STW]		08:20AM	01:24P	м		5.25	5.2500000	
Summary -	PW003 [JOH	INSON, HA	RRY]								
									Accrual		
Paycode	1.000	N/A		Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
101 [PAW]	1[UNUSED]			5.00		5.00					
401 [STW]	1[UNUSED]			16.50		16.50					
TOTALS				21.50		21.50					

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

Supervisor Signature

Employee Signature

S

Page 4

leu

1/de

2033

07

C

Wester

4##, RC

rezeo

illace

zor

08/07/2023 - 08/13/2023 [7 days]

PW005 [PETERSON, DALE]

Employee ID PW005	DEPT(G2) PW	Pay Policy 300
Pay Type 3	Last Name PETERSON	First Name DALE

Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hrs
08/07/2023 Mon	301 [SEW]	06:55AM	11:59AM		5.0000000	
	601 [WAW]	12:28PM	03:28PM	8.00	3.0000000	
08/08/2023 Tue	301 [SEW]	06:55AM	12:07PM		5.000000	
	401 [STW]	12:34PM	03:36PM		3.0000000	
	601 [WAW]	06:54PM	08:54PM	10.00	2.0000000	
08/09/2023 Wed	301 [SEW]	06:59AM	12:02PM		5.0000000	
	601 [WAW]	12:28PM	03:25PM	8.00	3.0000000	
08/10/2023 Thu	301 [SEW]	06:47AM	12:00PM		5.2500000	
	601 [WAW]	12:26PM*	03:25PM	8.25	3.0000000	
08/11/2023 Fri	301 [SEW]	06:57AM	02:30PM*	7.50	5.7500000	1.750000

Summary - PW005 [PETERSON, DALE]

		The second second			an Rolling		Accrual		0
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]									44.00
3 [SICK]									8.00
6 [FH]									8.00
301 [SEW]	1[UNUSED]	26.00	1.75	27.75					
401 [STW]	1[UNUSED]	3.00		3.00					
601 [WAW]	1[UNUSED]	11.00		11.00					
TOTALS		40.00	1.75	41.75					60.00

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

Employee Signature

Χ____

Supervisor Signature

Dale Peterson Weekly Work Log 8-7-2023

Monday 8-7-2023:

Daily Labs at WWTP 2 Hours

Repaired chlorine supply line leak at Well 2, 2 Hours, Brush pickup 1 Hour, Submitted monthly water and wastewater reports 3 Hours.

Tuesday 8-1-2023:

Daily Labs at WWTP 2 Hours

Cold patched 3 manholes that have sunk and caused issues while plowing 1.5 Hours, Cleaned and treated Wells St Lift Station, heavy grease buildup all around and vacuumed out the top grease chunks 4 Hours, Cleaned the clarifiers and skimmers 1 Hours. Board Meeting 2 Hours

Wednesday 8-2-2023:

Daily Labs at WWTP 2 Hours

Troubleshot and adjusted chemical pump at Well 1 as it needed to run 2 Hours, Cleaned treated and vacuumed Collins St lift station 3 Hours,

Cleaned clarifiers and skimmers 1 Hour

Thursday: 8-3-2023:

Daily Labs at WWTP 2 Hours

Continued clearing of scrap from Village properties 2 Hours, Observed and trained on backup generator service and testing 2 Hours, Cleaned and scrubbed clarifier A, 2 Hours

Friday 8-4-2023:

Daily Labs at WWTP 2 Hours

Cleaned and scrubbed the clarifiers and skimmers 3 Hours, Employee Review 1.5 Hours

08/14/2023 - 08/20/2023 [7 days]

	Employee ID	PW005			DEPT(G2)	PW			Pay Policy	300	
	Pay Type	3			Last Name	PETERSON		First Name		DALE	
ime Card											
	Date		Paycode		IN	OUT		Daily Total		Reg Hrs	OT Hrs
08/	14/2023 Mon		603 [WAS]						4	.0000000	
			303 [SES]					8.	00 4	.0000000	
08	/15/2023 Tue		301 [SEW]	(06:56AM	11:58A	M		5	5.0000000	
			601 [WAW]		12:23PM	03:26P	M	8.	00 3	8.0000000	
08/	16/2023 Wed		301 [SEW]	(06:59AM	03:26P	м	8.	50 8	3.5000000	
08	/17/2023 Thu		601 [WAW]		06:55AM	12:14PM	M*		5	5.2500000	
			301 [SEW]	1	2:44PM*	03:27P	M	8.	00 2	2.7500000	
08	3/18/2023 Fri		301 [SEW]		06:58AM	02:00P	м	7.	00 7	.0000000	
Summary - I	PW005 [PE	TERSON,	DALE]								
				Beeller	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Accrual Used	Earned	Available
Paycode		N/A		Reg Hrs	011-01-2	Total His	FIIUI Dai	Aujusi	Used	Lameu	44.00
							8.00		8.00		
3 [SICK]							0.00		0.00		8.00
6 [FH]				23.25		23.25					0.00
301 [SEW]	1[UNUSED			4.00		4.00					
303 [SES]	1[UNUSED					8.25					
601 [WAW]	1[UNUSED	ני		8.25		0.25					

4.00

39.50

8.00

X_____

4.00

39.50

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

1[UNUSED]

Employee Signature

603 [WAS]

TOTALS

Supervisor Signature

8.00

Item 2.

52.00

Dale Peterson Weekly Work Log 8-14-2023

Mondav 8-14-2023:

Out Sick 8 Hours

Tuesday 8-15-2022

Daily Labs at WWTP 2 Hour

Continued looking at options to keep lift stations clean 3 Hours, Gathered report paperwork for water testing ? Hours.

Wednesday 8-16-202

Daily Labs at WWTP 2 Hours

Prove WW samples to Lancaster due to 8 hour hold time 2.5 Hours, Investigating I&I sources 3.5 Hours

Thursday: 8-17-2023:

Daily Labs at WWTP 2 Hours

Investigated source of rain I&I coming into the WWTP 2 Hours, Cleaned clarifier A 2 Hours, Replaced 2 GFI outlets and 3 bulbs on the UV disinfection system 1 Hour, Calculating sludge storage volume for cost analysis 1 Hour

Friday 8-4-2023:

Daily Labs at WWTP 2 Hours

Cleaned clarifier B 2.5 Hours, attempted 2 service locates 1 Hour, Filled pothole in Fire Station lot left by removal of old telephone pole 1.5 Hours, Organized report paperwork 1 Hour.

08/07/2023 - 08/13/2023 [7 days]

SP003 [LOSBY, BRADEN]

Employee ID SP003	DEPT(G2) ST	Pay Policy 400
Pay Type 3	Last Name LOSBY	First Name BRADEN

Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hrs
08/07/2023 Mon	401 [STW]	06:55AM	11:50AM		4.7500000	
	401 [STW]	12:22PM	03:30PM*	8.00	3.2500000	
08/08/2023 Tue	401 [STW]	06:55AM	11:57AM		5.000000	
	301 [SEW]	12:33PM	03:37PM		3.0000000	
	401 [STW]	07:48PM	08:54PM	9.25	1.2500000	
08/09/2023 Wed	401 [STW]	06:55AM*	11:49AM*		4.7500000	
	301 [SEW]	12:24PM*	03:37PM	7.75	3.0000000	
08/10/2023 Thu	401 [STW]	06:50AM	12:19PM		5.2500000	
	401 [STW]	12:53PM	03:27PM	7.75	2.5000000	
08/11/2023 Fri	401 [STW]	06:55AM	01:28PM		6.5000000	
	301 [SEW]	10:44PM	12:44AM*	8.50	0.7500000	1.250000
08/12/2023 Sat	301 [SEW]	08:18AM	10:18AM*	2.00		2.000000
08/13/2023 Sun	601 [WAW]	09:57AM	11:57AM*	2.00		2.000000

Summary - SP003 [LOSBY, BRADEN]

A STREET STREET	A CONTRACTOR OF						Accrual		
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]									34.50
3 [SICK]									24.00
6 [FH]									8.00
301 [SEW]	1[UNUSED]	6.75	3.25	10.00		-			
401 [STW]	1[UNUSED]	33.25		33.25					
601 [WAW]	1[UNUSED]		2.00	2.00					
TOTALS		40.00	5.25	45.25					66.50

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

X

Х

Employee Signature

Supervisor Signature

st 7 – August 13 2023 Braden Losby

Weekly Work Log August 7 – August 13 202
Monday August 7 2023:
Daily rounds. – 2 hours
Replacing check valve on well 2. – 3 hours
Pushing brush back with skid steer. – 1 hour
Attempting to strip yellow paint in front of the school 1 hour
Tuesday August 8 2023:
Daily rounds. – 2 hours
Working on my cdl training. – 1 hour
Cleaning out well st lift station on well st 4 hours
Board meeting. – 1.25 hours
Wednesday August 9 2023:
Daily rounds. – 2 hours
Fire call. – 2 hours
Cleaning out lift station on Collins st 4 hours
Thursday August 10 2023:
Daily rounds. – 2 hours
Changing trash at park. – 2 hours
Watering trees and gardens 2 hours
Cleaning out green shed. – 2 hours
Friday August 11 2023:
Daily rounds. – 2 hours
Meter replacement 0.5 hours
Brush hogging at sewer plant 2 hours
Weed eating. – 1.5 hours
Finance committee meeting. – 0.5 hours
Call in for plant alarm due to storm. – 2 hours
Saturday August 12 2023:
Weekend rounds. – 2 hours

Sunday August 13 2023:

Weekend rounds. - 2 hours

08/14/2023 - 08/20/2023 [7 days]

SP003 [LOSBY, BRADEN]

Employee ID	SP003	DEPT(G2)	ST	Pay Policy	400
Pay Type	3	Last Name	LOSBY	First Name	BRADEN

	Tin	ne	Ca	rd	
--	-----	----	----	----	--

Date	Paycode	IN	OUT	Daily Total	Reg Hrs	OT Hrs
08/14/2023 Mon	401 [STW]	06:55AM	11:45AM		4.7500000	
	401 [STW]	12:18PM	03:29PM	8.00	3.2500000	
08/15/2023 Tue	401 [STW]	06:55AM	03:17PM	8.25	8.2500000	
08/16/2023 Wed	401 [STW]	06:55AM	11:32AM		4.5000000	
	401 [STW]	12:03PM	03:34PM	8.00	3.5000000	
08/17/2023 Thu	401 [STW]	06:56AM	12:10PM		5.2500000	
	401 [STW]	12:42PM	03:30PM*	8.00	2.7500000	
08/18/2023 Fri	401 [STW]	06:48AM	01:30PM	6.50	6.5000000	
08/19/2023 Sat	401 [STW]	09:46AM	11:46AM*	2.00	1.2500000	0.7500000
08/20/2023 Sun	401 [STW]	08:47AM	10:47AM*	2.00		2.000000

Summary - SP003 [LOSBY, BRADEN]

					Accrual				
Paycode	N/A	Reg Hrs	OT1 - OT-2	Total Hrs	Prior Bal	Adjust	Used	Earned	Available
2 [VACA]									34.50
3 [SICK]									24.00
6 [FH]									8.00
401 [STW]	1[UNUSED]	40.00	2.75	42.75					
TOTALS		40.00	2.75	42.75		9			66.50

I CERTIFY THE ABOVE INFORMATION TO BE CORRECT

xP2

Employee Signature

X

Supervisor Signature

Weekly Work Log August 14 – August 20 2023

Braden Losby

Monday August 14 2023:

Daily rounds. – 2 hours

Cleaned the interior of the sewer truck. - 2 hours

Working on aluminum culvert lid. - 1 hour

Pushing back brush. - 1 hour

Clearing downed branches off level st. - 2 hours

Tuesday August 15 2023:

Daily rounds. – 2 hours

Power washing at park. - 6 hours

Wednesday August 16 2023:

Daily rounds. - 2 hours

Hardware store run for gfci outlets for sewer plant and install. - 2 hours

Cleaning dugout at park. - 1 hour

Street sweeping. - 2 hours

Popping manholes for influent issue. - 1 hour

Thursday August 17 2023:

Daily rounds. - 2 hours

Fire call. - 1 hour

Investigating influent issue. - 3 hours

Working on UV bulbs and electrical cabinet at plant. - 2 hours

Friday August 18 2023:

Daily rounds. - 2 hours

Cold patching at fire department. - 1 hour

Picking up brush. - 1 hour

Pushing and burning brush. - 2 hours

Saturday August 19 2023:

Weekend rounds. - 2 hours

Sunday August 20 2023:

Weekend rounds. - 2 hours

MONTHLY STATS Statistics from: 8/1/2023 12:00:00AM to 8/31/2023 11:59:00PM

Count of Reports Completed



Citation

Item 2.

Count of Incident Types



4.76% # of Reports: 1 Citation CURFEW

4.76% # of Reports: 1 Citation INATTENTIVE DRIVING

4.76% # of Reports: 1 Citation ISSUE OF A WORTHLESS CHECK

4.76% # of Reports: 1 Citation NO INSURANCE (PROOF)

4.76% # of Reports: 1 Citation OPERATING WHILE SUSPENED

4.76% # of Reports: 1 Citation POSSESION OF THC

4.76% # of Reports: 1 Citation POSSESSION OF PARAPHERNAILIA

14.29% # of Reports: 3 Citation EQUIPMENT VIOLATIONS

42.86% # of Reports: 9 Citation SPEEDING

Count of Incident Types



4.76% # of Reports: 1 Citation UNLICENSED DOG

Grand Total: 100.00% Total # of Incident Types Reported: 21 Total # of Reports: 16

Item 2.

Count of Incident Types



7.14% # of Reports: 1 Calls For Service ANIMAL COMPLAINT

7.14% # of Reports: 1 Calls For Service CRIMINAL DAMAGE TO PROPERTY

7.14% # of Reports: 1 Calls For Service ILLEGAL USE OF FIREWORKS

7.14% # of Reports: 1 Calls For Service OPERATING A MOTOR VEHICLE WITHOUT OWNER'S CONSENT

7.14% # of Reports: 1 Calls For Service TRAFFIC COMPLAINT CITIZEN COMP

7.14% # of Reports: 1 Calls For Service WELFARE CHECK

14.29% # of Reports: 2 Calls For Service SUSPICIOUS ACTIVITY

35.71% # of Reports: 5 Calls For Service ASSIST ICSO

Grand Total: 100.00% Total # of Incident Types Reported: 14 Total # of Reports: 14



14.29% # of Reports: 1 Incident Report ASSIST OTHER LAW ENFORCEMENT AGENCY : ASSIST BARNEVELD POLICE

14.29% # of Reports: 1 Incident Report ASSIST OTHER LAW ENFORCEMENT AGENCY : ASSIST MADISON PD

14.29% # of Reports: 1 Incident Report CRIMES : THEFT : THEFT FELONY

14.29% # of Reports: 1 Incident Report CRIMES : RECOVERY OF LOST OR STOLEN PROPERTY

14.29% # of Reports: 1 Incident Report SERVICE CALL : WELFARE CHECK

14.29% # of Reports: 1 Incident Report CRIMES : THEFT : VEHICLE (OPERATING W/O CONSENT)

Grand Total: 100.00% Total # of Incident Types Reported: 7 Total # of Reports: 4

Grand Total: 100.00% Total # of Incident Types Reported: **42**



Iowa County Sheriff's Office

109 East Leffler Street, Dodgeville, Wisconsin 53533 Phone | 608-930-9500 Fax | 608-935-0331 Crime Stoppers | 608-319-6703

Calls for Service Ridgeway

Michael W. Peterson, Sheriff Austin L. Durst, Chief Deputy

Printed on September 1, 2023

CFS #	CFS Date/Time	Description	Address	Description
CFS23014700	08/01/23 19:28:20	MVA	COUNTY ROAD Y / DUGWAY RD, RIDGEWAY, WI	Assignment Completed/Settled By Contact
CFS23014733	08/02/23 10:57:57	Disabled Vehicle	US HIGHWAY 151 / COUNTY ROAD YZ, DODGEVILLE	Assignment Completed/Settled By Contact
CFS23014829	08/03/23 20:40:13	Traffic Stop	COUNTY RD HHH / TOWN HALL RD, RIDGEWAY, WI	Assignment Completed/Settled By Contact
CFS23014964	08/05/23 20:44:58	Theft	619 MAIN ST, RIDGEWAY, WI 53582	Assignment Completed/Settled By Contact
CFS23015012	08/06/23 13:28:03	Assist LE	COUNTY ROAD H / DODGE ROAD, ARENA, WI 53503	Assignment Completed/Settled By Contact
CFS23015100	08/07/23 21:31:50	Suspicious	619 MAIN ST, RIDGEWAY, WI 53582	Assignment Completed/Settled By Contact
CFS23015148	08/08/23 19:26:23	Follow Up	458 ASHTER STREET, LONE ROCK	Assignment Completed/Settled By Contact
CFS23015149	08/08/23 20:49:34	Citizen Assist	703 MAIN STREET, RIDGEWAY, WI 53582	Assignment Completed/Settled By Contact
CFS23015453	08/13/23 19:59:45	Follow Up	907 HIGH POINT RD, DODGEVILLE	Assignment Completed/Settled By Contact
CFS23015633	08/16/23 18:29:38	Vandalism	105 HUGHITT ST, RIDGEWAY, WI 53582	Assignment Completed/Settled By Contact
CFS23015635	08/16/23 19:00:03	Assault	304 MAIN ST, HOLLANDALE, WI 53544	Assignment Completed/Settled By Contact
CFS23015637	08/16/23 19:11:50	Traffic Hazard	1205 N BEQUETTE ST, DODGEVILLE, WI 53533	Assignment Completed/Settled By Contact
CFS23015677	08/17/23 16:48:23	Traffic Stop	US HIGHWAY 18 / BENNETT RD, DODGEVILLE, WI 53533	Assignment Completed/Settled By Contact
CFS23015698	08/17/23 19:09:56	Traffic Stop	6045 US HIGHWAY 18-151, RIDGEWAY, WI 53582	Assignment Completed/Settled By Contact

CFS #	CFS Date/Time	Description	Address	Description
CFS23015700	08/17/23 19:22:59	Traffic Stop	COUNTY ROAD YZ / US HIGHWAY 18-151, DODGEVILLE, WI 53533	Assignment Completed/Settled By Contact
CFS23015707	08/17/23 20:29:25	Traffic Stop	6201 US HIGHWAY 18-151, RIDGEWAY	Assignment Completed/Settled By Contact
CFS23015710	08/17/23 21:04:37	Traffic Stop	5485 US HIGHWAY 18-151, DODGEVILLE, WI 53533	Assignment Completed/Settled By Contact
CFS23015714	08/17/23 21:48:16	Traffic Stop	5485 US HIGHWAY 18-151, DODGEVILLE, WI 53533	Assignment Completed/Settled By Contact
CFS23015866	08/19/23 21:08:38	Traffic Complaint	6289 US HIGHWAY 18-151, RIDGEWAY, WI 53582	Assignment Completed/Settled By Contact
CFS23015870	08/19/23 21:41:18	Traffic Complaint	US HIGHWAY 18-151 / COUNTY ROAD K, BARNEVELD, WI 53507	Assignment Completed/Settled By Contact
CFS23016002	08/21/23 20:07:46	Traffic Stop	COUNTY RD HHH / US HIGHWAY 18-151, RIDGEWAY, WI 53582	Assignment Completed/Settled By Contact
CFS23016010	08/21/23 22:38:20	Suspicious	STONIER LN, RIDGEWAY	Assignment Completed/Settled By Contact
CFS23016059	08/22/23 17:23:11	Stolen Property	2911 BLUE ASTER BLVD, SUN PRAIRIE	Assignment Completed/Settled By Contact
CFS23016145	08/23/23 20:11:23	Fraud	408 MAIN ST, RIDGEWAY, WI 53582	Assignment Completed/Settled By Contact
CFS23016179	08/24/23 10:50:59	Traffic Stop	COUNTY ROAD BB, DODGEVILLE, WI 53533	Assignment Completed/Settled By Contact
CFS23016191	08/24/23 13:38:37	Suspicious	110 CRETNEY ST, RIDGEWAY, WI 53582	Dispatcher handled
CFS23016262	08/25/23 17:43:39	Traffic Complaint	30 US HIGHWAY 151, MINERAL POINT, WI 53565	Assignment Completed/Settled By Contact
CFS23016272	08/25/23 19:31:51	Traffic Stop	1500 US HIGHWAY 151, MINERAL POINT, WI 53565	Assignment Completed/Settled By Contact
CFS23016279	08/25/23 20:09:57	Traffic Stop	800 COMPASSION WAY, DODGEVILLE, WI 53533	Assignment Completed/Settled By Contact
CFS23016289	08/25/23 21:10:42	Traffic Stop	US HIGHWAY 151 / US HIGHWAY 18-151, DODGEVILLE, WI 53533	, Assignment Completed/Settled By Contact
CFS23016444	08/27/23 21:26:23	Suspicious	102 TALLMAN CT, RIDGEWAY	Assignment Completed/Settled By Contact
CFS23016494	08/28/23 23:25:22	Traffic Complaint	7289 US HIGHWAY 18-151, BARNEVELD, WI 53507	Assignment Completed/Settled By Contact

Total Records: 32

CFS #	Ħ.
-------	----

Description

Address

Description



FW: WisDOT: Public input requested for US 18/151 bridge and curve repairs, lowa County

1 message

Craig Hardy <Craig.Hardy@iowacounty.org>

Thu, Aug 10, 2023 at 7:49 AM

Item 3.

To: "Town of Ridgeway (twnridge@mhtc.net)" <twnridge@mhtc.net>, "Megan <ieden (Town Brigham)" <megan@brighamtown.com>, "Michelle Walker/Village of Barneveld Clerk (barneveld@mhtc.net)"
barneveld@mhtc.net>, Village of Ridgeway - Hailey <Clerk@ridgewaywi.gov>, Greg Lee <glee@ci.dodgeville.wi.us>, "Sara Olson (Town Dodgeville)" <twnclerk@mhtc.net>

Cc: "Joe Thomas (joethomasrepair@gmail.com)" <joethomasrepair@gmail.com>, Neil Wenger <Neil.Wenger@iowacounty.org>, Tammy Fitzsimons <Tammy.Fitzsimons@iowacounty.org>, Jennifer Blanchard <Jennifer.Blanchard@iowacounty.org>, Kevin Peterson <Kevin.Peterson@iowacounty.org>

FYI – This project runs from CTH YZ thru the Pikes Peak road intersection and includes the Jones Street overpass in Barneveld. Am glad to see this come to fruition after many years of discussion and should help with the corridor slide-offs along 151. This is an opportunity to see the proposed improvements and provide feedback on the project.

Thanks

Please advise if I can be of further assistance.

Craig E Hardy

Iowa County Highway Commissioner

1215 N Bequette St.

Dodgeville, Wis. 53533

PH (608) 935-3381 X605

Mob. (608) 574-2935

From: Punzel, Hannah - DOT <hannah.punzel@dot.wi.gov>
Sent: Wednesday, August 9, 2023 4:55 PM
Subject: WisDOT: Public input requested for US 18/151 bridge and curve repairs, Iowa County



News Release

Wisconsin Department of Transportation
Hannah Punzel, WisDOT Communications (608) 246-7907, hannah.punzel@dot.wi.gov

Public input requested for US 18/151 bridge and curve repairs, Iowa County

The Wisconsin Department of Transportation (WisDOT) is seeking a public input on improvements along US 18/151 bridge and curve repair project, between the Town of Dodgeville and the Village of Barneveld, in Iowa County. Construction is currently scheduled for 2025.

A handout and materials detailing the US 18/151 project are available to view on the project website at https://wisconsindot.gov/Pages/projects/by-region/sw/us18151-dodgeville/default.aspx.

The project is proposed to improve safety by adding a High Friction Surface Treatment (HFST) on the bridge deck on US 18/151 eastbound over South Jones Street. Along with the bridge repairs, WisDOT is also developing preliminary plans for a HFST for seven pairs of curves along US 18/151 to decrease the number of run-off-the-road crashes. Additionally, there are four bridges within the curve limits that will also receive a HFST.

US 18/151 will remain open to traffic while being constructed under single lane closures and closures of side roads along the project corridor.

All persons interested in the project are encouraged to view the website and provide input by contacting project staff via email, postal mail, or phone. Your comments will assist WisDOT in developing a project that will serve the needs of the traveling public and community. Your input is welcome and appreciated throughout the design process. Questions and comments can be directed to WisDOT Project Manager Derek Potter at derek.potter@dot.wi.gov, (608) 246-3861, or mailed to the WisDOT Southwest Region, 2101 Wright Street, Madison, WI 53704.

-End-



WisDOT Division of Transportation System Development Southwest Region 2101 Wright Street Madison, WI 53704 Governor Tony Evers Secretary Craig Thompson wisconsindot.gov Telephone: (608) 246-3800 FAX: (608) 246-7996 Email: swr.dtsd@dot.wi.gov



RECEIVED

August 9th, 2023

AUG 142023

Village of Ridgeway

Re: US 18/151 High Friction Surface Treatment (HFST) Project Project ID's: 1204-00-30/60 & 1204-00-31/61 Dodgeville – Barneveld US 18/151 Iowa County

Dear Sir/Madam,

The Wisconsin Department of Transportation (WisDOT) is proposing to improve the safety of US 18/151 between the Town of Dodgeville and the Village of Barneveld in Iowa County. The purpose of the project is to address the safety concerns along seven pairs of curves in this segment, and the project is needed to reduce the high rate of run-off-the-road crashes. This project will also include a HFST overlay on the US 18/151 eastbound bridge deck over S Jones St in Barneveld to extend the service life of the deck.

In lieu of an in-person meeting, the project team is sending the enclosed newsletter, project display, and comment form. We are seeking your input on the proposed project.

If you have any questions or would like to find out more about the project, please contact me at the phone number or email below or visit the project website at https://wisconsindot.gov/Pages/projects/by-region/sw/us18151-dodgeville/default.aspx

Sincerely,

Derek Potter

Derek Potter, P.E. Project Manager WisDOT Southwest Region 2101 Wright Street Madison, WI 53704 (608) 246-3861 Derek.Potter@dot.wi.gov

Enclosures: Newsletter Project Location Map Comment Form



Project Overview

The Wisconsin Department of Transportation (WisDOT) is proposing to improve the safety of US 18/151 between Dodgeville and Barneveld by applying a High Friction Surface Treatment (HFST) to seven pairs of curves and a bridge deck (see next page for project location). Construction is currently scheduled for 2025. This newsletter has been prepared to provide you with an update on the proposed project and instructions for providing comments on the proposed design. All persons with an interest in the project are encouraged to provide comments by September 14th, 2023, either by phone, email, or mail using the attached comment form. Your input is welcome and appreciated throughout the design process.

Purpose and Need

The purpose of the project is to address the safety concerns of the horizontal curves on US 18/151 between Dodgeville and Barneveld. The project is needed to reduce the high rate of run-off-the-road crashes on the curves. This project will also address the deterioration of the bridge deck on US 18/151 EB over S Jones St to extend the service life of the bridge.

Summary of Proposed Improvements

The following alternatives were considered for this project:

- No Change This alternative would fail to address the high rate of run-off-the-road crashes and the deterioration of the bridge deck over EB S Jones St.
- High Friction Surface Treatment (HFST) This is the only alternative that will adequately
 address the purpose and need of the project.

The proposed improvements listed below and shown on the next page are for the HFST alternative. All construction will be limited to the travel lanes and bridge decks. Improvements include the following:

 Apply a High Friction Surface Treatment (HFST) overlay to the seven pairs of curves, including the four bridges located within these curves, and the US 18/151 bridge deck over S Jones St in Barneveld.

The project is still early in the design phase and WisDOT is looking for feedback for the proposed alternative being considered.

Traffic Impacts

US 18/151 would be reduced to one-lane of traffic in each direction during each stage of construction. Temporary ramp closures at the US 18/151 interchange, the Ridgevue Road interchange and the County ID interchange will be necessary at times, as will temporary closures at the intersections of County Y/YZ, Hi-Point Road, and W Brigham Road/Pikes Peak Road. Access would be maintained to property owners with driveways within the project limits. Construction is expected to last up to three months.

Project Schedule (subject to change)

2023
February 2025
Summer 2025

Preliminary Design, Environmental Document Final Plan Construction

August 2023

Contact information:

WisDOT Southwest Region 2101 Wright Street Madison, WI 53704 Website: www.wisconsindot.gov

Derek Potter, P.E. Project Manager WisDOT Southwest Region (608) 246-3861 Derek.Potter@dot.wi.gov

Paul Windsor Project Leader WisDOT Southwest Region (608) 246-3841 Paul.Windsor@dot.wi.gov

Hannah Punzel WisDOT Southwest Region Communications Manager (608) 246-7907 Hannah.Punzel@dot.wi.gov

> Mission statement: To provide leadership in the development and operation of a safe and efficient transportation s 75

Wisconsin Department of Transportation – Dedicated people creating transportation solutions through innovation and exceptional service.

.

Public Comment Form

RECEIVED AUG 1 4 2023 Village of Ridgeway

Project ID's 1204-00-30/60 & 1204-00-31/61 US 18/151 Dodgeville - Barneveld Iowa County

August 2023

Please place this form in the mail by September 14th, 2023, to the address on the back of this sheet. Comments can also be e-mailed to <u>Derek.Potter@dot.wi.gov</u>. Your comments assist us in developing a project that will serve the needs of the traveling public as well as the needs of the local community. Your input is welcome and appreciated throughout the design process.

Name:

Address:

Daytime Phone Number (optional):

Email Address (optional):

Please Print Comments (attach additional sheets if necessary)

The information in this document including names, addresses, phone numbers, e-mail addresses, and signatures is not confidential, and may be subject to disclosure upon request, pursuant to the requirements of the Wisconsin open records law, sections 19.31 - 19.39 of the Wisconsin Statutes.



PROJECT LOCATION MAP



1204-00-30/60	
USH 18/151	
S Jones St. Bridge B-25-37	
Construction Year: 2025	

1204-00-31/61

USH 18/151

Dodgeville to Barneveld

Construction Year: 2025

Iowa County

Item 3.

Item 3.

ABOUT US Westwood Wisconsin

Madison | Appleton



Firm Overview

Established in 1972, **Westwood Professional Services**, **Inc.** (Westwood) is a full service, multi-discipline professional engineering firm providing services to public and private clients throughout the United States. With over 1,350 employees nationwide and registrations in most states, Westwood is recognized as one of the top engineering firms in the country.

Westwood's survey team has over 50 years of experience providing services for Governmental agencies, municipalities, contractors, and the private sector. Our teams have completed surveys for all facets of boundary identification, topo graphic for design and construction staking. We are uniquely qualified based on our team's extensive experience and our long history of successfully delivering projects on-time and on-budget to local municipal agencies.

Our team values relationships, innovation, dedication, initiative, and growth. Above all, we are problem solvers. Whether it be for our clients, communities, or each other — we help others succeed. Westwood's mission is to provide solutions for your success.

What Sets Us Apart

At Westwood, we understand the quality of any project greatly impacts client satisfaction and the reputation of the firm, and we are dedicated to a company-wide commitment to quality control. Westwood offers the following benefits:

- Depth and availability in team and resources to ensure the project is completed on time and within budget
- Experts in government and regulatory collaboration, various external funding resources, and standard PS&E processing procedures
- Standardized QA/QC processes to ensure project compliance and expedite the overall review process
- Proactive and responsive communication, keeping all parties engaged and ensuring issues are addressed promptly.

Our diverse range of experience and personnel allows us to perform numerous projects simultaneously while still delivering the product on schedule.

When you call, we will answer!

Land Professional Surveyors Civil Engineers 170 Professional 500 andscape Electrical Architects **Engineers** Certified 152 170 Federal Land Surveyors Wisconsin Personnel Resources Licensed Professional **Professional Engineers** Land Surveyors 4 Water Resource Surveying Environmental **Engineers** Technicans Engineers and Scientists Structural **B** Geotech Engineer **Professionals 9** Materials Lab **Real Estate Professional** Professionals Professional Transportation Airport Geologist **Engineers** Engin 79

Westwood Personnel Resources

Firm Expertise

At Westwood, we recognize the complex needs of a thriving and growing City. From downtown revitalizations, regional water or wastewater systems and interstate highways, to local trails and neighborhood streets, utilities and parks, our commitment to design excellence remains constant. Our experience encompasses all aspects of civil design, land surveying, landscape architecture and planning projects, from overall design through construction administration and quality control initiatives. Having all of these services under one roof streamlines coordination and design efficiency.



Civil Engineering & Design

Our extensive knowledge base of industry best practices allows our engineers to work hand in hand with our surveyors, landscape architects, water resource professionals and traffic engineers

to provide the best possible service to our clients. Frequent collaboration and teamwork allow our engineers to manage firm and client resources more effectively and efficiently at the outset of each project.



Roadway Improvement

Westwood provides roadway assessment and design, as well as develops capital improvement plans for a wide array of municipalities and government agencies. We

tailor each assessment to the client's needs, whether they be a full-scale assessment or select areas within the jurisdiction. We evaluate existing conditions and provide recommendations for rehabilitation, including full roadway reconstruction, panel replacement for concrete roadways, appropriate types of pavement sealing, and spot repair and overlay.

Westwood is mindful of our clients' budgets and makes sure to provide realistic recommendations while outlining the pros and cons of each one. Westwood has provided detailed construction plans and construction administration support for roadways, both small and large, including residential streets to major thoroughfares with bridges.



Traffic/Transportation Engineering

From decades of experience on hundreds of projects, our traffic professionals bring a

specialized, yet 'big picture', perspective to provide relevant and practical consultation in traffic engineering. For clients needing a traffic engineering study, design of traffic devices, parking analysis, or transportation planning services, Westwood is ready to put our expertise to work and help simplify the design experience.



Aviation Engineering

Westwood's aviation experience ranges from large hubs to small regional aviation facilities throughout Wisconsin. From designing runways and taxiways, perimeter fencing, apron

expansions, and infrastructure to ILS relocations, stormwater drainage plans, and land acquisition services, our aviation team has been an integral part of the development process.



Land Surveying

Having an in-house surveying team is invaluable to our success. Our surveyors work on a daily basis with engineers. Therefore, they have a greater understanding of what is required for

an adequate engineering survey than other surveying firms. Having in-house survey affords us, as engineers, greater control over how efficiently survey data is collected and integrated into our designs.



Water Resources

Our water resources professionals are experts in working with waters of the US, along with urban drainage and stormwater management.

We understand the importance of providing tailored solutions that embrace the original vision of the communities in which we live. Our team uses the latest modeling software and analytical tools available so our clients can make informed decisions, helping them get the most out of every dollar during design, construction, and maintenance.



Utility Infrastructure

Our water and wastewater systems experience in Wisconsin includes civil engineering design and land surveying services for urban and rural lines ranging from 6 to 96 inches. Additionally,

we are experienced in trenchless designs such as boring/ tunneling, pipe bursting, cured-in-place pipe, pipe lining, swagelining, and horizontal directional drilling.



Environmental Engineering

Westwood provides a variety of environmental consulting services that can be tailored to specific needs. Our hydrogeologists and engineers have been extensively involved in

environmental assessments and the investigation and remediation of contamination at a variety of sites. Westwood has the expertise to provide services ranging from the initial environmental assessment through site closure.

Geotechnical Engineering

Withou is great investig

Without proper geotechnical research, there is greater risk for fatal flaws on a project. By investigating the subsurface conditions through soil borings, cone penetration tests, and other

tests, we can provide quality recommendations for the design and construction of project infrastructure.

Item 3.

OUR SERVICES Westwood Wisconsin

×

Madison | Appleton

X

SERVICES



Roads, grading, streams, stormwater management, water systems, overall site planning, traffic, and creating a complete design package are just a tiny bit of what our talented civil engineering team can design. From a small substation upgrade to a large public infrastructure project, our team will create optimal solutions within your scope, budget, and regulatory requirements. Whatever challenges your project is facing – schedule, coordination, entitlements, grading constraints, available infrastructure, or utility connections – our engineering team has the expertise to guide you to the end.

- Due Diligence and Feasibility Studies
- Concept Layout, Site Optimization, and Cost Estimates
- LID Programming and LEED Scoping
- Site Constraint Analysis
- Road Use Agreement Support
- Infrastructure Master Planning
- Drainage Analysis
- Site Engineering Concepts
- Preliminary Plan Approval
- Site Grading and Earthwork Balance
- Stormwater Management

- Water and Sanitary Sewer System Design
- Parking Lot and Circulation Design
- Erosion Control Design/NPDES Permitting
- Utility District Creation, Bond Application Reports
- Site Feasibility/Zoning Analysis
- Integrating and Coordinating Permitting Requirements
- Roadway Design
- Bridge Analysis and Design
- Final Construction Plans and Bidding Support
- GIS and Field Support Integration



Westwood Survey Services

We conduct thousands of surveys each year, including some of the largest developmental projects across the country. Our surveyors are known for their responsiveness and high-quality service, regardless of project size or complexity. Whether it's renewable energy, land development, public infrastructure, or electric utilities, our survey crews can help make the best of your projects.

- ALTA/NSPS Land Title Survey
- Topographic Survey
- Boundary Survey
- Base Mapping
- ROW Acquisition and Lease Exhibits
- Easement Exhibits and Preparation
- Legal Descriptions
- High-Resolution Aerial Mapping and LiDAR Ground Control
- Utility Crossing Matrix
- Desktop Survey Constraint Map
- Corridor Route Survey
- Right-of-Way Platting and Mapping

- Construction Staking & Verification
- Control Survey
- Section Corner Recovery and Monumentation
- Subdivision Platting
- Condominium Survey
- Construction Impact Survey
- Record Survey
- 3D Laser Surveying and Imaging
- Municipal Design Survey
- Geodetic Control Networks
- Hydraulic and Hydrologic Surveys
- Real-time Collaboration and Technology



Westwood is committed to bringing transportation projects of all sizes to life for our partners. From minor roadway maintenance projects to high-visibility full roadway reconstructions and expansions, our team of engineers have the expertise to make your project a success. This experience goes well beyond just designing projects, though, as our in-house experts understand the challenges local governments face related to budgets, schedules, stakeholder management, and regulation. We will work with you to identify state and federal grant programs that you may be eligible for, navigate applications and requirements, and develop a unique solution for your community; all while helping you capitalize on funding opportunities that make your budget go farther. Make your next project a success from idea through implementation by partnering with the team at Westwood.

- Detailed Plan Preparation
- Specification Development
- Cost Estimation
- Public Involvement Coordination
- Stakeholder Management
- Drainage Design
- Traffic Management Plans
- Utility Coordination
- Environmental Documentation
- Agency Coordination and Permitting
- Design Study Reporting
- Alternatives Analysis

- Bidding Support
- Funding Identification and Assistance
- Grant Compliance Guidance
- Roadway Preservation, Reconditioning, and Reconstruction Design
- Intersection, Interchange, and Roundabout Design
- Pavement Marking, Signage Plan, and Erosion Control Plan Preparation
- Traffic Safety Studies
- Roadway and Intersection Needs Analysis
- Pedestrian and Bicycle Accommodation Design



Westwood Traffic Engineering

Aging infrastructure, increasing traffic demands, and dense urban environments require unique transportation solutions. Our team combines traditional planning and traffic engineering services with state-of-the-art technologies to drive efficiency in the transportation system and improve mobility for all users. Our multi-disciplinary team of professionals can assess your needs through complex multi-modal planning and microsimulation evaluations and develop traffic signal and intelligent transportation systems, roadway, and intersection capacity enhancements to drive your operations and maintenance needs. We bring value through cost-effective solutions and services that are strategic yet

- practical. Iransportation Planning
- Roadway Design
- Multimodal Transportation
- Traffic Impact Studies
- Traffic Signal Warrant Analysis
- Traffic Signal Design and Timing
- Intelligent Transportation Systems
- Parking Analysis and Design
- Traffic Control Planning
- Roundabout Design and Analysis
- Urban Mobility Planning

- Site Access Studies
- Master Thoroughfare Planning
- Micro and Macro Simulation Modeling
- Traffic Management Plans
- Parking Demand Studies
- Roadway Lighting Design
- Signage and Markings Plans
- Roundabout Design
- Corridor Studies
- Traffic Signal Performance Measures



Infrastructure needs, passenger demand, and advancing technology are some circumstances airports need to accommodate. Our team of aviation professionals has the expertise and experience to solve these circumstances by understanding the need to minimize impacts to the operation of the airport facility and deliver a successful project. We provide a full spectrum of professional services, which include navigating FAA and State agency requirements, planning, and design through construction administration services. Westwood's projects have involved both airside and landside improvements and expansions, many of which have won state and national awards for engineering excellence.

- Airfield Geometric Layout and Design
- Pavement Design
- Landside Improvements
- Airfield Lighting and NAVAIDs
- Fire Rescue Training Facility Design and Development
- Construction Administration
- Hanger Development
- Permitting and Spill Prevention Plans

- Boundary, Approach Obstructions, General Topography Survey
- Environmental Assessments and Impacts Statements
- Stormwater Management and Design
- Landscape Enhancement
- Land Acquisitions
- Net Zero Power Facilities



Westwood Structural Engineering

Our team optimizes structural designs while mitigating risk by using our combined expertise in applicable codes and standards while collaborating with our in-house civil, geotechnical, transportation, water resource, innovation, and aviation teams. Our design experience with steel, reinforced concrete, timber, masonry, unique building materials, and structural sections ensures we deliver the solutions our clients need.

- Steel Design
- Reinforced Concrete Design
- Timber Design
- Equipment and Building Foundation Design
- Foundation Analysis and Reinforcement
- Pile Cap Connection Design
- Anchorage Design
- Corrosion Evaluations
- Retaining and Blast Walls
- Steel and Aluminum Lattice Tower Analysis
- Steel and Concrete Pole Analysis
- Telecommunication Structure Design and Analysis
- Distribution Underground Manhole Design
- Structural Retrofitting and Reinforcement Design
- Substation Steel Structures and Foundations Design
- Owner's Engineer

- Inspection Requirements
- Building Structural Design New or Alterations
- Equipment Supports and Access Platforms
- Material Handling, including Overhead Cranes
- Bridge Superstructure, Substructure, and Foundation Design
- Bridge Widening
- Prestressed/Post-tensioned Concrete Design
- Cut and Cover Structure Design
- Box Culvert Design
- Bridge and Culvert Evaluations
- Bridge and Culvert Rating Determination and Design
- Fall Protection and Walkability Analysis for Transmission Lattice Towerstern (NPDES)
- Spill Pollution Control and Countermeasures (SPCCs)

Westwood Water Resources

Our design philosophy is based on providing a holistic analysis to ensure new problems are not created or shifted somewhere else in the watershed. We find the sweet spot between protecting infrastructure and water quality while complying with permit regulations and staying in budget. As your advocate, we support you through studies, permitting, and development while integrating environmentally friendly solutions.

- Stormwater Master Planning
- Capital Improvement Design
- Funding / Grant Assistance
- Floodplain Analysis and Reclamation
- Stream Bank Stabilization and Restoration
- Permitting Assistance
- Erosion Management
- Stormwater Detention and Retention
- 2D Hydrodynamic Modeling
- Wetland Mitigation and Restoration
- Dam Breach Analysis
- Green Infrastructure Design

- Stormwater Quality Modeling and Design
- Stormwater Facilities Operations and Maintenance Design
- Stormwater Pollution Prevention Plans (SWPPPs)
- Spill Prevention, Control, and Countermeasure (SPCCs) Plans
- FEMA Flood Studies
- Hydrologic Modeling
- Hydrology Studies & Due Diligence Hydrology Review
- Conceptual and Technical Drainage Studies

Westv

Westwood Water Wastewater

Our design philosophy is based on providing a holistic analysis to ensure new problems are not created or shifted somewhere else in the watershed. We find the sweet spot between protecting infrastructure and water quality while complying with permit regulations and staying in budget. As your advocate, we support you through studies, permitting, and development while integrating environmentally friendly solutions.

- Water Network Analysis
- Sewer System Hydraulic Modeling
- Water Master Plan
- Sewer Master Plan
- Sewer Condition Assessment and Correction Design
- Sewer Bypass Plans
- Water Pump Station Design
- Water Storage Design
- Water Wells
- Sewer Lift Stations
- Construction Services
- Schedule Analysis
- Water Quality Modeling and Mitigation Plans
- Feasibility and Conceptual Studies

- Detailed Hydraulic Analyses of Systems
- Large Water Main and Sewer Interceptor Design
- Steel and Concrete Reservoir Design
- Reservoir Rehabilitation Design
- Pressure Reducing and Flow Control Station Design
- Pumping and Lift Station Design
- Low Pressure Sewer System Design
- Treatment Process Studies and Design
- Water and Sewer Network Analyses and Regional Master Planning
- Sewer Assessment Studies (NASSCO) and Rehabilitation Prioritization



Westwood Environmental Engineering

Environmental issues span wetlands, wildlife, plant ecology, cultural resources, visual issues, and more. The wide array of localities and regulatory requirements across the US require a partner with a broad range of environmental experience. If your project needs environmental services in the arid southwest or the complex water systems of the Mississippi River, you've come to the right place.

- Permitting
- Critical Issues Analysis
- Phase I and II Environmental Site Assessment (ESA)
- Waters of the US Support and Delineations
- Wetlands Studies
- Cultural Resource Survey
- Stormwater Pollution Prevention Plans (SWPPP)
- Wildlife Studies
- Threatened and Endangered Species Surveys
- Tree Surveys and Woodland Evaluations
- Bat Surveys
- Avian and Raptor Surveys

- Wind Resource Modeling
- Noise and Sound Studies
- Shadow Flicker
- Glare Studies
- Post-Construction Fatality Monitoring
- FAA Aviation Evaluations
- GIS Data Evaluation and Mapping
- Permit Monitoring and Reporting Services
- National Pollutant Discharge Elimination System (NPDES)
- Spill Pollution Control and Countermeasures (SPCCs)



Without proper geotechnical research, there is greater risk for fatal flaws on a project. By investigating the subsurface conditions through soil borings, cone penetration tests, and other tests, we can provide quality recommendations for the design and construction of project infrastructure.

- Geotechnical Investigations
- Solar Foundation Design
- Structural Design
- Embankments and Retaining Walls
- Crane Pads and Crane Walks

- Road Studies and Design
- Slope Stability Analyses
- Soil Testing and Reports
- Pile Load Testing
- Ground Resistivity Testing

2024-2025 LRIP Overview (Local Road Improvement Program)

LRIP Program General Information

- LRIP provides substantial funding for county, city, village, and town roadway improvements
- Record funding for 2024-2025 biennium
- Reimbursement program

All project costs will be assumed by the municipality or county up front and reimbursement can only be requested once the project is constructed, all contractors paid in full, and all required documentation submitted and approved

- LRIP Project Types
 - Pavement Replacement
 - Reconditioning
 - Resurfacing
 - Reconstruction
 - Structure
 - Purchase HMA Only
- Eligible Costs for Reimbursement
 - Projects with contractor construction

Feasibility studies, design, engineering, all construction costs, ROW, etc.

Purchase HMA Only Project

HMA Only allows for only the purchase of hot-mix asphalt materials to be reimbursed. Costs for any other project-related activities, including the actual paving of the asphalt, may not be reimbursed. Additional documentation and justification may be required for HMA Only projects.

City and Village LRIP Funding Programs

- MSIGT/MSILT (Municipal Street Improvement

 Entitlement) Up to 50% funding match, MSIGT
 for municipalities over 20,000 population, MSILT for
 municipalities under 20,000 population
- **MSID** (Discretionary) For projects over \$250K, up to 50% funding match, over \$8M in funds available
- **MSIS** (Supplemental) For projects over \$250K, up to 90% funding match, over \$25M in funds available
- MSID and MSIS Project Competitive Criteria

Safety, reconstruction, joint venture, multi-jurisdictional projects, high traffic volume, truck traffic, comprehensive planning, economic development, jurisdictional transfer, distinct characteristics or unique circumstances

MSID and MSIS Eligibility

Must meet general guidelines, changes are not permitted so applications must be fully refined, Engineer's certificate required at reimbursement, resurfacing and HMA-only discouraged.

ARIP | Agricultural Road Improvement Program

New this Biennium

- Estimated \$150M available this biennium
- Aimed at addressing agricultural areas and roads with substantial agricultural uses adjacent
- Guidelines, criteria, information, and applications will come out late fall or early next year but municipalities and counties should be ready to capitalize on this opportunity

Timeline

- Application
 - New applications due in LRIP Web and submitted to County Highway Commissioners by November 1st
- County-Level Committee Selection
 - Towns and Municipalities will meet as a committee within their county to select and prioritize projects, likely in October or November
 - Entitlement fund distribution will be voted on by this committee
 - Discretionary and Supplemental project applications will be ranked by this committee for submittal to state-wide competition
 - County Highway Commissioners will review applications submitted by municipalities in their county for completeness and submit to WisDOT by January 15th
- State-Level Selection (Discretionary and Supplemental Only)
 - Discretionary and Supplemental projects will be reviewed, ranked, and ultimately selected based on state-wide competition
- Approval
 - FY 2024 SMA (State Municipal Agreements) will be issued by April 1, 2024
 - FY 2025 SMAs will be issued by July 15th, 2024
- Construction
 - Construction can occur any time after SMA execution but be mindful not to begin any project activities that you want to include for reimbursement before that time
- Reimbursement
 - Construction, close out, contractor payment, and request for reimbursement with all required documentation must be submitted by June 30th, 2029



Be Proactive

Identify opportunity and develop project concept, limits, scope, applicable standards, and other factors before application.

Prepare a quality submittal

- Ensure application meets all requirements
- Check page limits, font size and any required forms particularly on Discretionary and Supplemental applications
- Review criteria to ensure project criteria is addressed for competitive submittal
- Tell the story/explain the need
- Ensure all sections are complete
- Include supporting documentation (but be mindful of the page limit)
- Leverage resources with LRIP expertise, such as Westwood Professional Services

Review and re-review before submitting

• Pay special attention to grammar and spelling

Submit on-time

- Pay attention to due date and time
- Submit application on LRIP Web ahead of due date to avoid possible technical issues
- Check online portal ahead of time to ensure you allow enough time to download extra forms if needed

Be mindful of all requirements once you've secured funding

Refer to the guidelines and requirements document if you have questions





Item 3.

FW: Tree Assistance for your community: WDNR urban/community forestry grants (deadline OCT 2nd)

1 message

 Wahl, Brian D - DNR <Brian.Wahl@wisconsin.gov>
 Wed, Sep 6, 2023 at 2:35 PM

 To: "Fields, Danielle E - MUN"
veclerk@blackearthwisconsin.com>, "Breunig, Nicki - MUN" <clerk_treasurer@vonf.wi.gov>, "publicworks@ridgewaywi.gov" clerk@ridgewaywi.gov>, "streetsandparks@ridgewaywi.gov"

I apologize if you've received this e-mail twice

Subject: WDNR urban/community forestry grants (deadline OCT 2nd)

My name is Brian Wahl and I am the WDNR Urban and Community Forester serving your area.

Your communities all have trees and these trees can be wonderful assets to your communities.

I know the past years have been tough with trees dying from Emerald Ash Borer (EAB), Oak Wilt, drought, storm damage etc. and you either need to **remove and/or replant**.

Maybe you have a park, a ball diamond or a new subdivision that could benefit from planting some new trees -

We can help with that -

One of the ways we can assist you is with our WDNR Urban Forestry Grants – (we have a few extra dollars this year so your chances are even higher that your application will be funded)

Note: There is a lot of info here – feel free to just short circuit all this and give me a call or an e-mail to discuss.

Below I provide a quick summary of our two grants, please take a look and then feel free to give me a call if you think you might be interested.

<u>Please note that a resolution* is required</u> to accompany the application. Here is a sample authorizing resolution: sample/template for authorizing resolution.

Simple enough, but I wanted you to have this since you would likely need to have this on your board/council meeting agenda for September. *If you already had your Sept. meeting, we will still take resolutions through mid October, applications though are a hard deadline of Oct 2nd.

The basics:

WDNR Urban and Community Forestry Grants (<u>Applications Due OCT 2nd</u>): https://dnr.wisconsin.gov/ topic/urbanforests/grants

The WDNR Urban Forestry Grants are **50/50 matching reimbursement** grants and come in two basic flavors.

Start-up grants: https://dnr.wisconsin.gov/topic/urbanforests/grants/startup

-Smaller grants - Max. grant reimbursement is \$5,000 (so total project max \$10,000)

-Geared to new or restarting community forestry programs/projects

-Relatively simple application - really you can have this done in an hour (if you need assistance - reach out)

-Typically less competitive with relatively good chances of application success if project aligns with priorities (<u>especially if</u> <u>you speak with me</u>)

Application and resolution* deadline is Oct. 2nd (grants awarded for 2024)

Regular UF Grants: https://dnr.wisconsin.gov/topic/urbanforests/grants/regular

-Larger – Max grant reimbursement is \$25,000 (so total project max \$50,000)

-Application a bit more in depth

-Competitive - historically and average of about 35% of applicants funded

-Application and resolution* deadline is Oct. 2nd (grants awarded for 2024)

Application guide (also attached): 2024 Wisconsin Urban Forestry Grant Application Guide(PDF)

The applications are sometimes tricky to get to open online (especially if you use CHROME) – I have attached a copy (hopefully it will be more cooperative). If you save the file to your desktop and open directly through Adobe – that seems to work best.

Please note the applications for the start-up and the competitive are the same, **but** the start-up option blocks you from filling out a majority of the application, therefore it is much shorter/simpler.

Other non-match tree grants:

Alliant Energy also has grants, though their "branching out" program – they are often required to be planted in a way that provides energy savings benefit to a building – **Application due NOV 1**st.

https://www.alliantenergy.com/communitysupportandsafety/communityprograms/onemilliontrees/ communitytreeplantingprogram

Again, any questions or concerns etc., please reach out. Don't struggle or debate - just ask.

I look forward to speaking with you and **helping simplify the process as much as possible** – I know you are all pressed for time. If you have any question, just e-mail me or call (it's ok if you haven't read through everything yet (most people don't \bigcirc)).

*we realize some of you have already had your meetings for September, therefore we will take the resolution up until we start to rank grants – so Mid October would be just fine. However, grant applications are a hard deadline of OCTOBER 2ND.

We are committed to service excellence.

Visit our survey at http://dnr.wi.gov/customersurvey to evaluate how I did.

Brian Wahl Urban Forestry Coordinator Wisconsin Department of Natural Resources

3911 Fish Hatchery Rd.

Fitchburg, WI 53711

CELL: 608 225-7943 Fax: (608)-275-3338 Brian.Wahl@wisconsin.gov



3 attachments

UFGrantsApplicationGuide2024.pdf

UFGrantsCombinedResolution (22).doc 50K

8700-298 WDNR Grant application 2024.pdf

Post Office Box 186 • Ridgeway, Wisconsin 53582

The Malcolm Stack Foundation, Inc.

SEP 112023 Village of Ridgeway

September 5, 2023

Ridgeway Nature & Garden Club c/o Mayme Keagy, Treasurer 6025 County Road ZZ Dodgeville, WI 53533

Dear Mayme:

On behalf of the Malcolm Stack Foundation, I am pleased to enclose a check in the amount of \$2,000 to benefit the Ridgeway Nature & Garden Club.

Each year, the Malcolm Stack Foundation provides scholarships to Iowa County youth that are dedicated to studying the sciences. When we have additional funds, we donate to local 501(c)3 organizations like yours that focus on the environment and outdoor education. The Ridgeway Nature & Garden Club came to our attention because of the work you are doing to promote pollinators and healthy soil with native plantings in SW Wisconsin.

Thank you for the work you are doing on behalf of our community. It is recognized and appreciated.

Sincerely,

Anne M. Connor, President The Malcolm Stack Foundation

Cc: Hailey Roessler, Village of Ridgeway



BOARD OF TRUSTEES MEETING MINUTES

August 08, 2023 at 7:00 PM

Ridgeway Community Center - Room 101/102 208 Jarvis Street, Ridgeway, WI 53582

CALL TO ORDER AND ROLL CALL

Meeting called to order by Chair Casper at 7:00 pm.

PRESENT: President Michele Casper, Trustee Cindy Niehaus, Trustee Kellee Venden, Trustee Ruth Nevins, Trustee Julene Garner, Dale Peterson-Director of Public Works, Braden Losby -Streets and Parks Superintendent (arrived 7:48 pm), Hailey Roessler-Clerk/Treasurer

ABSENT: Trustee Steve Vosberg, Trustee Rick Short, Marshal Michael Gorham

CONFIRMATION OF OPEN MEETING

Roessler indicated this was a properly noticed meeting posted on August 7, 2023, at the Ridgeway Community Center, and the Village website with notification sent via email/text to subscribers.

PLEDGE OF ALLEGIANCE

The pledge was recited.

PUBLIC COMMENT

There was no one wishing to speak.

CONSENT AGENDA

Motion by Venden, Seconded by Niehaus, to adopt the consent agenda as presented. Motion carried.

- 1. Minutes to be Adopted: Board of Trustees July 11, 2023, Finance Committee August 4, 2023
- 2. ACH Payments and General Fund Disbursements
- 3. Adoption of Agenda

ANNOUNCEMENTS, DEPARTMENT REPORTS, AND CORRESPONDENCE

Item 4.

4. Announcements

Roessler indicated Delta3 had mailed surveys to the whole village since Main Street serves the whole village but will be tracking Farwell/Kirby as a potential individual site eligible project. Roessler further indicated that the quote from WRS was at their seats and is based on the annual contribution rate, eligible employees, and their salary. Verizon has not gotten back to the office regarding their site contract. Nathan Mattison is working on removing the portable building and it should be out by September 12, 2023. Larry Bierke, Iowa County Administrator visited the village to see the community center, talk about the TID and discuss potential development sites within the village. Alliant Energy can install two solar lights at the dog park, 150 watts, comply with grant requirements at a cost of \$30.28 per month.

5. Department Reports

Received and filed.

6. Correspondence

Received and filed.

ITEMS FOR CONSIDERATION AND ACTION

7. 2023 Village Park Improvements - Katie MacDonald Principal Engineer Parkitecture

Katie MacDonald was present to discuss the scope of work at the park, that included the gravel parking area, ADA access, and ballpark lighting.

Motion by Nevins, Seconded by Venden, to authorize payment of Parkitecture Invoice 2 in the amount of \$7,368. Motion carried.

8. Village Green Improvements - Julene Garner

Trustee Garner presented two landscaping options for the village green space. Garner and Niehaus will continue working on their ideas for the space.

9. Liquor License Application - CCD Holdings/Dee Hunter

Roessler indicated that the applicant was noticed in the Dodgeville Chronicle and has no criminal history prohibiting approval of the license and currently owns and operates a business in Mt. Horeb. Hunter was present to introduce herself and trustees discussed her plans for the business.

Motion by Garner, Seconded by Niehaus, to approve the liquor license for CCD Holidings, Dee Hunter-Agent. Motion carried.

10. 705 Main Street - Phelan Properties - CDI Grant Application

Andy Phelan was present to discuss his application for CDI Grant from WEDC. He wants to renovate the exterior and interior of 705 Main Street. He will be meeting with Premier Builders regarding the scope of work for the application. The CDI grant will match project funding at 50% with the other 50% being private financing. The village would administer the

grant funds and have an agreement with Phelan Properties but would not be spending any money on the project itself. Trustees thanked Phelan for his work and were excited to see improvements in the downtown.

11. Village Assessor Proposals

Motion by Nevins, Seconded by Niehaus, to accept the proposal from Bruce Gardiner. Motion carried.

12. Water Meter Purchase

Motion by Niehaus, Seconded by Venden, to authorize up to \$11,680 for the purchase of meter bases, registers, and transmitters from Midwest Meter. Motion carried.

13. Chemical Pump for Well 1

Motion by Niehaus, Seconded by Garner, to authorize up to \$685 from Water Replacement Fund Cash to purchase a new chemical pump for Well #1. Motion carried.

14. Chapter 10 Section 8 Regulation of Fires and Review of Burn Permit Policy including applicable ordinances and fees

Chapter 10 Section 8 will be discussed in Public Works, Safety, and Health Committee.

15. MHTC Water Tower Lease

Motion by Garner, Seconded by Nevins, to accept the MHTC renewal lease. Motion carried.

16. Village Insurance Renewal

Motion by Venden, Seconded by Niehaus, to authorize village insurance renewal with the League of Wisconsin Municipalities. Motion carried.

17. Hi Point Steakhouse Fire Damage and Response

Trustees discussed if there was anything that the village could do to assist owners or employees in the aftermath of the fire. Trustees will remain open to assisting in whatever way possible.

18. Halloween Event Budget

The boo bash is planned for Sunday, October 29 at the Ridgeway Community Center.

Motion by Niehaus, Seconded by Venden, to authorize up to \$1,000 for treats, entertainment, and decorations for the boo bash. Motion carried.

19. September 16 Home Talent Alumni Baseball Game

Casper discussed the alumni event in Ridgeway. The game will be played at 3:00 pm with potential for music and fireworks at dusk.

Motion by Niehaus, Seconded by Nevins, to authorize up to \$2,000 for entertainment on the 16th. Motion carried.

20. General Fund Budget Amendment

Motion by Nevins, Seconded by Venden, to adopt Resolution 2023-07 General Fund Budget Amendment as presented. Motion carried. 21. Utility Clerk/Deputy Clerk Job Posting and Hiring Process

Motion by Niehaus, Seconded by Nevins, to approve publishing the Utility Clerk/Deputy Clerk job posting with an end date of August 25th. Motion carried.

22. 2022 Audit

2022 Draft Audit Documents were received and filed.

23. Convene in closed session pursuant to State Statute 19.85(1) (c) considering employment promotion, compensation, or performance evaluation data of any public employee over which the government body has jurisdiction.

Motion by Nevins, Seconded by Niehaus, to convene in closed session pursuant to State Statute 19.85(1) (c) considering employment promotion, compensation, or performance evaluation data of any public employee over which the government body has jurisdiction. Motion carried.

ADJOURNMENT

Motion by Garner, Seconded by Venden, to adjourn at 9:50 pm. Motion carried.



FINANCE COMMITTEE MEETING MINUTES

August 04, 2023 at 1:00 PM

Ridgeway Community Center - Room 101/102 208 Jarvis Street, Ridgeway, WI 53582

CALL TO ORDER AND ROLL CALL

Meeting called to order by Chair Casper at 2:05 pm.

PRESENT: Michele Casper, Kellee Venden, Steve Vosberg, Braden Losby -Streets and Parks Superintendent, Dale Peterson-Director of Public Works, Hailey Roessler-Clerk/Treasurer

CONFIRMATION OF OPEN MEETING

Roessler indicated this was a properly noticed meeting and the amended agenda was posted on July 26, 2023, at the Ridgeway Community Center, and the Village website with notification sent via email/text to subscribers.

ADOPTION OF MEETING AGENDA

Motion by Venden, Seconded by Vosberg, to adopt this meeting's agenda. Motion carried.

2024 BUDGET PLANNING

OPERATIONS AND MAINTENANCE FOR THE GENERAL FUND, WATER FUND, SEWER FUND, PUBLIC PROPERTY AND EVENTS FUND

Dale Peterson and Braden Losby discussed capital improvement planning for public works. Trustees advised staff members what estimates they would like to see. Marshal Gorham made edits to the proposed budget, made requests for 2024, and suggestions for capital improvement planning. Roessler mentioned that the employee cell phone policy should be reviewed as the Deputy Clerk and Streets and Parks position are both utilizing personal cell phones for work frequently. Costs of current cell phone plans were discussed.

ITEMS FOR CONSIDERATION AND POSSIBLE ACTION

1. Convene in closed session pursuant to State Statute 19.85(1) (c) considering employment promotion, compensation, or performance evaluation data of any public employee over which the government body has jurisdiction.

Motion by Venden, Seconded by Vosberg, to convene in closed session pursuant to State Statute 19.85(1) (c) considering employment promotion, compensation, or performance

evaluation data of any public employee over which the government body has jurisdiction. Motion carried at 4:15 pm.

ADJOURNMENT

Motion by Venden, Seconded by Vosberg, to adjourn at 5:35 pm. Motion carried.



FINANCE COMMITTEE MEETING MINUTES

August 11, 2023 at 1:00 PM

Ridgeway Community Center - Room 101/102 208 Jarvis Street, Ridgeway, WI 53582

CALL TO ORDER AND ROLL CALL

PRESENT: Michele Casper, Kellee Venden, Steve Vosberg

ADOPTION OF MEETING AGENDA

Motion by Venden, Seconded by Vosberg, to adopt the meeting agenda. Motion carried.

CONVENE IN CLOSED SESSION FOR EMPLOYEE REVIEWS

1. Convene in closed session pursuant to State Statute 19.85(1) (c) considering employment promotion, compensation, or performance evaluation data of any public employee over which the government body has jurisdiction:

1:00 pm to 1:30 pm Braden Losby

1:30 pm to 2:00 pm Dale Peterson

2:00 pm to 2:30 pm Michael Gorham

Motion by Venden, Seconded by Vosberg, to convene in closed session pursuant to State Statute 19.85(1) (c) considering employment promotion, compensation, or performance evaluation data of any public employee over which the government body has jurisdiction. Motion carried.

ADJOURNMENT

Motion by Venden, Seconded by Vosberg, to adjourn at 4:05 pm. Motion carried.

Item 4.



FINANCE COMMITTEE MEETING MINUTES

September 07, 2023 at 7:00 PM

Ridgeway Community Center - Room 101/102 208 Jarvis Street, Ridgeway, WI 53582

CALL TO ORDER AND ROLL CALL

Meeting commenced at 7:00 pm.

PRESENT: Michele Casper, Kellee Venden, Steve Vosberg, Hailey Roessler-Clerk/Treasurer

CONFIRMATION OF OPEN MEETING

The meeting was properly posted.

CONVENE IN CLOSED SESSION

 Convene in closed session pursuant to State Statute 19.85(1) (c) considering employment promotion, compensation, or performance evaluation data of any public employee over which the government body has jurisdiction: Utility Clerk/Deputy Clerk Interviews

Motion by Vosberg, Seconded by Venden, to convene in closed session pursuant to State Statute 19.85(1) (c) considering employment promotion, compensation, or performance evaluation data of any public employee over which the government body has jurisdiction: Utility Clerk/Deputy Clerk Interviews

ADJOURNMENT

Motion by Venden, Seconded by Vosberg, to adjourn at 8:50 pm. Motion carried.

9/07/2023	4:49 PM	Reprint Check	Register - F	ull Report	- Manual	Page: 1 ACCT
1-POO	LED CHECKING A	ACCOUNT **0307			Accounting Checks	
Pos	ted From:	8/01/2023	From Account	::		
	Thru:	8/31/2023	Thru Account	::		
Check Nbr	Check Date	Payee				Amount
184408	8/08/2023	HOLIDAY WHOLE	SALE			
ACC 1-POOLED CHECKING ACCOUNT **0307 Posted From: 8/01/2023 From Account: Thru: 8/31/2023 Thru Account: Check Nbr Check Date Payee 184408 8/08/2023 HOLIDAY WHOLESALE Inv1460577 dated 07.10.2023 150-00-55200-000-400 CONCESSION STAND INVENTORY EXP Inv1460577 dated 07.10.2023 Total Aug MO 8/31/2023 FIRSTNET - AT&T MOBILITY SmartPhone & Hotspot Service Manual Check 100-00-52100-325-000 POLICE - TELEPHONE SmartPhone & Hotspot Service Total Aug W6 8/30/2023 WISCONSIN DEPT. OF REVENUE Aug 2023 Payroll Tax 100-00-21513-000-000 STATE W/H TAXES PAYABLE Aug 2023 Payroll Tax						
			INVENTORY EXP			-80.48
11171-	auted o	7.10.2025			Total	-80.48
<u> </u>						-00.40
-			&T MOBILITY			
	_	Service			Manual Check	
			E			88.77
Silar	trione a notap	Of Service			Total	88.77
						00.77
-		WISCONSIN DEF	T. OF REVENUE			
-	-				Manual Check	
			AYABLE			1,334.86
					Total	1,334.86
						1,001.00
-		PRINCIPAL LIF	E INSURANCE C	OMPANY		
					Manual Check	
		EMPLOYEE BENEFITS				44.04
		EMPLOYEE BENEFITS				44.03
						11.00
100-00-52100-	125-000	POLICE - EMPLOYEE	BENEFITS			49.33
Micha	ael G					
			NEFITS			76.06
						12.68
						12.68
100-00-53311-	125-000	STREETS - EMPLOYE	E BENEFITS			27.26
Brade	en L					
					Total	266.08
Aug ACH	8/11/2023	FARMERS SAVIN	GS BANK			
-					Manual Check	
210-00-58290-	000-000	TIF INTEREST & FI	SCAL CHARGES			672.68

Cardinal Way Phase 2 - TID loan x5570

9/07/2023	4:49 PM	Reprint Check Register - Full Report	- Manual	Page: 2 ACCT
1-P00	LED CHECKING	ACCOUNT **0307	Accounting Checks	
Pos	ted From:	8/01/2023 From Account:		
	Thru:	8/31/2023 Thru Account:		
Check Nbr	Check Dat	e Payee		Amount
			Total	672.68
Aug ACH	8/14/202	3 SPECTRUM BUSINESS		
Charter	ACH 50%		Manual Check	
100-00-51980- Chart	760-000 Cer ACH 50%	FACILITIES UTILIITIES		57.50
100-00-51600-		VILLAGE HALL UTILITIES		57.49
Chart	cer ACH 50%			
			Total	114.99
Aug ACH	8/21/202	3 ALLIANT ENERGY		
95834200	000		Manual Check	
100-00-55200- 95834	765-000 120000	PARK - LIGHTS		222.93
-300-00-53610 43949	000-821 940000, 77246	OPERATION EXPENSES-WWTP		1,510.80
300-00-53610-		UTILITIES-LIFT STATIONS&SHOP		75.57
400-00-53700- 67282	000-620 200000, 88121			768.04
400-00-53610- 38077	000-823 720000	UTILITIES-TOWER&SHOP		44.44
100-00-53311- 04872	760-000 210000, 03996	STREETS - UTILITIES 50000		108.16
100-00-53420- 68503		STREET (HWY) LIGHTING		998.47
100-00-51980- 19722	760-000 296511	FACILITIES UTILIITIES		513.46
100-00-51600- 19722	100-000 296511	VILLAGE HALL UTILITIES		37.12
100-00-52100- 19722	760-000 296511	POLICE - UTILITIES		68.05
			Total	4,347.04
AUG ACH AUG ACH		3 FARMERS SAVINGS BANK	Manual Check	
100-00-51500- AUG #	220-000 ACH Fees	BANK & PAYROLL PROCESSING FEES		30.00
	-		Total	30.00

9/07/2023 4:49 PM	A Reprint Check Register - Full Repo	ort - Manual	Page: 3 ACCT
1-POOLED CHECK	ING ACCOUNT **0307	Accounting Checks	
Posted From:	8/01/2023 From Account:		
Thru	: 8/31/2023 Thru Account:		
Check Nbr Check	Date Payee		Amount
July MO 8/01/	2023 FIRSTNET - AT&T MOBILITY		
SmartPhone & Hots	spot Service	Manual Check	
100-00-52100-325-000	POLICE - TELEPHONE		88.77
SmartPhone & H	Notspot Service		
		Total	88.77
8 16 941 8/16/	2023 INTERNAL REVENUE SERVICE		
08.16.2023 SS Tax		Manual Check	
100-00-21511-000-000	941 TAXES PAYABLE		1,284.70
08.16.2023 SS			_,
100-00-21511-000-000	941 TAXES PAYABLE		300.42
08.16.2023 Med	licare		
100-00-21511-000-000	941 TAXES PAYABLE		782.43
08.16.2023 Fed	l Tax Withholding		
		Total	2,367.55
	2023 INTERNAL REVENUE SERVICE	<u> </u>	
08.30.2023 SS Tax		Manual Check	
100-00-21511-000-000	941 TAXES PAYABLE		1,299.64
08.30.2023 SS			1,255.01
100-00-21511-000-000	941 TAXES PAYABLE		303.92
08.30.2023 Med	licare		
100-00-21511-000-000	941 TAXES PAYABLE		837.78
08.30.2023 Fed	l Tax Withholding		
		Total	2,441.34
July ACH 8/08/	2023 MADISON GAS & ELECTRIC CO.		
206 Kirby St.		Manual Check	
300-00-53610-000-823	UTILITIES-LIFT STATIONS&SHOP		6.60
206 Kirby St.			
400-00-53610-000-823	UTILITIES-TOWER&SHOP		6.60
206 Kirby St.			
100-00-53311-760-000	STREETS - UTILITIES		13.20
206 Kirby St.			
100-00-51600-100-000	VILLAGE HALL UTILITIES		2.94
208 Jarvis St			
100-00-51980-760-000	FACILITIES UTILIITIES		40.65
208 Jarvis St			

9/07/2023	4:49 PM	Reprint Check Reg	ister - Full Report	- Manual	Page: 4 ACCT
1-P00I	LED CHECKING	ACCOUNT **0307		Accounting Checks	
Post	ed From:	8/01/2023 Fro	om Account:		
	Thru:	8/31/2023 Thr	ru Account:		
Check Nbr	Check Dat	e Payee			Amount
100-00-52100-7	760-000	POLICE - UTILITIES			5.39
208 J	arvis St				
				Total	75.38
July ACH	8/18/202	3 ASCENTIS CORPORAT	ION		
July ACH				Manual Check	
100-00-51500-2	240-000	SOFTWARE SUBSCRIPTION	S & FEES		36.65
July	ACH				
				Total	36.65
WWTP ACH	8/10/202	3 FRONTIER COMMUNIC	ATIONS		
WWTP pho				Manual Check	
300-00-53610-0	000-821	OPERATION EXPENSES-WW	TP		95.43
WWTP	phone line				
				Total	95.43
08.02 941	8/02/202	3 INTERNAL REVENUE	SERVICE		
	23 SS Tax			Manual Check	
100-00-21511-0 08.02	000-000 .2023 SS Tax	941 TAXES PAYABLE			1,359.68
100-00-21511-0 08.02	000-000 .2023 Medica	941 TAXES PAYABLE			317.98
100-00-21511-0	000-000	941 TAXES PAYABLE			894.40
08.02	.2023 Fed Ta	x Withholding			
				Total	2,572.06
July WWTP	8/01/202	3 FIRSTNET - AT&T M	OBILITY		
Mobile I	internet Serv	<i>r</i> ice		Manual Check	
300-00-53612-0	000-852	CONTRACTED SERVICES			16.75
Mobil	e Internet S	Service			
		CONTRACTED SERVICES			16.74
Mobil	e Internet S	ervice			
				Total	33.49
Office ACH	8/10/202	3 FRONTIER COMMUNIC	ATIONS		
Office T	wo Phone lir	nes		Manual Check	
		CLERK TELEPHONE			152.65
Offic	e Two Phone	lines			4
				Total	152.65

9/07/2023	4:49 PM	Reprint Check	Register - Full	Report	- Ma	nual	Page: ACCT	5
1-POOI	LED CHECKING A	ACCOUNT **0307			i	Accounting Checks		
Post	ed From:	8/01/2023	From Account:					
	Thru:	8/31/2023	Thru Account:					
Check Nbr	Check Date	Payee					Amount	
					G	Grand Total	14,637.2	26

9/07/2023	3 4:49 PI	M Repr	int Check Regi	ister - Full	. Report -	Manual		Page: ACCT	6
1-	-POOLED CHECK	ING ACCOUNT *	*0307			Accounting	Checks		
	Posted From:	8/01/20	3 From	m Account:					
	Thru	: 8/31/20	23 Thru	a Account:					
								Amount	
Total	Expenditure :	from Fund # 1	0 - GENERAL F	UND				11,390.0	56
Total	Expenditure :	from Fund # 1	0 - PUBLIC PR	OPERTY AND I	EVENTS			-80.	48
Total	Expenditure :	from Fund # 2	0 - TIF FUND					672.	68
Total	Expenditure :	from Fund # 3	0 - SEWER FUN	ID				1,761.8	87
Total	Expenditure :	from Fund # 4	0 - WATER FUN	ID				892.	53
			Тс	otal Expendi	ture from	all Funds		14,637.2	26

8/30/2023	2:43 PM	Check Reg	ister - Full Repor ALL Checks CHASE VISA CARI		Page : ACCT	1
Da	ated From:	:	From Account:			
	Thru:		Thru Account:			
Check Nbr	Check Date	Payee			Amour	nt
STORE 1/3 POS	E 8/07/2023 TAGE FOR UTILITY	RIDGEWAY POST (BILLS	DFFICE	Manual Check		
100-00-51420- 1/3	-310-000 CLI POSTAGE FOR UTILI	ERK OFFICE SUPPLI ITY BILLS	IES		6	6.00
300-00-53612- 1/3	-000-840 BI POSTAGE FOR UTIL	LLING & ACCOUNTIN ITY BILLS	1G		6	6.00
400-00-53612- 1/3	-000-840 BI POSTAGE FOR UTIL	LLING & ACCOUNTIN ITY BILLS	1G		6	6.00
				Total	19	8.00
STORE	z 7/29/2023	WAL-MART				
	AIRHEADS, SODA			Manual Check		
150-00-55200- FRIT	-000-400 CON OS, AIRHEADS, SOI	NCESSION STAND IN DA	WENTORY EXP		9	4.35
				Total	9	4.35
STORE CHAINS ,		FARM & FLEET		Manual Check		
100-00-55200- CHAI	-745-000 PA	RK - SUPPLIES			-8	0.15
100-00-55200- CHAI	-745-000 PAI NS, OIL, NO TAX	RK - SUPPLIES			7	5.97
				Total	-	4.18
STORE OIL, BU		FARM & FLEET		Manual Check		
100-00-55200- OIL,	-745-000 PA BULK HEX	RK - SUPPLIES			3	3.88
				Total	3	3.88
STORE	E 8/02/2023 5, COTTON SWABS	FARM & FLEET		Manual Check		
300-00-16110- STAP	-000-150 MA	TERIALS & SUPPLIE S	ES INVENTORY		2	2.48
				Total	2	2.48
STORE WD40, E	E 8/16/2023 LBOS, CONNECTORS	MT HOREB LUMBEN , BUSHINGS	R DO IT BEST	Manual Check		
400-00-16110- WD40	-000-154 MA	TERIALS & SUPPLIE DRS, BUSHINGS	ES INVENTORY		2	4.75

8/30/2023	2:43 PM	Check Register - Full Report - ALL Checks	Manual	Page: ACCT	2
_		CHASE VISA CARD			
Da	ated From: Thru:	From Account: Thru Account:			
Check Nbr		Payee		Amour	. +
		rayee		Alloui	10
			Total	2	4.75
STORE TRIP LE	E 8/16/2023 EVER, WEATHERPROOF	MT HOREB LUMBER DO IT BEST GFCI	Manual Check		
	-000-833 MAI P LEVER, WEATHERPR	NT OF TREATMENT SYSTEM OOF GFCI		162	2.41
			Total	162	2.41
STORE	E 8/21/2023	MT HOREB LUMBER DO IT BEST			
BUSHING	GS, CONNECTORS		Manual Check		
	-000-154 MAT HINGS, CONNECTORS	ERIALS & SUPPLIES INVENTORY		1:	2.62
			Total	1:	2.62
STORE	E 8/24/2023 CIAL LEARNER'S PER	WISCONSIN DEPT OF TRANSPORTATION-DMV MIT	Manual Check		
100-00-53311- COMM	-330-000 STR MERCIAL LEARNER'S	EETS - TRAINING & EDUCATION PERMIT		30	0.60
			Total	3	0.60
STORE	E 8/09/2023 CAMP, PD FOR BY M		Manual Check		
150-00-55500- PRIZ	-000-000 EVE SES FOR ATTENDANCE	NT EXPENSES		4	4.20
			Total	4	4.20
STORE	E 8/16/2023 CAMP/MSF GRANT	TARGET	Manual Check		
150-00-55500- BAIL		NT EXPENSES ISSORS, MONDO LLAMA		6	5.46
			Total	6	5.46
STORE	E 8/05/2023	WAL-MART			
DON'T F	KNOW, RECEIPT WASN	'T CODED	Manual Check		
100-00-52100- DON'	-315-000 POL T KNOW, RECEIPT W	ICE - MISC SUPPLIES ASN'T CODED		1	7.88
DOM			Total	1	7.88

STORE 8/26/2023 COSTCO

Manual Check
8/30/2023	2:43 PM	-	- Full Report - ALL Checks HASE VISA CARD	Manual		Page : ACCT	3
Da	ated From:		Account:				
	Thru:		Account:				
Check Nbr	Check Date	Payee				Amou	int
150-00-55200	000 400	-				1.	97.80
	-000-400 CC 1, Chips	NCESSION STAND INVENT	JKI EAP			1	97.80
	-				Iotal	1	97.80
STOR	E 8/09/2023	FARM & FLEET					
				Manual Che	eck		
300-00-53611 Weld	-000-833 MA 1, bonder, compou	INT OF TREATMENT SYST	EM				46.59
	-,,			i	Total		46.59
							10.05
STOR	E 8/22/2023	MENARDS					
				Manual Che	eck		~~ ~1
300-00-53611. svri	-000-833 MA .nge, gorilla wel	INT OF TREATMENT SYST d, blades	EM				32.21
- 100-00-55200·		RK - MATERIALS				1	52.22
	treated 12'						
					Total	18	84.43
ONLINE	z 8/11/2023	ADOBE					
Adobe I	Pro Subscription	(2 Devices) 23-24		Manual Che	eck		
100-00-52100		LICE - COMPUTER/SOFTWA	ARE			2	39.88
Adob	e Pro Subscripti	on (2 Devices) 23-24			Iotal	2	39.88
<u></u>							
ONLINE	E 8/02/2023 DIFIER FOR OFFICE	AMAZON		Manual Che	ack		
				Manual Circ	SCR		20.00
100-00-51420. DEHU	MIDIFIER FOR OFF	ERK MISCELLANEOUS ICE					39.09
					Total		39.09
ONLINE	E 8/17/2023	AMAZON					
TOILET				Manual Che	eck		
100-00-51980· TOTL	-763-000 FA .ET PAPER	CILITIES MATERIALS					63.49
1011					Total		63.49
ONLINE	E 8/05/2023	7 M7 7 ON					
	asher attachment	AMAZON s for shelter		Manual Che	eck		
- 100-00-55200·		RK - MATERIALS				1	16.87
	er washer attachm						

power washer attachments for shelter

8/30/2023	2:43 PM	Check Register - Full Report - Manual	Page: 4
		ALL Checks	ACCT
		CHASE VISA CARD	
Da	ted From:	From Account:	
	Thru:	Thru Account:	
Check Nbr	Check Date	Payee	Amount
		Total	116.87
		Grand Total	1,590.60

8/30/2023	2:43 P	м		Ch	eck Reg	gi:	ster - Full Report - Manual	Pa	age:	5
							ALL Checks	AC	CCT	
							CHASE VISA CARD			
	Dated From:	:				Fı	rom Account:			
	Thru	1:				Тł	hru Account:			
									Amount	
Total E	xpenditure	from Fur	ıd #	100 -	GENER	۱AS	- FUND		755.	73
Total E	xpenditure	from Fur	ıd #	150 -	PUBLI	c	PROPERTY AND EVENTS		401.	81
Total E	xpenditure	from Fur	nd #	300 -	SEWER	٤F	FUND		329.	69
Total E	xpenditure	from Fur	nd #	400 -	WATER	łΕ	FUND		103.	37
							Total Expenditure from all Funds		1,590.0	50

9/12/2023	4:16 PM		Register - Full Report - ALL ALL Checks	Page: 1 ACCT
		1-900	OLED CHECKING ACCOUNT **0307	
Dat	ted From: Thru:		From Account: Thru Account:	
Check Nbr	Check Date	Payee	inita necount.	Amount
		_		
184434 Inv80137	9/12/2023 796 dated 08.3			
400-00-53612-0		BILLING & ACCOUNT	ING	98.25
July	2023			
				Total 98.25
184435	9/12/2023	BAER INSURANC	E SERVICES, INC	
100-00-51938-0	000-000	GENERAL GOV'T INS	URANCE	154.67
300-00-53612-0	000-853	INSURANCE		154.66
400-00-53710-0	000-684	INSURANCE		154.67
				Total 464.00
	0 /10 /2022			
184435	9/12/2023	BAER INSURANC.	E SERVICES, INC Manual Ch	neck
100-00-51938-0	000-000	GENERAL GOV'T INS	URANCE	-154.67
300-00-53612-0	000-853	INSURANCE		-154.66
400-00-53710-0	000-684	INSURANCE		-154.67
				Total -464.00
184436	9/12/2023	CHASE CARD SE	RVICES	
100-00-21800-0	000-000	CREDIT CARD PAYAB	LE	755.73
150-00-21800-0	000-000	CREDIT CARD PAYAB	LE	401.81
300-00-21800-0	000-000	CREDIT CARD PAYAB	LE	329.69
400-00-21800-0	000-000	CREDIT CARD PAYAB	LE	103.37
				Total 1,590.60
184437 Tpy17962	9/12/2023 8 dated 08.16		ES	
300-00-53612-0		CONTRACTED SERVIC	ES	3,078.00

9/12/2023	4:16 PM	Check Register - Full Report - ALL ALL Checks 1-POOLED CHECKING ACCOUNT **0307		Page: ACCT	2
Da	ated From:	From Account:			
	Thru:	Thru Account:			
Check Nbr	Check Date	Рауее		Amoun	t
			Total	3,078	.00
184438 40# Sol	9/12/2023 ar Salt - Communi	CULLIGAN TOTAL WATER TREATMENT			
100-00-51980- 40#	-760-000 FAG Solar Salt - Comm	CILITIES UTILIITIES munity Center		12	2.89
100-00-51600- Wate	-100-000 VII r Service	LLAGE HALL UTILITIES		19	9.12
100-00-52100-	-315-000 POI	LICE - MISC SUPPLIES		(0.00
			Total	32	2.01
184439 Septemb	9/12/2023 Der 2023 - M Gorha	DEAN HEALTH PLAN am			
100-00-21530- Sept	-000-000 HEA ember 2023 - M Go	ALTH & DENTAL INS PAYABLE orham		1,123	. 52
100-00-21530- Sept	-000-000 HEA ember 2023 - H Rc	ALTH & DENTAL INS PAYABLE Dessler		883	3.21
100-00-21530- Sept	-000-000 HEA ember 2023 - M Jo	ALTH & DENTAL INS PAYABLE		-493	8.41
300-00-53612-		PLOYEE BENEFITS		920	0.46
400-00-53710-		PLOYEE BENEFITS		920).47
			Total	3,354	.25
184440 D21-009	9/12/2023 WWTF Permit Comp	DELTA 3 ENGINEERING, INC. pliance Inv20683			
-2300-00-53612 D21-		NTRACTED SERVICES compliance Inv20683		240	0.00
140-00-57331- D23-	-000-000 HIC 032 MAIN STREET I	SHWAY & STREET OUTLAY MPVMT. Inv20684		512	2.00
			Total	752	2.00
	9/12/2023 2023 - HR, MJ	DELTA DENTAL OF WISCONSIN			
100-00-21530- Octo	-000-000 HEA ber 2023 - HR, MJ	ALTH & DENTAL INS PAYABLE		76	5.82
	-000-854 EMI ber 2023 - DP	PLOYEE BENEFITS		51	L.86

9/12/2023	4:16 PM	Check Regi	ster - Full Report - ALL ALL Checks		Page: 3 ACCT
		1-POOLED	CHECKING ACCOUNT **0307		ACCI
Dat	ed From:	From	n Account:		
	Thru:		Account:		
Check Nbr	Check Date	Payee			Amount
400-00-53710-0 Octobe	000-686 EMP er 2023 - DP	LOYEE BENEFITS			51.86
				Total	180.54
184442 Advertis		DODGEVILLE CHRONIC tdated 8.31.2023	LE		
100-00-51980-0 DepCll	000-000 OTH k Job Posting 2	ER GENERAL GOV'T wks			190.80
				Total	190.80
184443 Oct 1 20	9/12/2023 23 Music Enterta	DON GREENWOOD			
150-00-59000-0 Oct 1	000-000 FAR 2023 Music Ente	MER'S MARKET EXPENS rtainment	E		100.00
				Total	100.00
184444	9/12/2023	DRS ENTERPRISES, I	TC		
100-00-53311-7	730-000 STR	EETS - FUEL			270.57
100-00-52100-4	10-000 POL	ICE - FUEL			292.02
100-00-55200-7	/30-000 PAR	K - FUEL			394.15
300-00-53610-0	000-822 FUE	L-AUTO			45.25
400-00-53610-0	000-822 FUE	L-AUTO			45.25
300-00-53611-0	000-833 MAI	NT OF TREATMENT SYS	тем		0.00
				Total	1,047.24
184445 Septembe	9/12/2023 r 2023 MG, HR, D	EDWARD D. JONES P, BL			
- 100-00-21520-0	000-000 RET	IREMENT PAYABLE			400.00
Septer	mber 2023 MG, HR	, DP, BL			

184446 9/12/2023 FABICK, CAT InvSIMS0059094 dated 8.31.23

9/12/2023	4:16 PM	Check Register - Full Report - ALL ALL Checks 1-POOLED CHECKING ACCOUNT **0307		Page: 4 ACCT
Da	ated From:	From Account:		
	Thru:	Thru Account:		
Check Nbr	Check Date	Рауее		Amount
300-00-53610- Apvd	-000-821 1 07.11.23 gene	OPERATION EXPENSES-WWTP rator svc		1,167.01
300-00-53610-	-000-827	OTHER SUPPLIES & EXPENSES		149.00
Wasp	o removal from	enclosure, radiator, c		
300-00-53611-		MAINT OF TREATMENT SYSTEM		232.60
Heat	er, hosess, ba	rbs, labor		
			Total	1,548.61
184447 Aug 202		FAHERTY, INC.		
100-00-53635 Aug	-000-000 2023	RECYCLING COLLECTION		1,592.85
100-00-53620 Aug		GARBAGE COLLECTION		2,479.68
			Total	4,072.53
184448 INV6713	9/12/2023 35 DATED 08.23.			
100-00-53311- SAW	-722-000 REPAIRS	STREETS - EQUIP REPAIR/MAINT		93.49
			Total	93.49
184449 INVOICE	9/12/2023 E P25306	KELBE BROTHERS EQUIPMENT		
100-00-53311- STRE	-722-000 Et sweeper att	STREETS - EQUIP REPAIR/MAINT ACHMENTS		350.66
			Total	350.66
184450 Inv2549) 9/12/2023 92 dated 09.01.	LV LABS WATER, LLC 2023		
	-000-682 5492 dated 09.	CONTRACTED SERVICES 01.2023		125.00
			Total	125.00
184451 Inv1646	9/12/2023 5 dated 08.07.2			
	-000-852 .646 dated 08.0	CONTRACTED SERVICES 7.2023		879.50
			Total	879.50

9/12/2023	4:16 PM	Check Register - Full Report - ALL ALL Checks 1-POOLED CHECKING ACCOUNT **0307		Page: 5 ACCT
Da	ted From:	From Account:		
Da	Thru:	Thru Account:		
Check Nbr	Check Date	Payee		Amount
184452 Inv0158	9/12/2023 600-81 dated 08.	MIDWEST METER, INC. 17.2023		
400-00-18200- Inv01	000-320 CC 158600-81 dated	NST IN PROGRESS-2021 TOWER 08.17.2023		11,470.00
			Total	11,470.00
184453 BL, 5 T	9/12/2023 s; Sweatshirt; 2			
100-00-53311- BL, S	755-000 SI 5 Ts; Sweatshirt	REETS - UNIFORMS ; 25% setup fee		157.50
400-00-53311- DP, 5	000-852 UN 5 Ts; Sweatshirt	IFORMS ; 25% setup fee		97.50
300-00-53311- DP, 5	000-852 UN 5 Ts; Sweatshirt	IIFORMS ; 25% setup fee		97.50
100-00-55140- нј, 3	120-000 FA 3 Ts; 25% setup	CILITIES - PAYROLL TAXES		73.50
100-00-51420- T, Sv	380-000 CI weatshirt	ERK MISCELLANEOUS		72.00
			Total	498.00
184454 INVOICE	9/12/2023 325335 DATED 09	MULCAHY SHAW WATER 0.01.2023		
300-00-53611- LAMP	000-833 MA SLEEVE KIT	INT OF TREATMENT SYSTEM		357.50
			Total	357.50
184455 Account	9/12/2023 40000283 Policy	MUNICIPAL PROPERTY INSURANCE FUND 5000160		
100-00-51938- 1/3 1		NERAL GOV'T INSURANCE 15.2023-08.15.2024		4,388.66
	000-853 IN ACT#40000283 08.	ISURANCE 15.2023-08.15.2024		4,388.67
	000-684 IN ACT#40000283 08.	ISURANCE 15.2023-08.15.2024		4,388.67
			Total	13,166.00
	9/12/2023 r 93057 8.31.23	PERFORMANCE FOODSERVICE statement		
	000-400 CC ice 493044 dated	NCESSION STAND INVENTORY EXP 08.03.2023		899.47

9/12/2023	4:16 PM	Check	k Register - Full Report - ALL ALL Checks		Page: 6 ACCT
		1-P0	OOLED CHECKING ACCOUNT **0307		ACCI
Da	ated From:		From Account:		
	Thru:		Thru Account:		
Check Nbr	Check Date	e Payee			Amount
150-00-55200- Inv4	-000-400 78443 dated 0	CONCESSION STAND 7.13.2023	INVENTORY EXP		281.44
150-00-55200- 6.30	-000-400 .2023 credit	CONCESSION STAND	INVENTORY EXP		-6.68
	-000-400 1.2023 credit	CONCESSION STAND	INVENTORY EXP		-19.10
				Total	1,155.13
184457 INV#263		3 PIONEER RESE 31.2023 CUST 1995			
	-000-833 OLV GALS	MAINT OF TREATMEN	NT SYSTEM		367.31
				Total	367.31
184458 Inv5072	9/12/202 4 oil, tie ro	3 RANDY'S SERV d, labor	ICE & TOWING		
	-400-000 0724 oil, tie	POLICE - VEHICLE rod, labor	EXPENSE		283.56
				Total	283.56
184459 1/3 206		3 RIDGEWAY UTI	LITIES		
100-00-53311- 1/3	-760-000 206 Kirby	STREETS - UTILIT	IES		24.23
300-00-53610- 1/3	-000-823 206 Kirby	UTILITIES-LIFT S	TATIONS & SHOP		24.23
400-00-53610- 1/3	-000-823 206 Kirby	UTILITIES-TOWER&	SHOP		24.22
100-00-55200- 299	-760-000 Hughett St.	PARK - UTILITIES			78.81
100-00-52100- 208	-760-000 Jarvis 10%	POLICE - UTILITI	ES		12.20
100-00-51980- 208	-760-000 Jarvis 75%	FACILITIES UTILI	ITIES		91.53
100-00-51600- 208	-100-000 Jarvis 15%	VILLAGE HALL UT	ILITIES		18.30
				Total	273.52

 184460
 9/12/2023
 SHEKINAH
 KING

 Oct
 15
 2023
 MUSIC
 ENTERTAINMENT

9/12/2023	4:16 PM	Check Register - Full Report - ALL ALL Checks		Page: 7 ACCT
		1-POOLED CHECKING ACCOUNT **0307		ACCI
Da	ated From:	From Account:		
	Thru:	Thru Account:		
Check Nbr	Check Date	Payee		Amount
150-00-59000- Oct	-000-000 FAF 15 2023 MUSIC ENT	RMER'S MARKET EXPENSE ERTAINMENT		100.00
			Total	100.00
184461	9/12/2023 3 FENNIMORE TRAIN			
-300-00-53710 BRAD	-000-689 TRA EN LOSBY, DALE PE	LINING & EDUCATION TERSON		40.00
			Total	40.00
	9/12/2023 810 dated 08.11.2	·		
	-000-000 LEG 284810 dated 08.1	AL EXPENSE 1.2023 Burn Ordinan		299.00
			Total	299.00
184463 Big Sky	9/12/2023 Band Music 09.16			
150-00-55500- Big		NT EXPENSES .16.23 Alumni Game		450.00
			Total	450.00
184464 Inv1118	9/12/2023 8 dated 08.05.202			
100-00-52100- Blau	-431-000 POI er Collared Shirt	ICE - UNIFORMS		58.49
100-00-52100- Earp		LICE - MISC SUPPLIES		74.99
			Total	133.48
184465 Inv2204	9/12/2023 date 08.14.23	UPLAND FIRE & SECURITY LLC		
100-00-51980- Fire	-762-000 FAC Alarm Service -	CILITIES MAINTENANCE portable removal		165.00
			Total	165.00
184466	9/12/2023	US CELLULAR		
100-00-51420-	-325-000 CLE	RK TELEPHONE		35.32

9/12/2023	4:16 PM	Check Register - Full Report - ALL ALL Checks		Page: ACCT	8
		1-POOLED CHECKING ACCOUNT **0307		ACCI	
Da	ated From:	From Account:			
	Thru:	Thru Account:			
Check Nbr	Check Date	Payee		Amour	nt
300-00-53610-	-000-823 UTIL	ITIES-LIFT STATIONS&SHOP		4	13.83
400-00-53610-	-000-823 UTIL	ITIES-TOWER&SHOP		2	23.59
			Total	10:	2.74
184467 Inv0009	9/12/2023 9587 dated 08.09.2	USA BLUE BOOK			
300-00-53611- dipp	-000-833 MAIN ercup	IT OF TREATMENT SYSTEM		12	8.95
400-00-53700- sten	-000-650 REPA ner pump, well#1 aj	IRS & MAINTENANCE pvd 08.08.23		684	4.95
300-00-53611- Orde	-000-833 MAIN rS03121921 dated 0	T OF TREATMENT SYSTEM 9.05.23		36	5.97
			Total	1,179	9.87
184468 Inv4718	9/12/2023 3488 dated 08.14.23	WIL-KIL PEST CONTROL			
300-00-53612- Inv4	-000-852 CONT 718488 dated 08.14	RACTED SERVICES .23		9	98.55
			Total	٥	
184469				9	98.55
CONSTRU	9/12/2023 JCTION DOCUMENTS	PARKITECTURE+PLANNING		9	8.55
140-00-57620-	JCTION DOCUMENTS	S OUTLAY		7,368	
140-00-57620-	OCTION DOCUMENTS	IS OUTLAY	Total		8.00
140-00-57620- CONS 	JCTION DOCUMENTS -000-000 PARK TRUCTION DOCUMENTS	IS OUTLAY	Total	7,368	8.00
140-00-57620- CONS 184470 Inv7513 100-00-51938-	OCTION DOCUMENTS -000-000 PARK TRUCTION DOCUMENTS 9 9/12/2023 8, Inv7526	ES OUTLAY BAER INSURANCE SERVICES, INC ERAL GOV'T INSURANCE	Total	7,368	8.00 8.00
140-00-57620- CONS 184470 Inv7513 100-00-51938- Work 300-00-53612-	UCTION DOCUMENTS -000-000 PARK TRUCTION DOCUMENTS 9 9/12/2023 3, Inv7526 -000-000 GENE Comp, Liability, 2	BAER INSURANCE SERVICES, INC RAL GOV'T INSURANCE Auto 2023-2024 RANCE	Total	7,368 7,368	8.00 8.00 6.83
140-00-57620- CONS 184470 Inv7513 100-00-51938- Work 300-00-53612- Work 400-00-53710-	JCTION DOCUMENTS -000-000 PARK TRUCTION DOCUMENTS 9 9/12/2023 3, Inv7526 -000-000 GENE Comp, Liability, 2 -000-853 INSU Comp, Liability, 2	ES OUTLAY BAER INSURANCE SERVICES, INC ERAL GOV'T INSURANCE Auto 2023-2024 FRANCE Auto 2023-2024 FRANCE	Total	7,368 7,368 4,186	8.00 8.00 6.83 1.67

184471 9/12/2023 CINTAS CORP.

9/12/2023	4:16 PM	Check Register - Full Report - ALL	Page: 9
		ALL Checks	ACCT
		1-POOLED CHECKING ACCOUNT **0307	
Da	ted From:	From Account:	
	Thru:	Thru Account:	
Check Nbr	Check Dat	e Payee	Amount
300-00-53311-	000-852	UNIFORMS	48.36
400-00-53311-	000-852	UNIFORMS	48.36
100-00-53311-	755-000	STREETS - UNIFORMS	66.40
100-00-51980-	760-000	FACILITIES UTILIITIES	109.18
		Total	272.30
		Grand Total	70,903.61

9/12/2023	4:16 PM			Check Register - Full Report - ALL	Page :	10
				ALL Checks	ACCT	
				1-POOLED CHECKING ACCOUNT **0307		
D	Dated From:			From Account:		
	Thru:			Thru Account:		
					Amou	nt
Total Ex	penditure fro	m Fund #	100	- GENERAL FUND	18,637	7.60
Total Ex	penditure fro	m Fund #	140	- CAPITAL PROJECTS FUND	7,880	0.00
Total Ex	penditure fro	m Fund #	150	- PUBLIC PROPERTY AND EVENTS	2,20	6.94
Total Ex	penditure fro	m Fund #	300	- SEWER FUND	18,575	5.91
Total Ex	penditure fro	m Fund #	400	- WATER FUND	23,603	3.16
				Total Expenditure from all Funds	70,903	3.61

Parkitecture + Planning

901 Deming Way, Ste 201 Madison, WI 53717 US www.parkitecture.org



INVOICE

BILL TO Village of Ridgeway ATTN: Hailey Roessler 208 Jarvis Street, Suite A Ridgeway, WI 53582			INVOICE DATE TERMS DUE DATE	3 09/04/2023 Net 30 10/04/2023
PROJECT NAME Ridgeway Park Improvements	P+P PROJECT # 22.055			
TASK DESCRIPTION			AMOUNT DUE	DUE
Charges				
Construction Documents - Ball Diamond Lighting			284.40	10.00 % of 2,844.00
Construction Documents - Parking and Pathway Design			1,189.20	10.00 % of 11,892.00
Post Design Assistance			0.00	0.00 of 3,734.00
Thank You For Supporting Parkitecture + Planning!		SUBTOTAL		1,473.60
		TAX		0.00
		TOTAL		1,473.60
		BALANCE DUE		\$1,473.60

COMMUNITY DEVELOPMENT INVESTMENT GRANT



BUILDING VIBRANT COMMUNITIES IN WISCONSIN®

The livelihood of our communities is directly linked to quality services and infrastructure that support economic competitiveness. Wisconsin communities that mobilize people and resources inspire positive and substantive local development. The Wisconsin Economic Development Corporation's (WEDC's) **Community Development Investment (CDI) Grant Program** helps transform communities by supporting local development that is as unique as the communities in Wisconsin.

HOW IT WORKS

The CDI Grant Program will support urban, small city and rural community re/development efforts by providing financial incentives for shovel-ready projects with emphasis on, but not limited to, downtown communitydriven efforts.

ELIGIBILITY REQUIREMENTS

Grant recipients must demonstrate significant, measurable benefits in job opportunities, property values and/or leveraged investment by local and private partners in at least one of the following efforts:

- Development of a significant destination attraction
- Rehabilitation and reuse of an underutilized or landmark building
- Infill development
- Historic preservation
- Infrastructure efforts providing substantial benefits to downtown residents/property owners
- Mixed-use development

LEARN MORE

For more information about becoming eligible for the Community Development Investment Grant Program, contact a Wisconsin Economic Development Corporation (WEDC) regional economic development director or call 855-INWIBIZ toll free.

You can find the list of regional directors and territories covered at wedc.org/regional.

ELIGIBLE ACTIVITIES

CDI Grant funds may be used for the following activities:

- Building renovation
- Historic preservation
- Demolition
- New construction
- Infrastructure improvements

COMMUNITY DEVELOPMENT INVESTMENT GRANT AGREEMENT BETWEEN PHELAN PROPERTIES, LLC AND THE VILLAGE OF RIDGEWAY

THIS COMMUNITY DEVELOPMENT INVESTMENT GRANT AGREEMENT ("Agreement"), made as of the _____ day of _____, 2023, by and between the Village of Ridgeway, Wisconsin ("Village"), and Phelan Properties, LLC, or its assignee, ("Developer"), (individually, each of the foregoing is a "Party" and collectively, the "Parties").

RECITALS

WHEREAS, Village of Ridgeway and PHELAN PROPERTIES, LLC have entered into an agreement, dated ______, 2023, related to the redevelopment of the property located at 705 Main Street, Ridgeway, Wisconsin, which is further defined in said agreement (the "Property"); and

WHEREAS, Developer intends to make improvements to the Property in conjunction with its planned use of the Property for commercial tenant rental (the "Project"); and

WHEREAS, the Wisconsin Economic Development Corporation ("WEDC"), has approved providing the Village up to <u>twenty five thousand</u> (\$25,000) in Community Development Investment Funds to assist with the costs of Project rehabilitation;

NOW, THEREFORE, in consideration of the foregoing Recitals, which are incorporated herein, and the following promises and mutual obligations of the Parties hereto, each of them does hereby covenant and agree, as follows:

ARTICLE I DEFINITIONS

Section 1.1 <u>Definitions</u>. For purposes of this Agreement, the following terms shall have the following meanings:

- (a) "Agreement" means this agreement, to include all documents required to be delivered contemporaneously with the execution and delivery of this Agreement, and the attached Exhibits, together with any future amendments executed in compliance with this Agreement.
- (b) "Application" means the materials submitted to WEDC relating to this allocation of CDI Funds.
- (c) "CDI Funds" means the grant monies the Village is eligible to receive from WEDC's Community Development Investment grant program.
- (d) "Effective Date" means the date on which this Agreement is fully executed by both parties.
- (e) "Eligible Project Costs" means costs for which CDI Funds and Matching Funds may be used, as outlined in this Agreement, which the Village incurs between the Project Start Date and Project End Date.

- (f) "Ineligible Costs" means costs for which CDI Funds and Matching Funds may not be used, means costs incurred prior to the Project Start Date; costs for acquisition; costs related to grant applications or bid preparation; costs which may be covered by other grant or statutory programs; permits; performance and payment bonds; contingencies; developer fees; insurance premiums; supplies and the purchase of moveable equipment; signage and advertising; financing fees, interest payments, or the assumption of debt; relocation fees; accounting, legal, appraisal, and architectural fees; project administration fees, including costs associated with WEDC compliance reporting, schedules of expenditures, and payment requests.
- (g) "Leverage" means funding provided for the Project other than CDI Funds and Matching Funds.
- (h) "Matching Funds" means non-CDI funds secured by the Developer to meet the required 1:1 funding requirement of the CDI Funds under this Agreement. No more than Thirty Percent (30%) of the Matching Funds may consist of other state and/or federal grants. Matching Funds may not be in-kind.
- (i) "Project" means the development of 705 Main Street in accordance with the Village's Application and the terms of this Agreement.
- (j) "Project End Date" means _____, the date by which the Project will be complete.
- (k) "Project Location" means the site or sites at which the Project will take place, specifically 705 Main Street, Ridgeway, Wisconsin.
- (1) "Project Start Date" means ______, the date on which the Project begins and the Developer may start incurring costs against CDI funds and Matching Funds.
- (n) "WEDC" means the Wisconsin Economic Development Corporation, together with its successors and assigns.

ARTICLE II UNDERTAKINGS BY VILLAGE AND DEVELOPER

Section 2.1 <u>Village Obligations</u>. Village undertakes the following obligations, in consideration of the obligations of Developer, in Section 2.2, below.

- (1) Village will provide the CDI Funds to the Developer to assist with the construction of the Project.
- (2) Village will release the CDI Funds to the Developer on a disbursement basis, within thirty (30) days of Village receiving the funds from WEDC. The Developer may request the CDI Funds in One (1) or more disbursements.

Section 2.2 <u>Developer Obligations</u>. Developer undertakes the following obligations, in consideration of Village obligations in Section 2.1, above.

- (1) Developer shall diligently pursue construction activities for the Project with the objective of completing all elements of the Project, on or before _____.
- (2) Developer agrees to complete the Project as contemplated by the Application and in accordance with the terms of this Agreement, as outlined in the following budget:

USES	SOUR	CES			TOTAL
Budget Code	Eligible Project Costs	CDI Funds	Public Funds	Private Funds	
0200	Construction	\$234,435	\$0	\$252,386	\$0
0190	Site Clearance	\$15,565	\$0	\$0	\$0
0237	Site Preparation	\$0	\$0	\$0	\$0
0415	Site utilities, sidewalks, parking	\$0	\$0	\$0	\$0
TOTAL		\$250,000	\$0	\$252,386	\$0

- (3) Developer agrees to use the CDI Funds and Matching Funds for Eligible Project Costs, incurred between the Project Start Date and Project End Date. Eligible Project Costs include: (i) Eligible Project Costs to be applied to CDI Funds include construction. (ii) Eligible Project Costs to be applied to Matching Funds include construction, site clearance, site preparation, site utilities, sidewalks, and parking
- (4) Developer agrees to not use the CDI Funds or Matching Funds for any Ineligible Costs.
- (5) Developer agrees to secure Matching Funds from non-WEDC sources sufficient to achieve the 1:1 match requirement of the CDI Funds under this Agreement. Matching funds must equal at least <u>Two Hundred Fifty Thousand</u> Dollars (\$250,000) in order for the Village to obtain the maximum amount of the CDI Funds and must be documented prior to the final disbursement.
- (6) Developer agrees to provide acknowledgement of WEDC's participation in the Project in any signage at the Project Location and any planning and feasibility documents related to the Project.
- (7) Developer agrees to provide to the Village a report detailing the dollar amount and purpose of the Eligible Project Costs included in the request for disbursement as well as the dollar amount and purpose of each expenditure that the Developer has contributed to the Project since the date of the previous disbursement of CDI Funds.
- (8) Developer agrees to provide to the Village documentation of the Eligible Project Costs incurred against the CDI Funds, and documentation of the Eligible Project Costs incurred against the Matching Funds, in an amount that is 3:1 of the CDI Funds being requested. Such documentation may include, but not be limited to, purchase orders or invoices.

(9) Developer shall request all CDI Funds no later than _____

- (10) Developer agrees to submit to Village, a schedule of expenditures of the grant or loan funds, including expenditures of any matching cash or in-kind match, signed by the director or principal officer of the Developer to attest to the accuracy of the schedule of expenditures.
- (11) Developer shall engage an independent certified public accountant to perform procedures, approved by WEDC and consistent with applicable professional standards of the American Institute of Certified Public Accountants, to determine whether the grant or loan funds and any matching cash or in-kind match were expended in accordance with the grant or loan contract. The Developer must make available for inspection the documents supporting the schedule of expenditures.
- (12) Developer shall provide information to the Village that is necessary to allow the Village to complete the semi-annual performance reports as required by WEDC.

ARTICLE III REMEDIES

Section 3.1 <u>Event of Default</u>. The occurrence of any one or more of the following events shall constitute an "Event of Default" for the purposes of this Agreement:

- (1) The Developer supplies false or misleading information to Village or WEDC in connection with this Agreement, without providing a satisfactory explanation, in Village's sole discretion, for the noncompliance.
- (2) The Developer fails to comply with or perform, in any material respect, any of its obligations under this Agreement, without providing a satisfactory explanation, in Village's sole discretion, for the noncompliance.

Section 3.2 <u>Remedies in Event of Default.</u> Upon the occurrence of any Event of Default, Village shall send a written notice of default to the Developer, setting forth with reasonable specifivillage the nature of the default. If the Developer fails to cure any such Event of Default to the reasonable satisfaction of Village within Thirty (30) calendar days, Village may extend the cure period if Village determines, in its sole discretion, that the Developer has begun to cure the Event of Default and diligently pursues such cure, or, without further written notice to the Developer, declare the Developer in default. The cure period shall in no event be extended more than Ninety (90) days.

(1) In the Event of Default, Village shall terminate the Agreement and recover from the Developer:

(i) One Hundred Percent (100%) of the funds disbursed to the Developer under this Agreement;

(ii) All court costs and attorneys' fees incurred by Village in terminating this Agreement and recovering the amounts owed by the Developer under this provision.

- (2) These amounts shall be paid to Village within Thirty (30) calendar days of demand by Village hereunder. If the Developer fails to pay these amounts to Village as and when due, the Developer will be liable for the full unpaid balance plus interest at the annual rate of up to Twelve Percent (12%) from the date of the notice of Event of Default.
- (3) Upon an Event of Default, Village shall, without further notice, withhold remaining disbursements of the CDI Funds.

Section 3.3 <u>Remedies are Cumulative</u>. Except as specified in this Agreement, all remedies provided herein shall be cumulative and the exercise of one remedy shall not preclude the use of any other or all of said remedies.

Section 3.4 <u>Failure to Enforce Not Waiver</u>. Failure to enforce any provision contained herein shall not be deemed a waiver of that Party's rights to enforce such provision or any other provision in the event of a subsequent default.

Section 3.5 <u>Mediation</u>. Prior to litigation, and as a condition precedent to bringing litigation, any Party deeming itself aggrieved under this Agreement shall be obligated to request nonbinding mediation of the dispute. Mediation shall proceed before a single mediator. The Parties shall agree upon a mediator and if they fail to do so within 30 days, either Party may apply to the Chief Judge of the

Circuit Court for Grant County, Wisconsin, for the designation of a mediator. In the event the Parties do not accept the mediator's recommendation, the aggrieved Party may then commence an action. However, the Parties shall agree to alternative dispute resolution, if ordered by the Court.

ARTICLE IV AMENDMENT

This Agreement may be rescinded, modified or amended, in whole or in part, by mutual agreement of the Parties hereto, their successors and/or assigns, only in writing signed by the Parties.

ARTICLE V MISCELLANEOUS PROVISIONS

Section 5.1 <u>Execution in Multiple Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same instrument.

Section 5.2 <u>Construction</u>. The Parties acknowledge and represent that this Agreement has been the subject of negotiation by all Parties and that all Parties together shall be construed to be the drafter hereof and this Agreement shall not be construed against any Party individually as drafter.

Section 5.3 <u>Legal Relationship</u>. Nothing in this Agreement shall be construed to create an employer/employee relationship, joint employer, joint venture or partnership relationship, or a principal/agent relationship.

Section 5.4 <u>Survival</u>. All agreements, representations, covenants and warranties made herein shall survive the execution of this Agreement and the making of the grants hereunder. This Agreement shall be binding upon the Parties, their respective successors and assigns.

Section 5.5 <u>No Waiver</u>. The failure of any Party to require strict performance of any provision of this Agreement will not constitute a waiver of the provision or of any other of that Party's rights under this Agreement. Rights and obligations under this Agreement may only be waived or modified in writing. A writing waiving a right must be signed by the Party waiving the right. If an obligation of a Party is being waived or released, the writing must be signed by the affected Parties. Waiver of one right, or release of one obligation, will not constitute a waiver or release of any other right or obligation of any Party. Waivers and releases shall affect only the specific right or obligation waived or released and will not affect the rights or obligations of any other Party that did not sign the waiver or release.

Section 5.6 <u>Severability of Provisions</u>. If any provision of this Agreement shall be held or declared to be invalid, illegal or unenforceable by reason of its being contrary to any applicable law, such provision shall be deemed to be deleted from this Agreement without impairing or prejudicing the validity, legality or enforceability of the remaining provisions.

Section 5.7 <u>Law Governing</u>. This Agreement will be governed and construed in accordance with the laws of the State of Wisconsin.

Section 5.8 <u>Notices and Demands</u>. Except as otherwise expressly provided in this Agreement, a notice, demand or other communication under this Agreement by any Party to any other shall be sufficiently given or delivered if it is dispatched by registered or certified mail, postage prepaid, return receipt requested, or delivered personally, and

(a) in the case of Developer is addressed to or delivered to:

PHELAN PROPERTIES, LLLC 705 Main Street Ridgeway, WI 53818 Attn: Andy and Melinda Phelan

(b) in the case of Village is addressed to or delivered to:

Ridgeway Village Hall 208 Jarvis Street Suite A Ridgeway, WI 53818 Attn: Village Clerk/Treasurer

or at such other, or additional, address with respect to any such Party as that Party may, from time to time, designate in writing and forward to the other, as provided in this Section.

Section 5.9 <u>Term.</u> This Agreement shall continue from the effective date above indicated until the later of ______ or December 31 of the year during which the Village is required to submit to WEDC documentation or reporting information regarding the use of the CDI Funds.

[SIGNATURE PAGES TO FOLLOW]

IN WITNESS WHEREOF, the Parties hereto have executed this Agreement as of the date above indicated.

PHELAN PROPERTIES, LLC

By: ______ Name: Andrew Phelan, Authorized Signatory

STATE OF WISCONSIN)) ss. IOWA COUNTY)

Personally came before me this _____ day of _____, 2023, the above-named Andrew Phelan, the authorized signatory of PHELAN PROPERTIES, LLC, to me known to be the person who executed the foregoing instrument and acknowledged the same.

Notary Public, State of Wisconsin

My Commission expires:

VILLAGE OF RIDGEWAY, WISCONSIN

By: _____

Name: Michele Casper Title: Village President

ATTEST:

By: _____

Name: Hailey Roessler Title: Village Clerk

STATE OF WISCONSIN)) ss. IOWA COUNTY)

Personally came before me this _____ day of _____, 2023, the above-named Michele Casper, Village President and Hailey Roessler, Village Clerk of the Village of Ridgeway, to me known to be the persons who executed the foregoing instrument and acknowledged the same.

Notary Public, State of Wisconsin

My Commission expires: _____



208 Jarvis Street | Suite A | Ridgeway, WI 53582

Iowa County, Wisconsin

RESOLUTION NO. 2023-___

A RESOLUTION AUTHORIZING SUBMISSION OF A COMMUNITY DEVELOPMENT INVESTMENT GRANT APPLICATION

WHEREAS, funding is available under the Community Development Investment (CDI) program, administered by the Wisconsin Economic Development Corporation (WEDC) for the purpose of supporting economic development for communities; and

WHEREAS, the process for securing these funds is competitive, Village staff would be working with a developer on a plan for executing this funding to provide much needed resources to incentivize development in the Village's Main Street Business District; and

WHEREAS, the need to redevelop the property at 705 Main Street, Ridgeway, WI has been identified within the Village of Ridgeway Comprehensive Plan and would be completed utilizing the Village of Ridgeway Zoning and Building Codes established for the B-1Retail Business District.

NOW, THEREFORE BE IT RESOLVED, by the Village Board of Trustees, Village of Ridgeway, does hereby approve and authorize the preparation and filing of a WEDC CDI grant application in the amount of \$_______ for the purpose of rehabilitating a property at 705 Main Street, Ridgeway, WI. Also, the Village Clerk/Treasurer is authorized to sign all necessary documents on behalf of the Village of Ridgeway; and that authority is granted to the Village Clerk/Treasurer to take the necessary steps to prepare and file the application for funds under this program in accordance with this resolution.

BE IT FURTHER RESOLVED THAT, if said grant is awarded, authorization is granted to execute any and all documents required by the WEDC, and such funds shall be accepted and expended pursuant to the terms of the grant award.

The above and foregoing Resolution was duly adopted by the Village Board of the Village of Ridgeway at its meeting held on ______, 2023, by a vote of ______ in favor, ______ opposed, and ______ not voting.

APPROVED:

ATTEST:

By _____ Michele B. Casper, Village President

By _____ Hailey E. Roessler, Village Clerk Specification Manual August 31, 2023

Ridgeway Community Park Improvements Hughette Road Ridgeway, WI 53582

Project # 22.055

Bid Due Date: September x, 2023



Plans Prepared By:

Parkitecture + Planning 901 Deming Way Madison, WI 53717 608.203.8203





ADVERTISEMENT TO BID

August 31, 2023

The Village of Ridgeway is soliciting Bids for the proposed park improvements at the Community Park. Bids will be received until **2:30 PM, September x, 2023** online via **QUEST CDN** only. Bids will be virtually opened and read aloud. All planholders will be provided with login information prior to bid opening via addendum.

PROJECT SCOPE:

The project scope includes but not limited to: demolition, erosion control, earthwork, concrete paving, site electrical, ball diamond lighting, and site restoration.

BID SUBMITTAL:

Bids for **"Ridgeway Community Park Improvements."** shall be submitted online only at **QUEST CDN**. The OWNER reserves the right to waive any informalities and to reject any or all Bids. The letting of the work described herein is subject to the provisions of Sections 61.54, 66.0901 and 66.0903, Wisconsin Statutes.

BID DOCUMENT AVAILABILITY:

Bidding Documents will be available from **QUEST CDN** online bid room (**project #xxxx**). Plan download fees are \$42. Bid documents will be available for review only by appointment at the Village offices, 208 Jarvis Street, Suite A, Ridgeway, WI 53582.

PRE-BID CONFERENCE:

There will be no pre-bid conference for this project. Contractor questions may be directed to Parkitecture + Planning, 608-203-8203.

BID BOND:

A satisfactory Bid Bond in the amount of 5% of the total bid payable to the OWNER, shall be submitted with the Bid via **QUEST CDN**.

SCHEDULE:

Selection of final bid is at the sole discretion of the OWNER, and shall reserve the right to waive irregularities should it be needed. All Bids shall remain valid for 60 days after bid opening.

Publication Dates: September xx and September xx, 2023

SECTION 1 INSTRUCTIONS TO BIDDERS

These Instructions to Bidders establish requirements for Bidding and Award of Contract.

- To demonstrate Bidder's qualifications to perform the Work, within five days of OWNER's request, Bidder shall
 submit written evidence such as financial data, previous experience, present commitments and other such
 data as may be called for below. Bidder must be prepared to submit evidence of Bidder's qualifications to do
 business in the State of Wisconsin prior to award of the Contract.
- Bidder has made themselves familiar with those reports of explorations and tests of subsurface conditions at or contiguous to the Site that A/E (ARCHITECT/ENGINEER) has used in preparing the Bidding Documents.
- On request, OWNER will provide Bidder access to the Site to conduct such examinations, investigations, explorations, tests, and studies as Bidder deems necessary for submission of a Bid.
- Bidder shall comply with instructions and requirements identified in the Bid.
- The submission of a Bid will constitute an incontrovertible representation by Bidder to perform and furnish the work required by the Bidding Documents and apply the specific means, methods, techniques, sequences, or procedures of construction (if any) that may be shown or indicated or expressly required by the Bidding Documents, that Bidder has given A/E written notice of all conflicts, errors, ambiguities, and discrepancies that Bidder has discovered in the Bidding Documents and the written resolution thereof by A/E is acceptable to Bidder, and that the Bidding Documents are generally sufficient to indicate and convey understanding of all terms and conditions for performing and furnishing the work.
- If a pre-Bid conference is to be held, it will be held at the time and place stated in the Advertisement to Bid.
- All questions about the meaning or intent of the Bidding Documents are to be submitted to A/E in writing. Interpretations or clarifications considered necessary by A/E in response to such questions will be issued by Addenda. All Addenda so issued shall become part of the Contract Documents. Only questions answered by Addenda will be binding. Oral and other interpretations or clarifications will be without legal effect. Addenda may also be issued to clarify, correct, or change the Bidding Documents as deemed advisable by OWNER or A/E. Receipt of all Addenda must be acknowledged in space provided in the Bid.
- A Certified Check, Back Cashier's Check or satisfactory Bid Bond in the amount stated in the Advertisement for Bids payable to the OWNER, shall accompany each Bid as a guarantee, that if the Bid is accepted, the Bidder will execute and file the Contract, Performance/Payment Bonds and Certification of Insurance, as required by the Contract Documents, within 15-days after the Notice of Award of Contract by the OWNER.
- The numbers of days within which, or the dates by which, the Work is to be (a) Substantially Completed and (b) completed and ready for final payment are set forth in the Agreement (or incorporated therein by reference to the attached Bid Form).
- The Contract, if awarded, will be on the basis of materials and equipment specified or described in the Bidding Documents without consideration of possible substitute or "or-equal" items. Whenever it is specified or described in the Bidding Documents that a substitute or "or-equal" item of material or equipment may be furnished or used by CONTRACTOR if acceptable to A/E, application for such acceptance will not be considered by A/E until after the Effective Date of the Agreement.
- Bidders shall submit a Bid on a unit price basis for each item of Work listed in the Bid Form. All blanks on the Bid Form must be completed in ink and the Bid signed or via digital format on QUEST CDN..
- Bids will be received for all divisions of the specifications and all other provisions of the Bidding Documents. Bids will not be considered or opened if they are received after the time set for opening Bids. Bids will be opened at the time and place indicated in the Advertisement to Bid and read aloud publicly.

SECTION 2

BID FORM

The following bid proposal indicates an agreement on behalf of the Bidder to complete work required to fulfill the project scope as outlined in the plans and specifications for the unit prices specified herein for the following project:

Ridgway Community Park Improvements

Village of Ridgeway 208 Jarvis Street, Suite A Ridgeway, WI 53582

Bid proposals shall remain valid for up to 60 days after date of the bid opening. Upon notice of award, bidder shall complete required Bond Forms and insurance documents. A copy shall be provided to the A/E for review prior to execution by the OWNER.

The OWNER shall consider all factors and act in their best interest when considering the lowest responsible bidder for the project.

The scope of work as outlined in the plans and specifications shall be substantially completed by Friday May 24, 2024.

The signature(s) of the bidding entity shall confirm that the Bidder has reviewed all pertinent documents and existing site conditions, fully reviewed the scope of the project, and understands the overall nature of the work.

Any questions regarding the following items shall have been sufficiently raised and resolved with the A/E and/or OWNER prior to bidding. Any clarifications and responses provided by the A/E and/or OWNER shall only be binding if issued in writing via addenda.

Signature below must be completed by an officer of the bidding entity who shall be authorized to bind said entity into contractual agreements.

Name & Address of Bidder
Contact Phone & Email
Type of Entity (Corporation, LLC, etc)
conducting business in the State of
Signature

NOTARY Signature_____ Sworn and subscribed to me on this ______ day of 2023. Name and Title_____ Expiration of Commission______

SECTION 2

BID FORM

Addenda Acknowledged _____, ____, ____,

Bidder to input unit prices below and include the total cost per item in the far right column. Bid Item descriptions are outlined in the Special Provisions Section of this Project Manual. The A/E shall review the computations for irregularities and notify bidder should errors be found.

ITEM	ITEM DESCRIPTION	QUANTITY	UNIT	UNIT PRICE	TOTAL
1.	Mobilization	1	LS	\$	\$
2.	Erosion Control	1	LS	\$	\$
3.	Demolition	1	LS	\$	\$
4.	Earthwork	1	LS	\$	\$
5.	Dense Graded Basecourse	720	TON	\$	\$
6.	Concrete Pavement – 5″ Thickened	595	SF	\$	\$
7.	Concrete Pavement – 5″	880	SF	\$	\$
8.	Tactile Warning Plates	28	SF	\$	\$
9.	Asphaltic Pavement –	145	TON	\$	\$
10.	Parking Lot Pavement Markings	1	LS	\$	\$
11.	Regulatory Signs	2	EA	\$	\$
12.	4" Storm Sewer	30	LF	\$	\$
13.	Storm Sewer Outfall	1	LS	\$	\$
14.	Trench Drain	24	LF	\$	\$
15.	Site Electrical and Exterior	1	LS	\$	\$
16.	Athletic Lighting Restoration	1	LS	\$	\$
TOTAL	BID (ITEMS 1 THROUGH 16)				
			D	ollars \$	
	(Words)				(Numbers)
	NATE BID ITEM ITEM DESCRIPTION	QUANTITY	UNIT	UNIT PRICE	TOTAL
A-1	Light Poles	1	LS	\$	\$
TOTAL	ALTERNATE BID (ITEM A-1 th	ru A-1)			
			Do	ollars \$	
	(Words)				(Numbers)

Item 8.

SECTION 3

BID BOND

Any singular reference to Bidder, Surety, Owner or other party shall be considered plural where applicable.

BIDDER (Name and Address):

SURETY (Name and Address of Principal Place of Business):

OWNER (Name and Address):

BID

Bid Due Date: Description *(Project Name and Include Location)*:

BOND

Bond Number: Date *(Not earlier than Bid due date)*: Penal sum

(Words)

(Figures)

\$

Surety and Bidder, intending to be legally bound hereby, subject to the terms set forth below, do each cause this Bid Bond to be duly executed by an authorized officer, agent, or representative.

(Soal)	SURETY		(Soal)
_ (3031)	Surety's Name and Corporate Seal		(Seal)
	By:		_
_		Signature (Attach Power of A	ttorney)
_		Print Name	-
_		Title	_
	Attest:		
—		Signature	_
_		Title	_
-	(Seal) 	(Seal) Surety's By: 	(Seal) Surety's Name and Corporate Seal By: Signature (Attach Power of A Print Name Title Attest: Signature Signature

EJCDC C-430 Bid Bond (Penal Sum Form)	
Prepared by the Engineers Joint Contract Documents Committee.	
Page 1 of 2	

Item 8.

1. Bidder and Surety, jointly and severally, bind themselves, their heirs, executors, administrators, successors, and assigns to pay to Owner upon default of Bidder the penal sum set forth on the face of this Bond. Payment of the penal sum is the extent of Bidder's and Surety's liability. Recovery of such penal sum under the terms of this Bond shall be Owner's sole and exclusive remedy upon default of Bidder.

2. Default of Bidder shall occur upon the failure of Bidder to deliver within the time required by the Bidding Documents (or any extension thereof agreed to in writing by Owner) the executed Agreement required by the Bidding Documents and any performance and payment bonds required by the Bidding Documents.

- 3. This obligation shall be null and void if:
 - 3.1 Owner accepts Bidder's Bid and Bidder delivers within the time required by the Bidding Documents (or any extension thereof agreed to in writing by Owner) the executed Agreement required by the Bidding Documents and any performance and payment bonds required by the Bidding Documents, or
 - 3.2 All Bids are rejected by Owner, or
 - 3.3 Owner fails to issue a Notice of Award to Bidder within the time specified in the Bidding Documents (or any extension thereof agreed to in writing by Bidder and, if applicable, consented to by Surety when required by Paragraph 5 hereof).

4. Payment under this Bond will be due and payable upon default of Bidder and within 30 calendar days after receipt by Bidder and Surety of written notice of default from Owner, which notice will be given with reasonable promptness, identifying this Bond and the Project and including a statement of the amount due.

5. Surety waives notice of any and all defenses based on or arising out of any time extension to issue Notice of Award agreed to in writing by Owner and Bidder, provided that the total time for issuing Notice of Award including extensions shall not in the aggregate exceed 120 days from Bid due date without Surety's written consent.

6. No suit or action shall be commenced under this Bond prior to 30 calendar days after the notice of default required in Paragraph 4 above is received by Bidder and Surety and in no case later than one year after Bid due date.

7. Any suit or action under this Bond shall be commenced only in a court of competent jurisdiction located in the state in which the Project is located.

8. Notices required hereunder shall be in writing and sent to Bidder and Surety at their respective addresses shown on the face of this Bond. Such notices may be sent by personal delivery, commercial courier, or by United States Registered or Certified Mail, return receipt requested, postage pre-paid, and shall be deemed to be effective upon receipt by the party concerned.

9. Surety shall cause to be attached to this Bond a current and effective Power of Attorney evidencing the authority of the officer, agent, or representative who executed this Bond on behalf of Surety to execute, seal, and deliver such Bond and bind the Surety thereby.

10. This Bond is intended to conform to all applicable statutory requirements. Any applicable requirement of any applicable statute that has been omitted from this Bond shall be deemed to be included herein as if set forth at length. If any provision of this Bond conflicts with any applicable statute, then the provision of said statute shall govern and the remainder of this Bond that is not in conflict therewith shall continue in full force and effect.

11. The term "Bid" as used herein includes a Bid, offer, or proposal as applicable.

EJCDC C-430 Bid Bond (Penal Sum Form)	
Prepared by the Engineers Joint Contract Documents Committee.	
Page 2 of 2	

SECTION 4

CONTRACT FOR CONSTRUCTION OF A SMALL PROJECT

This Contract is dated as of the _____ day of _____ in the year _____ by and

between _____ (Owner)

and _____ (Contractor).

Owner and Contractor hereby agree as follows:

ARTICLE 1 - THE WORK

- 1.1 Work
 - A. Work includes all labor, materials, equipment, services, and documentation necessary to construct the Project defined herein. The Work may include related services such as testing, start-up, and commissioning, all as required by the Contract Documents.
 - B. The Contractor shall complete all Work as specified or indicated in the Contract Documents. The Work is generally described as follows:
 - 1. Project: Ridgeway Community Park Improvements
 - 2. Scope of Work: The project scope includes but not limited to: demolition, erosion control, earthwork, concrete paving, site electrical, ball diamond lighting, and site restoration.

ARTICLE 2 - CONTRACT DOCUMENTS

- 2.1 Intent of Contract Documents
 - A. It is the intent of the Contract Documents to describe a functionally complete project. The Contract Documents do not indicate or describe all of the Work required to complete the Project. Additional details required for the correct installation of selected products are to be provided by the Contractor and coordinated with the Owner and A/E. This Contract supersedes prior negotiations, representations, and agreements, whether written or oral. The Contract Documents (plans and specifications) are complementary; what is required by one part of the Contract Documents is as binding as if required by other parts of the Contract Documents.
 - B. During the performance of the Work and until final payment, Contractor and Owner shall submit all matters in question concerning the requirements of the Contract Documents, or relating to the acceptability of the Work under the Contract Documents to the A/E. A/E will be the initial interpreter of the requirements of the Contract Documents, and judge of the acceptability of the Work thereunder.
 - C. A/E will render a written clarification, interpretation, or decision on the issue submitted, or initiate a modification to the Contract Documents.
 - D. Contractor, and its subcontractors and suppliers, shall not have or acquire any title to or ownership rights to any of the Drawings, Specifications, or other documents (including copies or electronic media editions) prepared by A/E or its consultants.
- 2.2 Contract Documents Defined
 - A. The Contract Documents consist of the following documents:
 - 1. This Contract (pages 1 to <u>8</u>, inclusive).

EJCDC [®] C-522, Contract for Construction of a Small Project.
Copyright © 2016 National Society of Professional Engineers, American Council of Engineering Companies,
and American Society of Civil Engineers. All rights reserved. Page 1 of 88

- 2. Performance bond (pages <u>1</u> to <u>3</u>, inclusive), plus attachments.
- 3. Payment bond (pages <u>1</u> to <u>3</u>, inclusive), plus attachments.
- 4. Special Provisions (pages <u>SP-1</u> to <u>SP-10</u>, inclusive).
- 5. Drawings (not attached but incorporated by reference) consisting of 55 sheets as listed in the Drawing Sheet Index.
- 6. Addenda (numbers _ to _, inclusive).
- 7. Exhibits to this Contract (enumerated as follows):
 - a. Contractor's Bid (pages _ to _, inclusive).
- 8. The following which may be delivered or issued on or after the Effective Date of the Contract:
 - a. Work Change Directives.
 - b. Change Orders.
 - c. Field Orders.

ARTICLE 3 - A/E

- 3.1 A/E
 - A. The A/E for this Project is Parkitecture + Planning, 901 Deming Way, Suite 201, Madison, WI 53717.
 - B. The term ENGINEER is used in the standard language of the General Conditions to represent the A/E referred to hereunder.

ARTICLE 4 - CONTRACT TIMES

- 4.1 Contract Times
 - A. Construction may commence on or about <u>October xx, 2023.</u>
 - B. The Work will be substantially completed on or before May 24, 2024.
- 4.2 Liquidated Damages
 - A. None.
- 4.3 Delays in Contractor's Progress
 - A. If Owner, A/E, or anyone for whom Owner is responsible, delays, disrupts, or interferes with the performance or progress of the Work, then Contractor shall be entitled to an equitable adjustment in the Contract Times and Contract Price. Contractor's entitlement to an adjustment of the Contract Times is conditioned on such adjustment being essential to Contractor's ability to complete the Work within the Contract Times.
 - B. Refer to Article 4 of the General Conditions for additional terms and requirements.

ARTICLE 5 - CONTRACT PRICE

- 5.1 Payment
 - Α. Owner shall pay Contractor in accordance with the Contract Documents at the unit price for each unit of Work as stated in Contractor's Bid Schedule, which is attached to this Contract as an exhibit.
 - The initial Contract Price based on the estimated quantities of work listed in the Bid 1. Schedule is \$
 - 2. The final Contract Price will be determined by multiplying each unit price times the actual quantity of that Work item completed. Actual quantities installed will be determined by the A/E.

ARTICLE 6 - BONDS AND INSURANCE

С.

- 6.1 Bonds
 - Refer to Article 6 of the General Conditions for terms and requirements. Α.
- 6.2 Insurance
 - Before starting Work, Contractor shall furnish evidence of insurance from companies that are Α. duly licensed or authorized in the jurisdiction in which the Project is located with a minimum AM Best rating of A-VII or better. Contractor shall provide insurance in accordance with the following:
 - Contractor shall provide coverage for not less than the following amounts, or greater 1. where required by Laws and Regulations:
 - State: Statutory Employer's Liability: Bodily Injury, each Accident 100,000 S Bodily Injury By Disease, each Employee \$ 100,000 Bodily Injury/Disease Aggregate 500,000 \$ Commercial General Liability: b. General Aggregate Ś 1,000,000 Products - Completed Operations Aggregate \$ 2,000,000 Personal and Advertising Injury Ś 500,000 Each Occurrence (Bodily Injury and Property Damage) Ś 500,000 Automobile Liability: Bodily Injury and Property Damage Combined Single Limit of Ś 1,000,000 Excess or Umbrella Liability: d. Per Occurrence 3,000,000 General Aggregate 3,000,000
 - Workers' Compensation: a.

- B. All insurance policies required to be purchased and maintained will contain a provision or endorsement that the coverage afforded will not be canceled or materially changed or renewal refused until at least 10 days prior written notice has been given to the insured and additional insured.
- C. Automobile liability insurance provided by Contractor shall provide coverage against claims for damages because of bodily injury or death of any person or property damage arising out of the ownership, maintenance, or use of any motor vehicle. The automobile liability policy shall be written on an occurrence basis.
- D. Contractor's commercial general liability policy shall be written on a 1996 or later ISO commercial general liability occurrence form and include the following coverages and endorsements:
 - 1. Products and completed operations coverage maintained for three years after final payment;
 - 2. Blanket contractual liability coverage to the extent permitted by law;
 - 3. Broad form property damage coverage; and
 - 4. Severability of interest; underground, explosion, and collapse coverage; personal injury coverage.
- E. The Contractor's commercial general liability and automobile liability, and umbrella or excess liability policies shall include and list Owner and A/E and the respective officers, directors, members, partners, employees, agents, consultants, and subcontractors of each as additional insureds; and the insurance afforded to these additional insureds shall provide primary coverage for all claims covered thereby (including as applicable those arising from both ongoing and completed operations) on a non-contributory basis.
- F. Umbrella or excess liability insurance shall be written over the underlying employer's liability, commercial general liability, and automobile liability insurance. Subject to industry-standard exclusions, the coverage afforded shall be procured on a "follow the form" basis as to each of the underlying policies. Contractor may demonstrate to Owner that Contractor has met the combined limits of insurance (underlying policy plus applicable umbrella) specified for employer's liability, commercial general liability, and automobile liability through the primary policies alone, or through combinations of the primary insurance policies and an umbrella or excess liability policy.

ARTICLE 7 - CONTRACTOR'S RESPONSIBILITIES

- 7.1 Supervision and Superintendence
 - A. Refer to Article 8 of the General Conditions for terms and requirements.
- 7.2 Other Work at the Site
 - A. In addition to and apart from the Work of the Contractor, other work may occur at or adjacent to the Site. Contractor shall take reasonable and customary measures to avoid damaging, delaying, disrupting, or interfering with the work of Owner, any other contractor, or any utility owner performing other work at or adjacent to the Site.
 - B. Refer to Article 8 of the General Conditions for additional terms and requirements.
- 7.3 Ouality Management
 - A. Contractor is fully responsible for managing quality to ensure Work is completed in accordance with the Contract Documents.
- 7.4 Licenses, Fees and Permits
 - A. Contractor shall obtain and pay for all construction permits and licenses unless otherwise provided in the Contract Documents.
- 7.5 Record Documents
 - A. Contractor shall maintain one printed record copy of all Drawings, Specifications, Addenda, Change Orders, Work Change Directives, Field Orders, written interpretations and clarifications, and approved shop drawings in a safe place at the Site. Contractor shall annotate them to show changes made during construction. Contractor shall deliver these record documents to A/E upon completion of the Work.
- 7.6 Shop Drawings, Samples, and Other Submittals
 - A. Refer to Article 7 of the General Conditions for terms and requirements.
- 7.7 Warranties and Guarantees
 - A. Refer to Article 6 of the General Conditions for terms and requirements.
- 7.8 Correction Period
 - A. If within one year after the date of substantial completion, any Work is found to be defective, or if the repair of any damages to the Site, adjacent areas that Contractor has arranged to use through construction easements or otherwise, and other adjacent areas used by Contractor as permitted by Laws and Regulations, is found to be defective, then Contractor shall promptly and without cost to Owner, correct such defective Work.
- 7.9 Indemnification
 - A. To the fullest extent permitted by Laws and Regulations, and in addition to any other obligations of Contractor under the Contract or otherwise, Contractor shall indemnify and hold harmless Owner and A/E, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to the performance of the Work, provided that any such claim, cost, loss, or damage is attributable to bodily injury, sickness, disease, or death, or to injury to or destruction of tangible property (other than the Work itself), including the loss of use resulting therefrom but only to the extent caused by any negligent act or omission of Contractor, any subcontractor, any supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work or anyone for whose acts they may be liable.

ARTICLE 8 - OWNER'S RESPONSIBILITIES

- 8.1 Owner's Responsibilities
 - A. Except as otherwise provided in the Contract Documents, Owner shall issue all communications to Contractor through A/E.
 - B. Owner shall make payments to Contractor as provided in this Contract.
 - C. Owner shall provide Site and easements required to construct the Project.
 - D. Refer to Article 9 of the General Conditions for additional terms and requirements.

ARTICLE 9 - A/E'S STATUS DURING CONSTRUCTION

- 9.1 A/E's Status
 - A. A/E will be Owner's representative during construction. The duties and responsibilities and the limitations of authority of A/E as Owner's representative during construction are set forth in this Contract.
 - B. Neither A/E's authority or responsibility under this Article 9 or under any other provision of the Contract, nor any decision made by A/E in good faith either to exercise or not exercise such authority or responsibility or the undertaking, exercise, or performance of any authority or responsibility by A/E, shall create, impose, or give rise to any duty in contract, tort, or otherwise owed by A/E to Contractor, any subcontractor, any supplier, any other individual or entity, or to any surety for or employee or agent of any of them.
 - C. Refer to Article 10 of the General Conditions for additional terms and requirements.

ARTICLE 10 - CLAIMS AND DISPUTE RESOLUTION

- 10.1 Claims Process
 - A. Refer to Article 12 of the General Conditions for terms and requirements.

ARTICLE 11 - TESTS AND INSPECTIONS; CORRECTION OF DEFECTIVE WORK

11.1 Refer to Article 14 of the General Conditions for terms and requirements.

ARTICLE 12 - PAYMENTS TO CONTRACTOR

- 12.1 Progress Payments
 - A. The Contractor shall prepare a schedule of values that will serve as the basis for progress payments. The schedule of values will be in a form of application for payment acceptable to A/E. The unit price breakdown submitted with the bid will be used for unit price work. Break lump sum items into units that will allow for measurement of Work in progress.
- 12.2 Applications for Payments:
 - A. Contractor shall submit an application for payment in a form acceptable to the A/E, no more frequently than monthly, to A/E. Applications for payment will be prepared and signed by Contractor. Contractor shall provide supporting documentation required by the Contract Documents. Payment will be paid for Work completed as of the date of the application for payment.
- 12.3 Retainage
 - A. Prior to Substantial Completion, progress payments will be made in an amount equal to the percentage indicated below but, in each case, less the aggregate of payments previously made and less such amounts as Owner may withhold, including but not limited to liquidated damages, in accordance with the Contract:
 - 1. <u>95</u> percent of the Work completed (with the balance being retainage); and
 - 2. <u>95</u> percent of cost of materials and equipment not incorporated in the Work (with the balance being retainage).
 - 3. At 50 percent completion, no additional amounts will be retained unless A/E certifies that the Work is not proceeding satisfactorily, but amounts previously retained will not be paid to Contractor. At 50 percent completion or any time thereafter when the progress of the Work is not satisfactory, additional amounts may be retained but in no event will total

retainage be more than 10 percent of the Work completed and materials and equipment delivered, suitably stored, and accompanied by required documentation.

- B. Upon Substantial Completion, Owner shall pay an amount sufficient to increase total payments to Contractor to <u>98</u> percent of the Contract Price (with the balance being retainage), less such amounts set off by Owner pursuant to Paragraph 14.4.C, and less A/E's estimate of the value of Work to be completed or corrected as shown on the punch list of items to be completed or corrected prior to final payment.
- 12.4 Review of Applications
 - A. Refer to Article 15 of the General Conditions for terms and requirements.
- 12.5 Substantial Completion
 - A. All Contract work shall be substantially complete by <u>5/24/2024</u>.
 - B. The Contractor shall notify Owner and A/E in writing that the Work is substantially complete and request the A/E issue a certificate of substantial completion when Contractor considers the Work ready for its intended use. Contractor shall at the same time submit to Owner and A/E an initial draft of punch list items to be completed or corrected before final payment.
 - C. Refer to Article 15 of the General Conditions for additional terms and requirements.
- 12.6 Final Inspection
 - A. Refer to Article 15 of the General Conditions for terms and requirements.
- 12.7 Final Payment
 - A. Refer to Article 15 of the General Conditions for terms and requirements.

ARTICLE 13 - SUSPENSION OF WORK AND TERMINATION

13.1 Refer to Article 16 of the General Conditions for terms and requirements.

ARTICLE 14 - CONTRACTOR'S REPRESENTATIONS

- 14.1 Contractor Representations
 - A. Contractor makes the following representations when entering into this Contract:
 - 1. Contractor has examined and carefully studied the Contract Documents, and any data and reference items identified in the Contract Documents.
 - 2. Contractor has visited the Site, conducted a thorough visual examination of the Site and adjacent areas, and become familiar with and is satisfied as to the general, local, and Site conditions that may affect cost, progress, and performance of the Work.
 - 3. Contractor is familiar with and is satisfied as to all Laws and Regulations that may affect cost, progress, and performance of the Work.
 - 4. Contractor has considered the information known to Contractor itself; information commonly known to contractors doing business in the locality of the Site; information and observations obtained from visits to the Site; the Contract Documents; and the Site-related reports and drawings identified in the Contract Documents, with respect to the effect of such information, observations, and documents on:
 - a. The cost, progress, and performance of the Work;
 - b. The means, methods, techniques, sequences, and procedures of construction to be employed by Contractor; and
 - c. Contractor's safety precautions and programs.

- 5. Based on the information and observations referred to in the preceding paragraph, Contractor agrees that no further examinations, investigations, explorations, tests, studies, or data are necessary for the performance of the Work at the Contract Price, within the Contract Times, and in accordance with the other terms and conditions of the Contract.
- 6. Contractor is aware of the general nature of work to be performed by Owner and others at the Site that relates to the Work as indicated in the Contract Documents.
- 7. Contractor has given A/E written notice of all conflicts, errors, ambiguities, or discrepancies that Contractor has discovered in the Contract Documents, and the written resolution thereof by A/E is acceptable to Contractor.
- 8. The Contract Documents are generally sufficient to indicate and convey understanding of all terms and conditions for performance and furnishing of the Work.
- 9. Contractor's entry into this Contract constitutes an incontrovertible representation by Contractor that, without exception, all prices in the Contract are premised upon performing and furnishing the Work required by the Contract Documents.

IN WITNESS WHEREOF, Owner and Contractor have signed this Contract.

This Contract will be effective as of the date first written on Page 1 of the Contract (which is the Effective Date of the Contract).

By:
Title:
<i>(If Contractor is a corporation, a partnership, or a joint venture, attach evidence of authority to sign.)</i>
Attest:
Title:
Address for giving notices:
License No.: (where applicable)

EJCDC® C-522, Contract for Construction of a Small Project. Copyright © 2016 National Society of Professional Engineers, American Council of Engineering Companies, and American Society of Civil Engineers. All rights reserved. Page 8 of 88

PAYMENT BOND

Any singular reference to Contractor, Surety, Owner, or other party shall be considered plural where applicable.

CONTRACTOR (Name and Address):

SURETY (*Name, and Address of Principal Place of Business*):

OWNER (*Name and Address*):

CONTRACT

Effective Date of Agreement: Amount: Description *(Name and Location)*:

BOND

Bond Number: Date (*Not earlier than Effective Date of Agreement*): Amount: Modifications to this Bond Form:

Surety and Contractor, intending to be legally bound hereby, subject to the terms set forth below, do each cause this Payment Bond to be duly executed by an authorized officer, agent, or representative.

CONTRACTOR AS PRINCIPAL

SURETY

	(Seal)	_	(Sea	I)
Contr	actor's Name and Corporate Seal	S	urety's Name and Corporate Seal	
By:		By:		
	Signature		Signature (Attach Power of Attorney)	
	Print Name		Print Name	
	Title		Title	
Attest:		Attes	t:	
	Signature		Signature	
			T:41-	
	Title		Title	
Nata:	Provide execution by additional parties of	uch ac	iaint vanturars if pacassary	

Note: Provide execution by additional parties, such as joint venturers, if necessary.

{MW001504;1}	EJCDC C-615(A) Payment Bond	March 2008
	Prepared by the Engineers Joint Contract Documents Committee.	
	Page 1 of 3	149

1. Contractor and Surety, jointly and severally, bind themselves, their heirs, executors, administrate litem 8. successors, and assigns to Owner to pay for labor, materials, and equipment furnished by Claimants for use in the performance of the Contract, which is incorporated herein by reference.

- 2. With respect to Owner, this obligation shall be null and void if Contractor:
 - 2.1 Promptly makes payment, directly or indirectly, for all sums due Claimants, and
 - 2.2 Defends, indemnifies, and holds harmless Owner from all claims, demands, liens, or suits alleging non-payment by Contractor by any person or entity who furnished labor, materials, or equipment for use in the performance of the Contract, provided Owner has promptly notified Contractor and Surety (at the addresses described in Paragraph 12) of any claims, demands, liens, or suits and tendered defense of such claims, demands, liens, or suits to Contractor and Surety, and provided there is no Owner Default.

3. With respect to Claimants, this obligation shall be null and void if Contractor promptly makes payment, directly or indirectly, for all sums due.

- 4. Surety shall have no obligation to Claimants under this Bond until:
 - 4.1 Claimants who are employed by or have a direct contract with Contractor have given notice to Surety (at the address described in Paragraph 12) and sent a copy, or notice thereof, to Owner, stating that a claim is being made under this Bond and, with substantial accuracy, the amount of the claim.
 - 4.2 Claimants who do not have a direct contract with Contractor:
 - 1. Have furnished written notice to Contractor and sent a copy, or notice thereof, to Owner, within 90 days after having last performed labor or last furnished materials or equipment included in the claim stating, with substantial accuracy, the amount of the claim and the name of the party to whom the materials or equipment were furnished or supplied, or for whom the labor was done or performed; and
 - 2. Have either received a rejection in whole or in part from Contractor, or not received within 30 days of furnishing the above notice any communication from Contractor by which Contractor had indicated the claim will be paid directly or indirectly; and
 - 3. Not having been paid within the above 30 days, have sent a written notice to Surety (at the address described in Paragraph 12) and sent a copy, or notice thereof, to Owner, stating that a claim is being made under this Bond and enclosing a copy of the previous written notice furnished to Contractor.

5. If a notice by a Claimant required by Paragraph 4 is provided by Owner to Contractor or to Surety, that is sufficient compliance.

6. Reserved.

7. Surety's total obligation shall not exceed the amount of this Bond, and the amount of this Bond shall be credited for any payments made in good faith by Surety.

8. Amounts owed by Owner to Contractor under the Contract shall be used for the performance of the Contract and to satisfy claims, if any, under any performance bond. By Contractor furnishing and Owner accepting this Bond, they agree that all funds earned by Contractor in the performance of the Contract are dedicated to satisfy obligations of Contractor and Surety under this Bond, subject to Owner's priority to use the funds for the completion of the Work.

9. Surety shall not be liable to Owner, Claimants, or others for obligations of Contractor that are unrelated to the Contract. Owner shall not be liable for payment of any costs or expenses of any Claimant under this Bond, and shall have under this Bond no obligations to make payments to, give notices on behalf of, or otherwise have obligations to Claimants under this Bond.

10. Surety hereby waives notice of any change, including changes of time, to the Contract or to related

{MW001504;1}	EJCDC C-615(A) Payment Bond	March 2008
	Prepared by the Engineers Joint Contract Documents Committee.	
	Page 2 of 3	150

subcontracts, purchase orders, and other obligations.

11. No suit or action shall be commenced by a Claimant under this Bond other than in a court of competent jurisdiction in the location in which the Work or part of the Work is located or after the expiration of one year from the date (1) on which the Claimant gave the notice required by Paragraph 4.1 or Paragraph 4.2.3, or (2) on which the last labor or service was performed by anyone or the last materials or equipment were furnished by anyone under the Contract, whichever of (1) or (2) first occurs. If the provisions of this paragraph are void or prohibited by law, the minimum period of limitation available to sureties as a defense in the jurisdiction of the suit shall be applicable.

12. Notice to Surety, Owner, or Contractor shall be mailed or delivered to the addresses shown on the signature page. Actual receipt of notice by Surety, Owner, or Contractor, however accomplished, shall be sufficient compliance as of the date received at the address shown on the signature page.

13. When this Bond has been furnished to comply with a statutory requirement in the location where the Contract was to be performed, any provision in this Bond conflicting with said statutory requirement shall be deemed deleted herefrom and provisions conforming to such statutory requirement shall be deemed incorporated herein. The intent is that this Bond shall be construed as a statutory Bond and not as a common law bond.

14. Upon request of any person or entity appearing to be a potential beneficiary of this Bond, Contractor shall promptly furnish a copy of this Bond or shall permit a copy to be made.

- 15. Definitions
 - 15.1 Claimant: An individual or entity having a direct contract with Contractor, or with a first-tier subcontractor of Contractor, to furnish labor, materials, or equipment for use in the performance of the Contract. The intent of this Bond shall be to include without limitation in the terms "labor, materials or equipment" that part of water, gas, power, light, heat, oil, gasoline, telephone service, or rental equipment used in the Contract, architectural and engineering services required for performance of the Work of Contractor and Contractor's subcontractors, and all other items for which a mechanic's lien may be asserted in the jurisdiction where the labor, materials, or equipment were furnished.
 - 15.2 Contract: The agreement between Owner and Contractor identified on the signature page, including all Contract Documents and changes thereto.
 - 15.3 Owner Default: Failure of Owner, which has neither been remedied nor waived, to pay Contractor as required by the Contract, or to perform and complete or otherwise comply with the other terms thereof.

FOR INFORMATION ONLY – (*Name, Address, and Telephone*) Surety Agency or Broker: Owner's Representative (*Engineer or other*):

151

PERFORMANCE BOND

Any singular reference to Contractor, Surety, Owner, or other party shall be considered plural where applicable.

CONTRACTOR (Name and Address): SURETY (Name, and Address of Principal Place of Business):

OWNER (*Name and Address*):

CONTRACT

Effective Date of Agreement: Amount: Description (*Name and Location*):

BOND

Bond Number: Date (*Not earlier than Effective Date of Agreement*): Amount: Modifications to this Bond Form:

Surety and Contractor, intending to be legally bound hereby, subject to the terms set forth below, do each cause this Performance Bond to be duly executed by an authorized officer, agent, or representative.

CONTRACTOR AS PRINCIPAL

SURETY

		(Seal)			(Seal)
Contrac	tor's Name and Corporate Seal	,	Suret	y's Name and Corporate Seal	
Ву:	Signature		Ву:	Signature (Attach Power of Attorney)	
	Print Name			Print Name	
	Title			Title	
Attest:	Signature		Attest:	Signature	
	Title			Title	

Note: Provide execution by additional parties, such as joint venturers, if necessary.

Contractor and Surety, jointly and severally, bind themselves, their heirs, executors, administrate successors, and assigns to Owner for the performance of the Contract, which is incorporated herein by reference.

1. If Contractor performs the Contract, Surety and Contractor have no obligation under this Bond, except to participate in conferences as provided in Paragraph 2.1.

- 2. If there is no Owner Default, Surety's obligation under this Bond shall arise after:
 - 2.1 Owner has notified Contractor and Surety, at the addresses described in Paragraph 9 below, that Owner is considering declaring a Contractor Default and has requested and attempted to arrange a conference with Contractor and Surety to be held not later than 15 days after receipt of such notice to discuss methods of performing the Contract. If Owner, Contractor, and Surety agree, Contractor shall be allowed a reasonable time to perform the Contract, but such an agreement shall not waive Owner's right, if any, subsequently to declare a Contractor Default; and
 - 2.2 Owner has declared a Contractor Default and formally terminated Contractor's right to complete the Contract. Such Contractor Default shall not be declared earlier than 20 days after Contractor and Surety have received notice as provided in Paragraph 2.1; and
 - 2.3 Owner has agreed to pay the Balance of the Contract Price to:
 - 1. Surety in accordance with the terms of the Contract; or
 - 2. Another contractor selected pursuant to Paragraph 3.3 to perform the Contract.

3. When Owner has satisfied the conditions of Paragraph 2, Surety shall promptly, and at Surety's expense, take one of the following actions:

- 3.1 Arrange for Contractor, with consent of Owner, to perform and complete the Contract; or
- 3.2 Undertake to perform and complete the Contract itself, through its agents or through independent contractors; or
- 3.3 Obtain bids or negotiated proposals from qualified contractors acceptable to Owner for a contract for performance and completion of the Contract, arrange for a contract to be prepared for execution by Owner and contractor selected with Owner's concurrence, to be secured with performance and payment bonds executed by a qualified surety equivalent to the bonds issued on the Contract, and pay to Owner the amount of damages as described in Paragraph 5 in excess of the Balance of the Contract Price incurred by Owner resulting from Contractor Default; or
- 3.4 Waive its right to perform and complete, arrange for completion, or obtain a new contractor, and with reasonable promptness under the circumstances:
 - 1. After investigation, determine the amount for which it may be liable to Owner and, as soon as practicable after the amount is determined, tender payment therefor to Owner; or
 - 2. Deny liability in whole or in part and notify Owner citing reasons therefor.

4. If Surety does not proceed as provided in Paragraph 3 with reasonable promptness, Surety shall be deemed to be in default on this Bond 15 days after receipt of an additional written notice from Owner to Surety demanding that Surety perform its obligations under this Bond, and Owner shall be entitled to enforce any remedy available to Owner. If Surety proceeds as provided in Paragraph 3.4, and Owner refuses the payment tendered or Surety has denied liability, in whole or in part, without further notice Owner shall be entitled to enforce any remedy available to Owner available to Owner.

5. After Owner has terminated Contractor's right to complete the Contract, and if Surety elects to act under Paragraph 3.1, 3.2, or 3.3 above, then the responsibilities of Surety to Owner shall not be greater than those of Contractor under the Contract, and the responsibilities of Owner to Surety shall not be greater than those of Owner under the Contract. To the limit of the amount of this Bond, but subject to commitment by Owner of the Balance of the Contract Price to mitigation of costs and damages on the

- 5.1 The responsibilities of Contractor for correction of defective Work and completion of the Contract;
- 5.2 Additional legal, design professional, and delay costs resulting from Contractor's Default, and resulting from the actions of or failure to act of Surety under Paragraph 3; and
- 5.3 Liquidated damages, or if no liquidated damages are specified in the Contract, actual damages caused by delayed performance or non-performance of Contractor.

6. Surety shall not be liable to Owner or others for obligations of Contractor that are unrelated to the Contract, and the Balance of the Contract Price shall not be reduced or set off on account of any such unrelated obligations. No right of action shall accrue on this Bond to any person or entity other than Owner or its heirs, executors, administrators, or successors.

7. Surety hereby waives notice of any change, including changes of time, to Contract or to related subcontracts, purchase orders, and other obligations.

8. Any proceeding, legal or equitable, under this Bond may be instituted in any court of competent jurisdiction in the location in which the Work or part of the Work is located, and shall be instituted within two years after Contractor Default or within two years after Contractor ceased working or within two years after Surety refuses or fails to perform its obligations under this Bond, whichever occurs first. If the provisions of this paragraph are void or prohibited by law, the minimum period of limitation available to sureties as a defense in the jurisdiction of the suit shall be applicable.

9. Notice to Surety, Owner, or Contractor shall be mailed or delivered to the address shown on the signature page.

10. When this Bond has been furnished to comply with a statutory requirement in the location where the Contract was to be performed, any provision in this Bond conflicting with said statutory requirement shall be deemed deleted herefrom and provisions conforming to such statutory requirement shall be deemed incorporated herein. The intent is that this Bond shall be construed as a statutory bond and not as a common law bond.

11. Definitions.

- 11.1 Balance of the Contract Price: The total amount payable by Owner to Contractor under the Contract after all proper adjustments have been made, including allowance to Contractor of any amounts received or to be received by Owner in settlement of insurance or other Claims for damages to which Contractor is entitled, reduced by all valid and proper payments made to or on behalf of Contractor under the Contract.
- 11.2 Contract: The agreement between Owner and Contractor identified on the signature page, including all Contract Documents and changes thereto.
- 11.3 Contractor Default: Failure of Contractor, which has neither been remedied nor waived, to perform or otherwise to comply with the terms of the Contract.
- 11.4 Owner Default: Failure of Owner, which has neither been remedied nor waived, to pay Contractor as required by the Contract or to perform and complete or otherwise comply with the other terms thereof.

FOR INFORMATION ONLY – (*Name, Address and Telephone*) Surety Agency or Broker: Owner's Representative (*Engineer or other party*):

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

TABLE OF CONTENTS

	Page			
Article 1	Article 1—Definitions and Terminology1			
1.01	Defined Terms1			
1.02	Terminology6			
Article 2	- Preliminary Matters			
2.01	Delivery of Performance and Payment Bonds; Evidence of Insurance7			
2.02	Copies of Documents			
2.03	Before Starting Construction7			
2.04	Preconstruction Conference; Designation of Authorized Representatives			
2.05	Acceptance of Schedules8			
2.06	Electronic Transmittals8			
Article 3	-Contract Documents: Intent, Requirements, Reuse9			
3.01	Intent9			
3.02	Reference Standards9			
3.03	Reporting and Resolving Discrepancies			
3.04	Requirements of the Contract Documents			
3.05	Reuse of Documents			
Article 4	-Commencement and Progress of the Work11			
4.01	Commencement of Contract Times; Notice to Proceed11			
4.02	Starting the Work11			
4.03	Reference Points			
4.04	Progress Schedule			
4.05	Delays in Contractor's Progress12			
Article 5	-Site; Subsurface and Physical Conditions; Hazardous Environmental Conditions			
5.01	Availability of Lands			
5.02	Use of Site and Other Areas14			
5.03	Subsurface and Physical Conditions15			
5.04	Differing Subsurface or Physical Conditions16			

5.05	Underground Facilities	17
5.06	Hazardous Environmental Conditions at Site	19
Article 6-	-Bonds and Insurance	21
6.01	Performance, Payment, and Other Bonds	21
6.02	Insurance—General Provisions	22
6.03	Contractor's Insurance	24
6.04	Builder's Risk and Other Property Insurance	25
6.05	Property Losses; Subrogation	25
6.06	Receipt and Application of Property Insurance Proceeds	27
Article 7-	-Contractor's Responsibilities	27
7.01	Contractor's Means and Methods of Construction	27
7.02	Supervision and Superintendence	27
7.03	Labor; Working Hours	27
7.04	Services, Materials, and Equipment	28
7.05	"Or Equals"	28
7.06	Substitutes	29
7.07	Concerning Subcontractors and Suppliers	31
7.08	Patent Fees and Royalties	32
7.09	Permits	33
7.10	Taxes	33
7.11	Laws and Regulations	33
7.12	Record Documents	33
7.13	Safety and Protection	34
7.14	Hazard Communication Programs	35
7.15	Emergencies	35
7.16	Submittals	35
7.17	Contractor's General Warranty and Guarantee	38
7.18	Indemnification	39
7.19	Delegation of Professional Design Services	39
Article 8-	-Other Work at the Site	40
8.01	Other Work	40
8.02	Coordination	41
8.03	Legal Relationships	41

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

TOC Page 2 of 5

A	rticle 9-	–Owner's Responsibilities	42
	9.01	Communications to Contractor	42
	9.02	Replacement of Engineer	42
	9.03	Furnish Data	42
	9.04	Pay When Due	42
	9.05	Lands and Easements; Reports, Tests, and Drawings	43
	9.06	Insurance	43
	9.07	Change Orders	43
	9.08	Inspections, Tests, and Approvals	43
	9.09	Limitations on Owner's Responsibilities	43
	9.10	Undisclosed Hazardous Environmental Condition	43
	9.11	Evidence of Financial Arrangements	43
	9.12	Safety Programs	43
A	rticle 10)—Engineer's Status During Construction	44
	10.01	Owner's Representative	44
	10.02	Visits to Site	44
	10.03	Resident Project Representative	44
	10.04	Engineer's Authority	44
	10.05	Determinations for Unit Price Work	45
	10.06	Decisions on Requirements of Contract Documents and Acceptability of Work	45
	10.07	Limitations on Engineer's Authority and Responsibilities	45
	10.08	Compliance with Safety Program	45
A	rticle 11	-Changes to the Contract	46
	11.01	Amending and Supplementing the Contract	46
	11.02	Change Orders	46
	11.03	Work Change Directives	46
	11.04	Field Orders	47
	11.05	Owner-Authorized Changes in the Work	47
	11.06	Unauthorized Changes in the Work	47
	11.07	Change of Contract Price	47
	11.08	Change of Contract Times	49
	11.09	Change Proposals	49
	11.10	Notification to Surety	50

EJCDC[®] C-700, Standard General Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

TOC Page 3 of 5

Article 12–	-Claims	50
12.01	Claims	50
Article 13–	-Cost of the Work; Allowances; Unit Price Work	51
13.01	Cost of the Work	51
13.02	Allowances	55
13.03	Unit Price Work	55
Article 14–	-Tests and Inspections; Correction, Removal, or Acceptance of Defective Work	56
14.01	Access to Work	56
14.02	Tests, Inspections, and Approvals	56
14.03	Defective Work	57
14.04	Acceptance of Defective Work	58
14.05	Uncovering Work	58
14.06	Owner May Stop the Work	58
14.07	Owner May Correct Defective Work	59
Article 15–	-Payments to Contractor; Set-Offs; Completion; Correction Period	59
15.01	Progress Payments	59
15.02	Contractor's Warranty of Title	62
15.03	Substantial Completion	62
15.04	Partial Use or Occupancy	63
15.05	Final Inspection	64
15.06	Final Payment	64
15.07	Waiver of Claims	65
15.08	Correction Period	66
Article 16–	-Suspension of Work and Termination	67
16.01	Owner May Suspend Work	67
16.02	Owner May Terminate for Cause	67
16.03	Owner May Terminate for Convenience	68
16.04	Contractor May Stop Work or Terminate	68
Article 17–	-Final Resolution of Disputes	69
17.01	Methods and Procedures	69
Article 18–	-Miscellaneous	69
18.01	Giving Notice	69
18.02	Computation of Times	69

EJCDC[®] C-700, Standard General Conditions of the Construction Contract.

Copyright[©] 2018 National Society of Professional Engineers, American Council of Engineering Companies,

and American Society of Civil Engineers. All rights reserved.

TOC Page 4 of 5

	Cumulative Remedies	70
18.04	Limitation of Damages	70
18.05	No Waiver	70
18.06	Survival of Obligations	70
18.07	Controlling Law	70
18.08	Assignment of Contract	70
18.09	Successors and Assigns	70
18.10	Headings	70

STANDARD GENERAL CONDITIONS OF THE CONSTRUCTION CONTRACT

ARTICLE 1—DEFINITIONS AND TERMINOLOGY

1.01 Defined Terms

- A. Wherever used in the Bidding Requirements or Contract Documents, a term printed with initial capital letters, including the term's singular and plural forms, will have the meaning indicated in the definitions below. In addition to terms specifically defined, terms with initial capital letters in the Contract Documents include references to identified articles and paragraphs, and the titles of other documents or forms.
 - 1. Addenda—Written or graphic instruments issued prior to the opening of Bids which clarify, correct, or change the Bidding Requirements or the proposed Contract Documents.
 - 2. Agreement—The written instrument, executed by Owner and Contractor, that sets forth the Contract Price and Contract Times, identifies the parties and the Engineer, and designates the specific items that are Contract Documents.
 - 3. *Application for Payment*—The document prepared by Contractor, in a form acceptable to Engineer, to request progress or final payments, and which is to be accompanied by such supporting documentation as is required by the Contract Documents.
 - 4. *Bid*—The offer of a Bidder submitted on the prescribed form setting forth the prices for the Work to be performed.
 - 5. Bidder—An individual or entity that submits a Bid to Owner.
 - 6. *Bidding Documents*—The Bidding Requirements, the proposed Contract Documents, and all Addenda.
 - 7. *Bidding Requirements*—The Advertisement or invitation to bid, Instructions to Bidders, Bid Bond or other Bid security, if any, the Bid Form, and the Bid with any attachments.
 - 8. Change Order—A document which is signed by Contractor and Owner and authorizes an addition, deletion, or revision in the Work or an adjustment in the Contract Price or the Contract Times, or other revision to the Contract, issued on or after the Effective Date of the Contract.
 - 9. Change Proposal—A written request by Contractor, duly submitted in compliance with the procedural requirements set forth herein, seeking an adjustment in Contract Price or Contract Times; contesting an initial decision by Engineer concerning the requirements of the Contract Documents or the acceptability of Work under the Contract Documents; challenging a set-off against payments due; or seeking other relief with respect to the terms of the Contract.
 - 10. Claim
 - *a.* A demand or assertion by Owner directly to Contractor, duly submitted in compliance with the procedural requirements set forth herein, seeking an adjustment of Contract Price or Contract Times; contesting an initial decision by Engineer concerning the

requirements of the Contract Documents or the acceptability of Work under the Contract Documents; contesting Engineer's decision regarding a Change Proposal; seeking resolution of a contractual issue that Engineer has declined to address; or seeking other relief with respect to the terms of the Contract.

- b. A demand or assertion by Contractor directly to Owner, duly submitted in compliance with the procedural requirements set forth herein, contesting Engineer's decision regarding a Change Proposal, or seeking resolution of a contractual issue that Engineer has declined to address.
- c. A demand or assertion by Owner or Contractor, duly submitted in compliance with the procedural requirements set forth herein, made pursuant to Paragraph 12.01.A.4, concerning disputes arising after Engineer has issued a recommendation of final payment.
- *d.* A demand for money or services by a third party is not a Claim.
- 11. Constituent of Concern—Asbestos, petroleum, radioactive materials, polychlorinated biphenyls (PCBs), lead-based paint (as defined by the HUD/EPA standard), hazardous waste, and any substance, product, waste, or other material of any nature whatsoever that is or becomes listed, regulated, or addressed pursuant to Laws and Regulations regulating, relating to, or imposing liability or standards of conduct concerning, any hazardous, toxic, or dangerous waste, substance, or material.
- 12. *Contract*—The entire and integrated written contract between Owner and Contractor concerning the Work.
- 13. Contract Documents—Those items so designated in the Agreement, and which together comprise the Contract.
- 14. *Contract Price*—The money that Owner has agreed to pay Contractor for completion of the Work in accordance with the Contract Documents.
- 15. *Contract Times*—The number of days or the dates by which Contractor shall: (a) achieve Milestones, if any; (b) achieve Substantial Completion; and (c) complete the Work.
- 16. *Contractor*—The individual or entity with which Owner has contracted for performance of the Work.
- 17. *Cost of the Work*—See Paragraph 13.01 for definition.
- 18. *Drawings*—The part of the Contract that graphically shows the scope, extent, and character of the Work to be performed by Contractor.
- 19. *Effective Date of the Contract*—The date, indicated in the Agreement, on which the Contract becomes effective.
- 20. *Electronic Document*—Any Project-related correspondence, attachments to correspondence, data, documents, drawings, information, or graphics, including but not limited to Shop Drawings and other Submittals, that are in an electronic or digital format.
- Electronic Means—Electronic mail (email), upload/download from a secure Project website, or other communications methods that allow: (a) the transmission or communication of Electronic Documents; (b) the documentation of transmissions, including sending and receipt; (c) printing of the transmitted Electronic Document by the

recipient; (d) the storage and archiving of the Electronic Document by sender and recipient; and (e) the use by recipient of the Electronic Document for purposes permitted by this Contract. Electronic Means does not include the use of text messaging, or of Facebook, Twitter, Instagram, or similar social media services for transmission of Electronic Documents.

- 22. Engineer—The individual or entity named as such in the Agreement.
- 23. *Field Order*—A written order issued by Engineer which requires minor changes in the Work but does not change the Contract Price or the Contract Times.
- 24. *Hazardous Environmental Condition*—The presence at the Site of Constituents of Concern in such quantities or circumstances that may present a danger to persons or property exposed thereto.
 - a. The presence at the Site of materials that are necessary for the execution of the Work, or that are to be incorporated into the Work, and that are controlled and contained pursuant to industry practices, Laws and Regulations, and the requirements of the Contract, is not a Hazardous Environmental Condition.
 - b. The presence of Constituents of Concern that are to be removed or remediated as part of the Work is not a Hazardous Environmental Condition.
 - c. The presence of Constituents of Concern as part of the routine, anticipated, and obvious working conditions at the Site, is not a Hazardous Environmental Condition.
- 25. Laws and Regulations; Laws or Regulations—Any and all applicable laws, statutes, rules, regulations, ordinances, codes, and binding decrees, resolutions, and orders of any and all governmental bodies, agencies, authorities, and courts having jurisdiction.
- 26. *Liens*—Charges, security interests, or encumbrances upon Contract-related funds, real property, or personal property.
- 27. Milestone—A principal event in the performance of the Work that the Contract requires Contractor to achieve by an intermediate completion date, or by a time prior to Substantial Completion of all the Work.
- Notice of Award—The written notice by Owner to a Bidder of Owner's acceptance of the Bid.
- Notice to Proceed—A written notice by Owner to Contractor fixing the date on which the Contract Times will commence to run and on which Contractor shall start to perform the Work.
- 30. *Owner*—The individual or entity with which Contractor has contracted regarding the Work, and which has agreed to pay Contractor for the performance of the Work, pursuant to the terms of the Contract.
- 31. *Progress Schedule*—A schedule, prepared and maintained by Contractor, describing the sequence and duration of the activities comprising Contractor's plan to accomplish the Work within the Contract Times.
- 32. *Project*—The total undertaking to be accomplished for Owner by engineers, contractors, and others, including planning, study, design, construction, testing, commissioning, and start-up, and of which the Work to be performed under the Contract Documents is a part.

- 33. *Resident Project Representative*—The authorized representative of Engineer assigned to assist Engineer at the Site. As used herein, the term Resident Project Representative (RPR) includes any assistants or field staff of Resident Project Representative.
- 34. *Samples*—Physical examples of materials, equipment, or workmanship that are representative of some portion of the Work and that establish the standards by which such portion of the Work will be judged.
- 35. *Schedule of Submittals*—A schedule, prepared and maintained by Contractor, of required submittals and the time requirements for Engineer's review of the submittals.
- 36. Schedule of Values—A schedule, prepared and maintained by Contractor, allocating portions of the Contract Price to various portions of the Work and used as the basis for reviewing Contractor's Applications for Payment.
- 37. Shop Drawings—All drawings, diagrams, illustrations, schedules, and other data or information that are specifically prepared or assembled by or for Contractor and submitted by Contractor to illustrate some portion of the Work. Shop Drawings, whether approved or not, are not Drawings and are not Contract Documents.
- 38. Site—Lands or areas indicated in the Contract Documents as being furnished by Owner upon which the Work is to be performed, including rights-of-way and easements, and such other lands or areas furnished by Owner which are designated for the use of Contractor.
- 39. *Specifications*—The part of the Contract that consists of written requirements for materials, equipment, systems, standards, and workmanship as applied to the Work, and certain administrative requirements and procedural matters applicable to the Work.
- 40. *Subcontractor*—An individual or entity having a direct contract with Contractor or with any other Subcontractor for the performance of a part of the Work.
- 41. Submittal—A written or graphic document, prepared by or for Contractor, which the Contract Documents require Contractor to submit to Engineer, or that is indicated as a Submittal in the Schedule of Submittals accepted by Engineer. Submittals may include Shop Drawings and Samples; schedules; product data; Owner-delegated designs; sustainable design information; information on special procedures; testing plans; results of tests and evaluations, source quality-control testing and inspections; warranties and certifications; Suppliers' instructions and reports; records of delivery of spare parts and tools; operations and maintenance data; Project photographic documentation; record documents; and other such documents required by the Contract Documents. Submittals, whether or not approved or accepted by Engineer, are not Contract Documents. Change Proposals, Change Orders, Claims, notices, Applications for Payment, and requests for interpretation or clarification are not Submittals.
- 42. Substantial Completion—The time at which the Work (or a specified part thereof) has progressed to the point where, in the opinion of Engineer, the Work (or a specified part thereof) is sufficiently complete, in accordance with the Contract Documents, so that the Work (or a specified part thereof) can be utilized for the purposes for which it is intended. The terms "substantially complete" and "substantially completed" as applied to all or part of the Work refer to Substantial Completion of such Work.

- 43. *Successful Bidder*—The Bidder to which the Owner makes an award of contract.
- 44. *Supplementary Conditions*—The part of the Contract that amends or supplements these General Conditions.
- 45. *Supplier*—A manufacturer, fabricator, supplier, distributor, or vendor having a direct contract with Contractor or with any Subcontractor to furnish materials or equipment to be incorporated in the Work by Contractor or a Subcontractor.
- 46. Technical Data
 - a. Those items expressly identified as Technical Data in the Supplementary Conditions, with respect to either (1) existing subsurface conditions at or adjacent to the Site, or existing physical conditions at or adjacent to the Site including existing surface or subsurface structures (except Underground Facilities) or (2) Hazardous Environmental Conditions at the Site.
 - b. If no such express identifications of Technical Data have been made with respect to conditions at the Site, then Technical Data is defined, with respect to conditions at the Site under Paragraphs 5.03, 5.04, and 5.06, as the data contained in boring logs, recorded measurements of subsurface water levels, assessments of the condition of subsurface facilities, laboratory test results, and other factual, objective information regarding conditions at the Site that are set forth in any geotechnical, environmental, or other Site or facilities conditions report prepared for the Project and made available to Contractor.
 - c. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data, and instead Underground Facilities are shown or indicated on the Drawings.
- 47. Underground Facilities—All active or not-in-service underground lines, pipelines, conduits, ducts, encasements, cables, wires, manholes, vaults, tanks, tunnels, or other such facilities or systems at the Site, including but not limited to those facilities or systems that produce, transmit, distribute, or convey telephone or other communications, cable television, fiber optic transmissions, power, electricity, light, heat, gases, oil, crude oil products, liquid petroleum products, water, steam, waste, wastewater, storm water, other liquids or chemicals, or traffic or other control systems. An abandoned facility or system is not an Underground Facility.
- 48. Unit Price Work—Work to be paid for on the basis of unit prices.
- 49. Work—The entire construction or the various separately identifiable parts thereof required to be provided under the Contract Documents. Work includes and is the result of performing or providing all labor, services, and documentation necessary to produce such construction; furnishing, installing, and incorporating all materials and equipment into such construction; and may include related services such as testing, start-up, and commissioning, all as required by the Contract Documents.
- 50. Work Change Directive—A written directive to Contractor issued on or after the Effective Date of the Contract, signed by Owner and recommended by Engineer, ordering an addition, deletion, or revision in the Work.

1.02 Terminology

- A. The words and terms discussed in Paragraphs 1.02.B, C, D, and E are not defined terms that require initial capital letters, but, when used in the Bidding Requirements or Contract Documents, have the indicated meaning.
- B. Intent of Certain Terms or Adjectives: The Contract Documents include the terms "as allowed," "as approved," "as ordered," "as directed" or terms of like effect or import to authorize an exercise of professional judgment by Engineer. In addition, the adjectives "reasonable," "suitable," "acceptable," "proper," "satisfactory," or adjectives of like effect or import are used to describe an action or determination of Engineer as to the Work. It is intended that such exercise of professional judgment, action, or determination will be solely to evaluate, in general, the Work for compliance with the information in the Contract Documents and with the design concept of the Project as a functioning whole as shown or indicated in the Contract Documents (unless there is a specific statement indicating otherwise). The use of any such term or adjective is not intended to and shall not be effective to assign to Engineer any duty or authority to supervise or direct the performance of the Work, or any duty or authority to undertake responsibility contrary to the provisions of Article 10 or any other provision of the Contract Documents.
- C. *Day*: The word "day" means a calendar day of 24 hours measured from midnight to the next midnight.
- D. *Defective*: The word "defective," when modifying the word "Work," refers to Work that is unsatisfactory, faulty, or deficient in that it:
 - 1. does not conform to the Contract Documents;
 - 2. does not meet the requirements of any applicable inspection, reference standard, test, or approval referred to in the Contract Documents; or
 - 3. has been damaged prior to Engineer's recommendation of final payment (unless responsibility for the protection thereof has been assumed by Owner at Substantial Completion in accordance with Paragraph 15.03 or Paragraph 15.04).
- E. Furnish, Install, Perform, Provide
 - 1. The word "furnish," when used in connection with services, materials, or equipment, means to supply and deliver said services, materials, or equipment to the Site (or some other specified location) ready for use or installation and in usable or operable condition.
 - 2. The word "install," when used in connection with services, materials, or equipment, means to put into use or place in final position said services, materials, or equipment complete and ready for intended use.
 - 3. The words "perform" or "provide," when used in connection with services, materials, or equipment, means to furnish and install said services, materials, or equipment complete and ready for intended use.
 - 4. If the Contract Documents establish an obligation of Contractor with respect to specific services, materials, or equipment, but do not expressly use any of the four words "furnish," "install," "perform," or "provide," then Contractor shall furnish and install said services, materials, or equipment complete and ready for intended use.

- F. Contract Price or Contract Times: References to a change in "Contract Price or Contract Times" or "Contract Times or Contract Price" or similar, indicate that such change applies to (1) Contract Price, (2) Contract Times, or (3) both Contract Price and Contract Times, as warranted, even if the term "or both" is not expressed.
- G. Unless stated otherwise in the Contract Documents, words or phrases that have a well-known technical or construction industry or trade meaning are used in the Contract Documents in accordance with such recognized meaning.

ARTICLE 2—PRELIMINARY MATTERS

2.01 Delivery of Performance and Payment Bonds; Evidence of Insurance

- A. *Performance and Payment Bonds*: When Contractor delivers the signed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner the performance bond and payment bond (if the Contract requires Contractor to furnish such bonds).
- B. *Evidence of Contractor's Insurance*: When Contractor delivers the signed counterparts of the Agreement to Owner, Contractor shall also deliver to Owner, with copies to each additional insured (as identified in the Contract), the certificates, endorsements, and other evidence of insurance required to be provided by Contractor in accordance with Article 6, except to the extent the Supplementary Conditions expressly establish other dates for delivery of specific insurance policies.
- C. *Evidence of Owner's Insurance*: After receipt of the signed counterparts of the Agreement and all required bonds and insurance documentation, Owner shall promptly deliver to Contractor, with copies to each additional insured (as identified in the Contract), the certificates and other evidence of insurance required to be provided by Owner under Article 6.

2.02 Copies of Documents

- A. Owner shall furnish to Contractor four printed copies of the Contract (including one fully signed counterpart of the Agreement), and one copy in electronic portable document format (PDF). Additional printed copies will be furnished upon request at the cost of reproduction.
- B. Owner shall maintain and safeguard at least one original printed record version of the Contract, including Drawings and Specifications signed and sealed by Engineer and other design professionals. Owner shall make such original printed record version of the Contract available to Contractor for review. Owner may delegate the responsibilities under this provision to Engineer.

2.03 Before Starting Construction

- A. *Preliminary Schedules*: Within 10 days after the Effective Date of the Contract (or as otherwise required by the Contract Documents), Contractor shall submit to Engineer for timely review:
 - 1. a preliminary Progress Schedule indicating the times (numbers of days or dates) for starting and completing the various stages of the Work, including any Milestones specified in the Contract;
 - 2. a preliminary Schedule of Submittals; and
 - 3. a preliminary Schedule of Values for all of the Work which includes quantities and prices of items which when added together equal the Contract Price and subdivides the Work

into component parts in sufficient detail to serve as the basis for progress payments during performance of the Work. Such prices will include an appropriate amount of overhead and profit applicable to each item of Work.

2.04 Preconstruction Conference; Designation of Authorized Representatives

- A. Before any Work at the Site is started, a conference attended by Owner, Contractor, Engineer, and others as appropriate will be held to establish a working understanding among the parties as to the Work, and to discuss the schedules referred to in Paragraph 2.03.A, procedures for handling Shop Drawings, Samples, and other Submittals, processing Applications for Payment, electronic or digital transmittals, and maintaining required records.
- B. At this conference Owner and Contractor each shall designate, in writing, a specific individual to act as its authorized representative with respect to the services and responsibilities under the Contract. Such individuals shall have the authority to transmit and receive information, render decisions relative to the Contract, and otherwise act on behalf of each respective party.

2.05 Acceptance of Schedules

- A. At least 10 days before submission of the first Application for Payment a conference, attended by Contractor, Engineer, and others as appropriate, will be held to review the schedules submitted in accordance with Paragraph 2.03.A. No progress payment will be made to Contractor until acceptable schedules are submitted to Engineer.
 - The Progress Schedule will be acceptable to Engineer if it provides an orderly progression of the Work to completion within the Contract Times. Such acceptance will not impose on Engineer responsibility for the Progress Schedule, for sequencing, scheduling, or progress of the Work, nor interfere with or relieve Contractor from Contractor's full responsibility therefor.
 - 2. Contractor's Schedule of Submittals will be acceptable to Engineer if it provides a workable arrangement for reviewing and processing the required submittals.
 - 3. Contractor's Schedule of Values will be acceptable to Engineer as to form and substance if it provides a reasonable allocation of the Contract Price to the component parts of the Work.
 - 4. If a schedule is not acceptable, Contractor will have an additional 10 days to revise and resubmit the schedule.

2.06 Electronic Transmittals

- A. Except as otherwise stated elsewhere in the Contract, the Owner, Engineer, and Contractor may send, and shall accept, Electronic Documents transmitted by Electronic Means.
- B. If the Contract does not establish protocols for Electronic Means, then Owner, Engineer, and Contractor shall jointly develop such protocols.
- C. Subject to any governing protocols for Electronic Means, when transmitting Electronic Documents by Electronic Means, the transmitting party makes no representations as to long-term compatibility, usability, or readability of the Electronic Documents resulting from the recipient's use of software application packages, operating systems, or computer hardware differing from those used in the drafting or transmittal of the Electronic Documents.

ARTICLE 3—CONTRACT DOCUMENTS: INTENT, REQUIREMENTS, REUSE

3.01 Intent

- A. The Contract Documents are complementary; what is required by one Contract Document is as binding as if required by all.
- B. It is the intent of the Contract Documents to describe a functionally complete Project (or part thereof) to be constructed in accordance with the Contract Documents.
- C. Unless otherwise stated in the Contract Documents, if there is a discrepancy between the electronic versions of the Contract Documents (including any printed copies derived from such electronic versions) and the printed record version, the printed record version will govern.
- D. The Contract supersedes prior negotiations, representations, and agreements, whether written or oral.
- E. Engineer will issue clarifications and interpretations of the Contract Documents as provided herein.
- F. Any provision or part of the Contract Documents held to be void or unenforceable under any Law or Regulation will be deemed stricken, and all remaining provisions will continue to be valid and binding upon Owner and Contractor, which agree that the Contract Documents will be reformed to replace such stricken provision or part thereof with a valid and enforceable provision that comes as close as possible to expressing the intention of the stricken provision.
- G. Nothing in the Contract Documents creates:
 - 1. any contractual relationship between Owner or Engineer and any Subcontractor, Supplier, or other individual or entity performing or furnishing any of the Work, for the benefit of such Subcontractor, Supplier, or other individual or entity; or
 - 2. any obligation on the part of Owner or Engineer to pay or to see to the payment of any money due any such Subcontractor, Supplier, or other individual or entity, except as may otherwise be required by Laws and Regulations.

3.02 Reference Standards

- A. Standards Specifications, Codes, Laws and Regulations
 - Reference in the Contract Documents to standard specifications, manuals, reference standards, or codes of any technical society, organization, or association, or to Laws or Regulations, whether such reference be specific or by implication, means the standard specification, manual, reference standard, code, or Laws or Regulations in effect at the time of opening of Bids (or on the Effective Date of the Contract if there were no Bids), except as may be otherwise specifically stated in the Contract Documents.
 - 2. No provision of any such standard specification, manual, reference standard, or code, and no instruction of a Supplier, will be effective to change the duties or responsibilities of Owner, Contractor, or Engineer from those set forth in the part of the Contract Documents prepared by or for Engineer. No such provision or instruction shall be effective to assign to Owner or Engineer any duty or authority to supervise or direct the performance of the Work, or any duty or authority to undertake responsibility

inconsistent with the provisions of the part of the Contract Documents prepared by or for Engineer.

3.03 Reporting and Resolving Discrepancies

- A. Reporting Discrepancies
 - 1. Contractor's Verification of Figures and Field Measurements: Before undertaking each part of the Work, Contractor shall carefully study the Contract Documents, and check and verify pertinent figures and dimensions therein, particularly with respect to applicable field measurements. Contractor shall promptly report in writing to Engineer any conflict, error, ambiguity, or discrepancy that Contractor discovers, or has actual knowledge of, and shall not proceed with any Work affected thereby until the conflict, error, ambiguity, or discrepancy is resolved by a clarification or interpretation by Engineer, or by an amendment or supplement to the Contract issued pursuant to Paragraph 11.01.
 - 2. Contractor's Review of Contract Documents: If, before or during the performance of the Work, Contractor discovers any conflict, error, ambiguity, or discrepancy within the Contract Documents, or between the Contract Documents and (a) any applicable Law or Regulation, (b) actual field conditions, (c) any standard specification, manual, reference standard, or code, or (d) any instruction of any Supplier, then Contractor shall promptly report it to Engineer in writing. Contractor shall not proceed with the Work affected thereby (except in an emergency as required by Paragraph 7.15) until the conflict, error, ambiguity, or discrepancy is resolved, by a clarification or interpretation by Engineer, or by an amendment or supplement to the Contract issued pursuant to Paragraph 11.01.
 - 3. Contractor shall not be liable to Owner or Engineer for failure to report any conflict, error, ambiguity, or discrepancy in the Contract Documents unless Contractor had actual knowledge thereof.
- B. Resolving Discrepancies
 - 1. Except as may be otherwise specifically stated in the Contract Documents, the provisions of the part of the Contract Documents prepared by or for Engineer take precedence in resolving any conflict, error, ambiguity, or discrepancy between such provisions of the Contract Documents and:
 - a. the provisions of any standard specification, manual, reference standard, or code, or the instruction of any Supplier (whether or not specifically incorporated by reference as a Contract Document); or
 - b. the provisions of any Laws or Regulations applicable to the performance of the Work (unless such an interpretation of the provisions of the Contract Documents would result in violation of such Law or Regulation).

3.04 Requirements of the Contract Documents

A. During the performance of the Work and until final payment, Contractor and Owner shall submit to the Engineer in writing all matters in question concerning the requirements of the Contract Documents (sometimes referred to as requests for information or interpretation— RFIs), or relating to the acceptability of the Work under the Contract Documents, as soon as possible after such matters arise. Engineer will be the initial interpreter of the requirements of the Contract Documents, and judge of the acceptability of the Work.

- B. Engineer will, with reasonable promptness, render a written clarification, interpretation, or decision on the issue submitted, or initiate an amendment or supplement to the Contract Documents. Engineer's written clarification, interpretation, or decision will be final and binding on Contractor, unless it appeals by submitting a Change Proposal, and on Owner, unless it appeals by filing a Claim.
- C. If a submitted matter in question concerns terms and conditions of the Contract Documents that do not involve (1) the performance or acceptability of the Work under the Contract Documents, (2) the design (as set forth in the Drawings, Specifications, or otherwise), or (3) other engineering or technical matters, then Engineer will promptly notify Owner and Contractor in writing that Engineer is unable to provide a decision or interpretation. If Owner and Contractor are unable to agree on resolution of such a matter in question, either party may pursue resolution as provided in Article 12.

3.05 *Reuse of Documents*

- A. Contractor and its Subcontractors and Suppliers shall not:
 - have or acquire any title to or ownership rights in any of the Drawings, Specifications, or other documents (or copies of any thereof) prepared by or bearing the seal of Engineer or its consultants, including electronic media versions, or reuse any such Drawings, Specifications, other documents, or copies thereof on extensions of the Project or any other project without written consent of Owner and Engineer and specific written verification or adaptation by Engineer; or
 - 2. have or acquire any title or ownership rights in any other Contract Documents, reuse any such Contract Documents for any purpose without Owner's express written consent, or violate any copyrights pertaining to such Contract Documents.
- B. The prohibitions of this Paragraph 3.05 will survive final payment, or termination of the Contract. Nothing herein precludes Contractor from retaining copies of the Contract Documents for record purposes.

ARTICLE 4—COMMENCEMENT AND PROGRESS OF THE WORK

4.01 Commencement of Contract Times; Notice to Proceed

- A. The Contract Times will commence to run on the 30th day after the Effective Date of the Contract or, if a Notice to Proceed is given, on the day indicated in the Notice to Proceed. A Notice to Proceed may be given at any time within 30 days after the Effective Date of the Contract. In no event will the Contract Times commence to run later than the 60th day after the day of Bid opening or the 30th day after the Effective Date of the Contract, whichever date is earlier.
- 4.02 *Starting the Work*
 - A. Contractor shall start to perform the Work on the date when the Contract Times commence to run. No Work may be done at the Site prior to such date.
- 4.03 Reference Points
 - A. Owner shall provide engineering surveys to establish reference points for construction which in Engineer's judgment are necessary to enable Contractor to proceed with the Work. Contractor shall be responsible for laying out the Work, shall protect and preserve the

established reference points and property monuments, and shall make no changes or relocations without the prior written approval of Owner. Contractor shall report to Engineer whenever any reference point or property monument is lost or destroyed or requires relocation because of necessary changes in grades or locations, and shall be responsible for the accurate replacement or relocation of such reference points or property monuments by professionally qualified personnel.

4.04 *Progress Schedule*

- A. Contractor shall adhere to the Progress Schedule established in accordance with Paragraph 2.05 as it may be adjusted from time to time as provided below.
 - 1. Contractor shall submit to Engineer for acceptance (to the extent indicated in Paragraph 2.05) proposed adjustments in the Progress Schedule that will not result in changing the Contract Times.
 - 2. Proposed adjustments in the Progress Schedule that will change the Contract Times must be submitted in accordance with the requirements of Article 11.
- B. Contractor shall carry on the Work and adhere to the Progress Schedule during all disputes or disagreements with Owner. No Work will be delayed or postponed pending resolution of any disputes or disagreements, or during any appeal process, except as permitted by Paragraph 16.04, or as Owner and Contractor may otherwise agree in writing.

4.05 Delays in Contractor's Progress

- A. If Owner, Engineer, or anyone for whom Owner is responsible, delays, disrupts, or interferes with the performance or progress of the Work, then Contractor shall be entitled to an equitable adjustment in Contract Price or Contract Times.
- B. Contractor shall not be entitled to an adjustment in Contract Price or Contract Times for delay, disruption, or interference caused by or within the control of Contractor. Delay, disruption, and interference attributable to and within the control of a Subcontractor or Supplier shall be deemed to be within the control of Contractor.
- C. If Contractor's performance or progress is delayed, disrupted, or interfered with by unanticipated causes not the fault of and beyond the control of Owner, Contractor, and those for which they are responsible, then Contractor shall be entitled to an equitable adjustment in Contract Times. Such an adjustment will be Contractor's sole and exclusive remedy for the delays, disruption, and interference described in this paragraph. Causes of delay, disruption, or interference that may give rise to an adjustment in Contract Times under this paragraph include but are not limited to the following:
 - 1. Severe and unavoidable natural catastrophes such as fires, floods, epidemics, and earthquakes;
 - 2. Abnormal weather conditions;
 - 3. Acts or failures to act of third-party utility owners or other third-party entities (other than those third-party utility owners or other third-party entities performing other work at or adjacent to the Site as arranged by or under contract with Owner, as contemplated in Article 8); and
 - 4. Acts of war or terrorism.

- D. Contractor's entitlement to an adjustment of Contract Times or Contract Price is limited as follows:
 - 1. Contractor's entitlement to an adjustment of the Contract Times is conditioned on the delay, disruption, or interference adversely affecting an activity on the critical path to completion of the Work, as of the time of the delay, disruption, or interference.
 - 2. Contractor shall not be entitled to an adjustment in Contract Price for any delay, disruption, or interference if such delay is concurrent with a delay, disruption, or interference caused by or within the control of Contractor. Such a concurrent delay by Contractor shall not preclude an adjustment of Contract Times to which Contractor is otherwise entitled.
 - 3. Adjustments of Contract Times or Contract Price are subject to the provisions of Article 11.
- E. Each Contractor request or Change Proposal seeking an increase in Contract Times or Contract Price must be supplemented by supporting data that sets forth in detail the following:
 - 1. The circumstances that form the basis for the requested adjustment;
 - 2. The date upon which each cause of delay, disruption, or interference began to affect the progress of the Work;
 - 3. The date upon which each cause of delay, disruption, or interference ceased to affect the progress of the Work;
 - 4. The number of days' increase in Contract Times claimed as a consequence of each such cause of delay, disruption, or interference; and
 - 5. The impact on Contract Price, in accordance with the provisions of Paragraph 11.07.

Contractor shall also furnish such additional supporting documentation as Owner or Engineer may require including, where appropriate, a revised progress schedule indicating all the activities affected by the delay, disruption, or interference, and an explanation of the effect of the delay, disruption, or interference on the critical path to completion of the Work.

- F. Delays, disruption, and interference to the performance or progress of the Work resulting from the existence of a differing subsurface or physical condition, an Underground Facility that was not shown or indicated by the Contract Documents, or not shown or indicated with reasonable accuracy, and those resulting from Hazardous Environmental Conditions, are governed by Article 5, together with the provisions of Paragraphs 4.05.D and 4.05.E.
- G. Paragraph 8.03 addresses delays, disruption, and interference to the performance or progress of the Work resulting from the performance of certain other work at or adjacent to the Site.

ARTICLE 5—SITE; SUBSURFACE AND PHYSICAL CONDITIONS; HAZARDOUS ENVIRONMENTAL CONDITIONS

- 5.01 Availability of Lands
 - A. Owner shall furnish the Site. Owner shall notify Contractor in writing of any encumbrances or restrictions not of general application but specifically related to use of the Site with which Contractor must comply in performing the Work.

- B. Upon reasonable written request, Owner shall furnish Contractor with a current statement of record legal title and legal description of the lands upon which permanent improvements are to be made and Owner's interest therein as necessary for giving notice of or filing a mechanic's or construction lien against such lands in accordance with applicable Laws and Regulations.
- C. Contractor shall provide for all additional lands and access thereto that may be required for temporary construction facilities or storage of materials and equipment.

5.02 Use of Site and Other Areas

- A. Limitation on Use of Site and Other Areas
 - 1. Contractor shall confine construction equipment, temporary construction facilities, the storage of materials and equipment, and the operations of workers to the Site, adjacent areas that Contractor has arranged to use through construction easements or otherwise, and other adjacent areas permitted by Laws and Regulations, and shall not unreasonably encumber the Site and such other adjacent areas with construction equipment or other materials or equipment. Contractor shall assume full responsibility for (a) damage to the Site; (b) damage to any such other adjacent areas used for Contractor's operations; (c) damage to any other adjacent land or areas, or to improvements, structures, utilities, or similar facilities located at such adjacent lands or areas; and (d) for injuries and losses sustained by the owners or occupants of any such land or areas; provided that such damage or injuries result from the performance of the Work or from other actions or conduct of the Contractor or those for which Contractor is responsible.
 - 2. If a damage or injury claim is made by the owner or occupant of any such land or area because of the performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible, Contractor shall (a) take immediate corrective or remedial action as required by Paragraph 7.13, or otherwise; (b) promptly attempt to settle the claim as to all parties through negotiations with such owner or occupant, or otherwise resolve the claim by arbitration or other dispute resolution proceeding, or in a court of competent jurisdiction; and (c) to the fullest extent permitted by Laws and Regulations, indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from and against any such claim, and against all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any claim or action, legal or equitable, brought by any such owner or occupant against Owner, Engineer, or any other party indemnified hereunder to the extent caused directly or indirectly, in whole or in part by, or based upon, Contractor's performance of the Work, or because of other actions or conduct of the Contractor or those for which Contractor is responsible.
- B. *Removal of Debris During Performance of the Work*: During the progress of the Work the Contractor shall keep the Site and other adjacent areas free from accumulations of waste materials, rubbish, and other debris. Removal and disposal of such waste materials, rubbish, and other debris will conform to applicable Laws and Regulations.
- C. *Cleaning*: Prior to Substantial Completion of the Work Contractor shall clean the Site and the Work and make it ready for utilization by Owner. At the completion of the Work Contractor shall remove from the Site and adjacent areas all tools, appliances, construction equipment

and machinery, and surplus materials and shall restore to original condition all property not designated for alteration by the Contract Documents.

D. Loading of Structures: Contractor shall not load nor permit any part of any structure to be loaded in any manner that will endanger the structure, nor shall Contractor subject any part of the Work or adjacent structures or land to stresses or pressures that will endanger them.

5.03 Subsurface and Physical Conditions

- A. *Reports and Drawings*: The Supplementary Conditions identify:
 - 1. Those reports of explorations and tests of subsurface conditions at or adjacent to the Site that contain Technical Data;
 - 2. Those drawings of existing physical conditions at or adjacent to the Site, including those drawings depicting existing surface or subsurface structures at or adjacent to the Site (except Underground Facilities), that contain Technical Data; and
 - 3. Technical Data contained in such reports and drawings.
- B. Underground Facilities: Underground Facilities are shown or indicated on the Drawings, pursuant to Paragraph 5.05, and not in the drawings referred to in Paragraph 5.03.A. Information and data regarding the presence or location of Underground Facilities are not intended to be categorized, identified, or defined as Technical Data.
- C. *Reliance by Contractor on Technical Data*: Contractor may rely upon the accuracy of the Technical Data expressly identified in the Supplementary Conditions with respect to such reports and drawings, but such reports and drawings are not Contract Documents. If no such express identification has been made, then Contractor may rely upon the accuracy of the Technical Data as defined in Paragraph 1.01.A.46.b.
- D. *Limitations of Other Data and Documents*: Except for such reliance on Technical Data, Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, with respect to:
 - 1. the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences, and procedures of construction to be employed by Contractor, and safety precautions and programs incident thereto;
 - 2. other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings;
 - 3. the contents of other Site-related documents made available to Contractor, such as record drawings from other projects at or adjacent to the Site, or Owner's archival documents concerning the Site; or
 - 4. any Contractor interpretation of or conclusion drawn from any Technical Data or any such other data, interpretations, opinions, or information.

5.04 Differing Subsurface or Physical Conditions

- A. *Notice by Contractor*: If Contractor believes that any subsurface or physical condition that is uncovered or revealed at the Site:
 - 1. is of such a nature as to establish that any Technical Data on which Contractor is entitled to rely as provided in Paragraph 5.03 is materially inaccurate;
 - 2. is of such a nature as to require a change in the Drawings or Specifications;
 - 3. differs materially from that shown or indicated in the Contract Documents; or
 - 4. is of an unusual nature, and differs materially from conditions ordinarily encountered and generally recognized as inherent in work of the character provided for in the Contract Documents;

then Contractor shall, promptly after becoming aware thereof and before further disturbing the subsurface or physical conditions or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing about such condition. Contractor shall not further disturb such condition or perform any Work in connection therewith (except with respect to an emergency) until receipt of a written statement permitting Contractor to do so.

- B. *Engineer's Review*: After receipt of written notice as required by the preceding paragraph, Engineer will promptly review the subsurface or physical condition in question; determine whether it is necessary for Owner to obtain additional exploration or tests with respect to the condition; conclude whether the condition falls within any one or more of the differing site condition categories in Paragraph 5.04.A; obtain any pertinent cost or schedule information from Contractor; prepare recommendations to Owner regarding the Contractor's resumption of Work in connection with the subsurface or physical condition in question and the need for any change in the Drawings or Specifications; and advise Owner in writing of Engineer's findings, conclusions, and recommendations.
- C. Owner's Statement to Contractor Regarding Site Condition: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the subsurface or physical condition in question, addressing the resumption of Work in connection with such condition, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations, in whole or in part.
- D. *Early Resumption of Work*: If at any time Engineer determines that Work in connection with the subsurface or physical condition in question may resume prior to completion of Engineer's review or Owner's issuance of its statement to Contractor, because the condition in question has been adequately documented, and analyzed on a preliminary basis, then the Engineer may at its discretion instruct Contractor to resume such Work.
- E. Possible Price and Times Adjustments
 - 1. Contractor shall be entitled to an equitable adjustment in Contract Price or Contract Times, to the extent that the existence of a differing subsurface or physical condition, or any related delay, disruption, or interference, causes an increase or decrease in

Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:

- a. Such condition must fall within any one or more of the categories described in Paragraph 5.04.A;
- b. With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03; and,
- c. Contractor's entitlement to an adjustment of the Contract Times is subject to the provisions of Paragraphs 4.05.D and 4.05.E.
- 2. Contractor shall not be entitled to any adjustment in the Contract Price or Contract Times with respect to a subsurface or physical condition if:
 - a. Contractor knew of the existence of such condition at the time Contractor made a commitment to Owner with respect to Contract Price and Contract Times by the submission of a Bid or becoming bound under a negotiated contract, or otherwise;
 - b. The existence of such condition reasonably could have been discovered or revealed as a result of any examination, investigation, exploration, test, or study of the Site and contiguous areas expressly required by the Bidding Requirements or Contract Documents to be conducted by or for Contractor prior to Contractor's making such commitment; or
 - c. Contractor failed to give the written notice required by Paragraph 5.04.A.
- 3. If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, then any such adjustment will be set forth in a Change Order.
- 4. Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the subsurface or physical condition in question.
- F. Underground Facilities; Hazardous Environmental Conditions: Paragraph 5.05 governs rights and responsibilities regarding the presence or location of Underground Facilities. Paragraph 5.06 governs rights and responsibilities regarding Hazardous Environmental Conditions. The provisions of Paragraphs 5.03 and 5.04 are not applicable to the presence or location of Underground Facilities, or to Hazardous Environmental Conditions.

5.05 Underground Facilities

- A. *Contractor's Responsibilities*: Unless it is otherwise expressly provided in the Supplementary Conditions, the cost of all of the following are included in the Contract Price, and Contractor shall have full responsibility for:
 - reviewing and checking all information and data regarding existing Underground Facilities at the Site;
 - complying with applicable state and local utility damage prevention Laws and Regulations;

- 3. verifying the actual location of those Underground Facilities shown or indicated in the Contract Documents as being within the area affected by the Work, by exposing such Underground Facilities during the course of construction;
- 4. coordination of the Work with the owners (including Owner) of such Underground Facilities, during construction; and
- the safety and protection of all existing Underground Facilities at the Site, and repairing any damage thereto resulting from the Work.
- B. Notice by Contractor: If Contractor believes that an Underground Facility that is uncovered or revealed at the Site was not shown or indicated on the Drawings, or was not shown or indicated on the Drawings with reasonable accuracy, then Contractor shall, promptly after becoming aware thereof and before further disturbing conditions affected thereby or performing any Work in connection therewith (except in an emergency as required by Paragraph 7.15), notify Owner and Engineer in writing regarding such Underground Facility.
- C. Engineer's Review: Engineer will:
 - 1. promptly review the Underground Facility and conclude whether such Underground Facility was not shown or indicated on the Drawings, or was not shown or indicated with reasonable accuracy;
 - 2. identify and communicate with the owner of the Underground Facility; prepare recommendations to Owner (and if necessary issue any preliminary instructions to Contractor) regarding the Contractor's resumption of Work in connection with the Underground Facility in question;
 - 3. obtain any pertinent cost or schedule information from Contractor; determine the extent, if any, to which a change is required in the Drawings or Specifications to reflect and document the consequences of the existence or location of the Underground Facility; and
 - 4. advise Owner in writing of Engineer's findings, conclusions, and recommendations.

During such time, Contractor shall be responsible for the safety and protection of such Underground Facility.

- D. Owner's Statement to Contractor Regarding Underground Facility: After receipt of Engineer's written findings, conclusions, and recommendations, Owner shall issue a written statement to Contractor (with a copy to Engineer) regarding the Underground Facility in question addressing the resumption of Work in connection with such Underground Facility, indicating whether any change in the Drawings or Specifications will be made, and adopting or rejecting Engineer's written findings, conclusions, and recommendations in whole or in part.
- E. Early Resumption of Work: If at any time Engineer determines that Work in connection with the Underground Facility may resume prior to completion of Engineer's review or Owner's issuance of its statement to Contractor, because the Underground Facility in question and conditions affected by its presence have been adequately documented, and analyzed on a preliminary basis, then the Engineer may at its discretion instruct Contractor to resume such Work.
- F. Possible Price and Times Adjustments
 - Contractor shall be entitled to an equitable adjustment in the Contract Price or Contract Times, to the extent that any existing Underground Facility at the Site that was not shown

or indicated on the Drawings, or was not shown or indicated with reasonable accuracy, or any related delay, disruption, or interference, causes an increase or decrease in Contractor's cost of, or time required for, performance of the Work; subject, however, to the following:

- a. With respect to Work that is paid for on a unit price basis, any adjustment in Contract Price will be subject to the provisions of Paragraph 13.03;
- b. Contractor's entitlement to an adjustment of the Contract Times is subject to the provisions of Paragraphs 4.05.D and 4.05.E; and
- c. Contractor gave the notice required in Paragraph 5.05.B.
- 2. If Owner and Contractor agree regarding Contractor's entitlement to and the amount or extent of any adjustment in the Contract Price or Contract Times, then any such adjustment will be set forth in a Change Order.
- 3. Contractor may submit a Change Proposal regarding its entitlement to or the amount or extent of any adjustment in the Contract Price or Contract Times, no later than 30 days after Owner's issuance of the Owner's written statement to Contractor regarding the Underground Facility in question.
- 4. The information and data shown or indicated on the Drawings with respect to existing Underground Facilities at the Site is based on information and data (a) furnished by the owners of such Underground Facilities, or by others, (b) obtained from available records, or (c) gathered in an investigation conducted in accordance with the current edition of ASCE 38, Standard Guideline for the Collection and Depiction of Existing Subsurface Utility Data, by the American Society of Civil Engineers. If such information or data is incorrect or incomplete, Contractor's remedies are limited to those set forth in this Paragraph 5.05.F.
- 5.06 Hazardous Environmental Conditions at Site
 - A. *Reports and Drawings*: The Supplementary Conditions identify:
 - 1. those reports known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site;
 - 2. drawings known to Owner relating to Hazardous Environmental Conditions that have been identified at or adjacent to the Site; and
 - 3. Technical Data contained in such reports and drawings.
 - B. Reliance by Contractor on Technical Data Authorized: Contractor may rely upon the accuracy of the Technical Data expressly identified in the Supplementary Conditions with respect to such reports and drawings, but such reports and drawings are not Contract Documents. If no such express identification has been made, then Contractor may rely on the accuracy of the Technical Data as defined in Paragraph 1.01.A.46.b. Except for such reliance on Technical Data, Contractor may not rely upon or make any claim against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, with respect to:
 - 1. the completeness of such reports and drawings for Contractor's purposes, including, but not limited to, any aspects of the means, methods, techniques, sequences and procedures

of construction to be employed by Contractor, and safety precautions and programs incident thereto;

- 2. other data, interpretations, opinions, and information contained in such reports or shown or indicated in such drawings; or
- 3. any Contractor interpretation of or conclusion drawn from any Technical Data or any such other data, interpretations, opinions or information.
- C. Contractor shall not be responsible for removing or remediating any Hazardous Environmental Condition encountered, uncovered, or revealed at the Site unless such removal or remediation is expressly identified in the Contract Documents to be within the scope of the Work.
- D. Contractor shall be responsible for controlling, containing, and duly removing all Constituents of Concern brought to the Site by Contractor, Subcontractors, Suppliers, or anyone else for whom Contractor is responsible, and for any associated costs; and for the costs of removing and remediating any Hazardous Environmental Condition created by the presence of any such Constituents of Concern.
- E. If Contractor encounters, uncovers, or reveals a Hazardous Environmental Condition whose removal or remediation is not expressly identified in the Contract Documents as being within the scope of the Work, or if Contractor or anyone for whom Contractor is responsible creates a Hazardous Environmental Condition, then Contractor shall immediately: (1) secure or otherwise isolate such condition; (2) stop all Work in connection with such condition and in any area affected thereby (except in an emergency as required by Paragraph 7.15); and (3) notify Owner and Engineer (and promptly thereafter confirm such notice in writing). Owner shall promptly consult with Engineer concerning the necessity for Owner to retain a qualified expert to evaluate such condition or take corrective action, if any. Promptly after consulting with Engineer, Owner shall take such actions as are necessary to permit Owner to timely obtain required permits and provide Contractor the written notice required by Paragraph 5.06.F. If Contractor or anyone for whom Contractor is responsible created the Hazardous Environmental Condition, and impose a set-off against payments to account for the associated costs.
- F. Contractor shall not resume Work in connection with such Hazardous Environmental Condition or in any affected area until after Owner has obtained any required permits related thereto, and delivered written notice to Contractor either (1) specifying that such condition and any affected area is or has been rendered safe for the resumption of Work, or (2) specifying any special conditions under which such Work may be resumed safely.
- G. If Owner and Contractor cannot agree as to entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times, as a result of such Work stoppage, such special conditions under which Work is agreed to be resumed by Contractor, or any costs or expenses incurred in response to the Hazardous Environmental Condition, then within 30 days of Owner's written notice regarding the resumption of Work, Contractor may submit a Change Proposal, or Owner may impose a set-off. Entitlement to any such adjustment is subject to the provisions of Paragraphs 4.05.D, 4.05.E, 11.07, and 11.08.
- H. If, after receipt of such written notice, Contractor does not agree to resume such Work based on a reasonable belief it is unsafe, or does not agree to resume such Work under such special

conditions, then Owner may order the portion of the Work that is in the area affected by such condition to be deleted from the Work, following the contractual change procedures in Article 11. Owner may have such deleted portion of the Work performed by Owner's own forces or others in accordance with Article 8.

- I. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court, arbitration, or other dispute resolution costs) arising out of or relating to a Hazardous Environmental Condition, provided that such Hazardous Environmental Condition (1) was not shown or indicated in the Drawings, Specifications, or other Contract Documents, identified as Technical Data entitled to limited reliance pursuant to Paragraph 5.06.B, or identified in the Contract Documents to be included within the scope of the Work, and (2) was not created by Contractor or by anyone for whom Contractor is responsible. Nothing in this Paragraph 5.06.I obligates Owner to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- J. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to the failure to control, contain, or remove a Constituent of Concern brought to the Site by Contractor or by anyone for whom Contractor is responsible, or to a Hazardous Environmental Condition created by Contractor or by anyone for whom Contractor to indemnify any individual or entity from and against the consequences of that individual's or entity's own negligence.
- K. The provisions of Paragraphs 5.03, 5.04, and 5.05 do not apply to the presence of Constituents of Concern or to a Hazardous Environmental Condition uncovered or revealed at the Site.

ARTICLE 6—BONDS AND INSURANCE

6.01 *Performance, Payment, and Other Bonds*

- A. Contractor shall furnish a performance bond and a payment bond, each in an amount at least equal to the Contract Price, as security for the faithful performance and payment of Contractor's obligations under the Contract. These bonds must remain in effect until one year after the date when final payment becomes due or until completion of the correction period specified in Paragraph 15.08, whichever is later, except as provided otherwise by Laws or Regulations, the terms of a prescribed bond form, the Supplementary Conditions, or other provisions of the Contract.
- B. Contractor shall also furnish such other bonds (if any) as are required by the Supplementary Conditions or other provisions of the Contract.
- C. All bonds must be in the form included in the Bidding Documents or otherwise specified by Owner prior to execution of the Contract, except as provided otherwise by Laws or
Regulations, and must be issued and signed by a surety named in "Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies" as published in Department Circular 570 (as amended and supplemented) by the Bureau of the Fiscal Service, U.S. Department of the Treasury. A bond signed by an agent or attorney-in-fact must be accompanied by a certified copy of that individual's authority to bind the surety. The evidence of authority must show that it is effective on the date the agent or attorney-in-fact signed the accompanying bond.

- D. Contractor shall obtain the required bonds from surety companies that are duly licensed or authorized, in the state or jurisdiction in which the Project is located, to issue bonds in the required amounts.
- E. If the surety on a bond furnished by Contractor is declared bankrupt or becomes insolvent, or the surety ceases to meet the requirements above, then Contractor shall promptly notify Owner and Engineer in writing and shall, within 20 days after the event giving rise to such notification, provide another bond and surety, both of which must comply with the bond and surety requirements above.
- F. If Contractor has failed to obtain a required bond, Owner may exclude the Contractor from the Site and exercise Owner's termination rights under Article 16.
- G. Upon request to Owner from any Subcontractor, Supplier, or other person or entity claiming to have furnished labor, services, materials, or equipment used in the performance of the Work, Owner shall provide a copy of the payment bond to such person or entity.
- H. Upon request to Contractor from any Subcontractor, Supplier, or other person or entity claiming to have furnished labor, services, materials, or equipment used in the performance of the Work, Contractor shall provide a copy of the payment bond to such person or entity.
- 6.02 Insurance—General Provisions
 - A. Owner and Contractor shall obtain and maintain insurance as required in this article and in the Supplementary Conditions.
 - B. All insurance required by the Contract to be purchased and maintained by Owner or Contractor shall be obtained from insurance companies that are duly licensed or authorized in the state or jurisdiction in which the Project is located to issue insurance policies for the required limits and coverages. Unless a different standard is indicated in the Supplementary Conditions, all companies that provide insurance policies required under this Contract shall have an A.M. Best rating of A-VII or better.
 - C. Alternative forms of insurance coverage, including but not limited to self-insurance and "Occupational Accident and Excess Employer's Indemnity Policies," are not sufficient to meet the insurance requirements of this Contract, unless expressly allowed in the Supplementary Conditions.
 - D. Contractor shall deliver to Owner, with copies to each additional insured identified in the Contract, certificates of insurance and endorsements establishing that Contractor has obtained and is maintaining the policies and coverages required by the Contract. Upon request by Owner or any other insured, Contractor shall also furnish other evidence of such required insurance, including but not limited to copies of policies, documentation of applicable self-insured retentions (if allowed) and deductibles, full disclosure of all relevant exclusions, and evidence of insurance required to be purchased and maintained by

Subcontractors or Suppliers. In any documentation furnished under this provision, Contractor, Subcontractors, and Suppliers may block out (redact) (1) any confidential premium or pricing information and (2) any wording specific to a project or jurisdiction other than those applicable to this Contract.

- E. Owner shall deliver to Contractor, with copies to each additional insured identified in the Contract, certificates of insurance and endorsements establishing that Owner has obtained and is maintaining the policies and coverages required of Owner by the Contract (if any). Upon request by Contractor or any other insured, Owner shall also provide other evidence of such required insurance (if any), including but not limited to copies of policies, documentation of applicable self-insured retentions (if allowed) and deductibles, and full disclosure of all relevant exclusions. In any documentation furnished under this provision, Owner may block out (redact) (1) any confidential premium or pricing information and (2) any wording specific to a project or jurisdiction other than those relevant to this Contract.
- F. Failure of Owner or Contractor to demand such certificates or other evidence of the other party's full compliance with these insurance requirements, or failure of Owner or Contractor to identify a deficiency in compliance from the evidence provided, will not be construed as a waiver of the other party's obligation to obtain and maintain such insurance.
- G. In addition to the liability insurance required to be provided by Contractor, the Owner, at Owner's option, may purchase and maintain Owner's own liability insurance. Owner's liability policies, if any, operate separately and independently from policies required to be provided by Contractor, and Contractor cannot rely upon Owner's liability policies for any of Contractor's obligations to the Owner, Engineer, or third parties.
- H. Contractor shall require:
 - 1. Subcontractors to purchase and maintain worker's compensation, commercial general liability, and other insurance that is appropriate for their participation in the Project, and to name as additional insureds Owner and Engineer (and any other individuals or entities identified in the Supplementary Conditions as additional insureds on Contractor's liability policies) on each Subcontractor's commercial general liability insurance policy; and
 - 2. Suppliers to purchase and maintain insurance that is appropriate for their participation in the Project.
- I. If either party does not purchase or maintain the insurance required of such party by the Contract, such party shall notify the other party in writing of such failure to purchase prior to the start of the Work, or of such failure to maintain prior to any change in the required coverage.
- J. If Contractor has failed to obtain and maintain required insurance, Contractor's entitlement to enter or remain at the Site will end immediately, and Owner may impose an appropriate set-off against payment for any associated costs (including but not limited to the cost of purchasing necessary insurance coverage), and exercise Owner's termination rights under Article 16.
- K. Without prejudice to any other right or remedy, if a party has failed to obtain required insurance, the other party may elect (but is in no way obligated) to obtain equivalent insurance to protect such other party's interests at the expense of the party who was required to provide such coverage, and the Contract Price will be adjusted accordingly.

- L. Owner does not represent that insurance coverage and limits established in this Contract necessarily will be adequate to protect Contractor or Contractor's interests. Contractor is responsible for determining whether such coverage and limits are adequate to protect its interests, and for obtaining and maintaining any additional insurance that Contractor deems necessary.
- M. The insurance and insurance limits required herein will not be deemed as a limitation on Contractor's liability, or that of its Subcontractors or Suppliers, under the indemnities granted to Owner and other individuals and entities in the Contract or otherwise.
- N. All the policies of insurance required to be purchased and maintained under this Contract will contain a provision or endorsement that the coverage afforded will not be canceled, or renewal refused, until at least 10 days prior written notice has been given to the purchasing policyholder. Within three days of receipt of any such written notice, the purchasing policyholder shall provide a copy of the notice to each other insured and Engineer.

6.03 Contractor's Insurance

- A. *Required Insurance*: Contractor shall purchase and maintain Worker's Compensation, Commercial General Liability, and other insurance pursuant to the specific requirements of the Supplementary Conditions.
- B. *General Provisions*: The policies of insurance required by this Paragraph 6.03 as supplemented must:
 - 1. include at least the specific coverages required;
 - 2. be written for not less than the limits provided, or those required by Laws or Regulations, whichever is greater;
 - 3. remain in effect at least until the Work is complete (as set forth in Paragraph 15.06.D), and longer if expressly required elsewhere in this Contract, and at all times thereafter when Contractor may be correcting, removing, or replacing defective Work as a warranty or correction obligation, or otherwise, or returning to the Site to conduct other tasks arising from the Contract;
 - 4. apply with respect to the performance of the Work, whether such performance is by Contractor, any Subcontractor or Supplier, or by anyone directly or indirectly employed by any of them to perform any of the Work, or by anyone for whose acts any of them may be liable; and
 - 5. include all necessary endorsements to support the stated requirements.
- C. Additional Insureds: The Contractor's commercial general liability, automobile liability, employer's liability, umbrella or excess, pollution liability, and unmanned aerial vehicle liability policies, if required by this Contract, must:
 - 1. include and list as additional insureds Owner and Engineer, and any individuals or entities identified as additional insureds in the Supplementary Conditions;
 - 2. include coverage for the respective officers, directors, members, partners, employees, and consultants of all such additional insureds;
 - 3. afford primary coverage to these additional insureds for all claims covered thereby (including as applicable those arising from both ongoing and completed operations);

- 4. not seek contribution from insurance maintained by the additional insured; and
- 5. as to commercial general liability insurance, apply to additional insureds with respect to liability caused in whole or in part by Contractor's acts or omissions, or the acts and omissions of those working on Contractor's behalf, in the performance of Contractor's operations.

6.04 Builder's Risk and Other Property Insurance

- A. Builder's Risk: Unless otherwise provided in the Supplementary Conditions, Contractor shall purchase and maintain builder's risk insurance upon the Work on a completed value basis, in the amount of the Work's full insurable replacement cost (subject to such deductible amounts as may be provided in the Supplementary Conditions or required by Laws and Regulations). The specific requirements applicable to the builder's risk insurance are set forth in the Supplementary Conditions.
- B. Property Insurance for Facilities of Owner Where Work Will Occur: Owner is responsible for obtaining and maintaining property insurance covering each existing structure, building, or facility in which any part of the Work will occur, or to which any part of the Work will attach or be adjoined. Such property insurance will be written on a special perils (all-risk) form, on a replacement cost basis, providing coverage consistent with that required for the builder's risk insurance, and will be maintained until the Work is complete, as set forth in Paragraph 15.06.D.
- C. Property Insurance for Substantially Complete Facilities: Promptly after Substantial Completion, and before actual occupancy or use of the substantially completed Work, Owner will obtain property insurance for such substantially completed Work, and maintain such property insurance at least until the Work is complete, as set forth in Paragraph 15.06.D. Such property insurance will be written on a special perils (all-risk) form, on a replacement cost basis, and provide coverage consistent with that required for the builder's risk insurance. The builder's risk insurance may terminate upon written confirmation of Owner's procurement of such property insurance.
- D. Partial Occupancy or Use by Owner: If Owner will occupy or use a portion or portions of the Work prior to Substantial Completion of all the Work, as provided in Paragraph 15.04, then Owner (directly, if it is the purchaser of the builder's risk policy, or through Contractor) will provide advance notice of such occupancy or use to the builder's risk insurer, and obtain an endorsement consenting to the continuation of coverage prior to commencing such partial occupancy or use.
- E. Insurance of Other Property; Additional Insurance: If the express insurance provisions of the Contract do not require or address the insurance of a property item or interest, then the entity or individual owning such property item will be responsible for insuring it. If Contractor elects to obtain other special insurance to be included in or supplement the builder's risk or property insurance policies provided under this Paragraph 6.04, it may do so at Contractor's expense.

6.05 Property Losses; Subrogation

A. The builder's risk insurance policy purchased and maintained in accordance with Paragraph 6.04 (or an installation floater policy if authorized by the Supplementary Conditions), will contain provisions to the effect that in the event of payment of any loss or damage the insurer will have no rights of recovery against any insureds thereunder, or against

Engineer or its consultants, or their officers, directors, members, partners, employees, agents, consultants, or subcontractors.

- 1. Owner and Contractor waive all rights against each other and the respective officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, for all losses and damages caused by, arising out of, or resulting from any of the perils, risks, or causes of loss covered by such policies and any other property insurance applicable to the Work; and, in addition, waive all such rights against Engineer, its consultants, all individuals or entities identified in the Supplementary Conditions as builder's risk or installation floater insureds, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, under such policies for losses and damages so caused.
- 2. None of the above waivers extends to the rights that any party making such waiver may have to the proceeds of insurance held by Owner or Contractor as trustee or fiduciary, or otherwise payable under any policy so issued.
- B. Any property insurance policy maintained by Owner covering any loss, damage, or consequential loss to Owner's existing structures, buildings, or facilities in which any part of the Work will occur, or to which any part of the Work will attach or adjoin; to adjacent structures, buildings, or facilities of Owner; or to part or all of the completed or substantially completed Work, during partial occupancy or use pursuant to Paragraph 15.04, after Substantial Completion pursuant to Paragraph 15.03, or after final payment pursuant to Paragraph 15.06, will contain provisions to the effect that in the event of payment of any loss or damage the insurer will have no rights of recovery against any insureds thereunder, or against Contractor, Subcontractors, or Engineer, or the officers, directors, members, partners, employees, agents, consultants, or subcontractors of each and any of them, and that the insured is allowed to waive the insurer's rights of subrogation in a written contract executed prior to the loss, damage, or consequential loss.
 - 1. Owner waives all rights against Contractor, Subcontractors, and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, for all losses and damages caused by, arising out of, or resulting from fire or any of the perils, risks, or causes of loss covered by such policies.
- C. The waivers in this Paragraph 6.05 include the waiver of rights due to business interruption, loss of use, or other consequential loss extending beyond direct physical loss or damage to Owner's property or the Work caused by, arising out of, or resulting from fire or other insured peril, risk, or cause of loss.
- D. Contractor shall be responsible for assuring that each Subcontract contains provisions whereby the Subcontractor waives all rights against Owner, Contractor, all individuals or entities identified in the Supplementary Conditions as insureds, the Engineer and its consultants, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, for all losses and damages caused by, arising out of, relating to, or resulting from fire or other peril, risk, or cause of loss covered by builder's risk insurance, installation floater, and any other property insurance applicable to the Work.

6.06 Receipt and Application of Property Insurance Proceeds

- A. Any insured loss under the builder's risk and other policies of property insurance required by Paragraph 6.04 will be adjusted and settled with the named insured that purchased the policy. Such named insured shall act as fiduciary for the other insureds, and give notice to such other insureds that adjustment and settlement of a claim is in progress. Any other insured may state its position regarding a claim for insured loss in writing within 15 days after notice of such claim.
- B. Proceeds for such insured losses may be made payable by the insurer either jointly to multiple insureds, or to the named insured that purchased the policy in its own right and as fiduciary for other insureds, subject to the requirements of any applicable mortgage clause. A named insured receiving insurance proceeds under the builder's risk and other policies of insurance required by Paragraph 6.04 shall maintain such proceeds in a segregated account, and distribute such proceeds in accordance with such agreement as the parties in interest may reach, or as otherwise required under the dispute resolution provisions of this Contract or applicable Laws and Regulations.
- C. If no other special agreement is reached, Contractor shall repair or replace the damaged Work, using allocated insurance proceeds.

ARTICLE 7—CONTRACTOR'S RESPONSIBILITIES

- 7.01 Contractor's Means and Methods of Construction
 - A. Contractor shall be solely responsible for the means, methods, techniques, sequences, and procedures of construction.
 - B. If the Contract Documents note, or Contractor determines, that professional engineering or other design services are needed to carry out Contractor's responsibilities for construction means, methods, techniques, sequences, and procedures, or for Site safety, then Contractor shall cause such services to be provided by a properly licensed design professional, at Contractor's expense. Such services are not Owner-delegated professional design services under this Contract, and neither Owner nor Engineer has any responsibility with respect to (1) Contractor's determination of the need for such services, (2) the qualifications or licensing of the design professionals retained or employed by Contractor, (3) the performance of such services, or (4) any errors, omissions, or defects in such services.
- 7.02 Supervision and Superintendence
 - A. Contractor shall supervise, inspect, and direct the Work competently and efficiently, devoting such attention thereto and applying such skills and expertise as may be necessary to perform the Work in accordance with the Contract Documents.
 - B. At all times during the progress of the Work, Contractor shall assign a competent resident superintendent who will not be replaced without written notice to Owner and Engineer except under extraordinary circumstances.
- 7.03 Labor; Working Hours
 - A. Contractor shall provide competent, suitably qualified personnel to survey and lay out the Work and perform construction as required by the Contract Documents. Contractor shall maintain good discipline and order at the Site.

- B. Contractor shall be fully responsible to Owner and Engineer for all acts and omissions of Contractor's employees; of Suppliers and Subcontractors, and their employees; and of any other individuals or entities performing or furnishing any of the Work, just as Contractor is responsible for Contractor's own acts and omissions.
- C. Except as otherwise required for the safety or protection of persons or the Work or property at the Site or adjacent thereto, and except as otherwise stated in the Contract Documents, all Work at the Site will be performed during regular working hours, Monday through Friday. Contractor will not perform Work on a Saturday, Sunday, or any legal holiday. Contractor may perform Work outside regular working hours or on Saturdays, Sundays, or legal holidays only with Owner's written consent, which will not be unreasonably withheld.
- 7.04 Services, Materials, and Equipment
 - A. Unless otherwise specified in the Contract Documents, Contractor shall provide and assume full responsibility for all services, materials, equipment, labor, transportation, construction equipment and machinery, tools, appliances, fuel, power, light, heat, telephone, water, sanitary facilities, temporary facilities, and all other facilities and incidentals necessary for the performance, testing, start up, and completion of the Work, whether or not such items are specifically called for in the Contract Documents.
 - B. All materials and equipment incorporated into the Work must be new and of good quality, except as otherwise provided in the Contract Documents. All special warranties and guarantees required by the Specifications will expressly run to the benefit of Owner. If required by Engineer, Contractor shall furnish satisfactory evidence (including reports of required tests) as to the source, kind, and quality of materials and equipment.
 - C. All materials and equipment must be stored, applied, installed, connected, erected, protected, used, cleaned, and conditioned in accordance with instructions of the applicable Supplier, except as otherwise may be provided in the Contract Documents.
- 7.05 "Or Equals"
 - A. *Contractor's Request; Governing Criteria*: Whenever an item of equipment or material is specified or described in the Contract Documents by using the names of one or more proprietary items or specific Suppliers, the Contract Price has been based upon Contractor furnishing such item as specified. The specification or description of such an item is intended to establish the type, function, appearance, and quality required. Unless the specification or description contains or is followed by words reading that no like, equivalent, or "or equal" item is permitted, Contractor may request that Engineer authorize the use of other items of equipment or material, or items from other proposed Suppliers, under the circumstances described below.
 - If Engineer in its sole discretion determines that an item of equipment or material proposed by Contractor is functionally equal to that named and sufficiently similar so that no change in related Work will be required, Engineer will deem it an "or equal" item. For the purposes of this paragraph, a proposed item of equipment or material will be considered functionally equal to an item so named if:
 - a. in the exercise of reasonable judgment Engineer determines that the proposed item:
 - 1) is at least equal in materials of construction, quality, durability, appearance, strength, and design characteristics;

- 2) will reliably perform at least equally well the function and achieve the results imposed by the design concept of the completed Project as a functioning whole;
- 3) has a proven record of performance and availability of responsive service; and
- 4) is not objectionable to Owner.
- b. Contractor certifies that, if the proposed item is approved and incorporated into the Work:
 - 1) there will be no increase in cost to the Owner or increase in Contract Times; and
 - 2) the item will conform substantially to the detailed requirements of the item named in the Contract Documents.
- B. *Contractor's Expense*: Contractor shall provide all data in support of any proposed "or equal" item at Contractor's expense.
- C. Engineer's Evaluation and Determination: Engineer will be allowed a reasonable time to evaluate each "or-equal" request. Engineer may require Contractor to furnish additional data about the proposed "or-equal" item. Engineer will be the sole judge of acceptability. No "or-equal" item will be ordered, furnished, installed, or utilized until Engineer's review is complete and Engineer determines that the proposed item is an "or-equal," which will be evidenced by an approved Shop Drawing or other written communication. Engineer will advise Contractor in writing of any negative determination.
- D. *Effect of Engineer's Determination*: Neither approval nor denial of an "or-equal" request will result in any change in Contract Price. The Engineer's denial of an "or-equal" request will be final and binding, and may not be reversed through an appeal under any provision of the Contract.
- E. *Treatment as a Substitution Request*: If Engineer determines that an item of equipment or material proposed by Contractor does not qualify as an "or-equal" item, Contractor may request that Engineer consider the item a proposed substitute pursuant to Paragraph 7.06.

7.06 Substitutes

- A. *Contractor's Request; Governing Criteria*: Unless the specification or description of an item of equipment or material required to be furnished under the Contract Documents contains or is followed by words reading that no substitution is permitted, Contractor may request that Engineer authorize the use of other items of equipment or material under the circumstances described below. To the extent possible such requests must be made before commencement of related construction at the Site.
 - Contractor shall submit sufficient information as provided below to allow Engineer to determine if the item of material or equipment proposed is functionally equivalent to that named and an acceptable substitute therefor. Engineer will not accept requests for review of proposed substitute items of equipment or material from anyone other than Contractor.
 - 2. The requirements for review by Engineer will be as set forth in Paragraph 7.06.B, as supplemented by the Specifications, and as Engineer may decide is appropriate under the circumstances.

- 3. Contractor shall make written application to Engineer for review of a proposed substitute item of equipment or material that Contractor seeks to furnish or use. The application:
 - a. will certify that the proposed substitute item will:
 - 1) perform adequately the functions and achieve the results called for by the general design;
 - 2) be similar in substance to the item specified; and
 - 3) be suited to the same use as the item specified.
 - b. will state:
 - 1) the extent, if any, to which the use of the proposed substitute item will necessitate a change in Contract Times;
 - 2) whether use of the proposed substitute item in the Work will require a change in any of the Contract Documents (or in the provisions of any other direct contract with Owner for other work on the Project) to adapt the design to the proposed substitute item; and
 - 3) whether incorporation or use of the proposed substitute item in connection with the Work is subject to payment of any license fee or royalty.
 - c. will identify:
 - 1) all variations of the proposed substitute item from the item specified; and
 - 2) available engineering, sales, maintenance, repair, and replacement services.
 - d. will contain an itemized estimate of all costs or credits that will result directly or indirectly from use of such substitute item, including but not limited to changes in Contract Price, shared savings, costs of redesign, and claims of other contractors affected by any resulting change.
- B. Engineer's Evaluation and Determination: Engineer will be allowed a reasonable time to evaluate each substitute request, and to obtain comments and direction from Owner. Engineer may require Contractor to furnish additional data about the proposed substitute item. Engineer will be the sole judge of acceptability. No substitute will be ordered, furnished, installed, or utilized until Engineer's review is complete and Engineer determines that the proposed item is an acceptable substitute. Engineer's determination will be evidenced by a Field Order or a proposed Change Order accounting for the substitution itself and all related impacts, including changes in Contract Price or Contract Times. Engineer will advise Contractor in writing of any negative determination.
- C. *Special Guarantee*: Owner may require Contractor to furnish at Contractor's expense a special performance guarantee or other surety with respect to any substitute.
- D. Reimbursement of Engineer's Cost: Engineer will record Engineer's costs in evaluating a substitute proposed or submitted by Contractor. Whether or not Engineer approves a substitute so proposed or submitted by Contractor, Contractor shall reimburse Owner for the reasonable charges of Engineer for evaluating each such proposed substitute. Contractor shall also reimburse Owner for the reasonable charges of Engineer for making changes in the Contract Documents (or in the provisions of any other direct contract with Owner) resulting from the acceptance of each proposed substitute.

- E. *Contractor's Expense*: Contractor shall provide all data in support of any proposed substitute at Contractor's expense.
- F. *Effect of Engineer's Determination*: If Engineer approves the substitution request, Contractor shall execute the proposed Change Order and proceed with the substitution. The Engineer's denial of a substitution request will be final and binding, and may not be reversed through an appeal under any provision of the Contract. Contractor may challenge the scope of reimbursement costs imposed under Paragraph 7.06.D, by timely submittal of a Change Proposal.

7.07 Concerning Subcontractors and Suppliers

- A. Contractor may retain Subcontractors and Suppliers for the performance of parts of the Work. Such Subcontractors and Suppliers must be acceptable to Owner. The Contractor's retention of a Subcontractor or Supplier for the performance of parts of the Work will not relieve Contractor's obligation to Owner to perform and complete the Work in accordance with the Contract Documents.
- B. Contractor shall retain specific Subcontractors and Suppliers for the performance of designated parts of the Work if required by the Contract to do so.
- C. Subsequent to the submittal of Contractor's Bid or final negotiation of the terms of the Contract, Owner may not require Contractor to retain any Subcontractor or Supplier to furnish or perform any of the Work against which Contractor has reasonable objection.
- D. Prior to entry into any binding subcontract or purchase order, Contractor shall submit to Owner the identity of the proposed Subcontractor or Supplier (unless Owner has already deemed such proposed Subcontractor or Supplier acceptable during the bidding process or otherwise). Such proposed Subcontractor or Supplier shall be deemed acceptable to Owner unless Owner raises a substantive, reasonable objection within 5 days.
- E. Owner may require the replacement of any Subcontractor or Supplier. Owner also may require Contractor to retain specific replacements; provided, however, that Owner may not require a replacement to which Contractor has a reasonable objection. If Contractor has submitted the identity of certain Subcontractors or Suppliers for acceptance by Owner, and Owner has accepted it (either in writing or by failing to make written objection thereto), then Owner may subsequently revoke the acceptance of any such Subcontractor or Supplier so identified solely on the basis of substantive, reasonable objection after due investigation. Contractor shall submit an acceptable replacement for the rejected Subcontractor or Supplier.
- F. If Owner requires the replacement of any Subcontractor or Supplier retained by Contractor to perform any part of the Work, then Contractor shall be entitled to an adjustment in Contract Price or Contract Times, with respect to the replacement; and Contractor shall initiate a Change Proposal for such adjustment within 30 days of Owner's requirement of replacement.
- G. No acceptance by Owner of any such Subcontractor or Supplier, whether initially or as a replacement, will constitute a waiver of the right of Owner to the completion of the Work in accordance with the Contract Documents.

- H. On a monthly basis, Contractor shall submit to Engineer a complete list of all Subcontractors and Suppliers having a direct contract with Contractor, and of all other Subcontractors and Suppliers known to Contractor at the time of submittal.
- I. Contractor shall be solely responsible for scheduling and coordinating the work of Subcontractors and Suppliers.
- J. The divisions and sections of the Specifications and the identifications of any Drawings do not control Contractor in dividing the Work among Subcontractors or Suppliers, or in delineating the Work to be performed by any specific trade.
- K. All Work performed for Contractor by a Subcontractor or Supplier must be pursuant to an appropriate contractual agreement that specifically binds the Subcontractor or Supplier to the applicable terms and conditions of the Contract for the benefit of Owner and Engineer.
- L. Owner may furnish to any Subcontractor or Supplier, to the extent practicable, information about amounts paid to Contractor for Work performed for Contractor by the Subcontractor or Supplier.
- M. Contractor shall restrict all Subcontractors and Suppliers from communicating with Engineer or Owner, except through Contractor or in case of an emergency, or as otherwise expressly allowed in this Contract.
- 7.08 Patent Fees and Royalties
 - A. Contractor shall pay all license fees and royalties and assume all costs incident to the use in the performance of the Work or the incorporation in the Work of any invention, design, process, product, or device which is the subject of patent rights or copyrights held by others. If an invention, design, process, product, or device is specified in the Contract Documents for use in the performance of the Work and if, to the actual knowledge of Owner or Engineer, its use is subject to patent rights or copyrights calling for the payment of any license fee or royalty to others, the existence of such rights will be disclosed in the Contract Documents.
 - B. To the fullest extent permitted by Laws and Regulations, Owner shall indemnify and hold harmless Contractor, and its officers, directors, members, partners, employees, agents, consultants, and subcontractors, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device specified in the Contract Documents, but not identified as being subject to payment of any license fee or royalty to others required by patent rights or copyrights.
 - C. To the fullest extent permitted by Laws and Regulations, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to any infringement of patent rights or copyrights incident to the use in the performance of the Work or resulting from the incorporation in the Work of any invention, design, process, product, or device not specified in the Contract Documents.

7.09 Permits

A. Unless otherwise provided in the Contract Documents, Contractor shall obtain and pay for all construction permits, licenses, and certificates of occupancy. Owner shall assist Contractor, when necessary, in obtaining such permits and licenses. Contractor shall pay all governmental charges and inspection fees necessary for the prosecution of the Work which are applicable at the time of the submission of Contractor's Bid (or when Contractor became bound under a negotiated contract). Owner shall pay all charges of utility owners for connections for providing permanent service to the Work.

7.10 Taxes

A. Contractor shall pay all sales, consumer, use, and other similar taxes required to be paid by Contractor in accordance with the Laws and Regulations of the place of the Project which are applicable during the performance of the Work.

7.11 Laws and Regulations

- A. Contractor shall give all notices required by and shall comply with all Laws and Regulations applicable to the performance of the Work. Neither Owner nor Engineer shall be responsible for monitoring Contractor's compliance with any Laws or Regulations.
- B. If Contractor performs any Work or takes any other action knowing or having reason to know that it is contrary to Laws or Regulations, Contractor shall bear all resulting costs and losses, and shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants, and subcontractors of each and any of them, from and against all claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such Work or other action. It is not Contractor's responsibility to make certain that the Work described in the Contract Documents is in accordance with Laws and Regulations, but this does not relieve Contractor of its obligations under Paragraph 3.03.
- C. Owner or Contractor may give written notice to the other party of any changes after the submission of Contractor's Bid (or after the date when Contractor became bound under a negotiated contract) in Laws or Regulations having an effect on the cost or time of performance of the Work, including but not limited to changes in Laws or Regulations having an effect on procuring permits and on sales, use, value-added, consumption, and other similar taxes. If Owner and Contractor are unable to agree on entitlement to or on the amount or extent, if any, of any adjustment in Contract Price or Contract Times resulting from such changes, then within 30 days of such written notice Contractor may submit a Change Proposal, or Owner may initiate a Claim.

7.12 Record Documents

A. Contractor shall maintain in a safe place at the Site one printed record copy of all Drawings, Specifications, Addenda, Change Orders, Work Change Directives, Field Orders, written interpretations and clarifications, and approved Shop Drawings. Contractor shall keep such record documents in good order and annotate them to show changes made during construction. These record documents, together with all approved Samples, will be available to Engineer for reference. Upon completion of the Work, Contractor shall deliver these record documents to Engineer.

7.13 Safety and Protection

- A. Contractor shall be solely responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with the Work. Such responsibility does not relieve Subcontractors of their responsibility for the safety of persons or property in the performance of their work, nor for compliance with applicable safety Laws and Regulations.
- B. Contractor shall designate a qualified and experienced safety representative whose duties and responsibilities are the prevention of Work-related accidents and the maintenance and supervision of safety precautions and programs.
- C. Contractor shall take all necessary precautions for the safety of, and shall provide the necessary protection to prevent damage, injury, or loss to:
 - 1. all persons on the Site or who may be affected by the Work;
 - 2. all the Work and materials and equipment to be incorporated therein, whether in storage on or off the Site; and
 - 3. other property at the Site or adjacent thereto, including trees, shrubs, lawns, walks, pavements, roadways, structures, other work in progress, utilities, and Underground Facilities not designated for removal, relocation, or replacement in the course of construction.
- D. All damage, injury, or loss to any property referred to in Paragraph 7.13.C.2 or 7.13.C.3 caused, directly or indirectly, in whole or in part, by Contractor, any Subcontractor, Supplier, or any other individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, shall be remedied by Contractor at its expense (except damage or loss attributable to the fault of Drawings or Specifications or to the acts or omissions of Owner or Engineer or anyone employed by any of them, or anyone for whose acts any of them may be liable, and not attributable, directly or indirectly, in whole or in part, to the fault or negligence of Contractor or any Subcontractor, Supplier, or other individual or entity directly or indirectly employed by any of them).
- E. Contractor shall comply with all applicable Laws and Regulations relating to the safety of persons or property, or to the protection of persons or property from damage, injury, or loss; and shall erect and maintain all necessary safeguards for such safety and protection.
- F. Contractor shall notify Owner; the owners of adjacent property; the owners of Underground Facilities and other utilities (if the identity of such owners is known to Contractor); and other contractors and utility owners performing work at or adjacent to the Site, in writing, when Contractor knows that prosecution of the Work may affect them, and shall cooperate with them in the protection, removal, relocation, and replacement of their property or work in progress.
- G. Contractor shall comply with the applicable requirements of Owner's safety programs, if any. Any Owner's safety programs that are applicable to the Work are identified or included in the Supplementary Conditions or Specifications.
- H. Contractor shall inform Owner and Engineer of the specific requirements of Contractor's safety program with which Owner's and Engineer's employees and representatives must comply while at the Site.

- Contractor's duties and responsibilities for safety and protection will continue until all the Work is completed, Engineer has issued a written notice to Owner and Contractor in accordance with Paragraph 15.06.C that the Work is acceptable, and Contractor has left the Site (except as otherwise expressly provided in connection with Substantial Completion).
- J. Contractor's duties and responsibilities for safety and protection will resume whenever Contractor or any Subcontractor or Supplier returns to the Site to fulfill warranty or correction obligations, or to conduct other tasks arising from the Contract Documents.

7.14 Hazard Communication Programs

A. Contractor shall be responsible for coordinating any exchange of safety data sheets (formerly known as material safety data sheets) or other hazard communication information required to be made available to or exchanged between or among employers at the Site in accordance with Laws or Regulations.

7.15 Emergencies

A. In emergencies affecting the safety or protection of persons or the Work or property at the Site or adjacent thereto, Contractor is obligated to act to prevent damage, injury, or loss. Contractor shall give Engineer prompt written notice if Contractor believes that any significant changes in the Work or variations from the Contract Documents have been caused by an emergency, or are required as a result of Contractor's response to an emergency. If Engineer determines that a change in the Contract Documents is required because of an emergency or Contractor's response, a Work Change Directive or Change Order will be issued.

7.16 Submittals

- A. Shop Drawing and Sample Requirements
 - 1. Before submitting a Shop Drawing or Sample, Contractor shall:
 - a. review and coordinate the Shop Drawing or Sample with other Shop Drawings and Samples and with the requirements of the Work and the Contract Documents;
 - b. determine and verify:
 - 1) all field measurements, quantities, dimensions, specified performance and design criteria, installation requirements, materials, catalog numbers, and similar information with respect to the Submittal;
 - 2) the suitability of all materials and equipment offered with respect to the indicated application, fabrication, shipping, handling, storage, assembly, and installation pertaining to the performance of the Work; and
 - all information relative to Contractor's responsibilities for means, methods, techniques, sequences, and procedures of construction, and safety precautions and programs incident thereto;
 - c. confirm that the Submittal is complete with respect to all related data included in the Submittal.
 - 2. Each Shop Drawing or Sample must bear a stamp or specific written certification that Contractor has satisfied Contractor's obligations under the Contract Documents with respect to Contractor's review of that Submittal, and that Contractor approves the Submittal.

- 3. With each Shop Drawing or Sample, Contractor shall give Engineer specific written notice of any variations that the Submittal may have from the requirements of the Contract Documents. This notice must be set forth in a written communication separate from the Submittal; and, in addition, in the case of a Shop Drawing by a specific notation made on the Shop Drawing itself.
- B. Submittal Procedures for Shop Drawings and Samples: Contractor shall label and submit Shop Drawings and Samples to Engineer for review and approval in accordance with the accepted Schedule of Submittals.
 - 1. Shop Drawings
 - a. Contractor shall submit the number of copies required in the Specifications.
 - b. Data shown on the Shop Drawings must be complete with respect to quantities, dimensions, specified performance and design criteria, materials, and similar data to show Engineer the services, materials, and equipment Contractor proposes to provide, and to enable Engineer to review the information for the limited purposes required by Paragraph 7.16.C.
 - 2. Samples
 - a. Contractor shall submit the number of Samples required in the Specifications.
 - b. Contractor shall clearly identify each Sample as to material, Supplier, pertinent data such as catalog numbers, the use for which intended and other data as Engineer may require to enable Engineer to review the Submittal for the limited purposes required by Paragraph 7.16.C.
 - 3. Where a Shop Drawing or Sample is required by the Contract Documents or the Schedule of Submittals, any related Work performed prior to Engineer's review and approval of the pertinent submittal will be at the sole expense and responsibility of Contractor.
- C. Engineer's Review of Shop Drawings and Samples
 - Engineer will provide timely review of Shop Drawings and Samples in accordance with the accepted Schedule of Submittals. Engineer's review and approval will be only to determine if the items covered by the Submittals will, after installation or incorporation in the Work, comply with the requirements of the Contract Documents, and be compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents.
 - 2. Engineer's review and approval will not extend to means, methods, techniques, sequences, or procedures of construction, or to safety precautions or programs incident thereto.
 - 3. Engineer's review and approval of a separate item as such will not indicate approval of the assembly in which the item functions.
 - 4. Engineer's review and approval of a Shop Drawing or Sample will not relieve Contractor from responsibility for any variation from the requirements of the Contract Documents unless Contractor has complied with the requirements of Paragraph 7.16.A.3 and Engineer has given written approval of each such variation by specific written notation thereof incorporated in or accompanying the Shop Drawing or Sample. Engineer will

document any such approved variation from the requirements of the Contract Documents in a Field Order or other appropriate Contract modification.

- 5. Engineer's review and approval of a Shop Drawing or Sample will not relieve Contractor from responsibility for complying with the requirements of Paragraphs 7.16.A and B.
- 6. Engineer's review and approval of a Shop Drawing or Sample, or of a variation from the requirements of the Contract Documents, will not, under any circumstances, change the Contract Times or Contract Price, unless such changes are included in a Change Order.
- 7. Neither Engineer's receipt, review, acceptance, or approval of a Shop Drawing or Sample will result in such item becoming a Contract Document.
- 8. Contractor shall perform the Work in compliance with the requirements and commitments set forth in approved Shop Drawings and Samples, subject to the provisions of Paragraph 7.16.C.4.
- D. Resubmittal Procedures for Shop Drawings and Samples
 - 1. Contractor shall make corrections required by Engineer and shall return the required number of corrected copies of Shop Drawings and submit, as required, new Samples for review and approval. Contractor shall direct specific attention in writing to revisions other than the corrections called for by Engineer on previous Submittals.
 - 2. Contractor shall furnish required Shop Drawing and Sample submittals with sufficient information and accuracy to obtain required approval of an item with no more than two resubmittals. Engineer will record Engineer's time for reviewing a third or subsequent resubmittal of a Shop Drawing or Sample, and Contractor shall be responsible for Engineer's charges to Owner for such time. Owner may impose a set-off against payments due Contractor to secure reimbursement for such charges.
 - 3. If Contractor requests a change of a previously approved Shop Drawing or Sample, Contractor shall be responsible for Engineer's charges to Owner for its review time, and Owner may impose a set-off against payments due Contractor to secure reimbursement for such charges, unless the need for such change is beyond the control of Contractor.
- E. Submittals Other than Shop Drawings, Samples, and Owner-Delegated Designs
 - 1. The following provisions apply to all Submittals other than Shop Drawings, Samples, and Owner-delegated designs:
 - a. Contractor shall submit all such Submittals to the Engineer in accordance with the Schedule of Submittals and pursuant to the applicable terms of the Contract Documents.
 - b. Engineer will provide timely review of all such Submittals in accordance with the Schedule of Submittals and return such Submittals with a notation of either Accepted or Not Accepted. Any such Submittal that is not returned within the time established in the Schedule of Submittals will be deemed accepted.
 - c. Engineer's review will be only to determine if the Submittal is acceptable under the requirements of the Contract Documents as to general form and content of the Submittal.

- d. If any such Submittal is not accepted, Contractor shall confer with Engineer regarding the reason for the non-acceptance, and resubmit an acceptable document.
- 2. Procedures for the submittal and acceptance of the Progress Schedule, the Schedule of Submittals, and the Schedule of Values are set forth in Paragraphs 2.03. 2.04, and 2.05.
- F. Owner-delegated Designs: Submittals pursuant to Owner-delegated designs are governed by the provisions of Paragraph 7.19.

7.17 Contractor's General Warranty and Guarantee

- A. Contractor warrants and guarantees to Owner that all Work will be in accordance with the Contract Documents and will not be defective. Engineer is entitled to rely on Contractor's warranty and guarantee.
- B. Owner's rights under this warranty and guarantee are in addition to, and are not limited by, Owner's rights under the correction period provisions of Paragraph 15.08. The time in which Owner may enforce its warranty and guarantee rights under this Paragraph 7.17 is limited only by applicable Laws and Regulations restricting actions to enforce such rights; provided, however, that after the end of the correction period under Paragraph 15.08:
 - 1. Owner shall give Contractor written notice of any defective Work within 60 days of the discovery that such Work is defective; and
 - 2. Such notice will be deemed the start of an event giving rise to a Claim under Paragraph 12.01.B, such that any related Claim must be brought within 30 days of the notice.
- C. Contractor's warranty and guarantee hereunder excludes defects or damage caused by:
 - 1. abuse, or improper modification, maintenance, or operation, by persons other than Contractor, Subcontractors, Suppliers, or any other individual or entity for whom Contractor is responsible; or
 - 2. normal wear and tear under normal usage.
- D. Contractor's obligation to perform and complete the Work in accordance with the Contract Documents is absolute. None of the following will constitute an acceptance of Work that is not in accordance with the Contract Documents, a release of Contractor's obligation to perform the Work in accordance with the Contract Documents, or a release of Owner's warranty and guarantee rights under this Paragraph 7.17:
 - 1. Observations by Engineer;
 - 2. Recommendation by Engineer or payment by Owner of any progress or final payment;
 - 3. The issuance of a certificate of Substantial Completion by Engineer or any payment related thereto by Owner;
 - 4. Use or occupancy of the Work or any part thereof by Owner;
 - 5. Any review and approval of a Shop Drawing or Sample submittal;
 - 6. The issuance of a notice of acceptability by Engineer;
 - 7. The end of the correction period established in Paragraph 15.08;
 - 8. Any inspection, test, or approval by others; or

- 9. Any correction of defective Work by Owner.
- E. If the Contract requires the Contractor to accept the assignment of a contract entered into by Owner, then the specific warranties, guarantees, and correction obligations contained in the assigned contract will govern with respect to Contractor's performance obligations to Owner for the Work described in the assigned contract.

7.18 Indemnification

- A. To the fullest extent permitted by Laws and Regulations, and in addition to any other obligations of Contractor under the Contract or otherwise, Contractor shall indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them, from losses, damages, costs, and judgments (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals, and all court or arbitration or other dispute resolution costs) arising from third-party claims or actions relating to or resulting from the performance or furnishing of the Work, provided that any such claim, action, loss, cost, judgment or damage is attributable to bodily injury, sickness, disease, or death, or to damage to or destruction of tangible property (other than the Work itself), including the loss of use resulting therefrom, but only to the extent caused by any negligent act or omission of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable.
- B. In any and all claims against Owner or Engineer, or any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, by any employee (or the survivor or personal representative of such employee) of Contractor, any Subcontractor, any Supplier, or any individual or entity directly or indirectly employed by any of them to perform any of the Work, or anyone for whose acts any of them may be liable, the indemnification obligation under Paragraph 7.18.A will not be limited in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for Contractor or any such Subcontractor, Supplier, or other individual or entity under workers' compensation acts, disability benefit acts, or other employee benefit acts.

7.19 Delegation of Professional Design Services

- A. Owner may require Contractor to provide professional design services for a portion of the Work by express delegation in the Contract Documents. Such delegation will specify the performance and design criteria that such services must satisfy, and the Submittals that Contractor must furnish to Engineer with respect to the Owner-delegated design.
- B. Contractor shall cause such Owner-delegated professional design services to be provided pursuant to the professional standard of care by a properly licensed design professional, whose signature and seal must appear on all drawings, calculations, specifications, certifications, and Submittals prepared by such design professional. Such design professional must issue all certifications of design required by Laws and Regulations.
- C. If a Shop Drawing or other Submittal related to the Owner-delegated design is prepared by Contractor, a Subcontractor, or others for submittal to Engineer, then such Shop Drawing or other Submittal must bear the written approval of Contractor's design professional when submitted by Contractor to Engineer.

- D. Owner and Engineer shall be entitled to rely upon the adequacy, accuracy, and completeness of the services, certifications, and approvals performed or provided by the design professionals retained or employed by Contractor under an Owner-delegated design, subject to the professional standard of care and the performance and design criteria stated in the Contract Documents.
- E. Pursuant to this Paragraph 7.19, Engineer's review, approval, and other determinations regarding design drawings, calculations, specifications, certifications, and other Submittals furnished by Contractor pursuant to an Owner-delegated design will be only for the following limited purposes:
 - 1. Checking for conformance with the requirements of this Paragraph 7.19;
 - 2. Confirming that Contractor (through its design professionals) has used the performance and design criteria specified in the Contract Documents; and
 - 3. Establishing that the design furnished by Contractor is consistent with the design concept expressed in the Contract Documents.
- F. Contractor shall not be responsible for the adequacy of performance or design criteria specified by Owner or Engineer.
- G. Contractor is not required to provide professional services in violation of applicable Laws and Regulations.

ARTICLE 8—OTHER WORK AT THE SITE

- 8.01 Other Work
 - A. In addition to and apart from the Work under the Contract Documents, the Owner may perform other work at or adjacent to the Site. Such other work may be performed by Owner's employees, or through contracts between the Owner and third parties. Owner may also arrange to have third-party utility owners perform work on their utilities and facilities at or adjacent to the Site.
 - B. If Owner performs other work at or adjacent to the Site with Owner's employees, or through contracts for such other work, then Owner shall give Contractor written notice thereof prior to starting any such other work. If Owner has advance information regarding the start of any third-party utility work that Owner has arranged to take place at or adjacent to the Site, Owner shall provide such information to Contractor.
 - C. Contractor shall afford proper and safe access to the Site to each contractor that performs such other work, each utility owner performing other work, and Owner, if Owner is performing other work with Owner's employees, and provide a reasonable opportunity for the introduction and storage of materials and equipment and the execution of such other work.
 - D. Contractor shall do all cutting, fitting, and patching of the Work that may be required to properly connect or otherwise make its several parts come together and properly integrate with such other work. Contractor shall not endanger any work of others by cutting, excavating, or otherwise altering such work; provided, however, that Contractor may cut or alter others' work with the written consent of Engineer and the others whose work will be affected.

- E. If the proper execution or results of any part of Contractor's Work depends upon work performed by others, Contractor shall inspect such other work and promptly report to Engineer in writing any delays, defects, or deficiencies in such other work that render it unavailable or unsuitable for the proper execution and results of Contractor's Work. Contractor's failure to so report will constitute an acceptance of such other work as fit and proper for integration with Contractor's Work except for latent defects and deficiencies in such other work.
- F. The provisions of this article are not applicable to work that is performed by third-party utilities or other third-party entities without a contract with Owner, or that is performed without having been arranged by Owner. If such work occurs, then any related delay, disruption, or interference incurred by Contractor is governed by the provisions of Paragraph 4.05.C.3.

8.02 Coordination

- A. If Owner intends to contract with others for the performance of other work at or adjacent to the Site, to perform other work at or adjacent to the Site with Owner's employees, or to arrange to have utility owners perform work at or adjacent to the Site, the following will be set forth in the Supplementary Conditions or provided to Contractor prior to the start of any such other work:
 - 1. The identity of the individual or entity that will have authority and responsibility for coordination of the activities among the various contractors;
 - 2. An itemization of the specific matters to be covered by such authority and responsibility; and
 - 3. The extent of such authority and responsibilities.
- B. Unless otherwise provided in the Supplementary Conditions, Owner shall have sole authority and responsibility for such coordination.

8.03 Legal Relationships

A. If, in the course of performing other work for Owner at or adjacent to the Site, the Owner's employees, any other contractor working for Owner, or any utility owner that Owner has arranged to perform work, causes damage to the Work or to the property of Contractor or its Subcontractors, or delays, disrupts, interferes with, or increases the scope or cost of the performance of the Work, through actions or inaction, then Contractor shall be entitled to an equitable adjustment in the Contract Price or the Contract Times. Contractor must submit any Change Proposal seeking an equitable adjustment in the Contract Price or the Contract Times under this paragraph within 30 days of the damaging, delaying, disrupting, or interfering event. The entitlement to, and extent of, any such equitable adjustment will take into account information (if any) regarding such other work that was provided to Contractor in the Contract Documents prior to the submittal of the Bid or the final negotiation of the terms of the Contract, and any remedies available to Contractor under Laws or Regulations concerning utility action or inaction. When applicable, any such equitable adjustment in Contract Price will be conditioned on Contractor assigning to Owner all Contractor's rights against such other contractor or utility owner with respect to the damage, delay, disruption, or interference that is the subject of the adjustment. Contractor's entitlement to an adjustment of the Contract Times or Contract Price is subject to the provisions of Paragraphs 4.05.D and 4.05.E.

- B. Contractor shall take reasonable and customary measures to avoid damaging, delaying, disrupting, or interfering with the work of Owner, any other contractor, or any utility owner performing other work at or adjacent to the Site.
 - 1. If Contractor fails to take such measures and as a result damages, delays, disrupts, or interferes with the work of any such other contractor or utility owner, then Owner may impose a set-off against payments due Contractor, and assign to such other contractor or utility owner the Owner's contractual rights against Contractor with respect to the breach of the obligations set forth in this Paragraph 8.03.B.
 - 2. When Owner is performing other work at or adjacent to the Site with Owner's employees, Contractor shall be liable to Owner for damage to such other work, and for the reasonable direct delay, disruption, and interference costs incurred by Owner as a result of Contractor's failure to take reasonable and customary measures with respect to Owner's other work. In response to such damage, delay, disruption, or interference, Owner may impose a set-off against payments due Contractor.
- C. If Contractor damages, delays, disrupts, or interferes with the work of any other contractor, or any utility owner performing other work at or adjacent to the Site, through Contractor's failure to take reasonable and customary measures to avoid such impacts, or if any claim arising out of Contractor's actions, inactions, or negligence in performance of the Work at or adjacent to the Site is made by any such other contractor or utility owner against Contractor, Owner, or Engineer, then Contractor shall (1) promptly attempt to settle the claim as to all parties through negotiations with such other contractor or utility owner, or otherwise resolve the claim by arbitration or other dispute resolution proceeding or at law, and (2) indemnify and hold harmless Owner and Engineer, and the officers, directors, members, partners, employees, agents, consultants and subcontractors of each and any of them from and against any such claims, and against all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such damage, delay, disruption, or interference.

ARTICLE 9—OWNER'S RESPONSIBILITIES

- 9.01 Communications to Contractor
 - A. Except as otherwise provided in these General Conditions, Owner shall issue all communications to Contractor through Engineer.
- 9.02 Replacement of Engineer
 - A. Owner may at its discretion appoint an engineer to replace Engineer, provided Contractor makes no reasonable objection to the replacement engineer. The replacement engineer's status under the Contract Documents will be that of the former Engineer.
- 9.03 Furnish Data
 - A. Owner shall promptly furnish the data required of Owner under the Contract Documents.
- 9.04 Pay When Due
 - A. Owner shall make payments to Contractor when they are due as provided in the Agreement.

- 9.05 Lands and Easements; Reports, Tests, and Drawings
 - A. Owner's duties with respect to providing lands and easements are set forth in Paragraph 5.01.
 - B. Owner's duties with respect to providing engineering surveys to establish reference points are set forth in Paragraph 4.03.
 - C. Article 5 refers to Owner's identifying and making available to Contractor copies of reports of explorations and tests of conditions at the Site, and drawings of physical conditions relating to existing surface or subsurface structures at the Site.
- 9.06 Insurance
 - A. Owner's responsibilities, if any, with respect to purchasing and maintaining liability and property insurance are set forth in Article 6.
- 9.07 Change Orders
 - A. Owner's responsibilities with respect to Change Orders are set forth in Article 11.
- 9.08 Inspections, Tests, and Approvals
 - A. Owner's responsibility with respect to certain inspections, tests, and approvals is set forth in Paragraph 14.02.B.
- 9.09 Limitations on Owner's Responsibilities
 - A. The Owner shall not supervise, direct, or have control or authority over, nor be responsible for, Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Owner will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- 9.10 Undisclosed Hazardous Environmental Condition
 - A. Owner's responsibility in respect to an undisclosed Hazardous Environmental Condition is set forth in Paragraph 5.06.
- 9.11 *Evidence of Financial Arrangements*
 - A. Upon request of Contractor, Owner shall furnish Contractor reasonable evidence that financial arrangements have been made to satisfy Owner's obligations under the Contract (including obligations under proposed changes in the Work).
- 9.12 Safety Programs
 - A. While at the Site, Owner's employees and representatives shall comply with the specific applicable requirements of Contractor's safety programs of which Owner has been informed.
 - B. Owner shall furnish copies of any applicable Owner safety programs to Contractor.

ARTICLE 10—ENGINEER'S STATUS DURING CONSTRUCTION

10.01 Owner's Representative

A. Engineer will be Owner's representative during the construction period. The duties and responsibilities and the limitations of authority of Engineer as Owner's representative during construction are set forth in the Contract.

10.02 Visits to Site

- A. Engineer will make visits to the Site at intervals appropriate to the various stages of construction as Engineer deems necessary in order to observe, as an experienced and qualified design professional, the progress that has been made and the quality of the various aspects of Contractor's executed Work. Based on information obtained during such visits and observations, Engineer, for the benefit of Owner, will determine, in general, if the Work is proceeding in accordance with the Contract Documents. Engineer will not be required to make exhaustive or continuous inspections on the Site to check the quality or quantity of the Work. Engineer's efforts will be directed toward providing for Owner a greater degree of confidence that the completed Work will conform generally to the Contract Documents. On the basis of such visits and observations, Engineer will endeavor to guard Owner against defective Work.
- B. Engineer's visits and observations are subject to all the limitations on Engineer's authority and responsibility set forth in Paragraph 10.07. Particularly, but without limitation, during or as a result of Engineer's visits or observations of Contractor's Work, Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work.

10.03 Resident Project Representative

- A. If Owner and Engineer have agreed that Engineer will furnish a Resident Project Representative to represent Engineer at the Site and assist Engineer in observing the progress and quality of the Work, then the authority and responsibilities of any such Resident Project Representative will be as provided in the Supplementary Conditions, and limitations on the responsibilities thereof will be as provided in the Supplementary Conditions and in Paragraph 10.07.
- B. If Owner designates an individual or entity who is not Engineer's consultant, agent, or employee to represent Owner at the Site, then the responsibilities and authority of such individual or entity will be as provided in the Supplementary Conditions.

10.04 Engineer's Authority

- A. Engineer has the authority to reject Work in accordance with Article 14.
- B. Engineer's authority as to Submittals is set forth in Paragraph 7.16.
- C. Engineer's authority as to design drawings, calculations, specifications, certifications and other Submittals from Contractor in response to Owner's delegation (if any) to Contractor of professional design services, is set forth in Paragraph 7.19.
- D. Engineer's authority as to changes in the Work is set forth in Article 11.

E. Engineer's authority as to Applications for Payment is set forth in Article 15.

10.05 Determinations for Unit Price Work

- A. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor as set forth in Paragraph 13.03.
- 10.06 Decisions on Requirements of Contract Documents and Acceptability of Work
 - A. Engineer will render decisions regarding the requirements of the Contract Documents, and judge the acceptability of the Work, pursuant to the specific procedures set forth herein for initial interpretations, Change Proposals, and acceptance of the Work. In rendering such decisions and judgments, Engineer will not show partiality to Owner or Contractor, and will not be liable to Owner, Contractor, or others in connection with any proceedings, interpretations, decisions, or judgments conducted or rendered in good faith.

10.07 Limitations on Engineer's Authority and Responsibilities

- A. Neither Engineer's authority or responsibility under this Article 10 or under any other provision of the Contract, nor any decision made by Engineer in good faith either to exercise or not exercise such authority or responsibility or the undertaking, exercise, or performance of any authority or responsibility by Engineer, will create, impose, or give rise to any duty in contract, tort, or otherwise owed by Engineer to Contractor, any Subcontractor, any Supplier, any other individual or entity, or to any surety for or employee or agent of any of them.
- B. Engineer will not supervise, direct, control, or have authority over or be responsible for Contractor's means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto, or for any failure of Contractor to comply with Laws and Regulations applicable to the performance of the Work. Engineer will not be responsible for Contractor's failure to perform the Work in accordance with the Contract Documents.
- C. Engineer will not be responsible for the acts or omissions of Contractor or of any Subcontractor, any Supplier, or of any other individual or entity performing any of the Work.
- D. Engineer's review of the final Application for Payment and accompanying documentation, and all maintenance and operating instructions, schedules, guarantees, bonds, certificates of inspection, tests and approvals, and other documentation required to be delivered by Contractor under Paragraph 15.06.A, will only be to determine generally that their content complies with the requirements of, and in the case of certificates of inspections, tests, and approvals, that the results certified indicate compliance with the Contract Documents.
- E. The limitations upon authority and responsibility set forth in this Paragraph 10.07 also apply to the Resident Project Representative, if any.

10.08 Compliance with Safety Program

A. While at the Site, Engineer's employees and representatives will comply with the specific applicable requirements of Owner's and Contractor's safety programs of which Engineer has been informed.

ARTICLE 11—CHANGES TO THE CONTRACT

11.01 Amending and Supplementing the Contract

- A. The Contract may be amended or supplemented by a Change Order, a Work Change Directive, or a Field Order.
- B. If an amendment or supplement to the Contract includes a change in the Contract Price or the Contract Times, such amendment or supplement must be set forth in a Change Order.
- C. All changes to the Contract that involve (1) the performance or acceptability of the Work, (2) the design (as set forth in the Drawings, Specifications, or otherwise), or (3) other engineering or technical matters, must be supported by Engineer's recommendation. Owner and Contractor may amend other terms and conditions of the Contract without the recommendation of the Engineer.
- 11.02 Change Orders
 - A. Owner and Contractor shall execute appropriate Change Orders covering:
 - 1. Changes in Contract Price or Contract Times which are agreed to by the parties, including any undisputed sum or amount of time for Work actually performed in accordance with a Work Change Directive;
 - 2. Changes in Contract Price resulting from an Owner set-off, unless Contractor has duly contested such set-off;
 - 3. Changes in the Work which are: (a) ordered by Owner pursuant to Paragraph 11.05, (b) required because of Owner's acceptance of defective Work under Paragraph 14.04 or Owner's correction of defective Work under Paragraph 14.07, or (c) agreed to by the parties, subject to the need for Engineer's recommendation if the change in the Work involves the design (as set forth in the Drawings, Specifications, or otherwise) or other engineering or technical matters; and
 - 4. Changes that embody the substance of any final and binding results under: Paragraph 11.03.B, resolving the impact of a Work Change Directive; Paragraph 11.09, concerning Change Proposals; Article 12, Claims; Paragraph 13.02.D, final adjustments resulting from allowances; Paragraph 13.03.D, final adjustments relating to determination of quantities for Unit Price Work; and similar provisions.
 - B. If Owner or Contractor refuses to execute a Change Order that is required to be executed under the terms of Paragraph 11.02.A, it will be deemed to be of full force and effect, as if fully executed.

11.03 Work Change Directives

A. A Work Change Directive will not change the Contract Price or the Contract Times but is evidence that the parties expect that the modification ordered or documented by a Work Change Directive will be incorporated in a subsequently issued Change Order, following negotiations by the parties as to the Work Change Directive's effect, if any, on the Contract Price and Contract Times; or, if negotiations are unsuccessful, by a determination under the terms of the Contract Documents governing adjustments, expressly including Paragraph 11.07 regarding change of Contract Price.

- B. If Owner has issued a Work Change Directive and:
 - 1. Contractor believes that an adjustment in Contract Times or Contract Price is necessary, then Contractor shall submit any Change Proposal seeking such an adjustment no later than 30 days after the completion of the Work set out in the Work Change Directive.
 - 2. Owner believes that an adjustment in Contract Times or Contract Price is necessary, then Owner shall submit any Claim seeking such an adjustment no later than 60 days after issuance of the Work Change Directive.

11.04 Field Orders

- A. Engineer may authorize minor changes in the Work if the changes do not involve an adjustment in the Contract Price or the Contract Times and are compatible with the design concept of the completed Project as a functioning whole as indicated by the Contract Documents. Such changes will be accomplished by a Field Order and will be binding on Owner and also on Contractor, which shall perform the Work involved promptly.
- B. If Contractor believes that a Field Order justifies an adjustment in the Contract Price or Contract Times, then before proceeding with the Work at issue, Contractor shall submit a Change Proposal as provided herein.
- 11.05 *Owner-Authorized Changes in the Work*
 - A. Without invalidating the Contract and without notice to any surety, Owner may, at any time or from time to time, order additions, deletions, or revisions in the Work. Changes involving the design (as set forth in the Drawings, Specifications, or otherwise) or other engineering or technical matters will be supported by Engineer's recommendation.
 - B. Such changes in the Work may be accomplished by a Change Order, if Owner and Contractor have agreed as to the effect, if any, of the changes on Contract Times or Contract Price; or by a Work Change Directive. Upon receipt of any such document, Contractor shall promptly proceed with the Work involved; or, in the case of a deletion in the Work, promptly cease construction activities with respect to such deleted Work. Added or revised Work must be performed under the applicable conditions of the Contract Documents.
 - C. Nothing in this Paragraph 11.05 obligates Contractor to undertake work that Contractor reasonably concludes cannot be performed in a manner consistent with Contractor's safety obligations under the Contract Documents or Laws and Regulations.

11.06 Unauthorized Changes in the Work

- A. Contractor shall not be entitled to an increase in the Contract Price or an extension of the Contract Times with respect to any work performed that is not required by the Contract Documents, as amended, modified, or supplemented, except in the case of an emergency as provided in Paragraph 7.15 or in the case of uncovering Work as provided in Paragraph 14.05.C.2.
- 11.07 Change of Contract Price
 - A. The Contract Price may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Price must comply with the provisions of Paragraph 11.09. Any Claim for an adjustment of Contract Price must comply with the provisions of Article 12.
 - B. An adjustment in the Contract Price will be determined as follows:

- 1. Where the Work involved is covered by unit prices contained in the Contract Documents, then by application of such unit prices to the quantities of the items involved (subject to the provisions of Paragraph 13.03);
- 2. Where the Work involved is not covered by unit prices contained in the Contract Documents, then by a mutually agreed lump sum (which may include an allowance for overhead and profit not necessarily in accordance with Paragraph 11.07.C.2); or
- 3. Where the Work involved is not covered by unit prices contained in the Contract Documents and the parties do not reach mutual agreement to a lump sum, then on the basis of the Cost of the Work (determined as provided in Paragraph 13.01) plus a Contractor's fee for overhead and profit (determined as provided in Paragraph 11.07.C).
- C. *Contractor's Fee*: When applicable, the Contractor's fee for overhead and profit will be determined as follows:
 - 1. A mutually acceptable fixed fee; or
 - 2. If a fixed fee is not agreed upon, then a fee based on the following percentages of the various portions of the Cost of the Work:
 - a. For costs incurred under Paragraphs 13.01.B.1 and 13.01.B.2, the Contractor's fee will be 15 percent;
 - b. For costs incurred under Paragraph 13.01.B.3, the Contractor's fee will be 5 percent;
 - c. Where one or more tiers of subcontracts are on the basis of Cost of the Work plus a fee and no fixed fee is agreed upon, the intent of Paragraphs 11.07.C.2.a and 11.07.C.2.b is that the Contractor's fee will be based on: (1) a fee of 15 percent of the costs incurred under Paragraphs 13.01.B.1 and 13.01.B.2 by the Subcontractor that actually performs the Work, at whatever tier, and (2) with respect to Contractor itself and to any Subcontractors of a tier higher than that of the Subcontractor that actually performs the Work, a fee of 5 percent of the amount (fee plus underlying costs incurred) attributable to the next lower tier Subcontractor; provided, however, that for any such subcontracted Work the maximum total fee to be paid by Owner will be no greater than 27 percent of the costs incurred by the Subcontractor that actually performs the Work;
 - d. No fee will be payable on the basis of costs itemized under Paragraphs 13.01.B.4, 13.01.B.5, and 13.01.C;
 - e. The amount of credit to be allowed by Contractor to Owner for any change which results in a net decrease in Cost of the Work will be the amount of the actual net decrease in Cost of the Work and a deduction of an additional amount equal to 5 percent of such actual net decrease in Cost of the Work; and
 - f. When both additions and credits are involved in any one change or Change Proposal, the adjustment in Contractor's fee will be computed by determining the sum of the costs in each of the cost categories in Paragraph 13.01.B (specifically, payroll costs, Paragraph 13.01.B.1; incorporated materials and equipment costs, Paragraph 13.01.B.2; Subcontract costs, Paragraph 13.01.B.3; special consultants costs, Paragraph 13.01.B.4; and other costs, Paragraph 13.01.B.5) and applying to each such cost category sum the appropriate fee from Paragraphs 11.07.C.2.a through 11.07.C.2.e, inclusive.

11.08 Change of Contract Times

- A. The Contract Times may only be changed by a Change Order. Any Change Proposal for an adjustment in the Contract Times must comply with the provisions of Paragraph 11.09. Any Claim for an adjustment in the Contract Times must comply with the provisions of Article 12.
- B. Delay, disruption, and interference in the Work, and any related changes in Contract Times, are addressed in and governed by Paragraph 4.05.

11.09 Change Proposals

- A. Purpose and Content: Contractor shall submit a Change Proposal to Engineer to request an adjustment in the Contract Times or Contract Price; contest an initial decision by Engineer concerning the requirements of the Contract Documents or relating to the acceptability of the Work under the Contract Documents; challenge a set-off against payment due; or seek other relief under the Contract. The Change Proposal will specify any proposed change in Contract Times or Contract Price, or other proposed relief, and explain the reason for the proposed change, with citations to any governing or applicable provisions of the Contract Documents. Each Change Proposal will address only one issue, or a set of closely related issues.
- B. Change Proposal Procedures
 - 1. Submittal: Contractor shall submit each Change Proposal to Engineer within 30 days after the start of the event giving rise thereto, or after such initial decision.
 - 2. Supporting Data: The Contractor shall submit supporting data, including the proposed change in Contract Price or Contract Time (if any), to the Engineer and Owner within 15 days after the submittal of the Change Proposal.
 - a. Change Proposals based on or related to delay, interruption, or interference must comply with the provisions of Paragraphs 4.05.D and 4.05.E.
 - b. Change proposals related to a change of Contract Price must include full and detailed accounts of materials incorporated into the Work and labor and equipment used for the subject Work.

The supporting data must be accompanied by a written statement that the supporting data are accurate and complete, and that any requested time or price adjustment is the entire adjustment to which Contractor believes it is entitled as a result of said event.

- 3. Engineer's Initial Review: Engineer will advise Owner regarding the Change Proposal, and consider any comments or response from Owner regarding the Change Proposal. If in its discretion Engineer concludes that additional supporting data is needed before conducting a full review and making a decision regarding the Change Proposal, then Engineer may request that Contractor submit such additional supporting data by a date specified by Engineer, prior to Engineer beginning its full review of the Change Proposal.
- 4. Engineer's Full Review and Action on the Change Proposal: Upon receipt of Contractor's supporting data (including any additional data requested by Engineer), Engineer will conduct a full review of each Change Proposal and, within 30 days after such receipt of the Contractor's supporting data, either approve the Change Proposal in whole, deny it in whole, or approve it in part and deny it in part. Such actions must be in writing, with a copy provided to Owner and Contractor. If Engineer does not take action on the Change

Proposal within 30 days, then either Owner or Contractor may at any time thereafter submit a letter to the other party indicating that as a result of Engineer's inaction the Change Proposal is deemed denied, thereby commencing the time for appeal of the denial under Article 12.

- 5. *Binding Decision*: Engineer's decision is final and binding upon Owner and Contractor, unless Owner or Contractor appeals the decision by filing a Claim under Article 12.
- C. Resolution of Certain Change Proposals: If the Change Proposal does not involve the design (as set forth in the Drawings, Specifications, or otherwise), the acceptability of the Work, or other engineering or technical matters, then Engineer will notify the parties in writing that the Engineer is unable to resolve the Change Proposal. For purposes of further resolution of such a Change Proposal, such notice will be deemed a denial, and Contractor may choose to seek resolution under the terms of Article 12.
- D. *Post-Completion*: Contractor shall not submit any Change Proposals after Engineer issues a written recommendation of final payment pursuant to Paragraph 15.06.B.

11.10 Notification to Surety

A. If the provisions of any bond require notice to be given to a surety of any change affecting the general scope of the Work or the provisions of the Contract Documents (including, but not limited to, Contract Price or Contract Times), the giving of any such notice will be Contractor's responsibility. The amount of each applicable bond will be adjusted to reflect the effect of any such change.

ARTICLE 12—CLAIMS

12.01 Claims

- A. *Claims Process*: The following disputes between Owner and Contractor are subject to the Claims process set forth in this article:
 - 1. Appeals by Owner or Contractor of Engineer's decisions regarding Change Proposals;
 - 2. Owner demands for adjustments in the Contract Price or Contract Times, or other relief under the Contract Documents;
 - 3. Disputes that Engineer has been unable to address because they do not involve the design (as set forth in the Drawings, Specifications, or otherwise), the acceptability of the Work, or other engineering or technical matters; and
 - 4. Subject to the waiver provisions of Paragraph 15.07, any dispute arising after Engineer has issued a written recommendation of final payment pursuant to Paragraph 15.06.B.
- B. Submittal of Claim: The party submitting a Claim shall deliver it directly to the other party to the Contract promptly (but in no event later than 30 days) after the start of the event giving rise thereto; in the case of appeals regarding Change Proposals within 30 days of the decision under appeal. The party submitting the Claim shall also furnish a copy to the Engineer, for its information only. The responsibility to substantiate a Claim rests with the party making the Claim. In the case of a Claim by Contractor seeking an increase in the Contract Times or Contract Price, Contractor shall certify that the Claim is made in good faith, that the supporting data are accurate and complete, and that to the best of Contractor's knowledge

and belief the amount of time or money requested accurately reflects the full amount to which Contractor is entitled.

- C. *Review and Resolution*: The party receiving a Claim shall review it thoroughly, giving full consideration to its merits. The two parties shall seek to resolve the Claim through the exchange of information and direct negotiations. The parties may extend the time for resolving the Claim by mutual agreement. All actions taken on a Claim will be stated in writing and submitted to the other party, with a copy to Engineer.
- D. Mediation
 - 1. At any time after initiation of a Claim, Owner and Contractor may mutually agree to mediation of the underlying dispute. The agreement to mediate will stay the Claim submittal and response process.
 - 2. If Owner and Contractor agree to mediation, then after 60 days from such agreement, either Owner or Contractor may unilaterally terminate the mediation process, and the Claim submittal and decision process will resume as of the date of the termination. If the mediation proceeds but is unsuccessful in resolving the dispute, the Claim submittal and decision process will resume as of the date of the mediation, as determined by the mediator.
 - 3. Owner and Contractor shall each pay one-half of the mediator's fees and costs.
- E. *Partial Approval*: If the party receiving a Claim approves the Claim in part and denies it in part, such action will be final and binding unless within 30 days of such action the other party invokes the procedure set forth in Article 17 for final resolution of disputes.
- F. Denial of Claim: If efforts to resolve a Claim are not successful, the party receiving the Claim may deny it by giving written notice of denial to the other party. If the receiving party does not take action on the Claim within 90 days, then either Owner or Contractor may at any time thereafter submit a letter to the other party indicating that as a result of the inaction, the Claim is deemed denied, thereby commencing the time for appeal of the denial. A denial of the Claim will be final and binding unless within 30 days of the denial the other party invokes the procedure set forth in Article 17 for the final resolution of disputes.
- G. *Final and Binding Results*: If the parties reach a mutual agreement regarding a Claim, whether through approval of the Claim, direct negotiations, mediation, or otherwise; or if a Claim is approved in part and denied in part, or denied in full, and such actions become final and binding; then the results of the agreement or action on the Claim will be incorporated in a Change Order or other written document to the extent they affect the Contract, including the Work, the Contract Times, or the Contract Price.

ARTICLE 13—COST OF THE WORK; ALLOWANCES; UNIT PRICE WORK

- 13.01 *Cost of the Work*
 - A. *Purposes for Determination of Cost of the Work*: The term Cost of the Work means the sum of all costs necessary for the proper performance of the Work at issue, as further defined below. The provisions of this Paragraph 13.01 are used for two distinct purposes:
 - 1. To determine Cost of the Work when Cost of the Work is a component of the Contract Price, under cost-plus-fee, time-and-materials, or other cost-based terms; or

- 2. When needed to determine the value of a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price. When the value of any such adjustment is determined on the basis of Cost of the Work, Contractor is entitled only to those additional or incremental costs required because of the change in the Work or because of the event giving rise to the adjustment.
- B. *Costs Included*: Except as otherwise may be agreed to in writing by Owner, costs included in the Cost of the Work will be in amounts no higher than those commonly incurred in the locality of the Project, will not include any of the costs itemized in Paragraph 13.01.C, and will include only the following items:
 - 1. Payroll costs for employees in the direct employ of Contractor in the performance of the Work under schedules of job classifications agreed upon by Owner and Contractor in advance of the subject Work. Such employees include, without limitation, superintendents, foremen, safety managers, safety representatives, and other personnel employed full time on the Work. Payroll costs for employees not employed full time on the Work will be apportioned on the basis of their time spent on the Work. Payroll costs include, but are not limited to, salaries and wages plus the cost of fringe benefits, which include social security contributions, unemployment, excise, and payroll taxes, workers' compensation, health and retirement benefits, sick leave, and vacation and holiday pay applicable thereto. The expenses of performing Work outside of regular working hours, on Saturday, Sunday, or legal holidays, will be included in the above to the extent authorized by Owner.
 - 2. Cost of all materials and equipment furnished and incorporated in the Work, including costs of transportation and storage thereof, and Suppliers' field services required in connection therewith. All cash discounts accrue to Contractor unless Owner deposits funds with Contractor with which to make payments, in which case the cash discounts will accrue to Owner. All trade discounts, rebates, and refunds and returns from sale of surplus materials and equipment will accrue to Owner, and Contractor shall make provisions so that they may be obtained.
 - 3. Payments made by Contractor to Subcontractors for Work performed by Subcontractors. If required by Owner, Contractor shall obtain competitive bids from subcontractors acceptable to Owner and Contractor and shall deliver such bids to Owner, which will then determine, with the advice of Engineer, which bids, if any, will be acceptable. If any subcontract provides that the Subcontractor is to be paid on the basis of Cost of the Work plus a fee, the Subcontractor's Cost of the Work and fee will be determined in the same manner as Contractor's Cost of the Work and fee as provided in this Paragraph 13.01.
 - 4. Costs of special consultants (including but not limited to engineers, architects, testing laboratories, surveyors, attorneys, and accountants) employed or retained for services specifically related to the Work.
 - 5. Other costs consisting of the following:
 - a. The proportion of necessary transportation, travel, and subsistence expenses of Contractor's employees incurred in discharge of duties connected with the Work.
 - b. Cost, including transportation and maintenance, of all materials, supplies, equipment, machinery, appliances, office, and temporary facilities at the Site, which are

consumed in the performance of the Work, and cost, less market value, of such items used but not consumed which remain the property of Contractor.

- In establishing included costs for materials such as scaffolding, plating, or sheeting, consideration will be given to the actual or the estimated life of the material for use on other projects; or rental rates may be established on the basis of purchase or salvage value of such items, whichever is less. Contractor will not be eligible for compensation for such items in an amount that exceeds the purchase cost of such item.
- c. Construction Equipment Rental
 - 1) Rentals of all construction equipment and machinery, and the parts thereof, in accordance with rental agreements approved by Owner as to price (including any surcharge or special rates applicable to overtime use of the construction equipment or machinery), and the costs of transportation, loading, unloading, assembly, dismantling, and removal thereof. All such costs will be in accordance with the terms of said rental agreements. The rental of any such equipment, machinery, or parts must cease when the use thereof is no longer necessary for the Work.
 - 2) Costs for equipment and machinery owned by Contractor or a Contractor-related entity will be paid at a rate shown for such equipment in the equipment rental rate book specified in the Supplementary Conditions. An hourly rate will be computed by dividing the monthly rates by 176. These computed rates will include all operating costs.
 - 3) With respect to Work that is the result of a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price ("changed Work"), included costs will be based on the time the equipment or machinery is in use on the changed Work and the costs of transportation, loading, unloading, assembly, dismantling, and removal when directly attributable to the changed Work. The cost of any such equipment or machinery, or parts thereof, must cease to accrue when the use thereof is no longer necessary for the changed Work.
- d. Sales, consumer, use, and other similar taxes related to the Work, and for which Contractor is liable, as imposed by Laws and Regulations.
- e. Deposits lost for causes other than negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, and royalty payments and fees for permits and licenses.
- f. Losses and damages (and related expenses) caused by damage to the Work, not compensated by insurance or otherwise, sustained by Contractor in connection with the performance of the Work (except losses and damages within the deductible amounts of builder's risk or other property insurance established in accordance with Paragraph 6.04), provided such losses and damages have resulted from causes other than the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable. Such losses include settlements made with the written consent and approval of Owner. No such losses, damages, and expenses will be included in the Cost of the Work for the purpose of determining Contractor's fee.

- g. The cost of utilities, fuel, and sanitary facilities at the Site.
- h. Minor expenses such as communication service at the Site, express and courier services, and similar petty cash items in connection with the Work.
- i. The costs of premiums for all bonds and insurance that Contractor is required by the Contract Documents to purchase and maintain.
- C. *Costs Excluded*: The term Cost of the Work does not include any of the following items:
 - 1. Payroll costs and other compensation of Contractor's officers, executives, principals, general managers, engineers, architects, estimators, attorneys, auditors, accountants, purchasing and contracting agents, expediters, timekeepers, clerks, and other personnel employed by Contractor, whether at the Site or in Contractor's principal or branch office for general administration of the Work and not specifically included in the agreed upon schedule of job classifications referred to in Paragraph 13.01.B.1 or specifically covered by Paragraph 13.01.B.4. The payroll costs and other compensation excluded here are to be considered administrative costs covered by the Contractor's fee.
 - 2. The cost of purchasing, renting, or furnishing small tools and hand tools.
 - 3. Expenses of Contractor's principal and branch offices other than Contractor's office at the Site.
 - 4. Any part of Contractor's capital expenses, including interest on Contractor's capital employed for the Work and charges against Contractor for delinquent payments.
 - 5. Costs due to the negligence of Contractor, any Subcontractor, or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable, including but not limited to, the correction of defective Work, disposal of materials or equipment wrongly supplied, and making good any damage to property.
 - 6. Expenses incurred in preparing and advancing Claims.
 - 7. Other overhead or general expense costs of any kind and the costs of any item not specifically and expressly included in Paragraph 13.01.B.
- D. Contractor's Fee
 - 1. When the Work as a whole is performed on the basis of cost-plus-a-fee, then:
 - a. Contractor's fee for the Work set forth in the Contract Documents as of the Effective Date of the Contract will be determined as set forth in the Agreement.
 - b. for any Work covered by a Change Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price on the basis of Cost of the Work, Contractor's fee will be determined as follows:
 - 1) When the fee for the Work as a whole is a percentage of the Cost of the Work, the fee will automatically adjust as the Cost of the Work changes.
 - 2) When the fee for the Work as a whole is a fixed fee, the fee for any additions or deletions will be determined in accordance with Paragraph 11.07.C.2.
 - 2. When the Work as a whole is performed on the basis of a stipulated sum, or any other basis other than cost-plus-a-fee, then Contractor's fee for any Work covered by a Change

Order, Change Proposal, Claim, set-off, or other adjustment in Contract Price on the basis of Cost of the Work will be determined in accordance with Paragraph 11.07.C.2.

E. Documentation and Audit: Whenever the Cost of the Work for any purpose is to be determined pursuant to this Article 13, Contractor and pertinent Subcontractors will establish and maintain records of the costs in accordance with generally accepted accounting practices. Subject to prior written notice, Owner will be afforded reasonable access, during normal business hours, to all Contractor's accounts, records, books, correspondence, instructions, drawings, receipts, vouchers, memoranda, and similar data relating to the Cost of the Work and Contractor's fee. Contractor shall preserve all such documents for a period of three years after the final payment by Owner. Pertinent Subcontractors will afford such access to Owner, and preserve such documents, to the same extent required of Contractor.

13.02 Allowances

- A. It is understood that Contractor has included in the Contract Price all allowances so named in the Contract Documents and shall cause the Work so covered to be performed for such sums and by such persons or entities as may be acceptable to Owner and Engineer.
- B. Cash Allowances: Contractor agrees that:
 - 1. the cash allowances include the cost to Contractor (less any applicable trade discounts) of materials and equipment required by the allowances to be delivered at the Site, and all applicable taxes; and
 - 2. Contractor's costs for unloading and handling on the Site, labor, installation, overhead, profit, and other expenses contemplated for the cash allowances have been included in the Contract Price and not in the allowances, and no demand for additional payment for any of the foregoing will be valid.
- C. *Owner's Contingency Allowance*: Contractor agrees that an Owner's contingency allowance, if any, is for the sole use of Owner to cover unanticipated costs.
- D. Prior to final payment, an appropriate Change Order will be issued as recommended by Engineer to reflect actual amounts due Contractor for Work covered by allowances, and the Contract Price will be correspondingly adjusted.

13.03 Unit Price Work

- A. Where the Contract Documents provide that all or part of the Work is to be Unit Price Work, initially the Contract Price will be deemed to include for all Unit Price Work an amount equal to the sum of the unit price for each separately identified item of Unit Price Work times the estimated quantity of each item as indicated in the Agreement.
- B. The estimated quantities of items of Unit Price Work are not guaranteed and are solely for the purpose of comparison of Bids and determining an initial Contract Price. Payments to Contractor for Unit Price Work will be based on actual quantities.
- C. Each unit price will be deemed to include an amount considered by Contractor to be adequate to cover Contractor's overhead and profit for each separately identified item.
- D. Engineer will determine the actual quantities and classifications of Unit Price Work performed by Contractor. Engineer will review with Contractor the Engineer's preliminary determinations on such matters before rendering a written decision thereon (by recommendation of an Application for Payment or otherwise). Engineer's written decision

thereon will be final and binding (except as modified by Engineer to reflect changed factual conditions or more accurate data) upon Owner and Contractor, and the final adjustment of Contract Price will be set forth in a Change Order, subject to the provisions of the following paragraph.

- E. Adjustments in Unit Price
 - 1. Contractor or Owner shall be entitled to an adjustment in the unit price with respect to an item of Unit Price Work if:
 - a. the quantity of the item of Unit Price Work performed by Contractor differs materially and significantly from the estimated quantity of such item indicated in the Agreement; and
 - b. Contractor's unit costs to perform the item of Unit Price Work have changed materially and significantly as a result of the quantity change.
 - 2. The adjustment in unit price will account for and be coordinated with any related changes in quantities of other items of Work, and in Contractor's costs to perform such other Work, such that the resulting overall change in Contract Price is equitable to Owner and Contractor.
 - 3. Adjusted unit prices will apply to all units of that item.

ARTICLE 14—TESTS AND INSPECTIONS; CORRECTION, REMOVAL, OR ACCEPTANCE OF DEFECTIVE WORK

- 14.01 Access to Work
 - A. Owner, Engineer, their consultants and other representatives and personnel of Owner, independent testing laboratories, and authorities having jurisdiction have access to the Site and the Work at reasonable times for their observation, inspection, and testing. Contractor shall provide them proper and safe conditions for such access and advise them of Contractor's safety procedures and programs so that they may comply with such procedures and programs as applicable.

14.02 Tests, Inspections, and Approvals

- A. Contractor shall give Engineer timely notice of readiness of the Work (or specific parts thereof) for all required inspections and tests, and shall cooperate with inspection and testing personnel to facilitate required inspections and tests.
- B. Owner shall retain and pay for the services of an independent inspector, testing laboratory, or other qualified individual or entity to perform all inspections and tests expressly required by the Contract Documents to be furnished and paid for by Owner, except that costs incurred in connection with tests or inspections of covered Work will be governed by the provisions of Paragraph 14.05.
- C. If Laws or Regulations of any public body having jurisdiction require any Work (or part thereof) specifically to be inspected, tested, or approved by an employee or other representative of such public body, Contractor shall assume full responsibility for arranging and obtaining such inspections, tests, or approvals, pay all costs in connection therewith, and furnish Engineer the required certificates of inspection or approval.

- D. Contractor shall be responsible for arranging, obtaining, and paying for all inspections and tests required:
 - 1. by the Contract Documents, unless the Contract Documents expressly allocate responsibility for a specific inspection or test to Owner;
 - 2. to attain Owner's and Engineer's acceptance of materials or equipment to be incorporated in the Work;
 - 3. by manufacturers of equipment furnished under the Contract Documents;
 - 4. for testing, adjusting, and balancing of mechanical, electrical, and other equipment to be incorporated into the Work; and
 - 5. for acceptance of materials, mix designs, or equipment submitted for approval prior to Contractor's purchase thereof for incorporation in the Work.

Such inspections and tests will be performed by independent inspectors, testing laboratories, or other qualified individuals or entities acceptable to Owner and Engineer.

- E. If the Contract Documents require the Work (or part thereof) to be approved by Owner, Engineer, or another designated individual or entity, then Contractor shall assume full responsibility for arranging and obtaining such approvals.
- F. If any Work (or the work of others) that is to be inspected, tested, or approved is covered by Contractor without written concurrence of Engineer, Contractor shall, if requested by Engineer, uncover such Work for observation. Such uncovering will be at Contractor's expense unless Contractor had given Engineer timely notice of Contractor's intention to cover the same and Engineer had not acted with reasonable promptness in response to such notice.

14.03 Defective Work

- A. *Contractor's Obligation*: It is Contractor's obligation to assure that the Work is not defective.
- B. *Engineer's Authority*: Engineer has the authority to determine whether Work is defective, and to reject defective Work.
- C. *Notice of Defects*: Prompt written notice of all defective Work of which Owner or Engineer has actual knowledge will be given to Contractor.
- D. *Correction, or Removal and Replacement*: Promptly after receipt of written notice of defective Work, Contractor shall correct all such defective Work, whether or not fabricated, installed, or completed, or, if Engineer has rejected the defective Work, remove it from the Project and replace it with Work that is not defective.
- E. *Preservation of Warranties*: When correcting defective Work, Contractor shall take no action that would void or otherwise impair Owner's special warranty and guarantee, if any, on said Work.
- F. Costs and Damages: In addition to its correction, removal, and replacement obligations with respect to defective Work, Contractor shall pay all claims, costs, losses, and damages arising out of or relating to defective Work, including but not limited to the cost of the inspection, testing, correction, removal, replacement, or reconstruction of such defective Work, fines levied against Owner by governmental authorities because the Work is defective, and the costs of repair or replacement of work of others resulting from defective Work. Prior to final payment, if Owner and Contractor are unable to agree as to the measure of such claims, costs,
losses, and damages resulting from defective Work, then Owner may impose a reasonable set-off against payments due under Article 15.

- 14.04 Acceptance of Defective Work
 - A. If, instead of requiring correction or removal and replacement of defective Work, Owner prefers to accept it, Owner may do so (subject, if such acceptance occurs prior to final payment, to Engineer's confirmation that such acceptance is in general accord with the design intent and applicable engineering principles, and will not endanger public safety). Contractor shall pay all claims, costs, losses, and damages attributable to Owner's evaluation of and determination to accept such defective Work (such costs to be approved by Engineer as to reasonableness), and for the diminished value of the Work to the extent not otherwise paid by Contractor. If any such acceptance occurs prior to final payment, the necessary revisions in the Contract Documents with respect to the Work will be incorporated in a Change Order. If the parties are unable to agree as to the decrease in the Contract Price, reflecting the diminished value of Work so accepted, then Owner may impose a reasonable set-off against payments due under Article 15. If the acceptance of defective Work occurs after final payment, Contractor shall pay an appropriate amount to Owner.

14.05 Uncovering Work

- A. Engineer has the authority to require additional inspection or testing of the Work, whether or not the Work is fabricated, installed, or completed.
- B. If any Work is covered contrary to the written request of Engineer, then Contractor shall, if requested by Engineer, uncover such Work for Engineer's observation, and then replace the covering, all at Contractor's expense.
- C. If Engineer considers it necessary or advisable that covered Work be observed by Engineer or inspected or tested by others, then Contractor, at Engineer's request, shall uncover, expose, or otherwise make available for observation, inspection, or testing as Engineer may require, that portion of the Work in question, and provide all necessary labor, material, and equipment.
 - If it is found that the uncovered Work is defective, Contractor shall be responsible for all claims, costs, losses, and damages arising out of or relating to such uncovering, exposure, observation, inspection, and testing, and of satisfactory replacement or reconstruction (including but not limited to all costs of repair or replacement of work of others); and pending Contractor's full discharge of this responsibility the Owner shall be entitled to impose a reasonable set-off against payments due under Article 15.
 - 2. If the uncovered Work is not found to be defective, Contractor shall be allowed an increase in the Contract Price or an extension of the Contract Times, directly attributable to such uncovering, exposure, observation, inspection, testing, replacement, and reconstruction. If the parties are unable to agree as to the amount or extent thereof, then Contractor may submit a Change Proposal within 30 days of the determination that the Work is not defective.

14.06 *Owner May Stop the Work*

A. If the Work is defective, or Contractor fails to supply sufficient skilled workers or suitable materials or equipment, or fails to perform the Work in such a way that the completed Work will conform to the Contract Documents, then Owner may order Contractor to stop the Work,

or any portion thereof, until the cause for such order has been eliminated; however, this right of Owner to stop the Work will not give rise to any duty on the part of Owner to exercise this right for the benefit of Contractor, any Subcontractor, any Supplier, any other individual or entity, or any surety for, or employee or agent of any of them.

14.07 **Owner May Correct Defective Work**

- A. If Contractor fails within a reasonable time after written notice from Engineer to correct defective Work, or to remove and replace defective Work as required by Engineer, then Owner may, after 7 days' written notice to Contractor, correct or remedy any such deficiency.
- B. In exercising the rights and remedies under this Paragraph 14.07, Owner shall proceed expeditiously. In connection with such corrective or remedial action, Owner may exclude Contractor from all or part of the Site, take possession of all or part of the Work and suspend Contractor's services related thereto, and incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere. Contractor shall allow Owner, Owner's representatives, agents and employees, Owner's other contractors, and Engineer and Engineer's consultants access to the Site to enable Owner to exercise the rights and remedies under this paragraph.
- C. All claims, costs, losses, and damages incurred or sustained by Owner in exercising the rights and remedies under this Paragraph 14.07 will be charged against Contractor as set-offs against payments due under Article 15. Such claims, costs, losses and damages will include but not be limited to all costs of repair, or replacement of work of others destroyed or damaged by correction, removal, or replacement of Contractor's defective Work.
- D. Contractor shall not be allowed an extension of the Contract Times because of any delay in the performance of the Work attributable to the exercise by Owner of Owner's rights and remedies under this Paragraph 14.07.

ARTICLE 15—PAYMENTS TO CONTRACTOR; SET-OFFS; COMPLETION; CORRECTION PERIOD

- 15.01 Progress Payments
 - A. Basis for Progress Payments: The Schedule of Values established as provided in Article 2 will serve as the basis for progress payments and will be incorporated into a form of Application for Payment acceptable to Engineer. Progress payments for Unit Price Work will be based on the number of units completed during the pay period, as determined under the provisions of Paragraph 13.03. Progress payments for cost-based Work will be based on Cost of the Work completed by Contractor during the pay period.
 - B. Applications for Payments
 - 1. At least 20 days before the date established in the Agreement for each progress payment (but not more often than once a month), Contractor shall submit to Engineer for review an Application for Payment filled out and signed by Contractor covering the Work completed as of the date of the Application and accompanied by such supporting documentation as is required by the Contract Documents.
 - If payment is requested on the basis of materials and equipment not incorporated in the 2. Work but delivered and suitably stored at the Site or at another location agreed to in writing, the Application for Payment must also be accompanied by: (a) a bill of sale, invoice, copies of subcontract or purchase order payments, or other documentation

establishing full payment by Contractor for the materials and equipment; (b) at Owner's request, documentation warranting that Owner has received the materials and equipment free and clear of all Liens; and (c) evidence that the materials and equipment are covered by appropriate property insurance, a warehouse bond, or other arrangements to protect Owner's interest therein, all of which must be satisfactory to Owner.

- 3. Beginning with the second Application for Payment, each Application must include an affidavit of Contractor stating that all previous progress payments received by Contractor have been applied to discharge Contractor's legitimate obligations associated with prior Applications for Payment.
- 4. The amount of retainage with respect to progress payments will be as stipulated in the Agreement.
- C. Review of Applications
 - Engineer will, within 10 days after receipt of each Application for Payment, including each resubmittal, either indicate in writing a recommendation of payment and present the Application to Owner, or return the Application to Contractor indicating in writing Engineer's reasons for refusing to recommend payment. In the latter case, Contractor may make the necessary corrections and resubmit the Application.
 - 2. Engineer's recommendation of any payment requested in an Application for Payment will constitute a representation by Engineer to Owner, based on Engineer's observations of the executed Work as an experienced and qualified design professional, and on Engineer's review of the Application for Payment and the accompanying data and schedules, that to the best of Engineer's knowledge, information and belief:
 - a. the Work has progressed to the point indicated;
 - b. the quality of the Work is generally in accordance with the Contract Documents (subject to an evaluation of the Work as a functioning whole prior to or upon Substantial Completion, the results of any subsequent tests called for in the Contract Documents, a final determination of quantities and classifications for Unit Price Work under Paragraph 13.03, and any other qualifications stated in the recommendation); and
 - c. the conditions precedent to Contractor's being entitled to such payment appear to have been fulfilled in so far as it is Engineer's responsibility to observe the Work.
 - 3. By recommending any such payment Engineer will not thereby be deemed to have represented that:
 - a. inspections made to check the quality or the quantity of the Work as it has been performed have been exhaustive, extended to every aspect of the Work in progress, or involved detailed inspections of the Work beyond the responsibilities specifically assigned to Engineer in the Contract; or
 - b. there may not be other matters or issues between the parties that might entitle Contractor to be paid additionally by Owner or entitle Owner to withhold payment to Contractor.

- 4. Neither Engineer's review of Contractor's Work for the purposes of recommending payments nor Engineer's recommendation of any payment, including final payment, will impose responsibility on Engineer:
 - a. to supervise, direct, or control the Work;
 - b. for the means, methods, techniques, sequences, or procedures of construction, or the safety precautions and programs incident thereto;
 - c. for Contractor's failure to comply with Laws and Regulations applicable to Contractor's performance of the Work;
 - d. to make any examination to ascertain how or for what purposes Contractor has used the money paid by Owner; or
 - e. to determine that title to any of the Work, materials, or equipment has passed to Owner free and clear of any Liens.
- 5. Engineer may refuse to recommend the whole or any part of any payment if, in Engineer's opinion, it would be incorrect to make the representations to Owner stated in Paragraph 15.01.C.2.
- 6. Engineer will recommend reductions in payment (set-offs) necessary in Engineer's opinion to protect Owner from loss because:
 - a. the Work is defective, requiring correction or replacement;
 - b. the Contract Price has been reduced by Change Orders;
 - c. Owner has been required to correct defective Work in accordance with Paragraph 14.07, or has accepted defective Work pursuant to Paragraph 14.04;
 - d. Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible; or
 - e. Engineer has actual knowledge of the occurrence of any of the events that would constitute a default by Contractor and therefore justify termination for cause under the Contract Documents.
- D. Payment Becomes Due
 - 1. Ten days after presentation of the Application for Payment to Owner with Engineer's recommendation, the amount recommended (subject to any Owner set-offs) will become due, and when due will be paid by Owner to Contractor.
- E. Reductions in Payment by Owner
 - 1. In addition to any reductions in payment (set-offs) recommended by Engineer, Owner is entitled to impose a set-off against payment based on any of the following:
 - a. Claims have been made against Owner based on Contractor's conduct in the performance or furnishing of the Work, or Owner has incurred costs, losses, or damages resulting from Contractor's conduct in the performance or furnishing of the Work, including but not limited to claims, costs, losses, or damages from workplace injuries, adjacent property damage, non-compliance with Laws and Regulations, and patent infringement;

- b. Contractor has failed to take reasonable and customary measures to avoid damage, delay, disruption, and interference with other work at or adjacent to the Site;
- c. Contractor has failed to provide and maintain required bonds or insurance;
- d. Owner has been required to remove or remediate a Hazardous Environmental Condition for which Contractor is responsible;
- e. Owner has incurred extra charges or engineering costs related to submittal reviews, evaluations of proposed substitutes, tests and inspections, or return visits to manufacturing or assembly facilities;
- f. The Work is defective, requiring correction or replacement;
- g. Owner has been required to correct defective Work in accordance with Paragraph 14.07, or has accepted defective Work pursuant to Paragraph 14.04;
- h. The Contract Price has been reduced by Change Orders;
- i. An event has occurred that would constitute a default by Contractor and therefore justify a termination for cause;
- j. Liquidated or other damages have accrued as a result of Contractor's failure to achieve Milestones, Substantial Completion, or final completion of the Work;
- k. Liens have been filed in connection with the Work, except where Contractor has delivered a specific bond satisfactory to Owner to secure the satisfaction and discharge of such Liens; or
- I. Other items entitle Owner to a set-off against the amount recommended.
- 2. If Owner imposes any set-off against payment, whether based on its own knowledge or on the written recommendations of Engineer, Owner will give Contractor immediate written notice (with a copy to Engineer) stating the reasons for such action and the specific amount of the reduction, and promptly pay Contractor any amount remaining after deduction of the amount so withheld. Owner shall promptly pay Contractor the amount so withheld, or any adjustment thereto agreed to by Owner and Contractor, if Contractor remedies the reasons for such action. The reduction imposed will be binding on Contractor unless it duly submits a Change Proposal contesting the reduction.
- 3. Upon a subsequent determination that Owner's refusal of payment was not justified, the amount wrongfully withheld will be treated as an amount due as determined by Paragraph 15.01.D.1 and subject to interest as provided in the Agreement.

15.02 Contractor's Warranty of Title

A. Contractor warrants and guarantees that title to all Work, materials, and equipment furnished under the Contract will pass to Owner free and clear of (1) all Liens and other title defects, and (2) all patent, licensing, copyright, or royalty obligations, no later than 7 days after the time of payment by Owner.

15.03 Substantial Completion

A. When Contractor considers the entire Work ready for its intended use Contractor shall notify Owner and Engineer in writing that the entire Work is substantially complete and request that Engineer issue a certificate of Substantial Completion. Contractor shall at the same time submit to Owner and Engineer an initial draft of punch list items to be completed or corrected before final payment.

- B. Promptly after Contractor's notification, Owner, Contractor, and Engineer shall make an inspection of the Work to determine the status of completion. If Engineer does not consider the Work substantially complete, Engineer will notify Contractor in writing giving the reasons therefor.
- C. If Engineer considers the Work substantially complete, Engineer will deliver to Owner a preliminary certificate of Substantial Completion which will fix the date of Substantial Completion. Engineer shall attach to the certificate a punch list of items to be completed or corrected before final payment. Owner shall have 7 days after receipt of the preliminary certificate during which to make written objection to Engineer as to any provisions of the certificate or attached punch list. If, after considering the objections to the provisions of the preliminary certificate, Engineer concludes that the Work is not substantially complete, Engineer will, within 14 days after submission of the preliminary certificate to Owner, notify Contractor in writing that the Work is not substantially complete, stating the reasons therefor. If Owner does not object to the provisions of the certificate, or if despite consideration of Owner's objections Engineer concludes that the Work is substantially complete, then Engineer will, within said 14 days, execute and deliver to Owner and Contractor a final certificate of Substantial Completion (with a revised punch list of items to be completed or corrected) reflecting such changes from the preliminary certificate as Engineer believes justified after consideration of any objections from Owner.
- D. At the time of receipt of the preliminary certificate of Substantial Completion, Owner and Contractor will confer regarding Owner's use or occupancy of the Work following Substantial Completion, review the builder's risk insurance policy with respect to the end of the builder's risk coverage, and confirm the transition to coverage of the Work under a permanent property insurance policy held by Owner. Unless Owner and Contractor agree otherwise in writing, Owner shall bear responsibility for security, operation, protection of the Work, property insurance, maintenance, heat, and utilities upon Owner's use or occupancy of the Work.
- E. After Substantial Completion the Contractor shall promptly begin work on the punch list of items to be completed or corrected prior to final payment. In appropriate cases Contractor may submit monthly Applications for Payment for completed punch list items, following the progress payment procedures set forth above.
- F. Owner shall have the right to exclude Contractor from the Site after the date of Substantial Completion subject to allowing Contractor reasonable access to remove its property and complete or correct items on the punch list.

15.04 Partial Use or Occupancy

A. Prior to Substantial Completion of all the Work, Owner may use or occupy any substantially completed part of the Work which has specifically been identified in the Contract Documents, or which Owner, Engineer, and Contractor agree constitutes a separately functioning and usable part of the Work that can be used by Owner for its intended purpose without

significant interference with Contractor's performance of the remainder of the Work, subject to the following conditions:

- At any time, Owner may request in writing that Contractor permit Owner to use or occupy any such part of the Work that Owner believes to be substantially complete. If and when Contractor agrees that such part of the Work is substantially complete, Contractor, Owner, and Engineer will follow the procedures of Paragraph 15.03.A through 15.03.E for that part of the Work.
- 2. At any time, Contractor may notify Owner and Engineer in writing that Contractor considers any such part of the Work substantially complete and request Engineer to issue a certificate of Substantial Completion for that part of the Work.
- 3. Within a reasonable time after either such request, Owner, Contractor, and Engineer shall make an inspection of that part of the Work to determine its status of completion. If Engineer does not consider that part of the Work to be substantially complete, Engineer will notify Owner and Contractor in writing giving the reasons therefor. If Engineer considers that part of the Work to be substantially complete, the provisions of Paragraph 15.03 will apply with respect to certification of Substantial Completion of that part of the Work and the division of responsibility in respect thereof and access thereto.
- 4. No use or occupancy or separate operation of part of the Work may occur prior to compliance with the requirements of Paragraph 6.04 regarding builder's risk or other property insurance.
- 15.05 Final Inspection
 - A. Upon written notice from Contractor that the entire Work or an agreed portion thereof is complete, Engineer will promptly make a final inspection with Owner and Contractor and will notify Contractor in writing of all particulars in which this inspection reveals that the Work, or agreed portion thereof, is incomplete or defective. Contractor shall immediately take such measures as are necessary to complete such Work or remedy such deficiencies.
- 15.06 Final Payment
 - A. Application for Payment
 - 1. After Contractor has, in the opinion of Engineer, satisfactorily completed all corrections identified during the final inspection and has delivered, in accordance with the Contract Documents, all maintenance and operating instructions, schedules, guarantees, bonds, certificates or other evidence of insurance, certificates of inspection, annotated record documents (as provided in Paragraph 7.12), and other documents, Contractor may make application for final payment.
 - 2. The final Application for Payment must be accompanied (except as previously delivered) by:
 - a. all documentation called for in the Contract Documents;
 - b. consent of the surety, if any, to final payment;
 - c. satisfactory evidence that all title issues have been resolved such that title to all Work, materials, and equipment has passed to Owner free and clear of any Liens or other title defects, or will so pass upon final payment.

- d. a list of all duly pending Change Proposals and Claims; and
- e. complete and legally effective releases or waivers (satisfactory to Owner) of all Lien rights arising out of the Work, and of Liens filed in connection with the Work.
- 3. In lieu of the releases or waivers of Liens specified in Paragraph 15.06.A.2 and as approved by Owner, Contractor may furnish receipts or releases in full and an affidavit of Contractor that: (a) the releases and receipts include all labor, services, material, and equipment for which a Lien could be filed; and (b) all payrolls, material and equipment bills, and other indebtedness connected with the Work for which Owner might in any way be responsible, or which might in any way result in liens or other burdens on Owner's property, have been paid or otherwise satisfied. If any Subcontractor or Supplier fails to furnish such a release or receipt in full, Contractor may furnish a bond or other collateral satisfactory to Owner to indemnify Owner against any Lien, or Owner at its option may issue joint checks payable to Contractor and specified Subcontractors and Suppliers.
- B. Engineer's Review of Final Application and Recommendation of Payment: If, on the basis of Engineer's observation of the Work during construction and final inspection, and Engineer's review of the final Application for Payment and accompanying documentation as required by the Contract Documents, Engineer is satisfied that the Work has been completed and Contractor's other obligations under the Contract have been fulfilled, Engineer will, within 10 days after receipt of the final Application for Payment, indicate in writing Engineer's recommendation of final payment and present the final Application for Payment to Owner for payment. Such recommendation will account for any set-offs against payment that are necessary in Engineer's opinion to protect Owner from loss for the reasons stated above with respect to progress payments. Otherwise, Engineer will return the Application for Payment to Contractor, indicating in writing the reasons for refusing to recommend final payment, in which case Contractor shall make the necessary corrections and resubmit the Application for Payment.
- C. Notice of Acceptability: In support of its recommendation of payment of the final Application for Payment, Engineer will also give written notice to Owner and Contractor that the Work is acceptable, subject to stated limitations in the notice and to the provisions of Paragraph 15.07.
- D. *Completion of Work*: The Work is complete (subject to surviving obligations) when it is ready for final payment as established by the Engineer's written recommendation of final payment and issuance of notice of the acceptability of the Work.
- E. *Final Payment Becomes Due*: Upon receipt from Engineer of the final Application for Payment and accompanying documentation, Owner shall set off against the amount recommended by Engineer for final payment any further sum to which Owner is entitled, including but not limited to set-offs for liquidated damages and set-offs allowed under the provisions of this Contract with respect to progress payments. Owner shall pay the resulting balance due to Contractor within 30 days of Owner's receipt of the final Application for Payment from Engineer.
- 15.07 Waiver of Claims
 - A. By making final payment, Owner waives its claim or right to liquidated damages or other damages for late completion by Contractor, except as set forth in an outstanding Claim,

appeal under the provisions of Article 17, set-off, or express reservation of rights by Owner. Owner reserves all other claims or rights after final payment.

B. The acceptance of final payment by Contractor will constitute a waiver by Contractor of all claims and rights against Owner other than those pending matters that have been duly submitted as a Claim, or appealed under the provisions of Article 17.

15.08 Correction Period

- A. If within one year after the date of Substantial Completion (or such longer period of time as may be prescribed by the Supplementary Conditions or the terms of any applicable special guarantee required by the Contract Documents), Owner gives Contractor written notice that any Work has been found to be defective, or that Contractor's repair of any damages to the Site or adjacent areas has been found to be defective, then after receipt of such notice of defect Contractor shall promptly, without cost to Owner and in accordance with Owner's written instructions:
 - 1. correct the defective repairs to the Site or such adjacent areas;
 - 2. correct such defective Work;
 - 3. remove the defective Work from the Project and replace it with Work that is not defective, if the defective Work has been rejected by Owner, and
 - 4. satisfactorily correct or repair or remove and replace any damage to other Work, to the work of others, or to other land or areas resulting from the corrective measures.
- B. Owner shall give any such notice of defect within 60 days of the discovery that such Work or repairs is defective. If such notice is given within such 60 days but after the end of the correction period, the notice will be deemed a notice of defective Work under Paragraph 7.17.B.
- C. If, after receipt of a notice of defect within 60 days and within the correction period, Contractor does not promptly comply with the terms of Owner's written instructions, or in an emergency where delay would cause serious risk of loss or damage, Owner may have the defective Work corrected or repaired or may have the rejected Work removed and replaced. Contractor shall pay all costs, losses, and damages (including but not limited to all fees and charges of engineers, architects, attorneys, and other professionals and all court or arbitration or other dispute resolution costs) arising out of or relating to such correction or repair or such removal and replacement (including but not limited to all costs of repair or replacement of work of others). Contractor's failure to pay such costs, losses, and damages within 10 days of invoice from Owner will be deemed the start of an event giving rise to a Claim under Paragraph 12.01.B, such that any related Claim must be brought within 30 days of the failure to pay.
- D. In special circumstances where a particular item of equipment is placed in continuous service before Substantial Completion of all the Work, the correction period for that item may start to run from an earlier date if so provided in the Specifications.
- E. Where defective Work (and damage to other Work resulting therefrom) has been corrected or removed and replaced under this paragraph, the correction period hereunder with respect to such Work will be extended for an additional period of one year after such correction or removal and replacement has been satisfactorily completed.

F. Contractor's obligations under this paragraph are in addition to all other obligations and warranties. The provisions of this paragraph are not to be construed as a substitute for, or a waiver of, the provisions of any applicable statute of limitation or repose.

ARTICLE 16—SUSPENSION OF WORK AND TERMINATION

- 16.01 Owner May Suspend Work
 - A. At any time and without cause, Owner may suspend the Work or any portion thereof for a period of not more than 90 consecutive days by written notice to Contractor and Engineer. Such notice will fix the date on which Work will be resumed. Contractor shall resume the Work on the date so fixed. Contractor shall be entitled to an adjustment in the Contract Price or an extension of the Contract Times directly attributable to any such suspension. Any Change Proposal seeking such adjustments must be submitted no later than 30 days after the date fixed for resumption of Work.

16.02 Owner May Terminate for Cause

- A. The occurrence of any one or more of the following events will constitute a default by Contractor and justify termination for cause:
 - 1. Contractor's persistent failure to perform the Work in accordance with the Contract Documents (including, but not limited to, failure to supply sufficient skilled workers or suitable materials or equipment, or failure to adhere to the Progress Schedule);
 - 2. Failure of Contractor to perform or otherwise to comply with a material term of the Contract Documents;
 - 3. Contractor's disregard of Laws or Regulations of any public body having jurisdiction; or
 - 4. Contractor's repeated disregard of the authority of Owner or Engineer.
- B. If one or more of the events identified in Paragraph 16.02.A occurs, then after giving Contractor (and any surety) 10 days' written notice that Owner is considering a declaration that Contractor is in default and termination of the Contract, Owner may proceed to:
 - 1. declare Contractor to be in default, and give Contractor (and any surety) written notice that the Contract is terminated; and
 - 2. enforce the rights available to Owner under any applicable performance bond.
- C. Subject to the terms and operation of any applicable performance bond, if Owner has terminated the Contract for cause, Owner may exclude Contractor from the Site, take possession of the Work, incorporate in the Work all materials and equipment stored at the Site or for which Owner has paid Contractor but which are stored elsewhere, and complete the Work as Owner may deem expedient.
- D. Owner may not proceed with termination of the Contract under Paragraph 16.02.B if Contractor within 7 days of receipt of notice of intent to terminate begins to correct its failure to perform and proceeds diligently to cure such failure.
- E. If Owner proceeds as provided in Paragraph 16.02.B, Contractor shall not be entitled to receive any further payment until the Work is completed. If the unpaid balance of the Contract Price exceeds the cost to complete the Work, including all related claims, costs, losses, and damages (including but not limited to all fees and charges of engineers, architects,

attorneys, and other professionals) sustained by Owner, such excess will be paid to Contractor. If the cost to complete the Work including such related claims, costs, losses, and damages exceeds such unpaid balance, Contractor shall pay the difference to Owner. Such claims, costs, losses, and damages incurred by Owner will be reviewed by Engineer as to their reasonableness and, when so approved by Engineer, incorporated in a Change Order. When exercising any rights or remedies under this paragraph, Owner shall not be required to obtain the lowest price for the Work performed.

- F. Where Contractor's services have been so terminated by Owner, the termination will not affect any rights or remedies of Owner against Contractor then existing or which may thereafter accrue, or any rights or remedies of Owner against Contractor or any surety under any payment bond or performance bond. Any retention or payment of money due Contractor by Owner will not release Contractor from liability.
- G. If and to the extent that Contractor has provided a performance bond under the provisions of Paragraph 6.01.A, the provisions of that bond will govern over any inconsistent provisions of Paragraphs 16.02.B and 16.02.D.

16.03 Owner May Terminate for Convenience

- A. Upon 7 days' written notice to Contractor and Engineer, Owner may, without cause and without prejudice to any other right or remedy of Owner, terminate the Contract. In such case, Contractor shall be paid for (without duplication of any items):
 - 1. completed and acceptable Work executed in accordance with the Contract Documents prior to the effective date of termination, including fair and reasonable sums for overhead and profit on such Work;
 - 2. expenses sustained prior to the effective date of termination in performing services and furnishing labor, materials, or equipment as required by the Contract Documents in connection with uncompleted Work, plus fair and reasonable sums for overhead and profit on such expenses; and
 - 3. other reasonable expenses directly attributable to termination, including costs incurred to prepare a termination for convenience cost proposal.
- B. Contractor shall not be paid for any loss of anticipated profits or revenue, post-termination overhead costs, or other economic loss arising out of or resulting from such termination.

16.04 Contractor May Stop Work or Terminate

- A. If, through no act or fault of Contractor, (1) the Work is suspended for more than 90 consecutive days by Owner or under an order of court or other public authority, or (2) Engineer fails to act on any Application for Payment within 30 days after it is submitted, or (3) Owner fails for 30 days to pay Contractor any sum finally determined to be due, then Contractor may, upon 7 days' written notice to Owner and Engineer, and provided Owner or Engineer do not remedy such suspension or failure within that time, terminate the contract and recover from Owner payment on the same terms as provided in Paragraph 16.03.
- B. In lieu of terminating the Contract and without prejudice to any other right or remedy, if Engineer has failed to act on an Application for Payment within 30 days after it is submitted, or Owner has failed for 30 days to pay Contractor any sum finally determined to be due, Contractor may, 7 days after written notice to Owner and Engineer, stop the Work until payment is made of all such amounts due Contractor, including interest thereon. The

provisions of this paragraph are not intended to preclude Contractor from submitting a Change Proposal for an adjustment in Contract Price or Contract Times or otherwise for expenses or damage directly attributable to Contractor's stopping the Work as permitted by this paragraph.

ARTICLE 17—FINAL RESOLUTION OF DISPUTES

17.01 Methods and Procedures

- A. *Disputes Subject to Final Resolution*: The following disputed matters are subject to final resolution under the provisions of this article:
 - 1. A timely appeal of an approval in part and denial in part of a Claim, or of a denial in full, pursuant to Article 12; and
 - 2. Disputes between Owner and Contractor concerning the Work, or obligations under the Contract Documents, that arise after final payment has been made.
- B. *Final Resolution of Disputes*: For any dispute subject to resolution under this article, Owner or Contractor may:
 - 1. elect in writing to invoke the dispute resolution process provided for in the Supplementary Conditions;
 - agree with the other party to submit the dispute to another dispute resolution process; or
 - 3. if no dispute resolution process is provided for in the Supplementary Conditions or mutually agreed to, give written notice to the other party of the intent to submit the dispute to a court of competent jurisdiction.

ARTICLE 18—MISCELLANEOUS

18.01 *Giving Notice*

- A. Whenever any provision of the Contract requires the giving of written notice to Owner, Engineer, or Contractor, it will be deemed to have been validly given only if delivered:
 - 1. in person, by a commercial courier service or otherwise, to the recipient's place of business;
 - 2. by registered or certified mail, postage prepaid, to the recipient's place of business; or
 - 3. by e-mail to the recipient, with the words "Formal Notice" or similar in the e-mail's subject line.

18.02 Computation of Times

A. When any period of time is referred to in the Contract by days, it will be computed to exclude the first and include the last day of such period. If the last day of any such period falls on a Saturday or Sunday or on a day made a legal holiday by the law of the applicable jurisdiction, such day will be omitted from the computation.

18.03 Cumulative Remedies

A. The duties and obligations imposed by these General Conditions and the rights and remedies available hereunder to the parties hereto are in addition to, and are not to be construed in any way as a limitation of, any rights and remedies available to any or all of them which are otherwise imposed or available by Laws or Regulations, by special warranty or guarantee, or by other provisions of the Contract. The provisions of this paragraph will be as effective as if repeated specifically in the Contract Documents in connection with each particular duty, obligation, right, and remedy to which they apply.

18.04 Limitation of Damages

A. With respect to any and all Change Proposals, Claims, disputes subject to final resolution, and other matters at issue, neither Owner nor Engineer, nor any of their officers, directors, members, partners, employees, agents, consultants, or subcontractors, shall be liable to Contractor for any claims, costs, losses, or damages sustained by Contractor on or in connection with any other project or anticipated project.

18.05 No Waiver

- A. A party's non-enforcement of any provision will not constitute a waiver of that provision, nor will it affect the enforceability of that provision or of the remainder of this Contract.
- 18.06 Survival of Obligations
 - A. All representations, indemnifications, warranties, and guarantees made in, required by, or given in accordance with the Contract, as well as all continuing obligations indicated in the Contract, will survive final payment, completion, and acceptance of the Work or termination of the Contract or of the services of Contractor.
- 18.07 Controlling Law
 - A. This Contract is to be governed by the law of the state in which the Project is located.

18.08 Assignment of Contract

A. Unless expressly agreed to elsewhere in the Contract, no assignment by a party to this Contract of any rights under or interests in the Contract will be binding on the other party without the written consent of the party sought to be bound; and, specifically but without limitation, money that may become due and money that is due may not be assigned without such consent (except to the extent that the effect of this restriction may be limited by law), and unless specifically stated to the contrary in any written consent to an assignment, no assignment will release or discharge the assignor from any duty or responsibility under the Contract.

18.09 Successors and Assigns

A. Owner and Contractor each binds itself, its successors, assigns, and legal representatives to the other party hereto, its successors, assigns, and legal representatives in respect to all covenants, agreements, and obligations contained in the Contract Documents.

18.10 Headings

A. Article and paragraph headings are inserted for convenience only and do not constitute parts of these General Conditions.

Item 8.

SPECIAL PROVISIONS

Project Information

This project is being funded by the Village of Ridgeway (Owner) with assistance from the Wisconsin Department of Natural Resources. All scope items and work listed below AND as required to complete the project shall be accounted for in the Contractors total bid price. Contractor shall minimize disturbance to the existing park as much as possible. Public access will be permitted during construction in areas other than the designated work zone.

Reference

The work included in this project scope shall be in completed accordance with the plans. The scope items shall be described in the following section and executed according to the referenced standards. The latest edition of the "State of Wisconsin, Department of Transportation, Standard Specifications for Highway and Structure Construction" shall serve as the referenced standard.

1 – Mobilization

<u>Scope</u>: This work shall include the cost of moving in equipment and supplies to the site, the establishment of necessary stockpiles and trailer, and moving the same off site. This item also includes construction staking and layout required by the contractor.

<u>Reference:</u> Section 619 of the standard specifications.

<u>Payment</u>: This work will be measured and paid for at the contract lump sum price. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

2 – Erosion Control

<u>Scope</u>: This work consists of furnishing, installing, and maintaining tracking pad, inlet protection, and silt fence/sock at the locations designated on the Plans and where directed by the A/E, removing and disposing of sediment deposits, and removing fence at the completion of the work.

<u>Reference</u>: Section 628 of the standard specifications, WDNR Conservation Practice Standards No. 1070, and WDNR Conservation Practice Standard No. 1057.

<u>Payment</u>: This work will be measured and paid for at the contract lump sum price. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

3 – Demolition

<u>Scope</u>: This work consists of removing and disposing of miscellaneous items at the locations designated on the Plans and where directed by the A/E. Items include but are not limited to saw cutting, clearing and grubbing, tree protection, abandoning and relocating utilities, pavement removal, and miscellaneous item removals.

<u>Payment</u>: This work will be measured and paid for at the contract lump sum price. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

4 – Earthwork

<u>Scope</u>: This work consists of stripping and stockpiling of topsoil, excavating, filling, grading, importing new fill, compacting, preparation of sub-grade, disposal of excess materials, temporary drainage, berming, and related earthwork for the project.

<u>Payment</u>: This work will be measured and paid for at the contract lump sum price. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

5 – Dense Graded Basecourse

<u>Scope</u>: This work describes furnishing and installing 1-1/4["] dense graded basecourse under asphalt and concrete pavement areas as shown in the details and plans.

<u>Reference</u>: Section 305 of the standard specifications.

<u>Payment</u>: This work will be measured and paid for at the contract unit price per ton. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

6 – Concrete Pavement – 5" Thickened

<u>Scope</u>: This work consists of furnishing and installing concrete pavement including forming, reinforcing, and finishing at locations designated on the Plans in accordance with the standard specifications. ADA ramps and thickened edge areas shall be incidental to this item.

<u>Reference:</u> Section 602 of the standard specifications.

<u>Payment</u>: This work will be measured and paid for at the contract unit price per square foot. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

7 – Concrete Pavement – 5"

<u>Scope</u>: This work consists of furnishing and installing concrete pavement including forming, reinforcing, and finishing at locations designated on the Plans in accordance with the standard specifications. ADA ramps shall be incidental to this item.

<u>Reference:</u> Section 602 of the standard specifications.

<u>Payment</u>: This work will be measured and paid for at the contract unit price per square foot. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

8 – Tactile Warning Plates

<u>Scope</u>: This work describes detectable warning fields in accordance with the details shown in the Plans, and as hereinafter provided. All preparation required for this work is incidental to this item.

Reference: Section 602 of the Standard Specifications

<u>Payment</u>: This work will be measured and paid for at the contract price per square foot. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

9 – Asphaltic Pavement – Parking Lot

<u>Scope</u>: This work describes constructing asphalt pavement and pavement patching in accordance with the details shown in the plans and as hereinafter provided.

<u>Reference:</u> Section 460 of the WISDOT Specifications.

<u>Payment</u>: This work will measured and paid for at the contract unit price per ton. This shall include compensation for furnishing all materials, including preparing, sawing, hauling, mixing, placing, shaping, compacting, asphaltic materials and for all labor, equipment, tools and incidentals necessary to complete this item of work.

10 – Pavement Markings

<u>Scope</u>: This work describes pavement markings in accordance with the details shown in the Plans, and as hereinafter provided. Pavement markings include all striping and symbols and shall be white reflective water based paint. All preparation required for this work is incidental to this item.

Reference: Section 646 and 614 of the Standard Specifications

<u>Payment</u>: This work will be measured and paid for at the contract lump sum price. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

11 – Regulatory Signs

<u>Scope</u>: This work describes the furnishing and installing of regulatory signs as shown on Plans and manufacturers recommendations. Post shall be galvanized steel and set plumb.

<u>Reference:</u> Section 637 of the Standard Specifications and per manufacturer's recommendations.

<u>Payment</u>: This work will be measured and paid for at the contract unit price per each. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

12 – 4" Storm Sewer

<u>Scope</u>: This work describes furnishing and installing PVC stormwater piping as shown on the plans.

<u>Reference</u>: Per Section 612 of the standard specifications.

<u>Payment</u>: This work will be measured and paid for at the contract unit price per linear foot. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

13 – Storm Sewer Outfall

<u>Scope</u>: This work describes furnishing and installing the concrete endwall outfall, riprap, joint ties, geotextile fabric, rodent guard, and all fittings required to connect to existing stormwater structure as shown on the plans.

Reference: Per Section 612 of the standard specifications.

<u>Payment</u>: This work will be measured and paid for at the contract lump sum price This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

14 – Trench Drain

Scope: This work describes furnishing and installing linear trench drain as shown on the plans.

<u>Reference</u>: Per Section 612 of the standard specifications. Product = NDS Pro Dura Slope Trench Drain or approved equal.

<u>Payment</u>: This work will be measured and paid for at the contract unit price per linear foot. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

15 – Site Electrical and Exterior Athletic Lighting

Scope: This work consists of demolition, modifications, and new electrical construction in and on the electrical building, including all lighting, power, and distribution equipment where required and noted on plans, and as required for a complete installation. Site electrical shall also consist of modifications to the existing electric utility service and associated appurtenances where noted on the plans and as required for a complete installation. Demolition work shall include, but not be limited to, the removal, disconnection, and disposal of electrical wiring, fixtures, equipment, and associated appurtenances where required and noted on the plans. The work encompasses the demolition of both above-ground, concealed, and below-grade electrical systems. Contractor shall take measures to prevent the release of harmful substances, dust, or pollutants into the environment during demolition. Exterior athletic lighting shall consist of furnishing and installing light fixtures, lighting control devices, cables and conduits, fittings, bonding, all new electrical equipment and modifications where required and noted on plans, and miscellaneous work to provide power to the baseball field athletic lighting as shown on the plans whether or not specifically called for. This item also includes any modifications required to the existing service and coordination with the Village as required.

Reference: All electrical work and equipment to be installed in compliance with NEC and Section 651, 652, 653, 655, 656, 657, and 659 in a neat and clean manner. All efforts shall be made to locate devices in a neat and orderly manner such as to not draw attention to them. Owner or A/E shall be consulted prior to installation to confirm manner and location of component placement. This shall include all locations of equipment and devices in and on the electrical building. All outdoor components shall be rated for outdoor wet-use operation and tamper resistant where required by the NEC. Refer to plans and details for placement locations, heights, and material types.

<u>Payment</u>: This work will be measured and paid for at the contract lump sum price. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

16 – Restoration

<u>Scope</u>: This work consists of restoration of areas disturbed during the project and includes furnishing and installing topsoil, seed, and straw matting. Watering shall be considered incidental.

Reference: Per WisDOT Section 625, 627, 628, 629, 630, 631

<u>LAWN SEED</u>: Seed Mixture shall be WisDOT #40, per Section 630. Seed shall be free of non-seed debris and invasive weed species.

<u>ESTABLISHMENT PERIOD</u>: Contractor to maintain new turf areas until a substantial growth is observed and approved by the Owner or A/E. Any bare areas greater than 5" square shall be re-seeded as necessary until growth is acceptable. Contractor shall water as required to produce substantial growth, temporary irrigation systems as needed shall be incidental to this item. Mowing shall occur as needed until final project acceptance. Turf should be kept at below 3" to avoid lay down.

<u>Payment</u>: This work will be measured and paid for at the contract lump sum price. This shall include compensation for accepting delivery, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

Alternate Bid Items

A1 – Removal and Installation of Light Poles

Scope: This work is a bid alternate only and consists of demolition and removal of existing light poles and installation of Owner-furnished light poles where required and noted on the plans and as required

Division 26 technical specifications for additional base bid scope of work.

Reference: All electrical work and equipment to be installed in compliance with NEC and Section 651, 652, 653, 655, 656, 657, and 659 in a neat and clean manner. All efforts shall be made to locate devices in a neat and orderly manner such as to not draw attention to them. Owner or A/E shall be consulted prior to installation to confirm manner and location of component placement. This shall include all locations of equipment and devices in and on the electrical building. All outdoor components shall be rated for outdoor wet-use operation and tamper resistant where required by the NEC. Refer to plans and details for placement locations, heights, and material types.

for a complete installation. Refer to Site Electrical and Exterior Athletic Lighting Special Provision and the

<u>Payment</u>: This work will be measured and paid for at the contract lump sum price. This shall include compensation for furnishing, delivering, installing, maintenance, and removal of materials, and for all labor, equipment, and incidentals necessary to complete this item of work.

SECTION 260010 - SUPPLEMENTAL REQUIREMENTS FOR ELECTRICAL

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section Includes:
 - 1. Supplemental requirements applicable to Work specified in Division 26.

1.2 REFERENCES

- A. Abbreviations and Acronyms for Electrical Terms and Units of Measure:
 - 1. A: Ampere, unit of electrical current.
 - 2. AC or ac: Alternating current.
 - 3. AFCI: Arc-fault circuit interrupter.
 - 4. AIC: Ampere interrupting capacity.
 - 5. AL, Al, or ALUM: Aluminum.
 - 6. AWG: American wire gauge; see ASTM B258.
 - 7. BIL: Basic impulse insulation level.
 - 8. CAD: Computer-aided design or drafting.
 - 9. CB: Circuit breaker.
 - 10. CU or Cu: Copper.
 - 11. CU-AL or AL-CU: Copper-aluminum.
 - 12. dB: Decibel, a unitless logarithmic ratio of two electrical, acoustical, or optical power values.
 - 13. dB(A-weighted) or dB(A): Decibel acoustical sound pressure level with A-weighting applied in accordance with IEC 61672-1.
 - 14. dB(adjusted) or dBa: Decibel weighted absolute noise power with respect to 3.16 pW (minus 85 dBm).
 - 15. dBm: Decibel absolute power with respect to 1 mW.
 - 16. DC or dc: Direct current.
 - 17. fc: Footcandle, a unit of illuminance equal to one lumen per square foot.
 - 18. FLC: Full-load current.
 - 19. ft.: Foot.
 - 20. GEC: Grounding electrode conductor.
 - 21. GFCI: Ground-fault circuit interrupter.
 - 22. GFPE: Ground-fault protection of equipment.
 - 23. GND: Ground.
 - 24. HDPE: High-density polyethylene.
 - 25. HP or hp: Horsepower.
 - 26. Hz: Hertz.
 - 27. inch: Inch. To avoid confusion, the abbreviation "in." is not used.

- IP: Ingress protection rating (enclosures); Internet protocol (communications).
- 29. kAIC: Kiloampere interrupting capacity.
- 30. kcmil or MCM: One thousand circular mils.
- 31. kV: Kilovolt.
- 32. kVA: Kilovolt-ampere.
- 33. kVAr or kVAR: Kilovolt-ampere reactive.
- 34. kW: Kilowatt.
- 35. kWh: Kilowatt-hour.
- 36. lb: Pound (weight).
- 37. LED: Light-emitting diode.
- 38. LRC: Locked-rotor current.
- 39. MLO: Main lugs only.
- 40. MVA: Megavolt-ampere.
- 41. mW: Milliwatt.
- 42. MW: Megawatt.
- 43. MWh: Megawatt-hour.
- 44. NC: Normally closed.
- 45. NiCd: Nickel cadmium.
- 46. NIU: Network interface unit.
- 47. NO: Normally open.
- 48. NPT: National (American) standard pipe taper.
- 49. OCPD: Overcurrent protective device.
- 50. ONT: Optical network terminal.
- 51. PF or pf: Power factor.
- 52. PHEV: Plug-in hybrid electric vehicle.
- 53. PVC: Polyvinyl chloride.
- 54. pW: Picowatt.
- 55. RFI: Radio-frequency interference (electrical); Request for interpretation (contract).
- 56. RMS or rms: Root-mean-square.
- 57. RPM or rpm: Revolutions per minute.
- 58. SPD: Surge protective device.
- 59. sq.: Square.
- 60. SWD: Switching duty.
- 61. TEFC: Totally enclosed fan-cooled.
- 62. TR: Tamper resistant.
- 63. TVSS: Transient voltage surge suppressor.
- 64. UL: Underwriters Laboratories, Inc. (standards) or UL LLC (services).
- 65. UL CCN: UL Category Control Number.
- 66. UPS: Uninterruptible power supply.
- 67. USB: Universal serial bus.

- 68. UV: Ultraviolet.
- 69. V: Volt, unit of electromotive force.
- 70. V(ac): Volt, alternating current.
- 71. V(dc): Volt, direct current.
- 72. VA: Volt-ampere, unit of complex electrical power.
- 73. VAr: Volt-ampere reactive, unit of reactive electrical power.
- 74. W: Watt, unit of real electrical power.
- 75. Wh: Watt-hour, unit of electrical energy usage.
- 76. WR: Weather resistant.
- B. Abbreviations and Acronyms for Electrical Raceway Types:
 - 1. EMT: Electrical metallic tubing.
 - 2. EMT-A: Aluminum electrical metallic tubing.
 - 3. EMT-S: Steel electrical metallic tubing.
 - 4. EMT-SS: Stainless steel electrical metallic tubing.
 - 5. ENT: Electrical nonmetallic tubing.
 - 6. EPEC: Electrical HDPE underground conduit.
 - 7. EPEC-40: Schedule 40 electrical HDPE underground conduit.
 - 8. EPEC-80: Schedule 80 electrical HDPE underground conduit.
 - 9. EPEC-A: Type A electrical HDPE underground conduit.
 - 10. EPEC-B: Type B electrical HDPE underground conduit.
 - 11. ERMC: Electrical rigid metal conduit.
 - 12. ERMC-A: Aluminum electrical rigid metal conduit.
 - 13. ERMC-S: Steel electrical rigid metal conduit.
 - 14. ERMC-S-G: Galvanized-steel electrical rigid metal conduit.
 - 15. ERMC-S-PVC: PVC-coated-steel electrical rigid metal conduit.
 - 16. ERMC-SS: Stainless steel electrical rigid metal conduit.
 - 17. FMC: Flexible metal conduit.
 - 18. FMC-A: Aluminum flexible metal conduit.
 - 19. FMC-S: Steel flexible metal conduit.
 - 20. FMT: Steel flexible metallic tubing.
 - 21. FNMC: Flexible nonmetallic conduit. See LFNC.
 - 22. HDPE: See EPEC.
 - 23. IMC: Steel electrical intermediate metal conduit.
 - 24. LFMC: Liquidtight flexible metal conduit.
 - 25. LFMC-A: Aluminum liquidtight flexible metal conduit.
 - 26. LFMC-S: Steel liquidtight flexible metal conduit.
 - 27. LFMC-SS: Stainless steel liquidtight flexible metal conduit.
 - 28. LFNC: Liquidtight flexible nonmetallic conduit.
 - 29. LFNC-A: Layered (Type A) liquidtight flexible nonmetallic conduit.
 - 30. LFNC-B: Integral (Type B) liquidtight flexible nonmetallic conduit.
 - 31. LFNC-C: Corrugated (Type C) liquidtight flexible nonmetallic conduit.

Community Park Improvements 20235700

- 32. PVC: Rigid PVC conduit.
- 33. PVC-40: Schedule 40 rigid PVC conduit.
- 34. PVC-80: Schedule 80 rigid PVC Conduit.
- 35. PVC-A: Type A rigid PVC concrete-encased conduit.
- 36. PVC-EB: Type EB rigid PVC concrete-encased underground conduit.
- 37. RGS: See ERMC-S-G.
- 38. RMC: See ERMC.
- 39. RTRC: Reinforced thermosetting resin conduit.
- 40. RTRC-AG: Low-halogen, aboveground reinforced thermosetting resin conduit.
- 41. RTRC-AG-HW: Heavy wall, low-halogen, aboveground reinforced thermosetting resin conduit.
- 42. RTRC-AG-SW: Standard wall, low-halogen, aboveground reinforced thermosetting resin conduit.
- 43. RTRC-AG-XW: Extra heavy wall, low-halogen, aboveground reinforced thermosetting resin conduit.
- 44. RTRC-BG: Low-halogen, belowground reinforced thermosetting resin conduit.
- C. Abbreviations and Acronyms for Electrical Cable Types:
 - 1. AC: Armored cable.
 - 2. MC: Metal-clad cable.
 - 3. MC-HL: Metal-clad cable, hazardous location.
 - 4. MTW: Moisture-, heat-, and oil-resistant thermoplastic cable (machine tool wiring).
 - 5. MV: Medium-voltage cable.
 - 6. RHH: Thermoset rubber, heat-resistant cable (high heat).
 - 7. RHW: Thermoset rubber, moisture-resistant cable.
 - 8. SE: Service-entrance cable.
 - 9. SER: Service-entrance cable, round.
 - 10. SEU: Service-entrance cable, flat.
 - 11. THW: Thermoplastic, heat- and moisture-resistant cable.
 - 12. THHN: Thermoplastic, heat-resistant cable with nylon jacket outer sheath.
 - 13. THHW: Thermoplastic, heat- and moisture-resistant cable.
 - 14. THWN: Thermoplastic, moisture- and heat-resistant cable with nylon jacket outer sheath.
 - 15. TW: Thermoplastic, moisture-resistant cable.
 - 16. UF: Underground feeder and branch-circuit cable.
 - 17. USE: Underground service-entrance cable.
 - 18. XHH: Cross-linked polyethylene, heat-resistant cable.
 - 19. XHHW: Cross-linked polyethylene, heat- and moisture-resistant cable.
- D. Definitions:

- 1. Basic Impulse Insulation Level: Reference insulation level expressed in impulse crest voltage with a standard wave not longer than 1.5 times 50 microseconds and 1.5 times 40 microseconds.
- 2. Direct Buried: Installed underground without encasement in concrete or other protective material.
- 3. Enclosure: The case or housing of an apparatus, or the fence or wall(s) surrounding an installation, to prevent personnel from accidentally contacting energized parts or to protect the equipment from physical damage. Types of enclosures and enclosure covers include the following:
 - a. Cabinet: An enclosure that is designed for either surface mounting or flush mounting and is provided with a frame, mat, or trim in which a swinging door or doors are or can be hung.
 - b. Concrete Box: A box intended for use in poured concrete.
 - c. Conduit Body: A means for providing access to the interior of a conduit or tubing system through one or more removable covers at a junction or terminal point. In the United States, conduit bodies are listed in accordance with outlet box requirements.
 - d. Conduit Box: A box having threaded openings or knockouts for conduit, EMT, or fittings.
 - e. Cutout Box: An enclosure designed for surface mounting that has swinging doors or covers secured directly to and telescoping with the walls of the enclosure.
 - f. Device Box: A box with provisions for mounting a wiring device directly to the box.
 - g. Extension Ring: A ring intended to extend the sides of an outlet box or device box to increase the box depth, volume, or both.
 - h. Floor Box: A box mounted in the floor intended for use with a floor box cover and other components to complete the floor box enclosure.
 - i. Junction Box: A box with a blank cover that joins different runs of raceway or cable and provides space for connection and branching of the enclosed conductors.
 - j. Outlet Box: A box that provides access to a wiring system having pryout openings, knockouts, threaded entries, or hubs in either the sides or the back, or both, for the entrance of conduit, conduit or cable fittings, or cables, with provisions for mounting an outlet box cover, but without provisions for mounting a wiring device directly to the box.
 - k. Pedestal Floor Box Cover: A floor box cover that, when installed as intended, provides a means for typically vertical or near-vertical mounting of receptacle outlets above the floor's finished surface.
 - I. Pull Box: A box with a blank cover that joins different runs of raceway and provides access for pulling or replacing the enclosed cables or conductors.
 - m. Ring: A sleeve, which is not necessarily round, used for positioning a recessed wiring device flush with the plaster, concrete, drywall, or other wall surface.
 - n. Ring Cover: A box cover, with raised center portion to accommodate a specific wall or ceiling thickness, for mounting wiring devices or luminaires flush with the surface.

- o. Termination Box: An enclosure designed for installation of termination base assemblies consisting of bus bars, terminal strips, or terminal blocks with provision for wire connectors to accommodate incoming or outgoing conductors, or both.
- 4. Luminaire: A complete lighting unit consisting of a light source such as a lamp, together with the parts designed to position the light source and connect it to the power supply. It may also include parts to protect the light source or the ballast or to distribute the light.
- 5. Receptacle: A fixed connecting device arranged for insertion of a power cord plug. Also called a power jack.
- 6. Receptacle Outlet: One or more receptacles mounted in a box with a suitable protective cover.
- 7. Sheath: A continuous metallic covering for conductors or cables.
- 8. UL Category Control Number: An alphabetic or alphanumeric code used to identify product categories covered by UL's Listing, Classification, and Recognition Services.
- 9. Voltage Class: For specified circuits and equipment, voltage classes are defined as follows:
 - a. Control Voltage: Listed and labeled for use in remote-control, signaling, and power-limited circuits supplied by Class 2 or Class 3 power supplies having rated output not greater than 150 V and 5 A, allowing use of alternate wiring methods complying with NFPA 70, Article 725.
 - b. Low Voltage: Listed and labeled for use in circuits supplied by Class 1 or other power supplies having rated output not greater than 1000 V, requiring use of wiring methods complying with NFPA 70, Article 300, Part I.

1.3 COORDINATION

- A. Interruption of Existing Electrical Service: Do not interrupt electrical service to facilities occupied by Owner or others unless permitted under the following conditions:
 - 1. Notify Owner and Engineer, in writing, no fewer than seven (7) days in advance of proposed interruption of electrical service.
 - 2. Do not proceed with interruption of electrical service without Owner and Engineer's written permission.

1.4 SEQUENCING

- A. Conduct and submit results of power system studies before submitting Product Data and Shop Drawings for electrical equipment.
- 1.5 ACTION SUBMITTALS
 - A. Coordination Drawings for Structural Supports: Show coordination of structural supports for equipment and devices, including restraints and bracing for control of seismic and wind loads, with other systems, equipment, and structural supports in the vicinity.

1.6 INFORMATIONAL SUBMITTALS

- A. Electrical Installation Schedule: At preconstruction meeting, and periodically thereafter as dates change, provide schedule for electrical installation Work to Owner and Design-Builder including, but not limited to, milestone dates for the following activities:
 - 1. Submission of action submittals specified in Division 26.
 - 2. Orders placed for major electrical equipment.
 - 3. Arrival of major electrical equipment on-site.
 - 4. Utility service outages.
 - 5. Utility service inspection and activation.
 - 6. System startup, testing, and commissioning activities for LED sports lighting.
 - 7. Requests for special inspections.
 - 8. Requests for inspections by authorities having jurisdiction.
- B. Certificates:
- C. Qualification Statements:
 - 1. For qualified regional manufacturer.
 - 2. For structural professional engineer.
 - 3. For electrical professional engineer.
 - 4. For lighting professional engineer.
 - 5. For EPM specialist.
 - 6. For welder.
 - 7. For ERMC-S-PVC raceway Installer.
 - 8. For exterior athletic lighting Installer.
 - 9. For lighting testing and inspecting agency.

1.7 CLOSEOUT SUBMITTALS

- A. EPM Program Binders:
 - 1. Complete Set: On USB media that is clearly and permanently labeled with attached placard on lanyard to prevent misplacement.
- B. Operation and Maintenance Data:
 - 1. Provide emergency, operation, and maintenance manuals for each system, equipment, and device listed below:
 - a. LED sports lighting.
 - 2. Include the following information:
 - a. Manufacturer's operating specifications.
 - b. User's guides for software and hardware.

Supplemental Requirements For Electrical 260010 - 7

- c. Lists of spare parts, replacement components, tools, and consumable supplies recommended to be stored at Project site.
- d. Detailed instructions covering operation under both normal and abnormal conditions.
- e. Manufacturer's instructions for setting field-adjustable components.
- f. Manufacturer's instructions for testing, adjusting, and reprogramming microprocessor controls.
- g. EPSS: Manufacturer's system checklists, maintenance schedule, and maintenance log sheets in accordance with NFPA 110.
- h. Exterior pole inspection and repair procedures.

1.8 FIELD CONDITIONS

A. Modeling, analysis, product selection, installation, and quality control for Work specified in Division 26 must comply with requirements specified in Section 260011 "Facility Performance Requirements for Electrical."

PART 2 - PRODUCTS

2.1 SUBSTITUTION LIMITATIONS FOR ELECTRICAL EQUIPMENT

- A. Substitution requests for electrical equipment will be entertained under the following conditions:
 - 1. Substitution requests may be submitted for consideration prior to the Electrical Preconstruction Conference if accompanied by value analysis data indicating that substitution will comply with Project performance requirements while significantly increasing value for Owner throughout life of facility.
 - 2. Substitution requests may be submitted for consideration concurrently with submission of power system study reports when those reports indicate that substitution is necessary for safety of maintenance personnel and facility occupants.
 - 3. Contractor is responsible for sequencing and scheduling power system studies and electrical equipment procurement. After the Electrical Preconstruction Conference, insufficient lead time for electrical equipment delivery will not be considered a valid reason for substitution.

2.2 FACILITY ELECTRICAL PREVENTIVE MAINTENANCE (EPM) PROGRAM BINDERS

- A. Description: Set of binders containing operation and maintenance data for facility's electrical equipment that was compiled during analysis of installed electrical Work for Facility EPM Program development.
- B. Applicable Standards:
 - 1. Regulatory Requirements: Comply with recommendations in NFPA 70B.
 - 2. General Characteristics:
 - a. Volume 1 Introduction:

- 1) Summarize how Facility EPM Program Analysis was performed, how data were collected, and how volumes are organized.
- Describe Facility EPM Program and provide recommended policies and procedures for implementing the program and keeping it current.
- Provide place for Owner to identify contact information for employees responsible for implementing and maintaining Facility EPM Program.
- b. Volume 2 Facility Safety, Hazards Awareness, and Emergency Procedures:
 - 1) Include training requirements for employees and contractors.
- c. Volume 3 Operating Procedures for Electrical Equipment and Controls.
- d. Volume 4 Facility Diagrams and Schedules:
 - 1) Include essential wiring diagrams.
- e. Volume 5 Inventory of Facility Equipment Using Electrical Power:
 - 1) Include identifying designations and nameplate data.
 - 2) Include warranty and maintenance contract information.
- f. Volume 6 Inventory of Facility Tools, Supplies, and Personnel Protective Equipment:
 - 1) Include list of spare parts, replacement components, tools, and consumable supplies recommended to be stored at facility.
 - 2) Include calibration and servicing data for each item.
- g. Volume 7 Inspection, Testing, and Repair (IT&R) Plan:
 - 1) Include tables showing frequency of activities for each item.
 - 2) Include annual schedule with activities mapped to specific days of the year.
 - 3) Include light fixture inspection and repair procedures.
- h. Volume 9 Spare Parts List:
 - 1) Include list of all parts required to perform IT&R procedures.
 - 2) Identify quantities of which parts are recommended to be stored on-site.
 - 3) Include source contact information and budget cost for each item.
- i. Volume 10 Construction Project Closeout Record Documentation:
 - 1) Include records of electrical system startup and commissioning activities.
 - 2) Include records of baseline inspections and tests.
 - 3) Include records of baseline settings for adjustable equipment and devices.

PART 3 - EXECUTION

- 3.1 DEVELOPMENT OF FACILITY EPM PROGRAM
 - A. Facility EPM Program must be developed by qualified EPM specialist.
 - B. Conduct Facility EPM Program analysis in accordance with NFPA 70B recommendations.
 - C. Compile operation and maintenance data from Facility EPM Program analysis and submit EPM Program Binders.
- 3.2 INSTALLATION OF ELECTRICAL WORK
 - A. Unless more stringent requirements are specified in the Contract Documents or manufacturers' written instructions, comply with NFPA 70 and NECA NEIS 1 for installation of Work specified in Division 26. Consult Design-Builder for resolution of conflicting requirements.
- 3.3 SYSTEM STARTUP
 - A. Commissioning Activities:
 - 1. LED sports lighting shall be commissioned in accordance with manufacturer's standards.
- 3.4 FIELD QUALITY CONTROL
 - A. Administrant for Field Tests and Inspections of Lighting Installations:
 - 1. Administer and perform tests and inspections with assistance of factoryauthorized service representative.

3.5 CLOSEOUT ACTIVITIES

- A. Demonstration:
 - 1. With assistance from factory-authorized service representatives, demonstrate to Owner's maintenance and clerical personnel how to operate the following systems and equipment:
 - a. LED sports lighting.
 - 2. Allow Owner to record demonstrations.
- B. Training:
 - 1. With assistance from factory-authorized service representatives, train Owner's maintenance personnel on the following topics:
 - a. How to implement EPM Program.

END OF SECTION 260010

SECTION 260519 - LOW-VOLTAGE ELECTRICAL POWER CONDUCTORS AND CABLES

PART 1 - GENERAL

- 1.1 REGULATORY REQUIREMENTS
 - A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
 - B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.

1.2 SUMMARY

- A. Section Includes:
 - 1. Copper building wire rated 600 V or less.
 - 2. Connectors, splices, and terminations rated 600 V and less.
- B. Related Requirements:
 - 1. Section 260010 "Supplemental Requirements for Electrical" for additional abbreviations, definitions, submittals, qualifications, testing agencies, and other Project requirements applicable to Work specified in this Section.

1.3 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- 1.4 INFORMATIONAL SUBMITTALS
 - A. Field quality-control reports.

PART 2 - PRODUCTS

- 2.1 COPPER BUILDING WIRE
 - A. Description: Flexible, insulated and uninsulated, drawn copper current-carrying conductor with an overall insulation layer or jacket, or both, rated 600 V or less.
 - B. Standards:
 - 1. Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and use.
 - 2. Conductor and Cable Marking: Comply with wire and cable marking according to UL's "Wire and Cable Marking and Application Guide."

Low-Voltage Electrical Power Conductors And Cables 260519 - 1

- C. Conductors: Copper, complying with ASTM B3 for bare annealed copper and with ASTM B8 for stranded conductors.
- D. Conductor Insulation:
 - 1. Type NM: Comply with UL 83 and UL 719.
 - 2. Type USE-2 and Type SE: Comply with UL 854.
 - 3. Type THHN and Type THWN-2: Comply with UL 83.
 - 4. Type THW and Type THW-2: Comply with NEMA WC-70/ICEA S-95-658 and UL 83.
 - 5. Type XHHW-2: Comply with UL 44.
- 2.2 CONNECTORS AND SPLICES
 - A. Description: Factory-fabricated connectors, splices, and lugs of size, ampacity rating, material, type, and class for application and service indicated; listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and use.
 - B. Jacketed Cable Connectors: For steel and aluminum jacketed cables, zinc die-cast with set screws, designed to connect conductors specified in this Section.
 - C. Lugs: One piece, seamless, designed to terminate conductors specified in this Section.
 - 1. Material: Copper.
 - 2. Type: One or Two hole with standard barrels.
 - 3. Termination: Compression lugs for #6 AWG and larger, crimp lugs for #8 AWG and smaller.

PART 3 - EXECUTION

- 3.1 CONDUCTOR MATERIAL APPLICATIONS
 - A. Feeders:
 - 1. Copper. Conductors must be solid for No. 10 AWG and smaller; stranded for No. 8 AWG and larger.
 - B. Branch Circuits:
 - 1. Copper, Solid for No. 10 AWG and smaller; stranded for No. 8 AWG and larger.
- 3.2 CONDUCTOR INSULATION AND MULTICONDUCTOR CABLE APPLICATIONS AND WIRING METHODS
 - A. Service Entrance: Type THHN/THWN-2, single conductors in raceway or Type XHHW-2, single conductors in raceway.
 - B. Feeders Concealed in Ceilings, Walls, Partitions, and Crawlspaces: Type THHN/THWN-2, single conductors in raceway.

Low-Voltage Electrical Power Conductors And Cables 260519 - 2

- C. Feeders Concealed in Concrete, below Slabs-on-Grade, and Underground: Type THHN/THWN-2, single conductors in raceway or Type XHHW-2, single conductors in raceway.
- D. Exposed Branch Circuits, Including in Crawlspaces: Type THHN/THWN-2, single conductors in raceway or Metal-clad cable, Type MC.
- E. Branch Circuits Concealed in Ceilings, Walls, and Partitions: Type THHN/THWN-2, single conductors in raceway.
- F. Branch Circuits Concealed in Concrete, below Slabs-on-Grade, and Underground: Type THHN/THWN-2, single conductors in raceway or Type XHHW-2, single conductors in raceway.

3.3 INSTALLATION OF CONDUCTORS AND CABLES

- A. Conceal cables in finished walls, ceilings, and floors unless otherwise indicated.
- B. Complete raceway installation between conductor and cable termination points according to Section 260533 "Raceways and Boxes for Electrical Systems" prior to pulling conductors and cables.
- C. Use manufacturer-approved pulling compound or lubricant where necessary; compound used must not deteriorate conductor or insulation. Do not exceed manufacturer's recommended maximum pulling tensions and sidewall pressure values.
- D. Use pulling means, including fish tape, cable, rope, and basket-weave wire/cable grips, that will not damage cables or raceway.
- E. Install exposed cables parallel and perpendicular to surfaces of exposed structural members, and follow surface contours where possible.
- F. Support cables according to Section 260529 "Hangers and Supports for Electrical Systems."

3.4 CONNECTIONS

- A. Tighten electrical connectors and terminals according to manufacturer's published torque-tightening values. If manufacturer's torque values are not indicated, use those specified in UL 486A-486B.
- B. Make splices, terminations, and taps that are compatible with conductor material and that possess equivalent or better mechanical strength and insulation ratings than unspliced conductors.
 - 1. Use oxide inhibitor in each splice, termination, and tap for aluminum conductors.
- C. Wiring at Outlets: Install conductor at each outlet, with at least 6 inch of slack.
- D. Comply with requirements in [Section 284621.11 "Addressable Fire-Alarm Systems"] [Section 284621.13 "Conventional Fire-Alarm Systems"] for connecting, terminating, and identifying wires and cables.

3.5 IDENTIFICATION

- A. Identify and color-code conductors and cables according to Section 260553 "Identification for Electrical Systems."
- B. Identify each spare conductor at each end with identity number and location of other end of conductor, and identify as spare conductor.

3.6 SLEEVE AND SLEEVE-SEAL INSTALLATION FOR ELECTRICAL PENETRATIONS

A. Install sleeves and sleeve seals at penetrations of exterior floor and wall assemblies. Comply with requirements in Section 260544 "Sleeves and Sleeve Seals for Electrical Raceways and Cabling."

3.7 FIRESTOPPING

A. Apply firestopping to electrical penetrations of fire-rated floor and wall assemblies to restore original fire-resistance rating of assembly according to Section 078413 "Penetration Firestopping."

END OF SECTION 260519

SECTION 260526 - GROUNDING AND BONDING FOR ELECTRICAL SYSTEMS

PART 1 - GENERAL

1.1 SUMMARY

A. Section includes grounding and bonding systems and equipment.

1.2 ACTION SUBMITTALS

A. Product Data: For each type of product.

1.3 CLOSEOUT SUBMITTALS

- A. Operation and maintenance data.
 - 1. Plans showing as-built, dimensioned locations of grounding features specified in "Field Quality Control" Article, including the following:
 - a. Ground rods.

1.4 REGULATORY REQUIREMENTS

- A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
- B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.

PART 2 - PRODUCTS

- 2.1 SYSTEM DESCRIPTION
 - A. Electrical Components, Devices, and Accessories: Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and application.
 - B. Comply with UL 467 for grounding and bonding materials and equipment.

2.2 CONDUCTORS

- A. Insulated Conductors: Copper or tinned-copper wire or cable insulated for 600 V unless otherwise required by applicable Code or authorities having jurisdiction.
- B. Bare Copper Conductors:
 - 1. Solid Conductors: ASTM B 3.
 - 2. Stranded Conductors: ASTM B 8.
 - 3. Tinned Conductors: ASTM B 33.

Community Park Improvements 20235700

Grounding And Bonding For Electrical Systems 260526 - 1

- 4. Bonding Cable: 28 kcmil, 14 strands of No. 17 AWG conductor, 1/4 inch in diameter.
- 5. Bonding Conductor: No. 4 or No. 6 AWG, stranded conductor.
- 6. Bonding Jumper: Copper tape, braided conductors terminated with copper ferrules; 1-5/8 inches wide and 1/16 inch thick.
- 7. Tinned Bonding Jumper: Tinned-copper tape, braided conductors terminated with copper ferrules; 1-5/8 inches wide and 1/16 inch thick.
- C. Grounding Bus: Predrilled rectangular bars of annealed copper, 1/4 by 4 inches in cross section, with 9/32-inch holes spaced 1-1/8 inches apart. Stand-off insulators for mounting shall comply with UL 891 for use in switchboards, 600 V and shall be Lexan or PVC, impulse tested at 5000 V.

2.3 CONNECTORS

- A. Listed and labeled by an NRTL acceptable to authorities having jurisdiction for applications in which used and for specific types, sizes, and combinations of conductors and other items connected.
- B. Welded Connectors: Exothermic-welding kits of types recommended by kit manufacturer for materials being joined and installation conditions.
- C. Bus-Bar Connectors: Mechanical type, cast silicon bronze, solderless compressiontype wire terminals, and long-barrel, two-bolt connection to ground bus bar.
- D. Beam Clamps: Mechanical type, terminal, ground wire access from four directions, with dual, tin-plated or silicon bronze bolts.
- E. Cable-to-Cable Connectors: Compression type, copper or copper alloy.
- F. Cable Tray Ground Clamp: Mechanical type, zinc-plated malleable iron.
- G. Conduit Hubs: Mechanical type, terminal with threaded hub.
- H. Ground Rod Clamps: Mechanical type, copper or copper alloy, terminal with hex head bolt.
- I. Service Post Connectors: Mechanical type, bronze alloy terminal, in short- and long-stud lengths, capable of single and double conductor connections.
- J. Signal Reference Grid Clamp: Mechanical type, stamped-steel terminal with hex head screw.
- K. Straps: Solid copper, copper lugs. Rated for 600 A.
- L. Tower Ground Clamps: Mechanical type, copper or copper alloy, terminal two-piece clamp.
- M. U-Bolt Clamps: Mechanical type, copper or copper alloy, terminal listed for direct burial.
 - 1. U-bolt type with malleable-iron clamp and copper ground connector.

Community Park Improvements 20235700

Grounding And Bonding For Electrical Systems 260526 - 2

2.4 GROUNDING ELECTRODES

- A. Ground Rods: Copper-clad steel; 3/4 inch by 10 feet.
- B. Ground Plates: 1/4 inch thick, hot-dip galvanized.

PART 3 - EXECUTION

- 3.1 APPLICATIONS
 - A. Conductors: Install solid conductor for No. 8 AWG and smaller, and stranded conductors for No. 6 AWG and larger unless otherwise indicated.
 - B. Underground Grounding Conductors: Install bare copper conductor, No. 2/0 AWG minimum.
 - 1. Bury at least 24 inches below grade.
 - C. Grounding Bus: Install in Electrical Building where shown on drawings.
 - 1. Install bus horizontally, on insulated spacers 2 inches minimum from wall, 6 inches above finished floor unless otherwise indicated.
 - 2. Where indicated on both sides of doorways, route bus up to top of door frame, across top of doorway, and down; connect to horizontal bus.
 - D. Conductor Terminations and Connections:
 - 1. Pipe and Equipment Grounding Conductor Terminations: Bolted connectors.
 - 2. Underground Connections: Welded connectors.
 - 3. Connections to Structural Steel: Welded connectors.
- 3.2 GROUNDING AT THE SERVICE
 - A. Equipment grounding conductors and grounding electrode conductors shall be connected to the ground bus. Install a main bonding jumper between the neutral and ground buses.
- 3.3 GROUNDING UNDERGROUND DISTRIBUTION SYSTEM COMPONENTS
 - A. Comply with IEEE C2 grounding requirements.
 - B. Pad-Mounted Utility Transformers and Switches: Grounding by Utility.
- 3.4 EQUIPMENT GROUNDING
 - A. Install insulated equipment grounding conductors with all feeders and branch circuits.
 - B. Install insulated equipment grounding conductors with the following items, in addition to those required by NFPA 70:
 - 1. Feeders and branch circuits.
 - 2. Lighting circuits.

 Community Park Improvements
 Grounding And Bonding For Electrical Systems

 20235700
 260526 - 3

3. Receptacle circuits.

C. Poles Supporting Outdoor Lighting Fixtures: Install grounding electrode and a separate insulated equipment grounding conductor in addition to grounding conductor installed with branch-circuit conductors. See detail on plans.

3.5 INSTALLATION

- A. Grounding Conductors: Route along shortest and straightest paths possible unless otherwise indicated or required by Code. Avoid obstructing access or placing conductors where they may be subjected to strain, impact, or damage.
- B. Ground Rods: Drive rods until tops are 2 inches below final grade unless otherwise indicated.
 - 1. Interconnect ground rods with grounding electrode conductor below grade and as otherwise indicated. Make connections without exposing steel or damaging coating if any.
 - 2. For grounding electrode system, install at least three rods spaced at least onerod length from each other and located at least the same distance from other grounding electrodes, and connect to the service grounding electrode conductor.
- C. Bonding Straps and Jumpers: Install in locations accessible for inspection and maintenance except where routed through short lengths of conduit.
 - 1. Bonding to Structure: Bond straps directly to basic structure, taking care not to penetrate any adjacent parts.
 - 2. Bonding to Equipment Mounted on Vibration Isolation Hangers and Supports: Install bonding so vibration is not transmitted to rigidly mounted equipment.
 - 3. Use exothermic-welded connectors for outdoor locations; if a disconnect-type connection is required, use a bolted clamp.

3.6 FIELD QUALITY CONTROL

- A. Perform tests and inspections.
- B. Tests and Inspections:
 - 1. After installing grounding system but before permanent electrical circuits have been energized, test for compliance with requirements.
 - Inspect physical and mechanical condition. Verify tightness of accessible, bolted, electrical connections with a calibrated torque wrench according to manufacturer's written instructions.
 - 3. Test completed grounding system at each location where a maximum groundresistance level is specified, at service disconnect enclosure grounding terminal, and at individual ground rods. Make tests at ground rods before any conductors are connected.

Community Park Improvements 20235700

Grounding And Bonding For Electrical Systems 260526 - 4

Iconica

Commented [KP1]: This section from previous spec section and copied over. Does it need to be here?

Commented [KP2]: This is a much-expanded section. Please review.
- a. Measure ground resistance no fewer than two full days after last trace of precipitation and without soil being moistened by any means other than natural drainage or seepage and without chemical treatment or other artificial means of reducing natural ground resistance.
- b. Perform tests by fall-of-potential method according to IEEE 81.
- 4. Prepare dimensioned Drawings locating each test well, ground rod and groundrod assembly, and other grounding electrodes. Identify each by letter in alphabetical order, and key to the record of tests and observations. Include the number of rods driven and their depth at each location, and include observations of weather and other phenomena that may affect test results. Describe measures taken to improve test results.
- C. Grounding system will be considered defective if it does not pass tests and inspections.
- D. Prepare test and inspection reports.
- E. Report measured ground resistances that exceed the following values:
 - 1. Power and Lighting Equipment or System with Capacity of 500 kVA and Less: 5 ohms.
- F. Excessive Ground Resistance: If resistance to ground exceeds specified values, notify Engineer in writing promptly and include recommendations to reduce ground resistance.

END OF SECTION 260526

SECTION 260529 - HANGERS AND SUPPORTS FOR ELECTRICAL SYSTEMS

PART 1 - GENERAL

1.1 SUMMARY

- A. Section Includes:
 - 1. Steel slotted support systems.
 - 2. Conduit and cable support devices.
 - 3. Support for conductors in vertical conduit.
 - 4. Structural steel for fabricated supports and restraints.
 - Mounting, anchoring, and attachment components, including powderactuated fasteners, mechanical expansion anchors, concrete inserts, clamps, through bolts, toggle bolts, and hanger rods.
 - 6. Fabricated metal equipment support assemblies.

1.2 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Shop Drawings: For fabrication and installation details for electrical hangers and support systems.
 - 1. Hangers. Include product data for components.
 - 2. Slotted support systems.
 - 3. Equipment supports.
- 1.3 INFORMATIONAL SUBMITTALS
 - A. Welding certificates.
- 1.4 QUALITY ASSURANCE
 - A. Welding Qualifications: Qualify procedures and personnel according to the following:
 - 1. AWS D1.1/D1.1M.
 - 2. AWS D1.2/D1.2M.
- 1.5 REGULATORY REQUIREMENTS
 - A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
 - B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.

Community Park Improvements	Hangers And Supports For Electrical Systems
20235700	260529 - 1

Iconica

PART 2 - PRODUCTS

2.1 SUPPORT, ANCHORAGE, AND ATTACHMENT COMPONENTS

- A. Steel Slotted Support Systems: Preformed steel channels and angles with minimum 13/32-inch- diameter holes at a maximum of 8 inches o.c. in at least one surface.
 - 1. Standard: Comply with MFMA-4 factory-fabricated components for field assembly.
 - 2. Material for Channel, Fittings, and Accessories: Plain steel
 - 3. Channel Width: 1-1/4 inches (31.75 mm)
 - 4. Metallic Coatings: Hot-dip galvanized after fabrication and applied according to MFMA-4.
 - Nonmetallic Coatings: Manufacturer's standard PVC, polyurethane, or polyester coating applied according to MFMA-4.
 - 6. Protect finishes on exposed surfaces from damage by applying a strippable, temporary protective covering before shipping.
- B. Conduit and Cable Support Devices: Steel hangers, clamps, and associated fittings, designed for types and sizes of raceway or cable to be supported.
- C. Support for Conductors in Vertical Conduit: Factory-fabricated assembly consisting of threaded body and insulating wedging plug or plugs for non-armored electrical conductors or cables in riser conduits. Plugs shall have number, size, and shape of conductor gripping pieces as required to suit individual conductors or cables supported. Body shall be made of malleable iron.
- D. Structural Steel for Fabricated Supports and Restraints: ASTM A 36/A 36M steel plates, shapes, and bars; black and galvanized.
- E. Mounting, Anchoring, and Attachment Components: Items for fastening electrical items or their supports to building surfaces include the following:
 - 1. Powder-Actuated Fasteners: Threaded-steel stud, for use in hardened portland cement concrete, steel, or wood, with tension, shear, and pullout capacities appropriate for supported loads and building materials where used.
 - Mechanical-Expansion Anchors: Insert-wedge-type, zinc-coated steel, for use in hardened portland cement concrete, with tension, shear, and pullout capacities appropriate for supported loads and building materials where used.
 - 3. Clamps for Attachment to Steel Structural Elements: MSS SP-58 units are suitable for attached structural element.
 - Through Bolts: Structural type, hex head, and high strength. Comply with ASTM F 3125/F 3125M, Grade A325.
 - 5. Toggle Bolts: All steel springhead type.
 - 6. Hanger Rods: Threaded steel.
- 2.2 FABRICATED METAL EQUIPMENT SUPPORT ASSEMBLIES
 - A. Description: Welded or bolted structural-steel shapes, shop or field fabricated to fit dimensions of supported equipment.

 Community Park Improvements
 Hangers And Supports For Electrical Systems

 20235700
 260529 - 2

Iconica

B. Materials: Comply with requirements in Section 055000 "Metal Fabrications" for steel shapes and plates.

2.3 EXTERIOR LOCATIONS

A. Support, anchorage, attachment components, and fabricated metal equipment support assemblies in exterior locations shall be constructed of stainless steel.

PART 3 - EXECUTION

- 3.1 APPLICATION
 - A. Comply with the following standards for application and installation requirements of hangers and supports, except where requirements on Drawings or in this Section are stricter:
 - 1. NECA 1.
 - 2. NECA 101
 - B. Comply with requirements in Section 078413 "Penetration Firestopping" for firestopping materials and installation for penetrations through fire-rated walls, ceilings, and assemblies.
 - C. Comply with requirements for raceways and boxes specified in Section 260533 "Raceways and Boxes for Electrical Systems."
 - D. Maximum Support Spacing and Minimum Hanger Rod Size for Raceways: Space supports for EMT, IMC, and RMC as required by NFPA 70. Minimum rod size shall be 1/4 inch in diameter.
 - E. Multiple Raceways or Cables: Install trapeze-type supports fabricated with steel slotted support system, sized so capacity can be increased by at least 25 percent in future without exceeding specified design load limits.
 - 1. Secure raceways and cables to these supports with single-bolt conduit clamps.
 - F. Spring-steel clamps designed for supporting single conduits without bolts may be used for 1-1/2-inch and smaller raceways serving branch circuits and communication systems above suspended ceilings, and for fastening raceways to trapeze supports.
- 3.2 SUPPORT INSTALLATION
 - A. Comply with NECA 1 and NECA 101 for installation requirements except as specified in this article.
 - B. Raceway Support Methods: In addition to methods described in NECA 1, EMT, IMC and RMC may be supported by openings through structure members, according to NFPA 70.

Community Park Improvements 20235700

Hangers And Supports For Electrical Systems 260529 - 3

Iconica

Commented [CM1]: Commented [CM2]:

- C. Strength of Support Assemblies: Where not indicated, select sizes of components so strength will be adequate to carry present and future static loads within specified loading limits. Minimum static design load used for strength determination shall be weight of supported components plus 200 lb.
- D. Mounting and Anchorage of Surface-Mounted Equipment and Components: Anchor and fasten electrical items and their supports to building structural elements by the following methods unless otherwise indicated by code:
 - 1. To Wood: Fasten with lag screws or through bolts.
 - 2. To New Concrete: Bolt to concrete inserts.
 - 3. To Masonry: Approved toggle-type bolts on hollow masonry units and expansion anchor fasteners on solid masonry units.
 - 4. To Existing Concrete: Expansion anchor fasteners.
 - 5. Instead of expansion anchors, powder-actuated driven threaded studs provided with lock washers and nuts may be used in existing standard-weight concrete 4 inches thick or greater. Do not use for anchorage to lightweightaggregate concrete or for slabs less than 4 inches thick.
 - 6. To Steel: Beam clamps (MSS SP-58, Type 19, 21, 23, 25, or 27), complying with MSS SP-69.
 - 7. To Light Steel: Sheet metal screws.
 - Items Mounted on Hollow Walls and Nonstructural Building Surfaces: Mount cabinets, panelboards, disconnect switches, control enclosures, pull and junction boxes, transformers, and other devices on slotted-channel racks attached to substrate.
- E. Drill holes for expansion anchors in concrete at locations and to depths that avoid the need for reinforcing bars.

3.3 INSTALLATION OF FABRICATED METAL SUPPORTS

- A. Cut, fit, and place miscellaneous metal supports accurately in location, alignment, and elevation to support and anchor electrical materials and equipment.
- B. Field Welding: Comply with AWS D1.1/D1.1M.

3.4 PAINTING

A. Galvanized Surfaces: Clean welds, bolted connections, and abraded areas and apply galvanizing-repair paint to comply with ASTM A 780.

3.5 TOUCH-UP COMPOUND FOR STEEL AND STAINLESS STEEL

A. Manufacturer-recommended touch-up compound shall be used on all steel and stainless steel exposed bare because of nicks, cuts, abrasions, cutting, and finishing. Apply minimum number of coats as recommended by manufacturer.

END OF SECTION 260529

Community Park Improvements 20235700

Hangers And Supports For Electrical Systems 260529 - 4

Iconica

Commented [KP3]: This section in previous spec version but not in this updated one.

SECTION 260533 - RACEWAY AND BOXES FOR ELECTRICAL SYSTEMS

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section Includes:
 - 1. Type EMT-SS raceways and elbows.
 - 2. Type ERMC-SS raceways, elbows, couplings, and nipples.
 - 3. Type ERMC-S raceways, elbows, couplings, and nipples.
 - 4. Type ERMC-S-PVC raceways, elbows, couplings, and nipples.
 - 5. Type PVC raceways and fittings.
 - 6. Fittings for conduit, tubing, and cable.
 - 7. Threaded metal joint compound.
 - 8. Solvent cements.
 - 9. Surface metal raceways and fittings.
 - 10. Wireways and auxiliary gutters.
 - 11. Metallic outlet boxes, device boxes, and covers.
 - 12. Nonmetallic outlet boxes, device boxes, and covers.
 - 13. Termination boxes.
 - 14. Cabinets, cutout boxes, junction boxes, and pull boxes.
 - 15. Cover plates for device boxes.
 - 16. Hoods for outlet boxes.
 - B. Related Requirements:
 - 1. Section 260010 "Supplemental Requirements for Electrical" for additional abbreviations, definitions, submittals, qualifications, testing agencies, and other Project requirements applicable to Work specified in this Section.

1.2 ACTION SUBMITTALS

A. Product Data: For each type of product to be used.

1.3 REGULATORY REQUIREMENTS

- A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
- B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.

PART 2 - PRODUCTS

- 2.1 TYPE ERMC-SS RACEWAYS, ELBOWS, COUPLINGS, AND NIPPLES
 - A. Stainless Steel Electrical Rigid Metal Conduit (ERMC-SS), Elbows, Couplings, and Nipples:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 6A and UL Category Control Number DYWV.
 - 2) Material: Stainless steel.
- 2.2 TYPE ERMC-S RACEWAYS, ELBOWS, COUPLINGS, AND NIPPLES
 - A. Galvanized-Steel Electrical Rigid Metal Conduit (ERMC-S-G), Elbows, Couplings, and Nipples:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 6 and UL Category Control Number DYIX.
 - 2) Exterior Coating: Zinc.
 - 3) Interior Coating: Zinc.
 - B. PVC-Coated-Steel Electrical Rigid Metal Conduit (ERMC-S-PVC), Elbows, Couplings, and Nipples:
 - 1. Manufacturers:
 - a. Atkore Calbond.
 - b. Plasti-bond RedH₂OT by Robroy Industries.
 - c. Ocal-Blue by ABB/Thomas & Betts.
 - 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 6, ANSI C80.1, and NEMA RN-1.
 - 2) PVC-Coated Steel Electrical Rigid Metal Conduit:
 - a) Shall be listed to UL 6 and manufactured in accordance with ANSI C80.1 and NEMA RN 1.
 - b) All installers shall be field-certified from the factory for installation and shall provide proof of certification upon request.

Raceway And Boxes For Electrical Systems 260533 - 2

Iconica

- c) Conduit shall be hot dip galvanized on the inside and outside covered by a protective zinc coating.
- d) Exterior PVC coating shall have a nominal thickness of 40 mil (0.040-inch) and interior Urethane coating shall have a nominal thickness of 2 mil (0.002-inch).
- e) Conduit shall be threaded on both ends.
- 3) Fittings for PVC-Coated Conduit:
 - a) Minimum coating thickness of 0.040 inch, with overlapping sleeves protecting threaded joints.
 - b) Conduit bodies must be Form 8 with an effective seal and a positive placement feature to ease and assure proper installation. Certified results confirming seal performance at 15 psig (positive) and 25 in. of mercury (vacuum) for 72 hours must be available. Conduit bodies must be supplied with plastic-encapsulated stainless steel cover screws.
 - c) Form 2 inch long or one pipe diameter long, whichever is less, PVC sleeve at openings of female fittings, except unions. Inside sleeve diameter must be matched to outside diameter of metal conduit.
 - d) PVC coating on the outside of conduit couplings must be protected from tool damage during installation.
 - e) Female threads on fittings and couplings must be protected by urethane coating.
 - f) Fittings must be from same manufacturer as conduit.
 - g) Beam clamps and U bolts must be formed and sized to fit outside diameter of coated conduit. Plastic-encapsulated nuts must cover the exposed portions of threads.
- 2.3 TYPE PVC RACEWAYS AND FITTINGS
 - A. Schedule 40 Rigid PVC Conduit (PVC-40) and Fittings:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 651 and UL Category Control Number DZYR.
 - 2) Dimensional Specifications: Schedule 40.
 - B. Schedule 80 Rigid PVC Conduit (PVC-80) and Fittings:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 651 and UL Category Control Number DZYR.

- 2) Dimensional Specifications: Schedule 80.
- C. Type A Rigid PVC Concrete-Encased Conduit (PVC-A) and Fittings:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 651 and UL Category Control Number DZYR.
 - 2) Dimensional Specifications: Type A.
- 2.4 FITTINGS FOR CONDUIT, TUBING, AND CABLE
 - A. Fittings for Type ERMC and Type PVC Raceways:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 514B and UL Category Control Number DWTT.
 - 2) Material: Steel.
 - 3) Coupling Method: Raintight compression coupling with distinctive color gland nut.
 - c. Options:
 - 1) Expansion and Deflection Fittings: UL 651 with flexible external bonding jumper.
- 2.5 ELECTRICALLY CONDUCTIVE CORROSION-RESISTANT COMPOUNDS FOR THREADED CONDUIT
 - A. Applicable Standards:
 - 1. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and approved by authorities having jurisdiction for application to threaded conduit assemblies.
 - 2. General Characteristics:
 - a. Reference Standards: UL 2419 and UL Category Control Number FOIZ.

2.6 SOLVENT CEMENTS

- A. Solvent Cements for Type PVC Raceways and Fittings:
 - 1. Applicable Standards:
 - a. General Characteristics:

- Reference Standards: As recommended by conduit manufacturer in accordance with UL 514B and UL Category Control Number DWTT.
- 2.7 SURFACE METAL RACEWAYS AND FITTINGS
 - A. Surface Metal Raceways and Fittings with Metal Covers:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 5 and UL Category Control Number RJBT.
 - c. Options:
 - 1) Galvanized steel base with snap-on covers.
 - 2) Manufacturer's standard enamel finish.
 - 3) Wiring Channels: Multiple channels must be capable of housing a standard 20 to 30 A NEMA device flush within the raceway.

2.8 WIREWAYS AND AUXILIARY GUTTERS

- A. Metal Wireways and Auxiliary Gutters:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 870 and UL Category Control Number ZOYX.
 - Fittings and Accessories: Include covers, couplings, offsets, elbows, expansion joints, adapters, hold-down straps, end caps, and other fittings to match and mate with wireways as required for complete system.
 - 3) Finish: Manufacturer's standard enamel finish.
 - c. Options:
 - 1) Degree of Protection: Type 3R unless otherwise indicated.
 - 2) Wireway Covers: Flanged-and-gasketed type unless otherwise indicated.

2.9 METALLIC OUTLET BOXES, DEVICE BOXES, AND COVERS

A. Metallic Outlet Boxes:

- 1. Description: Box having pryout openings, knockouts, threaded entries, or hubs in either the sides of the back, or both, for entrance of conduit, conduit or cable fittings, or cables, with provisions for mounting outlet box cover, but without provisions for mounting wiring device directly to box.
- 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 514A and UL Category Control Number QCIT.
 - c. Options:
 - 1) Material: Sheet steel.
 - 2) Sheet Metal Depth: Minimum 1.5 inch.
 - 3) Luminaire Outlet Boxes and Covers: Nonadjustable, listed and labeled for attachment of luminaire weighing up to 50 lb.
- B. Metallic Conduit Bodies:
 - 1. Description: Means for providing access to interior of conduit or tubing system through one or more removable covers at junction or terminal point. In the United States, conduit bodies are listed in accordance with outlet box requirements.
 - 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 514A and UL Category Control Number QCIT.
- C. Metallic Device Boxes:
 - 1. Description: Box with provisions for mounting wiring device directly to box.
 - 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 514A and UL Category Control Number QCIT.
 - c. Options:
 - 1) Material: Sheet steel.
 - 2) Sheet Metal Depth: minimum 1.5 inch.

- 3) Luminaire Outlet Boxes and Covers: Nonadjustable, listed and labeled for attachment of luminaire weighing up to 50 lb.
- D. Metallic Concrete Boxes and Covers:
 - 1. Description: Box intended for use in poured concrete.
 - 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 514A and UL Category Control Number QCIT.
- 2.10 TERMINATION BOXES
 - A. Description: Enclosure for termination base consisting of lengths of bus bars, terminal strips, or terminal blocks with provision for wire connectors to accommodate incoming or outgoing conductors or both.
 - B. Termination Boxes and Termination Bases for Installation on Line Side of Service Equipment:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 1773 and UL Category Control Number XCKT.
 - 2) Listed and labeled for installation on line side of service equipment.
 - C. Termination Boxes and Termination Bases for Installation on Load Side of Service Equipment:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 1773 and UL Category Control Number XCKT.
 - 2) Listed and labeled for installation on load side of service equipment.

2.11 CABINETS, CUTOUT BOXES, JUNCTION BOXES, AND PULL BOXES

- A. Indoor Sheet Metal Cabinets:
 - 1. Description: Enclosure provided with frame, mat, or trim in which swinging door or doors are or can be hung.

- 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL Category Control Number CYIV.
 - a) Non-Environmental Characteristics: UL 50.
 - b) Environmental Characteristics: UL 50E.
 - c. Options:
 - 1) Degree of Protection: Type 12.
- B. Indoor Sheet Metal Cutout Boxes:
 - 1. Description: Enclosure that has swinging doors or covers secured directly to and telescoping with walls of enclosure.
 - 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL Category Control Number CYIV.
 - a) Non-Environmental Characteristics: UL 50.
 - b) Environmental Characteristics: UL 50E.
 - c. Options:
 - 1) Degree of Protection: Type 12.
- C. Indoor Sheet Metal Junction and Pull Boxes:
 - 1. Description: Box with a blank cover that serves the purpose of joining different runs of raceway or cable.
 - 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL Category Control Number BGUZ.
 - a) Non-Environmental Characteristics: UL 50.
 - b) Environmental Characteristics: UL 50E.
 - c. Options:
 - 1) Degree of Protection: Type 12.

Community Park Improvements 20235700

- D. Indoor Polymeric Junction and Pull Boxes:
 - 1. Description: Box with a blank cover that serves the purpose of joining different runs of raceway or cable.
 - 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL Category Control Number BGUZ.
 - a) Non-Environmental Characteristics: UL 50.
 - b) Environmental Characteristics: UL 50E.
 - c. Options:
 - 1) Degree of Protection: Type 12.
- E. Outdoor Sheet Metal Junction and Pull Boxes:
 - 1. Description: Box with a blank cover that serves the purpose of joining different runs of raceway or cable.
 - 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL Category Control Number BGUZ.
 - a) Non-Environmental Characteristics: UL 50.
 - b) Environmental Characteristics: UL 50E.
 - c. Options:
 - 1) Degree of Protection: Type 3R.
- F. Outdoor Polymeric Junction and Pull Boxes:
 - 1. Description: Box with a blank cover that serves the purpose of joining different runs of raceway or cable.
 - 2. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL Category Control Number BGUZ.
 - a) Non-Environmental Characteristics: UL 50.

Raceway And Boxes For Electrical Systems 260533 - 9

Iconica

- b) Environmental Characteristics: UL 50E.
- c. Options:
 - 1) Degree of Protection: Type 3R.
- G. Polymer-Concrete Handholes and Boxes with Polymer-Concrete Cover: Molded of sand and aggregate, bound together with polymer resin, and reinforced with steel, fiberglass, or a combination of the two.
 - 1. Standard: Comply with SCTE 77.
 - Configuration: Designed for flush burial with open bottom unless otherwise indicated.
 - 3. Cover: Weatherproof, secured by tamper-resistant locking devices and having structural load rating consistent with enclosure and handhole location.
 - 4. Cover Finish: Nonskid finish shall have a minimum coefficient of friction of 0.50.
 - 5. Cover Legend: Molded lettering, "ELECTRIC.".
 - 6. Conduit Entrance Provisions: Conduit-terminating fittings shall mate with entering ducts for secure, fixed installation in enclosure wall.

2.12 COVER PLATES FOR DEVICES BOXES

- A. Metallic Cover Plates for Device Boxes:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:
 - 1) Reference Standards: UL 514D and UL Category Control Numbers QCIT and QCMZ.
 - 2) Wallplate-Securing Screws: Metal with head color to match wallplate finish.
 - c. Options:
 - 1) Damp and Wet Locations: Listed, labeled, and marked for location and use. Provide gaskets and accessories necessary for compliance with listing.
 - 2) Wallplate Material: Galvanized steel.

2.13 HOODS FOR OUTLET BOXES

- A. Extra-Duty, While-in-Use Hoods for Outlet Boxes:
 - 1. Applicable Standards:
 - a. Regulatory Requirements: Listed and labeled in accordance with NFPA 70 and marked for intended location and use.
 - b. General Characteristics:

- 1) Reference Standards: UL 514D and UL Category Control Numbers QCIT and QCMZ.
- 2) Marked "Extra-Duty" in accordance with UL 514D.
- 3) Receptacle, hood, cover plate, gaskets, and seals comply with UL 498 Supplement SA when mated with box or enclosure complying with UL 514A, UL 514C, or UL 50E.
- 4) Mounts to box using fasteners different from wiring device.
- c. Options:
 - 1) Provide clear weatherproof, "while-in-use" cover.
 - 2) Manufacturer may combine nonmetallic device box with hood as extra-duty rated assembly.

PART 3 - EXECUTION

- 3.1 SELECTION OF RACEWAYS
 - A. Unless more stringent requirements are specified in Contract Documents or manufacturers' written instructions, comply with NFPA 70 for selection of raceways. Consult Design-Builder for resolution of conflicting requirements.
 - B. Outdoors:
 - 1. Exposed Conduit: ERMC.
 - 2. Concealed Conduit, Aboveground: ERMC.
 - 3. Direct-Buried Conduit: PVC-40. PVC conduit under pavement or roadways shall be PVC-80
 - C. Indoors:
 - 1. Exposed and Subject to Physical Damage: ERMC. Raceway locations include the following:
 - a. Electrical Building.
 - 2. Damp or Wet Locations: ERMC.
 - D. Stub-ups to Above Recessed Ceilings: Provide EMT, IMC, or ERMC for raceways.
 - E. Raceway Fittings: Select fittings in accordance with NEMA FB 2.10 guidelines.
 - 1. ERMC and IMC: Provide threaded type fittings unless otherwise indicated.
- 3.2 SELECTION OF BOXES AND ENCLOSURES
 - A. Unless more stringent requirements are specified in Contract Documents or manufacturers' written instructions, comply with NFPA 70 for selection of boxes and enclosures. Consult Design-Builder for resolution of conflicting requirements.
 - B. Degree of Protection:
 - 1. Outdoors:
 - a. **Type 3R** unless otherwise indicated.

Community Park Improvements 20235700

- 2. Indoors:
 - a. Type 12 unless otherwise indicated.

3.3 INSTALLATION OF RACEWAYS

- A. Installation Standards:
 - 1. Unless more stringent requirements are specified in Contract Documents or manufacturers' written instructions, comply with NFPA 70 for installation of raceways. Consult Design-Builder for resolution of conflicting requirements.
 - 2. Comply with NFPA 70 limitations for types of raceways allowed in specific occupancies and number of floors.
 - 3. Comply with requirements in Section 260529 "Hangers and Supports for Electrical Systems" for hangers and supports.
 - 4. Comply with NECA NEIS 101 for installation of steel raceways.
 - 5. Comply with NECA NEIS 102 for installation of aluminum raceways.
 - 6. Comply with NECA NEIS 111 for installation of nonmetallic raceways.
 - 7. Install raceways square to the enclosure and terminate at enclosures without hubs with locknuts on both sides of enclosure wall. Install locknuts hand tight, plus one-quarter turn more.
 - 8. Terminate threaded conduits into threaded hubs or with locknuts on inside and outside of boxes or cabinets. Install bushings on conduits up to 1-1/4 inch trade size and insulated throat metal bushings on 1-1/2 inch trade size and larger conduits terminated with locknuts. Install insulated throat metal grounding bushings on service conduits.
 - 9. Raceway Terminations at Locations Subject to Moisture or Vibration:
 - a. Provide insulating bushings to protect conductors, including conductors smaller than No. 4 AWG. Install insulated throat metal grounding bushings on service conduits.
- B. General Requirements for Installation of Raceways:
 - 1. Complete raceway installation before starting conductor installation.
 - 2. Provide stub-ups through floors with coupling threaded inside for plugs, set flush with finished floor. Plug coupling until conduit is extended above floor to final destination or a minimum of 2 ft. above finished floor.
 - 3. Install no more than equivalent of three 90-degree bends in conduit run. Support within 12 inch of changes in direction.
 - 4. Make bends in raceway using large-radius preformed ells except for parallel bends. Field bending must be in accordance with NFPA 70 minimum radii requirements. Provide only equipment specifically designed for material and size involved.
 - 5. Conceal conduit within finished walls, ceilings, and floors unless otherwise indicated. Install conduits parallel or perpendicular to building lines.
 - 6. Support conduit within 12 inch of enclosures to which attached.

- 7. Install raceway sealing fittings at accessible locations in accordance with NFPA 70 and fill them with listed sealing compound. For concealed raceways, install fitting in flush steel box with blank cover plate having finish similar to that of adjacent plates or surfaces. Install raceway sealing fittings in accordance with NFPA 70.
- 8. Install devices to seal raceway interiors at accessible locations. Locate seals so no fittings or boxes are between the seal and the following changes of environments. Seal interior of raceways at the following points:
 - a. Where an underground service raceway enters a building or structure.
 - b. Conduit extending from interior to exterior of building.
 - c. Where otherwise required by NFPA 70.
- Do not install raceways or electrical items on "explosion-relief" walls or rotating equipment.
- 10. Do not install conduits within 2 inch of the bottom side of a metal deck roof.
- 11. Keep raceways at least 6 inch away from parallel runs of flues and steam or hot-water pipes. Install horizontal raceway runs above water and steam piping.
- 12. Cut conduit perpendicular to the length. For conduits 2 inch trade size and larger, use roll cutter or a guide to make cut straight and perpendicular to the length. Ream inside of conduit to remove burrs.
- 13. Install pull wires in empty raceways. Provide polypropylene or monofilament plastic line with not less than 200 lb tensile strength. Leave at least 12 inch of slack at both ends of pull wire. Cap underground raceways designated as spare above grade alongside raceways in use.
- C. Requirements for Installation of Specific Raceway Types:
 - 1. Type ERMC-A:
 - a. Do not install aluminum raceways or fittings in contact with concrete or earth.
 - 2. Types ERMC:
 - a. Threaded Conduit Joints, Exposed to Wet, Damp, Corrosive, or Outdoor Conditions: Apply listed compound that maintains electrical conductivity to threads of raceway and fittings before making up joints. Follow compound manufacturer's written instructions.
 - 3. Type ERMC-S-PVC:
 - a. Follow manufacturer's installation instructions for clamping, cutting, threading, bending, and assembly.
 - b. Provide PVC-coated sealing locknut for exposed male threads transitioning into female NPT threads that do not have sealing sleeves, including transitions from PVC couplings/female adapters to Type ERMC-S-PVC elbows in direct-burial applications. PVC-coated sealing locknuts must not be used in place of conduit hub. PVC-coated sealing locknut must cover exposed threads on Type ERMC-S-PVC raceway.

- c. Coat field-cut threads on PVC-coated raceway with manufacturerapproved corrosion-preventing conductive compound prior to assembly.
- d. The PVC-coated rigid conduit manufacturer's touch-up compound shall be used on all conduit interior and exterior bare steel exposed because of nicks, cuts, abrasions, thread cutting, and reaming.
- 4. Type PVC:
 - a. Do not install Type PVC conduit where ambient temperature exceeds 122 deg F. Conductor ratings must be limited to 75 deg C except where installed in a trench outside buildings with concrete encasement, where 90 deg C conductors are permitted.
 - b. Comply with manufacturer's written instructions for solvent welding and fittings.
- D. Raceway Fittings: Install fittings in accordance with NEMA FB 2.10 guidelines.
 - 1. ERMC-S-PVC: Provide only fittings listed for use with this type of conduit. Patch and seal joints, nicks, and scrapes in PVC coating after installing conduits and fittings. Provide sealant recommended by fitting manufacturer and apply in thickness and number of coats recommended by manufacturer.
 - 2. EMT: Provide setscrew, steel fittings. Comply with NEMA FB 2.10.
 - 3. Flexible Conduit: Provide only fittings listed for use with flexible conduit type. Comply with NEMA FB 2.20.
- E. Expansion-Joint Fittings:
 - 1. Install in runs of aboveground PVC that are located where environmental temperature change may exceed 30 deg F and that have straight-run length that exceeds 25 ft. Install in runs of aboveground ERMC and EMT conduit that are located where environmental temperature change may exceed 100 deg F and that have straight-run length that exceeds 100 ft..
 - 2. Install type and quantity of fittings that accommodate temperature change listed for the following locations:
 - a. Outdoor Locations Not Exposed to Direct Sunlight: 125 deg F temperature change.
 - b. Outdoor Locations Exposed to Direct Sunlight: 155 deg F temperature change.
 - c. Indoor Spaces Connected with Outdoors without Physical Separation: 125 deg F temperature change.
 - 3. Install fitting(s) that provide expansion and contraction for at least 0.00041 inch per foot of length of straight run per deg F of temperature change for PVC conduits. Install fitting(s) that provide expansion and contraction for at least 0.000078 inch per foot of length of straight run per deg F of temperature change for metal conduits.
 - 4. Install expansion fittings at locations where conduits cross building or structure expansion joints.

- 5. Install expansion-joint fitting with position, mounting, and piston setting selected in accordance with manufacturer's written instructions for conditions at specific location at time of installation. Install conduit supports to allow for expansion movement.
- F. Raceways Penetrating Rooms or Walls with Acoustical Requirements:
 - 1. Seal raceway openings on both sides of rooms or walls with acoustically rated putty or firestopping.
- 3.4 INSTALLATION OF SURFACE RACEWAYS
 - A. Install surface raceways only where indicated on Drawings.
 - B. Install surface raceway with a minimum 2 inch radius control at bend points.
 - C. Secure surface raceway with screws or other anchor-type devices at intervals not exceeding 48 inch and with no less than two supports per straight raceway section. Support surface raceway in accordance with manufacturer's written instructions. Tape and glue are unacceptable support methods.

3.5 INSTALLATION OF BOXES AND ENCLOSURES

- A. Provide boxes in wiring and raceway systems wherever required for pulling of wires, making connections, and mounting of devices or fixtures.
- B. Refer to Section 260553 Identification for Electrical Systems for box labeling requirements.
- C. Horizontally separate boxes mounted on opposite sides of walls so they are not in the same vertical channel.
- D. Locate boxes so that cover or plate will not span different building finishes.
- E. Support boxes in recessed ceilings independent of ceiling tiles and ceiling grid.
- F. Support boxes of three gangs or more from more than one side by spanning two framing members or mounting on brackets specifically designed for purpose.
- G. Fasten junction and pull boxes to, or support from, building structure. Do not support boxes by conduits.
- H. Set metal floor boxes level and flush with finished floor surface.
- I. Set nonmetallic floor boxes level. Trim after installation to fit flush with finished floor surface.
- J. Do not install aluminum boxes, enclosures, or fittings in contact with concrete or earth.
- K. Do not rely on locknuts to penetrate nonconductive coatings on enclosures. Remove coatings in the locknut area prior to assembling conduit to enclosure to ensure a continuous ground path.
- L. Boxes and Enclosures in Areas or Walls with Acoustical Requirements:

- 1. Seal openings and knockouts in back and sides of boxes and enclosures with acoustically rated putty.
- 2. Provide gaskets for wallplates and covers.

3.6 FIRESTOPPING

A. Install firestopping at penetrations of fire-rated floor and wall assemblies. Comply with requirements in Section 078413 "Penetration Firestopping."

3.7 PROTECTION

- A. Protect coatings, finishes, and cabinets from damage and deterioration.
 - 1. Repair damage to galvanized finishes with zinc-rich paint recommended by manufacturer.
 - 2. Repair damage to PVC coatings or paint finishes with matching touchup coating recommended by manufacturer.

3.8 CLEANING

A. Boxes: Remove construction dust and debris from device boxes, outlet boxes, and floor-mounted enclosures before installing wallplates, covers, and hoods.

END OF SECTION 260533

SECTION 260544 - SLEEVES AND SLEEVE SEALS FOR ELECTRICAL RACEWAYS AND CABLING

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section Includes:
 - 1. Round sleeves.
 - 2. Rectangular sleeves.
 - 3. Sleeve seal systems.
 - 4. Sleeve seal fittings.
 - 5. Grout.
 - 6. Silicone Sealants
 - B. Related Requirements:
 - 1. Section 260010 "Supplemental Requirements for Electrical" for additional abbreviations, definitions, submittals, qualifications, testing agencies, and other Project requirements applicable to Work specified in this Section.
 - 2. Section 078413 "Penetration Firestopping" for penetration firestopping installed in fire-resistance-rated walls, horizontal assemblies, and smoke barriers, with and without penetrating items.
- 1.2 ACTION SUBMITTALS
 - A. Product Data: For each type of product.

PART 2 - PRODUCTS

- 2.1 ROUND SLEEVES
 - A. Wall Sleeves, Steel:
 - 1. Description: ASTM A53/A53M, Type E, Grade B, Schedule 40, zinc coated, plain ends and integral waterstop.
 - B. Pipe Sleeves, PVC:
 - 1. Description: ASTM D1785, Schedule 40.
 - C. Sheet Metal Sleeves, Galvanized Steel, Round:
 - 1. Description: Galvanized-steel sheet; thickness not less than 0.0239 inch; round tube closed with welded longitudinal joint, with tabs for screw-fastening the sleeve to the board.
- 2.2 RECTANGULAR SLEEVES
 - A. Sheet Metal Sleeves, Galvanized Steel, Rectangular:
 - 1. Description:

Community Park Improvements Sleeves And Sleeve Seals For Electrical Raceways And Cabling 20235700 260544 - 1

- a. Material: Galvanized sheet steel.
- b. Minimum Metal Thickness:
 - 1) For sleeve cross-section rectangle perimeter less than 50 inch and with no side larger than 16 inch, thickness must be 0.052 inch.
 - 2) For sleeve cross-section rectangle perimeter not less than 50 inch or with one or more sides larger than 16 inch, thickness must be 0.138 inch.

2.3 SLEEVE SEAL SYSTEMS

- A. Description: Modular sealing device, designed for field assembly, to fill annular space between sleeve and raceway or cable or between raceway and cable.
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Advance Products & Systems, Inc.
 - b. Metraflex Company (The).
 - c. Proco Products, Inc.
 - 2. Sealing Elements: EPDM rubber interlocking links shaped to fit surface of pipe. Include type and number required for pipe material and size of pipe.
 - 3. Pressure Plates: Stainless steel.
 - 4. Connecting Bolts and Nuts: Stainless steel of length required to secure pressure plates to sealing elements.

2.4 SLEEVE-SEAL FITTINGS

- A. Description: Manufactured plastic, sleeve-type, waterstop assembly made for embedding in concrete slab or wall. Unit shall have plastic or rubber waterstop collar with center opening to match piping OD.
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. HOLDRITE.

2.5 GROUT

- A. Description: Nonshrink; recommended for interior and exterior sealing openings in non-fire-rated walls or floors.
 - 1. Standard: ASTM C1107/C1107M, Grade B, post-hardening and volumeadjusting, dry, hydraulic-cement grout.
 - 2. Design Mix: 5000 psi, 28-day compressive strength.
 - 3. Packaging: Premixed and factory packaged.

2.6 SILICONE SEALANTS

A. Silicone Sealants: Single-component, silicone-based, neutral-curing elastomeric sealants of grade indicated below.

- 1. Grade: Pourable (self-leveling) formulation for openings in floors and other horizontal surfaces that are not fire rated.
- B. Silicone Foams: Multicomponent, silicone-based liquid elastomers that, when mixed, expand and cure in place to produce a flexible, nonshrinking foam.

PART 3 - EXECUTION

- 3.1 INSTALLATION OF SLEEVES FOR NON-FIRE-RATED ELECTRICAL PENETRATIONS
 - A. Comply with NECA 1 and NEMA VE 2 (for cable tray and cable penetrations).
- 3.2 Interior Penetrations of Non-Fire-Rated Walls and Floors:
 - 1. Seal space outside of sleeves with mortar or grout. Pack sealing material solidly between sleeve and wall or floor so no voids remain. Tool exposed surfaces smooth; protect material while curing.
 - 2. Seal annular space between sleeve and raceway or cable, using joint sealant appropriate for size, depth, and location of joint. Comply with requirements in Section 079200 "Joint Sealants."
 - 3. Use pipe sleeves unless penetration arrangement requires rectangular sleeved opening.
 - 4. Size pipe sleeves to provide 1/4 inch annular clear space between sleeve and raceway or cable, unless sleeve seal system is to be installed.
 - 5. Install sleeves for wall penetrations unless core-drilled holes or formed openings are used. Install sleeves during erection of walls. Cut sleeves to length for mounting flush with both surfaces of walls. Deburr after cutting.
 - 6. Install sleeves for floor penetrations. Extend sleeves installed in floors 2 inches above finished floor level. Install sleeves during erection of floors.
 - B. Sleeves for Conduits Penetrating Non-Fire-Rated Wall Assemblies:
 - 1. Use circular metal sleeves unless penetration arrangement requires rectangular sleeved opening.
 - 2. Seal space outside of sleeves with approved joint compound for wall assemblies.
 - C. Roof-Penetration Sleeves: Seal penetration of individual raceways and cables with flexible boot-type flashing units applied in coordination with roofing work.
 - D. Aboveground, Exterior-Wall Penetrations: Seal penetrations using steel pipe sleeves and mechanical sleeve seal systems. Size sleeves to allow for 1 inch annular clear space between pipe and sleeve for installing mechanical sleeve seals.
 - E. Underground, Exterior-Wall and Floor Penetrations:
 - 1. Install steel pipe sleeves with integral waterstops. Size sleeves to allow for 1 inch annular clear space between raceway or cable and sleeve for installing sleeve seal system. Install sleeve during construction of floor or wall.

2. Install steel pipe sleeves. Size sleeves to allow for 1 inch annular clear space between raceway or cable and sleeve for installing sleeve seal system. Grout sleeve into wall or floor opening.

3.3 INSTALLATION OF RECTANGULAR SLEEVES AND SLEEVE SEALS

- A. Install sleeves in existing walls without compromising structural integrity of walls. Do not cut structural elements without reinforcing the wall to maintain the designed weight bearing and wall stiffness.
- B. Install conduits and cable with no crossings within the sleeve.
- C. Fill opening around conduits and cables with expanding foam without leaving voids.
- D. Provide metal sheet covering at both wall surfaces and finish to match surrounding surfaces. Metal sheet must be same material as sleeve.

3.4 INSTALLATION OF SLEEVE SEAL SYSTEMS

- A. Install sleeve seal systems in sleeves in exterior concrete walls and slabs-on-grade at raceway entries into building.
- B. Install type and number of sealing elements recommended by manufacturer for raceway or cable material and size. Position raceway or cable in center of sleeve. Assemble mechanical sleeve seals and install in annular space between raceway or cable and sleeve. Tighten bolts against pressure plates that cause sealing elements to expand and make watertight seal.
- 3.5 SLEEVE-SEAL-FITTING INSTALLATION
 - A. Install sleeve-seal fittings in new walls and slabs as they are constructed.
 - B. Assemble fitting components of length to be flush with both surfaces of concrete slabs and walls. Position waterstop flange to be centered in concrete slab or wall.
 - C. Secure nailing flanges to concrete forms.
 - D. Using grout, seal the space around outside of sleeve-seal fittings.

END OF SECTION 260544

SECTION 260553 - IDENTIFICATION FOR ELECTRICAL SYSTEMS

PART 1 - GENERAL

- 1.1 RELATED DOCUMENTS
 - A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

1.2 SUMMARY

- A. Section Includes:
 - 1. Color and legend requirements for raceways, conductors, and warning labels and signs.
 - 2. Labels.
 - 3. Bands and tubes.
 - 4. Tapes and stencils.
 - 5. Tags.
 - 6. Signs.
 - 7. Cable ties.
 - 8. Paint for identification.
 - 9. Fasteners for labels and signs.
- 1.3 ACTION SUBMITTALS
 - A. Product Data: For each type of product.

PART 2 - PRODUCTS

- 2.1 PERFORMANCE REQUIREMENTS
 - A. Comply with ASME A13.1.
 - B. Comply with NFPA 70.
 - C. Comply with 29 CFR 1910.144 and 29 CFR 1910.145.
 - D. Comply with ANSI Z535.4 for safety signs and labels.
 - E. Comply with NFPA 70E and Section 260573.19 "Arc-Flash Hazard Analysis" requirements for arc-flash warning labels.
 - F. Adhesive-attached labeling materials, including label stocks, laminating adhesives, and inks used by label printers, shall comply with UL 969.
 - G. Thermal Movements: Allow for thermal movements from ambient and surface temperature changes.

- 1. Temperature Change: 120 deg F (67 deg C), ambient; 180 deg F (100 deg C), material surfaces.
- 2.2 COLOR AND LEGEND REQUIREMENTS
 - A. Raceways and Cables Carrying Circuits at 600 V or Less:
 - 1. Black letters on an orange field.
 - 2. Legend: Indicate voltage and system or service type.
 - B. Color-Coding for Phase Identification, 600 V or Less: Use colors listed below for ungrounded feeder and branch-circuit conductors.
 - 1. Color shall be factory applied.
 - 2. Colors for 240-V Circuits:
 - a. Phase A: Black.
 - b. Phase B: Red.
 - c. Ground: Green.
 - d. Neutral: White.
 - 3. Color for Equipment Grounds: Bare copper, Green or Green with a yellow stripe.
 - C. Warning Label Colors:
 - 1. Identify system voltage with black letters on an orange background.
 - D. Warning labels and signs shall include, but are not limited to, the following legends:
 - 1. Multiple Power Source Warning: "DANGER ELECTRICAL SHOCK HAZARD EQUIPMENT HAS MULTIPLE POWER SOURCES."
 - Workspace Clearance Warning: "WARNING OSHA REGULATION AREA IN FRONT OF ELECTRICAL EQUIPMENT MUST BE KEPT CLEAR FOR 36 INCHES (915 MM)."
 - E. Equipment Identification Labels:
 - 1. Black letters on a white field.
- 2.3 LABELS
 - A. Vinyl Wraparound Labels: Preprinted, flexible labels laminated with a clear, weather- and chemical-resistant coating and matching wraparound clear adhesive tape for securing label ends.
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Brady Corporation.
 - b. Panduit Corp.

- c. Seton Identification Products.
- B. Self-Adhesive Wraparound Labels: Preprinted, 3-mil- thick, vinyl flexible label with acrylic pressure-sensitive adhesive.
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Brady Corporation.
 - b. Panduit Corp.
 - c. Seton Identification Products.
 - 2. Marker for Labels: Permanent, waterproof, black ink marker recommended by tag manufacturer.
 - 3. Marker for Labels: Machine-printed, permanent, waterproof, black ink recommended by printer manufacturer.
- C. Self-Adhesive Labels: Vinyl, thermal, transfer-printed, 3-mil- thick, multicolor, weather- and UV-resistant, pressure-sensitive adhesive labels, configured for intended use and location.
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Brady Corporation.
 - b. Panduit Corp.
 - c. Seton Identification Products.
 - 2. Minimum Nominal Size:
 - a. 1-1/2 by 6 inches for raceway and conductors.
 - b. 3-1/2 by 5 inches for equipment.
 - c. As required by authorities having jurisdiction.

2.4 BANDS AND TUBES

1.

- B. Heat-Shrink Preprinted Tubes: Flame-retardant polyolefin tubes with machineprinted identification labels, sized to suit diameters of and shrunk to fit firmly around item being identified. Full shrink recovery occurs at a maximum of 200 deg F. Comply with UL 224.
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Brady Corporation.
 - b. Panduit Corp.
 - c. Seton Identification Products.

2.5 TAPES AND STENCILS

- A. Marker Tapes: Vinyl or vinyl-cloth, self-adhesive wraparound type, with circuit identification legend machine printed by thermal transfer or equivalent process.
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Hellermann Tyton.
 - b. Ideal Industries, Inc.
 - c. Panduit Corp.
- B. Self-Adhesive Vinyl Tape: Colored, heavy duty, waterproof, fade resistant; not less than 3 mils thick by 1 to 2 inches wide; compounded for outdoor use.
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Brady Corporation.
 - b. Carlton Industries, LP.
 - c. Marking Services, Inc.
- C. Floor Marking Tape: 2-inch- wide, 5-mil pressure-sensitive vinyl tape, with black and white stripes and clear vinyl overlay.
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Carlton Industries, LP.
 - b. Seton Identification Products.
- D. Underground-Line Warning Tape:
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Brady Corporation.
 - b. Ideal Industries, Inc.
 - c. Marking Services, Inc.
 - d. Seton Identification Products.
 - 2. Tape:
 - a. Recommended by manufacturer for the method of installation and suitable to identify and locate underground electrical and communications utility lines.
 - b. Printing on tape shall be permanent and shall not be damaged by burial operations.

- c. Tape material and ink shall be chemically inert and not subject to degradation when exposed to acids, alkalis, and other destructive substances commonly found in soils.
- 3. Color and Printing:
 - a. Comply with ANSI Z535.1, ANSI Z535.2, ANSI Z535.3, ANSI Z535.4, and ANSI Z535.5.
 - b. Inscriptions for Red-Colored Tapes: "ELECTRIC LINE, HIGH VOLTAGE".
 - c. Inscriptions for Orange-Colored Tapes: "TELEPHONE CABLE, CATV CABLE, COMMUNICATIONS CABLE, OPTICAL FIBER CABLE".
- 4. Tag: Type I:
 - a. Pigmented polyolefin, bright colored, continuous-printed on one side with the inscription of the utility, compounded for direct-burial service.
 - b. Width: 3 inches.
 - c. Thickness: 4 mils.
 - d. Weight: 18.5 lb/1000 sq. ft.
 - e. Tensile according to ASTM D 882: 30 lbf and 2500 psi.
- E. Stenciled Legend: In nonfading, waterproof, black ink or paint. Minimum letter height shall be 1 inch (25 mm)
- 2.6 TAGS
 - A. Write-on Tags:
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Carlton Industries, LP.
 - b. Seton Identification Products.
 - 2. Polyester Tags: 0.010 inch (0.25 mm) thck, with corrosion-resistant grommet and cable tie for attachment.
 - 3. Marker for Tags: Permanent, waterproof, black ink marker recommended by tag manufacturer.

2.7 SIGNS

- A. Laminated Acrylic or Melamine Plastic Signs:
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Brady Corporation.
 - b. Carlton Industries, LP.
 - c. Marking Services, Inc.
 - 2. Engraved legend.

Community Park Improvements 20235700

- 3. Thickness:
 - a. For signs up to 20 sq. in., minimum 1/16 inch thick.
 - b. For signs larger than 20 sq. in., 1/8 inch thick.
 - c. Engraved legend with white letters on a dark gray background.
 - d. Self-adhesive.
 - e. Framed with mitered acrylic molding and arranged for attachment at applicable equipment.

2.8 CABLE TIES

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - 1. HellermannTyton.
 - 2. Ideal Industries, Inc.
 - 3. Marking Services, Inc.
 - 4. Panduit Corp.
 - B. General-Purpose Cable Ties: Fungus inert, self-extinguishing, one piece, self-locking, and Type 6/6 nylon.
 - 1. Minimum Width: 3/16 inch.
 - 2. Tensile Strength at 73 Deg F according to ASTM D 638: 12,000 psi.
 - 3. Temperature Range: Minus 40 to plus 185 deg F.
 - 4. Color: Black, except where used for color-coding.
 - C. UV-Stabilized Cable Ties: Fungus inert, designed for continuous exposure to exterior sunlight, self-extinguishing, one piece, self-locking, and Type 6/6 nylon.
 - 1. Minimum Width: 3/16 inch.
 - 2. Tensile Strength at 73 Deg F according to ASTM D 638: 12,000 psi.
 - 3. Temperature Range: Minus 40 to plus 185 deg F.
 - 4. Color: Black.
 - D. Plenum-Rated Cable Ties: Self-extinguishing, UV stabilized, one piece, and self-locking.
 - 1. Minimum Width: 3/16 inch.
 - 2. Tensile Strength at 73 Deg F according to ASTM D 638: 7000 psi.
 - 3. UL 94 Flame Rating: 94V-0.
 - 4. Temperature Range: Minus 50 to plus 284 deg F.
 - 5. Color: Black.

2.9 MISCELLANEOUS IDENTIFICATION PRODUCTS

A. Paint: Comply with requirements in painting Sections for paint materials and application requirements. Retain paint system applicable for surface material and location (exterior or interior).

B. Fasteners for Labels and Signs: Self-tapping, stainless-steel screws or stainlesssteel machine screws with nuts and flat and lock washers.

PART 3 - EXECUTION

- 3.1 INSTALLATION
 - A. Verify and coordinate identification names, abbreviations, colors, and other features with requirements in other Sections requiring identification applications, Drawings, Shop Drawings, manufacturer's wiring diagrams, and operation and maintenance manual. Use consistent designations throughout Project.
 - B. Install identifying devices before installing acoustical ceilings and similar concealment.
 - C. Verify identity of each item before installing identification products.
 - D. Coordinate identification with Project Drawings, manufacturer's wiring diagrams, and operation and maintenance manual.
 - E. Apply identification devices to surfaces that require finish after completing finish work.
 - F. Install signs with approved legend to facilitate proper identification, operation, and maintenance of electrical systems and connected items.
 - G. Self-Adhesive Identification Products: Before applying electrical identification products, clean substrates of substances that could impair bond, using materials and methods recommended by manufacturer of identification product.
 - H. System Identification for Raceways and Cables under 600 V: Identification shall completely encircle cable or conduit. Place identification of two-color markings in contact, side by side.
 - 1. Secure tight to surface of conductor, cable, or raceway.
 - I. Auxiliary Electrical Systems Conductor Identification: Identify field-installed alarm, control, and signal connections.
 - J. Elevated Components: Increase sizes of labels, signs, and letters to those appropriate for viewing from the floor.
 - K. Accessible Fittings for Raceways: Identify the covers of each junction and pull box of the following systems with the wiring system legend and system voltage.
 - L. Vinyl Wraparound Labels:
 - 1. Secure tight to surface at a location with high visibility and accessibility.
 - 2. Attach labels that are not self-adhesive type with clear vinyl tape, with adhesive appropriate to the location and substrate.
 - M. Self-Adhesive Wraparound Labels: Secure tight to surface of raceway or cable at a location with high visibility and accessibility.

- N. Self-Adhesive Labels:
 - 1. On each unit of equipment, install unique designation label that is consistent with wiring diagrams, schedules, and operation and maintenance manual.
 - 2. Unless otherwise indicated, provide a single line of text with 1/2-inch- high letters on 1-1/2-inch- high label; where two lines of text are required, use labels 2 inches high.
- O. Heat-Shrink, Preprinted Tubes: Secure tight to surface at a location with high visibility and accessibility.
- P. Marker Tapes: Secure tight to surface at a location with high visibility and accessibility.
- Q. Self-Adhesive Vinyl Tape: Secure tight to surface at a location with high visibility and accessibility.
 - 1. Field-Applied, Color-Coding Conductor Tape: Apply in half-lapped turns for a minimum distance of 6 inches where splices or taps are made. Apply last two turns of tape with no tension to prevent possible unwinding.
- R. Tape and Stencil: Comply with requirements in painting Sections for surface preparation and paint application.
- S. Floor Marking Tape: Apply stripes to finished surfaces following manufacturer's written instructions.
- T. Underground Line Warning Tape:
 - 1. During backfilling of trenches, install continuous underground-line warning tape directly above cable or raceway at 6 to 8 inches below finished grade. Use multiple tapes where width of multiple lines installed in a common trench exceeds 16 inches overall.
 - 2. Install underground-line warning tape for direct-buried cables and cables in raceways.
- U. Write-on Tags:
 - 1. Place in a location with high visibility and accessibility.
 - 2. Secure using cable ties.
- V. Baked-Enamel Signs:
 - 1. Attach signs that are not self-adhesive type with mechanical fasteners appropriate to the location and substrate.
 - 2. Unless otherwise indicated, provide a single line of text with 1/2-inch- high letters on minimum 1-1/2-inch- high sign; where two lines of text are required, use signs minimum 2 inches high.
- W. Metal-Backed Butyrate Signs:
 - 1. Attach signs that are not self-adhesive type with mechanical fasteners appropriate to the location and substrate.

- 2. Unless otherwise indicated, provide a single line of text with 1/2-inch- high letters on minimum 1-1/2-inch- high sign; where two lines of text are required, use signs minimum 2 inches high.
- X. Laminated Acrylic or Melamine Plastic Signs:
 - 1. Attach signs and plastic labels that are not self-adhesive type with mechanical fasteners appropriate to the location and substrate.
 - 2. Unless otherwise indicated, provide a single line of text with 1/2-inch- high letters on minimum 1-1/2-inch- high sign; where two lines of text are required, use signs minimum 2 inches high.
- Y. Cable Ties: General purpose, for attaching tags, except as listed below:
 - 1. Outdoors: UV-stabilized nylon.
 - 2. In Spaces Handling Environmental Air: Plenum rated.

3.2 IDENTIFICATION SCHEDULE

- A. Install identification materials and devices at locations for most convenient viewing without interference with operation and maintenance of equipment. Install access doors or panels to provide view of identifying devices.
- B. Identify conductors, cables, and terminals in enclosures and at junctions, terminals, pull points, and locations of high visibility. Identify by system and circuit designation.
- C. Accessible Raceways and Metal-Clad Cables, 600 V or Less, for Service and Feeder Circuits, 100A and 120 V to Ground and greater: Identify with self-adhesive raceway labels.
 - 1. Locate identification at changes in direction, at penetrations of walls and floors, at 50-foot maximum intervals in straight runs, and at 25-foot maximum intervals in congested areas.
- D. Accessible Fittings for Raceways and Cables within Buildings: Identify the covers of each junction and pull box of the following systems with self-adhesive labels containing the wiring system legend and system voltage.
- E. Power-Circuit Conductor Identification, 600 V or Less: For conductors in pull and junction boxes, manholes, and handholes, use self-adhesive wraparound labels self-adhesive vinyl tape to identify the phase.
 - 1. Locate identification at changes in direction, at penetrations of walls and floors, at 50-foot maximum intervals in straight runs, and at 25-foot maximum intervals in congested areas.
- F. Conductors to Be Extended in the Future: Attach write-on tags to conductors and list source.
- G. Locations of Underground Lines: Underground-line warning tape for power, lighting, communication, and control wiring and optical-fiber cable.

- H. Workspace Indication: Apply floor marking tape to finished surfaces. Show working clearances in the direction of access to live parts. Workspace shall comply with NFPA 70 and 29 CFR 1926.403 unless otherwise indicated. Do not install at flush-mounted panelboards and similar equipment in finished spaces.
- I. Instructional Signs: Self-adhesive labels, including the color code for grounded and ungrounded conductors.
- J. Warning Labels for Indoor Cabinets, Boxes, and Enclosures for Power and Lighting: Self-adhesive equipment labels.
 - 1. Apply to exterior of door, cover, or other access.
 - 2. For equipment with multiple power or control sources, apply to door or cover of equipment, including, but not limited to, the following:
 - a. Controls with external control power connections.
- K. Arc Flash Warning Labeling: Self-adhesive labels.
 - 1. Comply with NFPA 70E and ANSI Z535.4.
- L. Operating Instruction Signs: Self-adhesive labels.
- M. Equipment Identification Labels:
 - 1. Indoor Equipment: Self-adhesive label.
 - 2. Outdoor Equipment: Stenciled legend 4 inches high.

3.3 BOX IDENTIFICATION

A. All boxes shall be labeled with permanent, machine printed nameplates using a label maker. Handwritten labels are not acceptable. Nameplates shall indicate circuit or load served, as well as the power source and highest voltage present on any conductor.

END OF SECTION 260553

SECTION 260923 - LIGHTING CONTROL DEVICES

PART 1 - GENERAL

1.1 REGULATORY REQUIREMENTS

- A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
- B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.
- 1.2 SUMMARY
 - A. Section Includes:
 - 1. Time switches.
 - 2. Photoelectric switches.
 - 3. Lighting contactors.
 - B. Related Requirements:
 - 1. Section 262726 "Wiring Devices" for manual light switches.
- 1.3 ACTION SUBMITTALS
 - A. Product Data: For each type of product.
 - B. Shop Drawings:
 - 1. Interconnection diagrams showing field-installed wiring.
 - 2. Include diagrams for power, signal, and control wiring.

1.4 INFORMATIONAL SUBMITTALS

- A. Field quality-control reports.
- B. Sample warranty.
1.5 CLOSEOUT SUBMITTALS

A. Operation and maintenance data.

1.6 WARRANTY

- A. Manufacturer's Warranty: Manufacturer and Installer agree to repair or replace lighting control devices that fail(s) in materials or workmanship within specified warranty period.
 - 1. Warranty Period: Two year(s) from date of Substantial Completion.

PART 2 - PRODUCTS

2.1 TIME SWITCHES

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - 1. Cooper Industries, Inc.
 - 2. Intermatic, Inc.
 - 3. Invensys Controls.
 - 4. Leviton Manufacturing Company, Inc.

2.2 OUTDOOR PHOTOELECTRIC SWITCHES

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - 1. Cooper Industries, Inc.
 - 2. Intermatic, Inc.
 - 3. Leviton Manufacturing Company, Inc.
- B. Description: Solid state, with DPST dry contacts rated for 1000 W incandescent, to operate connected relay, contactor coils, or microprocessor input; complying with UL 773A, and compatible with ballasts and LED lamps.
 - 1. Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and application.
 - 2. Light-Level Monitoring Range: 1.5 to 10 fc, with an adjustment for turnon and turn-off levels within that range, and a directional lens in front of the photocell to prevent fixed light sources from causing turn-off.

- 3. Time Delay: Fifteen-second minimum, to prevent false operation.
- 4. Surge Protection: Metal-oxide varistor.
- 5. Mounting: Twist lock complies with NEMA C136.10, with base-and-stem mounting or stem-and-swivel mounting accessories as required to direct sensor to the north sky exposure.
- 6. Failure Mode: Luminaire stays ON.
- C. Description: Solid state; one set of NO dry contacts rated for 24 V ac at 1 A, to operate connected load, complying with UL 773, and compatible with luminaire or power pack or lighting control panelboard.
 - 1. Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and application.
 - 2. Light-Level Monitoring Range: 1.5 to 10 fc, with an adjustment for turnon and turn-off levels within that range.
 - 3. Time Delay: Thirty-second minimum, to prevent false operation.
 - 4. Mounting: 1/2-inch threaded male conduit.
 - 5. Failure Mode: Luminaire stays ON.
 - Power Pack: Digital controller capable of accepting four RJ45 inputs with two outputs rated for 20-A incandescent or LED load at 120- and 277-V ac, for 16-Aballast or LED at 120- and 277-V ac, and for 1 hp at 120-V ac. Sensor has 24-V dc, Class 2 power source, as defined by NFPA 70.
 - a. With integral current monitoring.
 - b. Compatible with digital addressable lighting interface.
 - c. Plenum rated.

2.3 LIGHTING CONTACTORS

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - 1. ABB.
 - 2. Allen-Bradley/Rockwell Automation.
 - 3. ASCO.
 - 4. Eaton.
 - 5. Leviton Manufacturing Company, Inc.
 - 6. Schneider Electric USA (Square D).
- B. Description: Electrically operated and electrically held, combination-type lighting contactors with nonfused disconnect, complying with NEMA ICS 2 and UL 508.

- 1. Current Rating for Switching: Listing or rating consistent with type of load served, including tungsten filament, inductive, and high-inrush ballast (ballast with 15 percent or less THD of normal load current).
- 2. Fault Current Withstand Rating: Equal to or exceeding the available fault current at the point of installation.
- 3. Enclosure: Comply with NEMA 250.
- 4. Provide with control and pilot devices as scheduled, matching the NEMA type specified for the enclosure.
- 2.4 CONDUCTORS AND CABLES
 - A. Power Wiring to Supply Side of Remote-Control Power Sources: Not smaller than No. 12 AWG. Comply with requirements in Section 260519 "Low-Voltage Electrical Power Conductors and Cables."
 - B. Class 1 Control Cable: Multiconductor cable with stranded-copper conductors not smaller than No. 18 AWG. Comply with requirements in Section 260519 "Low-Voltage Electrical Power Conductors and Cables."

PART 3 - EXECUTION

- 3.1 INSTALLATION
 - A. Comply with NECA 1.
 - B. Examine lighting control devices before installation. Reject lighting control devices that are wet, moisture damaged, or mold damaged.
 - C. Coordinate layout and installation of ceiling-mounted devices with other construction that penetrates ceilings or is supported by them, including light fixtures, HVAC equipment, smoke detectors, fire-suppression systems, and partition assemblies.
 - D. Install and aim sensors in locations to achieve not less than 90-percent coverage of areas indicated. Do not exceed coverage limits specified in manufacturer's written instructions.
 - E. Mount electrically held lighting contactors with elastomeric isolator pads to eliminate structure-borne vibration unless contactors are installed in an enclosure with factory-installed vibration isolators.

3.2 WIRING INSTALLATION

- A. Wiring Method: Comply with Section 260519 "Low-Voltage Electrical Power Conductors and Cables." Minimum conduit size is 1/2 inch.
- B. Wiring within Enclosures: Separate power-limited and nonpower-limited conductors in accordance with conductor manufacturer's written instructions.
- C. Size conductors in accordance with lighting control device manufacturer's written instructions unless otherwise indicated.
- D. Splices, Taps, and Terminations: Make connections only on numbered terminal strips in junction, pull, and outlet boxes; terminal cabinets; and equipment enclosures.

3.3 IDENTIFICATION

- A. Identify components and power and control wiring in accordance with Section 260553 "Identification for Electrical Systems."
- B. Label time switches and contactors with a unique designation.

3.4 FIELD QUALITY CONTROL

- A. Perform the following tests and inspections:
 - 1. Operational Test: After installing time switches and sensors, and after electrical circuitry has been energized, start units to confirm proper unit operation.
 - 2. Test and adjust controls and safeties. Replace damaged and malfunctioning controls and equipment.
- B. Lighting control devices will be considered defective if they do not pass tests and inspections.
- C. Prepare test and inspection reports.

3.5 ADJUSTING

1. For daylighting controls, adjust set points and deadband controls to suit Owner's operations.

3.6 DEMONSTRATION

A. Engage a factory-authorized service representative to train Owner's maintenance personnel to adjust, operate, and maintain lighting control devices.

END OF SECTION 260923

293

SECTION 262416 - PANELBOARDS

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section Includes:
 - 1. Lighting and appliance branch-circuit panelboards.

1.2 DEFINITIONS

A. MCCB: Molded-case circuit breaker.

1.3 ACTION SUBMITTALS

- A. Product Data: For each type of panelboard.
- B. Shop Drawings: For each panelboard and related equipment.
 - 1. Include dimensioned plans, elevations, sections, and details.
 - Detail enclosure types including mounting and anchorage, environmental protection, knockouts, corner treatments, covers and doors, gaskets, hinges, and locks.
 - 3. Detail bus configuration, current, and voltage ratings.
 - 4. Short-circuit current rating of panelboards and overcurrent protective devices.
 - 5. Include evidence of NRTL listing for series rating of installed devices.
 - 6. Detail features, characteristics, ratings, and factory settings of individual overcurrent protective devices and auxiliary components.
 - 7. Include time-current coordination curves for each type and rating of overcurrent protective device included in panelboards.

1.4 CLOSEOUT SUBMITTALS

A. Operation and maintenance data.

1.5 FIELD CONDITIONS

- A. Service Conditions: NEMA PB 1, usual service conditions, as follows:
 - 1. Ambient temperatures within limits specified.
 - 2. Altitude not exceeding 6600 feet.

1.6 WARRANTY

- A. Manufacturer's Warranty: Manufacturer agrees to repair or replace panelboards that fail in materials or workmanship within specified warranty period.
 - 1. Panelboard Warranty Period: 18 months from date of Substantial Completion.

Community Park Improvements 20235700

Panelboards 262416 - 1

1.7 REGULATORY REQUIREMENTS

- A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
- B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.

PART 2 - PRODUCTS

- 2.1 PANELBOARDS COMMON REQUIREMENTS
 - A. Electrical Components, Devices, and Accessories: Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and application.
 - B. Comply with NEMA PB 1.
 - C. Comply with NFPA 70.
 - D. Enclosures: Flush and Surface-mounted, dead-front cabinets.
 - 1. Rated for environmental conditions at installed location.
 - a. Indoor Locations: NEMA 250, Type 12.
 - 2. Front: Secured to box with concealed trim clamps. For surface-mounted fronts, match box dimensions; for flush-mounted fronts, overlap box. Trims shall cover all live parts and shall have no exposed hardware.
 - 3. Hinged Front Cover: Entire front trim hinged to box and with standard door within hinged trim cover. Trims shall cover all live parts and shall have no exposed hardware.
 - E. Incoming Mains Location: Convertible between top and bottom
 - F. Phase, Neutral, and Ground Buses: Tin-plated aluminum.
 - G. Conductor Connectors: Suitable for use with conductor material and sizes.
 - 1. Material: Tin-plated aluminum.
 - 2. Main and Neutral Lugs: Mechanical type, with a lug on the neutral bar for each pole in the panelboard.
 - 3. Ground Lugs and Bus-Configured Terminators: Compression type, with a lug on the bar for each pole in the panelboard.
 - 4. Feed-Through Lugs: Compression type, suitable for use with conductor material. Locate at opposite end of bus from incoming lugs or main device. See schedules on plans.

- H. NRTL Label: Panelboards shall be labeled by an NRTL acceptable to authority having jurisdiction for use as service equipment with one or more main service disconnecting and overcurrent protective devices.
- I. Future Devices: Panelboards shall have mounting brackets, bus connections, filler plates, and necessary appurtenances required for future installation of devices.
- J. Panelboard Short-Circuit Current Rating: Fully rated to interrupt symmetrical short-circuit current available at terminals. Assembly listed by an NRTL for 100 percent interrupting capacity.
- 2.2 LIGHTING AND APPLIANCE BRANCH-CIRCUIT PANELBOARDS
 - A. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
 - 1. Eaton.
 - 2. Siemens Industry, Inc., Energy Management Division.
 - 3. Square D; by Schneider Electric.
 - B. Panelboards: NEMA PB 1, lighting and appliance branch-circuit type.
 - C. Mains: Circuit breaker or lugs only See schedule on plans.
 - D. Branch Overcurrent Protective Devices: Bolt-on circuit breakers, replaceable without disturbing adjacent units.
 - E. Doors: Concealed hinges; secured with flush latch with tumbler lock; keyed alike.
- 2.3 DISCONNECTING AND OVERCURRENT PROTECTIVE DEVICES
 - A. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
 - 1. Eaton.
 - 2. Siemens Industry, Inc., Energy Management Division.
 - 3. Square D; by Schneider Electric.
 - B. MCCB: Comply with UL 489, with interrupting capacity to meet available fault currents.
 - 1. Thermal-Magnetic Circuit Breakers:
 - a. Inverse time-current element for low-level overloads.
 - b. Instantaneous magnetic trip element for short circuits.
 - c. Adjustable magnetic trip setting for circuit-breaker frame sizes 250 A and larger.
 - 2. Adjustable Instantaneous-Trip Circuit Breakers: Magnetic trip element with front-mounted, field-adjustable trip setting.
 - 3. MCCB Features and Accessories:

Panelboards 262416 - 3

- a. Standard frame sizes, trip ratings, and number of poles.
- b. Breaker handle indicates tripped status.
- c. UL listed for reverse connection without restrictive line or load ratings.
- d. Lugs: Mechanical style, suitable for number, size, trip ratings, and conductor materials.
- e. Application Listing: Appropriate for application; Type SWD for switching fluorescent lighting loads; Type HID for feeding fluorescent and HID lighting circuits.
- f. Ground-Fault Protection: Integrally mounted relay and trip unit with adjustable pickup and time-delay settings, push-to-test feature, and ground-fault indicator.
- g. Shunt Trip: 120-V trip coil energized from separate circuit, set to trip at 75 percent of rated voltage.
- h. Handle Padlocking Device: Fixed attachment, for locking circuitbreaker handle in on or off position.
- i. Handle Clamp: Loose attachment, for holding circuit-breaker handle in on position.

2.4 IDENTIFICATION

- A. Panelboard Label: Manufacturer's name and trademark, voltage, amperage, number of phases, and number of poles shall be located on the interior of the panelboard door.
- B. Breaker Labels: Faceplate shall list current rating, UL and IEC certification standards, and AIC rating.
- C. Circuit Directory: Directory card inside panelboard door, mounted in metal frame with transparent protective cover.

PART 3 - EXECUTION

- 3.1 INSTALLATION
 - A. Comply with NECA 1.
 - B. Install panelboards and accessories according to NECA 407.
 - C. Mount top of trim 90 inches above finished floor unless otherwise indicated.
 - D. Mount panelboard cabinet plumb and rigid without distortion of box.
 - E. Install overcurrent protective devices and controllers not already factory installed.
 - 1. Set field-adjustable, circuit-breaker trip ranges.
 - F. Make grounding connections and bond neutral for services and separately derived systems to ground. Make connections to grounding electrodes, separate grounds for isolated ground bars, and connections to separate ground bars.
 - G. Install filler plates in unused spaces.

3.2 IDENTIFICATION

- A. Identify field-installed conductors, interconnecting wiring, and components; install warning signs complying with requirements in Section 260553 "Identification for Electrical Systems."
- B. Create a directory to indicate installed circuit loads; incorporate Owner's final room designations. Obtain approval before installing. Handwritten directories are not acceptable. Install directory inside panelboard door.
- C. Panelboard Nameplates: Label each panelboard with a nameplate complying with requirements for identification specified in Section 260553 "Identification for Electrical Systems."
- D. Device Nameplates: Label each branch circuit device in power panelboards with a nameplate complying with requirements for identification specified in Section 260553 "Identification for Electrical Systems."
- E. Install warning signs complying with requirements in Section 260553 "Identification for Electrical Systems" identifying source of remote circuit.

3.3 FIELD QUALITY CONTROL

- A. Perform tests and inspections.
- B. Acceptance Testing Preparation:
 - 1. Test insulation resistance for each panelboard bus, component, connecting supply, feeder, and control circuit.
 - 2. Test continuity of each circuit.
- C. Tests and Inspections:
 - 1. Perform each visual and mechanical inspection and electrical test for lowvoltage air circuit breakers stated in NETA Acceptance Testing Specification. Certify compliance with test parameters.
 - 2. Correct malfunctioning units on-site, where possible, and retest to demonstrate compliance; otherwise, replace with new units and retest.
- D. Panelboards will be considered defective if they do not pass tests and inspections.
- E. Prepare test and inspection reports, including a certified report that identifies panelboards included and that describes scanning results, with comparisons of the two scans. Include notation of deficiencies detected, remedial action taken, and observations after remedial action.

SECTION 262713 - ELECTRICITY METERING

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section includes work to accommodate utility company revenue meters.
- 1.2 ACTION SUBMITTALS
 - A. Product Data: For each type of product.
- 1.3 INFORMATIONAL SUBMITTALS
 - A. Field quality-control reports.
- 1.4 QUALITY ASSURANCE
 - A. Testing Agency Qualifications: An NRTL.
- 1.5 WARRANTY
 - A. Special Warranty: Manufacturer agrees to repair or replace components of metering equipment that fail in materials or workmanship within specified warranty period.
 - 1. Warranty Period: Cost to repair or replace any parts for two years from date of Substantial Completion.
- 1.6 COORDINATION
 - A. Electrical Service Connections: Coordinate with utility companies and utilityfurnished components.

PART 2 - PRODUCTS

- 2.1 SYSTEM DESCRIPTION
 - A. Electrical Components, Devices, and Accessories: Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and application.
 - B. Comply with UL 916.

2.2 UTILITY METERING INFRASTRUCTURE

- A. Install metering accessories furnished by the utility company, complying with its requirements.
- B. Utility-Furnished Meters: Installed by the Utility.
- C. Current-Transformer Cabinets: Comply with requirements of electrical-power utility company.

Community Park Improvements 20235700

Electricity Metering 262713 - 1

- D. Meter Sockets:
 - 1. Comply with requirements of electrical-power utility company.
 - 2. Meter Sockets: Steady-state and short-circuit current ratings shall meet indicated circuit ratings.

PART 3 - EXECUTION

3.1 INSTALLATION

- A. Comply with equipment installation requirements in NECA 1.
- B. Meters furnished and installed by utility company. Install raceways and equipment according to utility company's written instructions. Provide empty conduits for metering leads and extend grounding connections as required by utility company.
- C. Install modular meter center according to switchboard installation requirements in NECA 400.
- D. Install arc-flash labels as required by NFPA 70.
- E. Wiring Method:
 - 1. Comply with requirements in Section 260519 "Low-Voltage Electrical Power Conductors and Cables."
- F. Comply with requirements for identification specified in Section 260553 "Identification for Electrical Systems."

SECTION 262726 - WIRING DEVICES

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section Includes:
 - 1. Receptacles., 125 V, 20 A.
 - 2. GFCI receptacles, 125 V, 20 A.
 - 3. Wall switches.
 - 4. Wall plates.

1.2 ACTION SUBMITTALS

- A. Product Data: For each type of product.
- B. Shop Drawings: List of legends and description of materials and process used for premarking wall plates.
- C. Samples: One for each type of device and wall plate specified, in each color specified.
- 1.3 REGULATORY REQUIREMENTS
 - A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
 - B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.

PART 2 - PRODUCTS

- 2.1 GENERAL WIRING-DEVICE REQUIREMENTS
 - A. Wiring Devices, Components, and Accessories: Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and use.
 - B. Comply with NFPA 70.
 - C. RoHS compliant.
 - D. Comply with NEMA WD 1.
 - E. Device Color:
 - 1. Wiring Devices Connected to Normal Power System: As selected by Design-Builder unless otherwise indicated or required by NFPA 70 or device listing. Multiple colors for Designer devices may be required.
 - 2. Wiring Devices Connected to Essential Electrical System: Red.

Community Park Improvements 20235700

Wiring Devices 262726 - 1

- 3. SPD Devices: Blue.
- F. Wall Plate Color: For plastic covers, match device color.
- G. Source Limitations: Obtain each type of wiring device and associated wall plate from single source from single manufacturer.

2.2 GFCI RECEPTACLES, 125 V, 20 A

- A. Duplex GFCI Receptacles, 125 V, 20 A :
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Eaton (Arrow Hart).
 - b. Hubbell Incorporated; Wiring Device-Kellems.
 - c. Leviton Manufacturing Co., Inc.
 - d. Pass & Seymour/Legrand (Pass & Seymour).
 - 2. Description: Integral GFCI with "Test" and "Reset" buttons and LED indicator light. Two pole, three wire, and self-grounding. Designer Face.
 - 3. Configuration: NEMA WD 6, Configuration 5-20R.
 - 4. Type: Feed through.
 - 5. Standards: Comply with UL 498, UL 943 Class A, and FS W-C-596.
- B. Tamper-Resistant Duplex GFCI Receptacles, 125 V, 20 A:
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Eaton (Arrow Hart).
 - b. Hubbell Incorporated; Wiring Device-Kellems.
 - c. Leviton Manufacturing Co., Inc.
 - d. Pass & Seymour/Legrand (Pass & Seymour).
 - 2. Description: Integral GFCI with "Test" and "Reset" buttons and LED indicator light. Two pole, three wire, and self-grounding. Integral shutters that operate only when a plug is inserted in the receptacle. Rectangular face.
 - 3. Configuration: NEMA WD 6, Configuration 5-20R.
 - 4. Type: Feed through.
 - 5. Standards: Comply with UL 498, UL 943 Class A, and FS W-C-596.
 - 6. Marking: Listed and labeled as complying with NFPA 70, "Tamper-Resistant Receptacles" Article.
- C. Tamper- and Weather-Resistant, GFCI Duplex Receptacles, 125 V, 20 A:
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Eaton (Arrow Hart).
 - b. Hubbell Incorporated; Wiring Device-Kellems.
 - c. Leviton Manufacturing Co., Inc.
 - d. Pass & Seymour/Legrand (Pass & Seymour).

Wiring Devices 262726 - 2 Description: Integral GFCI with "Test" and "Reset" buttons and LED indicator light. Two pole, three wire, and self-grounding. Integral shutters that operate

only when a plug is inserted in the receptacle. Rectangular face.

- 3. Configuration: NEMA WD 6, Configuration 5-20R.
- 4. Type: Feed through.

2.

- 5. Standards: Comply with UL 498 and UL 943 Class A.
- 6. Marking: Listed and labeled as complying with NFPA 70, "Tamper-Resistant Receptacles" and "Receptacles in Damp or Wet Locations" articles.
- 2.3 DECORATOR-STYLE DEVICES, 20 A
 - A. Decorator Duplex Receptacles, 125 V, 20 A:
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Eaton (Arrow Hart).
 - b. Hubbell Incorporated; Wiring Device-Kellems.
 - c. Leviton Manufacturing Co., Inc.
 - d. Pass & Seymour/Legrand (Pass & Seymour).
 - 2. Description: Two pole, three wire, and self-grounding. Square face.
 - 3. Configuration: NEMA WD 6, Configuration 5-20R.
 - 4. Standards: Comply with UL 498.
 - B. Decorator Tamper-Resistant Duplex Receptacles, 125 V, 20 A:
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Eaton (Arrow Hart).
 - b. Hubbell Incorporated; Wiring Device-Kellems.
 - c. Leviton Manufacturing Co., Inc.
 - d. Pass & Seymour/Legrand (Pass & Seymour).
 - 2. Description: Two pole, three wire, and self-grounding. Integral shutters that operate only when a plug is inserted in the receptacle. Square face.
 - 3. Configuration: NEMA WD 6, Configuration 5-20R.
 - 4. Standards: Comply with UL 498.
 - 5. Marking: Listed and labeled as complying with NFPA 70, "Tamper-Resistant Receptacles" Article.
 - C. Decorator, Tamper- and Weather-Resistant, Duplex Receptacles, 125 V, 20 A:
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Eaton (Arrow Hart).
 - b. Hubbell Incorporated; Wiring Device-Kellems.
 - c. Leviton Manufacturing Co., Inc.
 - d. Pass & Seymour/Legrand (Pass & Seymour).
 - 2. Description: Two pole, three wire, and self-grounding. Integral shutters that operate only when a plug is inserted in the receptacle. Square face.

Wiring Devices 262726 - 3 Item 8.

- 3. Configuration: NEMA WD 6, Configuration 5-20R.
- 4. Standards: Comply with UL 498.
- 5. Marking: Listed and labeled as complying with NFPA 70, "Tamper-Resistant Receptacles" and "Receptacles in Damp or Wet Locations" articles.
- D. Decorator Single-Pole Switches, 120/277 V, 20 A:
 - 1. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - a. Eaton (Arrow Hart).
 - b. Hubbell Incorporated; Wiring Device-Kellems.
 - c. Leviton Manufacturing Co., Inc.
 - d. Pass & Seymour/Legrand (Pass & Seymour).
 - 2. Comply with UL 20.

2.4 WALL PLATES

- A. Single Source: Obtain wall plates from same manufacturer of wiring devices.
- B. Single and combination types shall match corresponding wiring devices.
 - 1. Plate-Securing Screws: Metal with head color to match plate finish.
 - 2. Material shall be galvanized steel.
- C. Wet-Location, Weatherproof Cover Plates: NEMA 250, complying with Type 3R, weather-resistant thermoplastic with lockable cover.

PART 3 - EXECUTION

3.1 INSTALLATION

- A. Comply with NECA 1, including mounting heights listed in that standard, unless otherwise indicated.
- B. Coordination with Other Trades:
 - 1. Keep outlet boxes free of plaster, drywall joint compound, mortar, cement, concrete, dust, paint, and other material that may contaminate the raceway system, conductors, and cables.
 - 2. Install device boxes in brick or block walls so that the cover plate does not cross a joint unless the joint is troweled flush with the face of the wall.
 - 3. Install wiring devices after all wall preparation, including painting, is complete.
- C. Device Installation:
 - 1. Connect devices to branch circuits using pigtails that are not less than 6 inches in length.
 - 2. When mounting into metal boxes, remove the fiber or plastic washers used to hold device-mounting screws in yokes, allowing metal-to-metal contact.
- D. Receptacle and Switch Labeling Requirements:

Community Park Improvements 20235700

- 1. Individual labels shall be placed on the back of all switch faceplates and receptacle faceplates indicating the panel and circuit from which the device is fed. Labels shall be permanent, machine printed using a label maker. Handwritten labels are not allowed.
- E. Receptacle Orientation:
 - 1. Install ground pin of vertically mounted receptacles down, and on horizontally mounted receptacles to the left.
 - 2. Install hospital-grade receptacles in patient-care areas with the ground pin or neutral blade at the top.
- F. Device Plates: Do not use oversized or extra-deep plates. Repair wall finishes and remount outlet boxes when standard device plates do not fit flush or do not cover rough wall opening.
- G. Arrangement of Devices: Unless otherwise indicated, mount flush, with long dimension vertical and with grounding terminal of receptacles on top. Group adjacent switches under single, multigang wall plates.
- H. Adjust locations of floor service outlets and service poles to suit arrangement of partitions and furnishings.
- 3.2 FIELD QUALITY CONTROL
 - A. Perform the following tests and inspections:
 - 1. Test Instruments: Use instruments that comply with UL 1436.
 - 2. Test Instrument for Receptacles: Digital wiring analyzer with digital readout or illuminated digital-display indicators of measurement.
 - B. Tests for Receptacles:
 - 1. Line Voltage: Acceptable range is 105 to 132 V.
 - 2. Percent Voltage Drop under 15-A Load: A value of 6 percent or higher is unacceptable.
 - 3. Ground Impedance: Values of up to 2 ohms are acceptable.
 - 4. GFCI Trip: Test for tripping values specified in UL 1436 and UL 943.
 - 5. Using the test plug, verify that the device and its outlet box are securely mounted.
 - C. Wiring device will be considered defective if it does not pass tests and inspections.
 - D. Prepare test and inspection reports.

SECTION 262813 - FUSES

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section Includes:
 - Cartridge fuses rated 600 V ac and less for use in the following:
 a. Enclosed switches.
- 1.2 ACTION SUBMITTALS
 - A. Product Data: For each type of product.
- 1.3 CLOSEOUT SUBMITTALS
 - A. Operation and maintenance data.

PART 2 - PRODUCTS

- 2.1 CARTRIDGE FUSES
 - A. Characteristics: NEMA FU 1, current-limiting, nonrenewable cartridge fuses with voltage ratings consistent with circuit voltages.
 - 1. Type RK-1: 250-V, zero- to 600-A rating, 200 kAIC.
 - 2. Type RK-5: 250-V, zero- to 600-A rating, 200 kAIC.
 - B. Electrical Components, Devices, and Accessories: Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and application.
 - C. Comply with NEMA FU 1 for cartridge fuses.
 - D. Comply with NFPA 70.
 - E. Coordinate fuse ratings with utilization equipment nameplate limitations of maximum fuse size and with system short-circuit current levels.

PART 3 - EXECUTION

3.1 INSTALLATION

A. Install fuses in fusible devices. Arrange fuses so rating information is readable without removing fuse.

3.2 IDENTIFICATION

A. Install labels complying with requirements for identification specified in Section 260553 "Identification for Electrical Systems" and indicating fuse replacement information inside of door of each fused switch and adjacent to each fuse block, socket, and holder.

SECTION 262816 - ENCLOSED SWITCHES AND CIRCUIT BREAKERS

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section Includes:
 - 1. Fusible switches.
 - 2. Nonfusible switches.
 - 3. Molded-case circuit breakers (MCCBs).
 - 4. Molded-case switches.
 - 5. Enclosures.

1.2 ACTION SUBMITTALS

- A. Product Data: For each type of enclosed switch, circuit breaker, accessory, and component indicated. Include nameplate ratings, dimensioned elevations, sections, weights, and manufacturers' technical data on features, performance, electrical characteristics, ratings, accessories, and finishes.
 - 1. Include time-current coordination curves (average melt) for each type and rating of overcurrent protective device; include selectable ranges for each type of overcurrent protective device. Provide in PDF electronic format.
- B. Shop Drawings: For enclosed switches and circuit breakers.
 - 1. Include plans, elevations, sections, details, and attachments to other work.
 - 2. Include wiring diagrams for power, signal, and control wiring.

1.3 CLOSEOUT SUBMITTALS

A. Operation and maintenance data.

1.4 QUALITY ASSURANCE

- A. Testing Agency Qualifications: Accredited by NETA.
 - 1. Testing Agency's Field Supervisor: Currently certified by NETA to supervise on-site testing.

1.5 WARRANTY

- A. Manufacturer's Warranty: Manufacturer and Installer agree to repair or replace components that fail in materials or workmanship within specified warranty period.
 - 1. Warranty Period: One year(s) from date of Substantial Completion.

PART 2 - PRODUCTS

2.1 GENERAL REQUIREMENTS

- A. Source Limitations: Obtain enclosed switches and circuit breakers, overcurrent protective devices, components, and accessories, within same product category, from single manufacturer.
- B. Product Selection for Restricted Space: Drawings indicate maximum dimensions for enclosed switches and circuit breakers, including clearances between enclosures, and adjacent surfaces and other items. Comply with indicated maximum dimensions.
- C. Electrical Components, Devices, and Accessories: Listed and labeled as defined in NFPA 70, by an NRTL, and marked for intended location and application.
- D. Comply with NFPA 70.

2.2 FUSIBLE SWITCHES

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - 1. Eaton.
 - 2. General Electric Company.
 - 3. Siemens Industry, Inc.
 - 4. Square D; by Schneider Electric.
 - B. Type HD, Heavy Duty:
 - 1. Single throw.
 - 2. Three pole.
 - 3. 240-V ac.
 - 4. 1200 A and smaller.
 - 5. UL 98 and NEMA KS 1, horsepower rated, with clips or bolt pads to accommodate specified fuses.
 - 6. Lockable handle with capability to accept three padlocks, and interlocked with cover in closed position.
 - C. Accessories:
 - 1. Equipment Ground Kit: Internally mounted and labeled for copper and aluminum ground conductors.
 - 2. Neutral Kit: Internally mounted; insulated, capable of being grounded and bonded; labeled for copper and aluminum neutral conductors.
 - 3. Class R Fuse Kit: Provides rejection of other fuse types when Class R fuses are specified.
 - 4. Service-Rated Switches: Labeled for use as service equipment.

2.3 NONFUSIBLE SWITCHES

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - 1. Eaton.
 - 2. General Electric Company.
 - 3. Siemens Industry, Inc.
 - 4. Square D; by Schneider Electric.
 - B. Type GD, General Duty, Three Pole, Single Throw, 240-V ac, 600 A and Smaller: UL 98 and NEMA KS 1, horsepower rated, lockable handle with capability to accept two padlocks, and interlocked with cover in closed position.
 - C. Type HD, Heavy Duty, Three Pole, Single Throw, 600-V ac, 1200 A and Smaller: UL 98 and NEMA KS 1, horsepower rated, lockable handle with capability to accept three padlocks, and interlocked with cover in closed position.
 - D. Accessories:
 - 1. Equipment Ground Kit: Internally mounted and labeled for copper and aluminum ground conductors.
 - 2. Neutral Kit: Internally mounted; insulated, capable of being grounded and bonded; labeled for copper and aluminum neutral conductors.
- 2.4 MOLDED-CASE CIRCUIT BREAKERS
 - A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - 1. Eaton.
 - 2. General Electric Company.
 - 3. Siemens Industry, Inc.
 - 4. Square D; by Schneider Electric.
 - B. Circuit breakers shall be constructed using glass-reinforced insulating material. Current carrying components shall be completely isolated from the handle and the accessory mounting area.
 - C. Circuit breakers shall have a toggle operating mechanism with common tripping of all poles, which provides quick-make, quick-break contact action. The circuit-breaker handle shall be over center, be trip free, and reside in a tripped position between on and off to provide local trip indication. Circuit-breaker escutcheon shall be clearly marked on and off in addition to providing international I/O markings. Equip circuit breaker with a push-to-trip button, located on the face of the circuit breaker to mechanically operate the circuit-breaker tripping mechanism for maintenance and testing purposes.

- D. The maximum ampere rating and UL, IEC, or other certification standards with applicable voltage systems and corresponding interrupting ratings shall be clearly marked on face of circuit breaker. Circuit breakers shall be 100 percent rated or series rated as indicated on the Drawings. Circuit breaker/circuit breaker or Fuse/circuit breaker combinations for series connected interrupting ratings shall be listed by UL as recognized component combinations. Any series rated combination used shall be marked on the end-use equipment along with the statement "Caution Series Rated System. _____ Amps Available. Identical Replacement Component Required."
- E. MCCBs shall be equipped with a device for locking in the isolated position.
- F. Lugs shall be suitable for 140 deg F rated wire on 125-A circuit breakers and below.
- G. Standards: Comply with UL 489 with interrupting capacity to comply with available fault currents.
- H. Thermal-Magnetic Circuit Breakers: Inverse time-current thermal element for lowlevel overloads and instantaneous magnetic trip element for short circuits. Adjustable magnetic trip setting for circuit-breaker frame sizes 250 A and larger.
- I. Features and Accessories:
 - 1. Standard frame sizes, trip ratings, and number of poles.
 - 2. Application Listing: Appropriate for application; Type SWD for switching fluorescent lighting loads; Type HID for feeding fluorescent and high-intensity discharge lighting circuits.
- 2.5 ENCLOSURES
 - A. Enclosed Switches and Circuit Breakers: UL 489, NEMA KS 1, NEMA 250, and UL 50, to comply with environmental conditions at installed location.
 - B. Enclosure Finish: The enclosure shall be finished with gray baked enamel paint, electrodeposited on cleaned, phosphatized steel (NEMA 250 Type 1), gray baked enamel paint, electrodeposited on cleaned, phosphatized galvannealed steel (NEMA 250 Types 3R, 12) or a brush finish on Type 304 stainless steel (NEMA 250 Type 4-4X stainless steel).
 - C. Conduit Entry: NEMA 250 Types 4, 4X, and 12 enclosures shall contain no knockouts.
 - D. Operating Mechanism: The circuit-breaker operating handle shall be directly operable through the front cover of the enclosure (NEMA 250 Type 1) or directly operable through the dead front trim of the enclosure (NEMA 250 Type 3R). The cover interlock mechanism shall have an externally operated override. The override shall not permanently disable the interlock mechanism, which shall return to the locked position once the override is released. The tool used to override the cover interlock mechanism shall not be required to enter the enclosure in order to override the interlock.

E. Enclosures designated as NEMA 250 Type 4, 4X stainless steel, 12, or 12K shall have a dual cover interlock mechanism to prevent unintentional opening of the enclosure cover when the circuit breaker is ON and to prevent turning the circuit breaker ON when the enclosure cover is open.

PART 3 - EXECUTION

- 3.1 ENCLOSURE ENVIRONMENTAL RATING APPLICATIONS
 - A. Enclosed Switches and Circuit Breakers: Provide enclosures at installed locations with the following environmental ratings.
 - 1. Indoor Locations: NEMA 250, Type 12.
 - 2. Outdoor Locations: NEMA 250, Type 3R or Type 4X.

3.2 INSTALLATION

- A. Interruption of Existing Electric Service: Do not interrupt electric service to facilities occupied by Owner or others unless permitted under the following conditions:
 - 1. Notify Design-Builder no fewer than seven days in advance of proposed interruption of electric service.
 - 2. Comply with NFPA 70E.
- B. Coordinate layout and installation of switches, circuit breakers, and components with equipment served and adjacent surfaces. Maintain required workspace clearances and required clearances for equipment access doors and panels.
- C. Install individual wall-mounted switches and circuit breakers with tops at uniform height unless otherwise indicated.
- D. Temporary Lifting Provisions: Remove temporary lifting of eyes, channels, and brackets and temporary blocking of moving parts from enclosures and components.
- E. Install fuses in fusible devices.
- F. Comply with NFPA 70 and NECA 1.
- G. Set field-adjustable circuit-breaker trip ranges as specified in Section 260573.16 "Coordination Studies."

3.3 IDENTIFICATION

- A. Comply with requirements in Section 260553 "Identification for Electrical Systems."
 - 1. Identify field-installed conductors, interconnecting wiring, and components; provide warning signs.
 - 2. Label each enclosure with engraved metal or laminated-plastic nameplate.

3.4 FIELD QUALITY CONTROL

- A. Perform tests and inspections.
- B. Tests and Inspections for Switches:
 - 1. Visual and Mechanical Inspection:
 - a. Inspect physical and mechanical condition.
 - b. Inspect anchorage, alignment, grounding, and clearances.
 - c. Verify that the unit is clean.
 - d. Verify blade alignment, blade penetration, travel stops, and mechanical operation.
 - e. Verify that fuse sizes and types match the Specifications and Drawings.
 - f. Verify that each fuse has adequate mechanical support and contact integrity.
 - g. Inspect bolted electrical connections for high resistance using one of the two following methods:
 - 1) Use a low-resistance ohmmeter.
 - a) Compare bolted connection resistance values to values of similar connections. Investigate values that deviate from those of similar bolted connections by more than 50 percent of the lowest value.
 - 2) Verify tightness of accessible bolted electrical connections by calibrated torque-wrench method in accordance with manufacturer's published data or NETA ATS Table 100.12.
 - a) Bolt-torque levels shall be in accordance with manufacturer's published data. In the absence of manufacturer's published data, use NETA ATS Table 100.12.
 - h. Verify that operation and sequencing of interlocking systems is as described in the Specifications and shown on the Drawings.
 - i. Verify correct phase barrier installation.
 - j. Verify lubrication of moving current-carrying parts and moving and sliding surfaces.
- C. Tests and Inspections for Molded Case Circuit Breakers:
 - 1. Visual and Mechanical Inspection:
 - a. Verify that equipment nameplate data are as described in the Specifications and shown on the Drawings.
 - b. Inspect physical and mechanical condition.
 - c. Inspect anchorage, alignment, grounding, and clearances.
 - d. Verify that the unit is clean.
 - e. Operate the circuit breaker to ensure smooth operation.

Community Park Improvements 20235700

Enclosed Switches And Circuit Breakers 262816 - 6

- f. Inspect bolted electrical connections for high resistance using one of the two following methods:
 - 1) Use a low-resistance ohmmeter.
 - a) Compare bolted connection resistance values to values of similar connections. Investigate values that deviate from those of similar bolted connections by more than 50 percent of the lowest value.
 - 2) Verify tightness of accessible bolted electrical connections by calibrated torque-wrench method in accordance with manufacturer's published data or NETA ATS Table 100.12.
 - a) Bolt-torque levels shall be in accordance with manufacturer's published data. In the absence of manufacturer's published data, use NETA ATS Table 100.12.
- g. Inspect operating mechanism, contacts, and chutes in unsealed units.
- h. Perform adjustments for final protective device settings in accordance with the coordination study.
- 2. Electrical Tests:
- 3. Correct malfunctioning units on-site, where possible, and retest to demonstrate compliance; otherwise, replace with new units and retest.
- 4. Test and adjust controls, remote monitoring, and safeties. Replace damaged and malfunctioning controls and equipment.
- D. Enclosed switches and circuit breakers will be considered defective if they do not pass tests and inspections.
- E. Prepare test and inspection reports.
 - 1. Test procedures used.
 - 2. Include identification of each enclosed switch and circuit breaker tested and describe test results.
 - 3. List deficiencies detected, remedial action taken, and observations after remedial action.

ltem 8.

SECTION 264313 - SURGE PROTECTION FOR LOW-VOLTAGE ELECTRICAL POWER CIRCUITS

PART 1 - GENERAL

- 1.1 SUMMARY
 - A. Section includes field-mounted SPDs for low-voltage (120 to 600 V) power distribution and control equipment.
- 1.2 ACTION SUBMITTALS
 - A. Product Data: For each type of product.
 - 1. Include rated capacities, operating characteristics, electrical characteristics, and furnished specialties and accessories.
 - 2. Copy of UL Category Code VZCA certification, as a minimum, listing the tested values for VPRs, Inominal ratings, MCOVs, type designations, OCPD requirements, model numbers, system voltages, and modes of protection.
- 1.3 CLOSEOUT SUBMITTALS
 - A. Maintenance data.
- 1.4 WARRANTY
 - A. Manufacturer's Warranty: Manufacturer agrees to replace or replace SPDs that fail in materials or workmanship within specified warranty period.
 - 1. Warranty Period: Five years from date of Substantial Completion.
- 1.5 REGULATORY REQUIREMENTS
 - A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
 - B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.

PART 2 - PRODUCTS

- 2.1 GENERAL SPD REQUIREMENTS
 - A. SPD with Accessories: Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and application.
 - B. Comply with NFPA 70.
 - C. Comply with UL 1449.

Community Park Improvements 20235700

Surge Protection For Low-Voltage Electrical Power Circuits 264313 - 1

D. MCOV of the SPD shall be the nominal system voltage.

2.2 SERVICE ENTRANCE SUPPRESSOR

- A. Manufacturers: Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - 1. ABB USA.
 - 2. Eaton.
 - 3. Leviton Manufacturing Co., Inc.
 - 4. Liebert; a brand of Emerson Electric Co.
 - 5. Schneider Electric USA, Inc.
 - 6. Square D; by Schneider Electric.
- B. SPDs: Comply with UL 1449, Type 2.
 - 1. SPDs with the following features and accessories intended for installation on the load side of the main distribution panel. Install outside the panelboard:
 - a. Internal thermal protection that disconnects the SPD before damaging internal suppressor components.
 - b. Indicator light display for protection status.
- C. Peak Surge Current Rating: The minimum single-pulse surge current withstand rating per phase shall not be less than 200 kA. The peak surge current rating shall be the arithmetic sum of the ratings of the individual MOVs in a given mode.
- D. Protection modes and UL 1449 VPR for 240/120 V, single-phase, three-wire circuits shall not exceed the following:
 - 1. Line to Neutral: 700 V.
 - 2. Line to Ground: 800 V.
 - 3. Line to Line: 1200 V.
- E. SCCR: Equal or exceed 200 kA.
- F. I nominal Rating: 20 kA.
- G. SPD shall include the following features:
 - 1. Audible diagnostic monitoring by way of an audible alarm function
 - 2. One set of NO/NC dry contracts for alarm conditions

2.3 ENCLOSURES

A. Indoor Enclosures: NEMA 250, Type 12.

PART 3 - EXECUTION

- 3.1 INSTALLATION
 - A. Comply with NECA 1.

Community Park Improvements 20235700

Surge Protection For Low-Voltage Electrical Power Circuits 264313 - 2

Iconica

- B. Install an OCPD or disconnect as required to comply with the UL listing of the SPD.
- C. Install SPDs with conductors between suppressor and points of attachment as short and straight as possible, and adjust circuit-breaker positions to achieve shortest and straightest leads. Do not splice and extend SPD leads unless specifically permitted by manufacturer. Do not exceed manufacturer's recommended lead length. Do not bond neutral and ground.
- D. Use crimped connectors and splices only. Wire nuts are unacceptable.
- E. Complete startup checks according to manufacturer's written instructions. Energize SPDs after power system has been energized, stabilized, and tested.

3.2 FIELD QUALITY CONTROL

- A. Perform the following tests and inspections with the assistance of a factoryauthorized service representative.
 - 1. Compare equipment nameplate data for compliance with Drawings and Specifications.
 - 2. Inspect anchorage, alignment, grounding, and clearances.
 - 3. Verify that electrical wiring installation complies with manufacturer's written installation requirements.
- B. An SPD will be considered defective if it does not pass tests and inspections.
- C. Prepare test and inspection reports.

3.3 DEMONSTRATION

A. Train Owner's maintenance personnel to operate and maintain SPDs.

SECTION 265119 - LED INTERIOR LIGHTING

PART 1 - GENERAL

1.1 REGULATORY REQUIREMENTS

- A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
- B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.

1.2 ACTION SUBMITTALS

- A. Product Data: For each type of product.
 - 1. Manufacturers' Certified Data: Photometric data certified by manufacturer's laboratory with a current accreditation under the National Voluntary Laboratory Accreditation Program for Energy Efficient Lighting Products.
 - 2. Testing Agency Certified Data: For indicated luminaires, photometric data certified by a qualified independent testing agency. Photometric data for remaining luminaires shall be certified by manufacturer.
- B. Shop Drawings: For nonstandard or custom luminaires.
 - 1. Include plans, elevations, sections, and mounting and attachment details.
 - 2. Include details of luminaire assemblies. Indicate dimensions, weights, loads, required clearances, method of field assembly, components, and location and size of each field connection.
 - 3. Include diagrams for power, signal, and control wiring.

1.3 INFORMATIONAL SUBMITTALS

A. Sample warranty.

1.4 CLOSEOUT SUBMITTALS

A. Operation and maintenance data.

1.5 QUALITY ASSURANCE

- A. Luminaire Photometric Data Testing Laboratory Qualifications: Luminaire manufacturer's laboratory that is accredited under the NVLAP for Energy Efficient Lighting Products.
- B. Luminaire Photometric Data Testing Laboratory Qualifications: Provided by an independent agency, with the experience and capability to conduct the testing indicated, that is an NRTL as defined by OSHA in 29 CFR 1910.7, accredited under the NVLAP for Energy Efficient Lighting Products, and complying with the applicable IES testing standards.
- C. Provide luminaires from a single manufacturer for each luminaire type.
- D. Each luminaire type shall be binned within a three-step MacAdam Ellipse to ensure color consistency among luminaires.

1.6 WARRANTY

- A. Warranty: Manufacturer and Installer agree to repair or replace components of luminaires that fail in materials or workmanship within specified warranty period.
- B. Warranty Period: Five year(s) from date of Substantial Completion.

PART 2 - PRODUCTS

2.1 PERFORMANCE REQUIREMENTS

- A. Ambient Temperature: 41 to 104 deg F.
 - 1. Relative Humidity: Zero to 95 percent.
- B. Altitude: Sea level to 2000 feet.

2.2 LUMINAIRE REQUIREMENTS

- A. Electrical Components, Devices, and Accessories: Listed and labeled as defined in NFPA 70, by a qualified testing agency, and marked for intended location and application.
- B. Factory-Applied Labels: Comply with UL 1598. Include recommended lamps. Locate labels where they will be readily visible to service personnel, but not seen from normal viewing angles when lamps are in place.
 - 1. Label shall include the following lamp characteristics:
 - a. "USE ONLY" and include specific lamp type.
 - b. Lamp diameter, shape, size, wattage, and coating.
 - c. CCT and CRI.
- C. Recessed luminaires shall comply with NEMA LE 4.
- D. NRTL Compliance: Luminaires for hazardous locations shall be listed and labeled for indicated class and division of hazard by an NRTL.
- 2.3 FIXTURE TYPES: See Schedule on plans.
- 2.4 MATERIALS
 - A. Metal Parts:
 - 1. Free of burrs and sharp corners and edges.
 - 2. Sheet metal components shall be steel unless otherwise indicated.
 - 3. Form and support to prevent warping and sagging.
 - B. Steel:
 - 1. ASTM A36/A36M for carbon structural steel.
 - 2. ASTM A568/A568M for sheet steel.
 - C. Stainless Steel:
 - 1. Manufacturer's standard grade.
 - 2. Manufacturer's standard type, ASTM A240/240M.
 - D. Galvanized Steel: ASTM A653/A653M.
 - E. Aluminum: ASTM B209.

2.5 METAL FINISHES

A. Variations in finishes are unacceptable in the same piece. Variations in finishes of adjoining components are acceptable if they are within the range of approved Samples and if they can be and are assembled or installed to minimize contrast.

2.6 LUMINAIRE SUPPORT

- A. Comply with requirements in Section 260529 "Hangers and Supports for Electrical Systems" for channel and angle iron supports and nonmetallic channel and angle supports.
- B. Single-Stem Hangers: 1/2-inch steel tubing with swivel ball fittings and ceiling canopy. Finish same as luminaire.
- C. Wires: ASTM A641/A641M, Class 3, soft temper, zinc-coated steel, 12 gage.
- D. Rod Hangers: 3/16-inch minimum diameter, cadmium-plated, threaded steel rod.
- E. Hook Hangers: Integrated assembly matched to luminaire, line voltage, and equipment with threaded attachment, cord, and locking-type plug.

PART 3 - EXECUTION

- 3.1 INSTALLATION
 - A. Comply with NECA 1.
 - B. Install luminaires level, plumb, and square with ceilings and walls unless otherwise indicated.
 - C. Install lamps in each luminaire.
 - D. Supports:
 - 1. Sized and rated for luminaire weight.
 - 2. Able to maintain luminaire position after cleaning and relamping.
 - 3. Provide support for luminaire without causing deflection of ceiling or wall.
 - 4. Luminaire-mounting devices shall be capable of supporting a horizontal force of 100 percent of luminaire weight and a vertical force of 400 percent of luminaire weight.

E. Comply with requirements in Section 260519 "Low-Voltage Electrical Power Conductors and Cables" for wiring connections.

3.2 IDENTIFICATION

A. Identify system components, wiring, cabling, and terminals. Comply with requirements for identification specified in Section 260553 "Identification for Electrical Systems."

3.3 FIELD QUALITY CONTROL

- A. Perform the following tests and inspections:
 - 1. Operational Test: After installing luminaires, switches, and accessories, and after electrical circuitry has been energized, test units to confirm proper operation.
 - 2. Test for Emergency Lighting: Interrupt power supply to demonstrate proper operation. Verify transfer from normal power to battery power and retransfer to normal.
- B. Luminaire will be considered defective if it does not pass operation tests and inspections.
- C. Prepare test and inspection reports.

SECTION 265668 - EXTERIOR ATHLETIC LIGHTING

PART 1 - GENERAL

1.1 REGULATORY REQUIREMENTS

- A. All work and materials are to conform in every detail to applicable rules and requirements of the Wisconsin State Electrical Code (SPS 316), the National Electrical Code (NFPA 70), other applicable National Fire Protection Association codes, the National Electrical Safety Code, and present manufacturing standards (including NEMA).
- B. All Division 26 work shall be done under the direction of a currently certified State of Wisconsin Certified Master Electrician.

1.2 SUMMARY

- A. Section includes lighting for the following outdoor sports venues:
 - 1. Baseball fields.
- B. This specification also includes information on the Bid Alternate for this project. As part of the Bid Alternate, Contractor shall install Owner-furnished, direct-bury wooden light poles and provide mounting hardware as required to mount exterior athletic lighting as specified herein and shown on the Contract Documents.

1.3 DEFINITIONS

- A. Coefficient of Variation (CV): A statistical measure of the weighted average of all relevant illumination values for the playing area, expressed as the ratio of the standard deviation for all illuminance values to the mean illuminance value.
- B. Fixture: See "Luminaire."
- C. Illuminance: The metric most commonly used to evaluate lighting systems. It is the density of luminous flux, or flow of light, reaching a surface divided by the area of that surface.
 - 1. Horizontal Illuminance: Measurement in foot-candles, on a horizontal surface 36 inches above ground unless otherwise indicated.
 - 2. Target Illuminance: Average maintained illuminance level, calculated by multiplying initial illuminance by LLF.

Exterior Athletic Lighting 265668 - 1

- D. LC: Lighting Certified.
- E. Light-Loss Factor (LLF): A factor used in calculating the level of illumination after a given period of time and under given conditions. It takes into account temperature, dirt accumulation on the luminaire, lamp depreciation, maintenance procedures, and atmospheric conditions. An LLF includes a recoverable light-loss factor.
- F. Luminaire: A complete lighting unit, internally lighted exit sign, or emergency lighting unit. Luminaires include lamps and the parts required to distribute light, position and protect lamps, and connect lamps to power supply. Note that "fixture" and "luminaire" may be used interchangeably and the "IES Lighting Handbook" uses "luminaire" over "fixture."
- G. Pole: Luminaire support structure, including tower used for large area illumination.
- 1.4 ACTION SUBMITTALS
 - A. Product Data:
 - 1. For each type of lighting product.
 - B. Delegated-Design Submittal: For exterior athletic lighting indicated to comply with performance requirements and design criteria, including analysis data.
 - 1. Drawings and specifications for construction of lighting system.
 - 2. Manufacturer's determination of LLF used in design calculations.
 - 3. Lighting system design calculations for the following:
 - a. Target illuminance.
 - b. Point calculations of horizontal illuminance and CV at minimum grid size and area.
 - c. Point calculations of horizontal illuminance in indicated areas of concern for spill light.
 - 4. Electrical system design calculations for the following:
 - a. Total connected and estimated peak-demand electrical load of lighting system.
 - b. Capacity of feeder required to supply lighting system.
 - 5. Wiring requirements, including required conductors, cables, and wiring methods.

1.5 INFORMATIONAL SUBMITTALS

A. Field quality-control reports.
- B. Sample warranty.
- C. Product Certificates:
 - 1. For support structures, including brackets, arms, appurtenances, bases, anchorages, and foundations.
- 1.6 CLOSEOUT SUBMITTALS
 - A. Operation and maintenance data.

1.7 WARRANTY

- A. Special Warranty: Manufacturer's standard form in which manufacturer agrees to repair or replace components of luminaires, lamps, and luminaire alignment products and to correct misalignment that occurs subsequent to successful acceptance tests. Manufacturer may exclude lightning damage, hail damage, vandalism, abuse, and unauthorized repairs and alterations from special warranty coverage.
- B. Warranty Period: Five years from date of Substantial Completion.

PART 2 - PRODUCTS

- A. Subject to compliance with requirements, available manufacturers offering products that may be incorporated into the Work include, but are not limited to the following:
 - 1. Wisconsin Lighting Lab, Inc. (WiLL), or approved equal.
- B. The drawings and specifications were prepared based on WiLL. Contractor shall include in the Bid and shall be responsible for the cost of any changes to accommodate other equipment including but not limited to structural, electrical, or civil work. Contractor shall also pay additional costs necessary for revision of drawings and/or specifications by Architect or Engineer.

2.2 PERFORMANCE REQUIREMENTS

- A. Illumination Criteria:
 - 1. Minimum average target illuminance level for each lighted area for each sports venue and for the indicated class of play in accordance with IES RP-6.
 - CV and maximum-to-minimum uniformity ratios for each lighted area equal to or less than those listed in IES RP-6 for the indicated class of play.

- 3. UG levels within each lighted area equal to or less than those listed in IES RP-6 for the indicated speed of sport.
- B. Illumination Calculations: Computer-analyzed point method complying with IES RP-6 to optimize selection, location, and aiming of luminaires.
 - 1. Grid Pattern Dimensions: For playing areas of each sport and areas of concern for spill-light control, correlate and reference calculated parameters to the grid areas. Each grid point represents the center of the grid area defined by the length and width of the grid spacing.
 - 2. Spill-Light Control: Minimize spill light for each playing area on adjacent and nearby areas.
 - 3. Glare Control: The baseball field shall use only downward-facing, full cut-off lens lighting, with same or less intensity than the replaced lighting for it's baseball field. All new lighting shall be directed away from woodlands located to the west and north of Ridgeway Park and fields. The definition for full cut-off lens is "zero intensity at or above horizontal (90° above nadir) and limited to a value not exceeding 10% of lamp lumens at or above 90°" (definition from Illuminating Engineering Society of North America).
 - 4. Determine LLF in accordance with IES RP-6 and manufacturer's test data.
 - a. LLF shall not be higher than 70 percent and may be lower when determined by manufacturer after application of the ballast output and optical system output in accordance with IES RP-6.
- C. Baseball Fields:
 - 1. IES RP-6: Class III 50/30 fc .
 - 2. Speed of Sport: Moderate.
 - 3. Grid Pattern Dimensions: 30 by 30 feet.
- D. Lighting Control: Providing the following functions, integrated into a single control station:
 - 1. Control Station: Master switch, manual push-button controls, and system status indicator lights.
 - 2. Timer with 2-minute dim warning.
 - 3. Sunrise, sunset, and/or time-triggered lighting events.
 - 4. Recurring events (daily/weekly).
 - 5. Calendar event scheduling.
- E. Electric Power Distribution Requirements:
 - 1. Electric Power: 240 V; single-phase.

2. Maximum Total Voltage Drop from Source to Load: 5 percent, including voltage drops in branch circuit, subfeeder, and feeder. Refer to wiring sizes on drawings for additional information.

2.3 LUMINAIRES, LAMPS, AND BALLASTS

- A. Luminaires: WiLL WS-PKG-KBX Package with pre-aimed KB8 LED fixtures and aluminum cross arms. This shall include the following:
 - 1. WS-KB8-940-57-MV-5N-BK-APY-WHP15NP.
 - 2. WS-PDH.
 - 3. WS-RPCE-8W-X-MV-LG-PBS.
 - 4. NP-A-ACr-BTP-4/2-STUD-180-SBF.
 - 5. WS-GFX.
- B. Listed and labeled, by an NRTL acceptable to authorities having jurisdiction, for compliance with UL 1598 for installation in wet locations.
- C. Doors, Frames, and Other Internal Access: Smooth operating, free from light leakage under operating conditions, and arranged to permit relamping without using tools. Arrange doors, frames, lenses, diffusers, and other pieces to prevent their accidental falling during relamping and when secured in operating position. Doors shall be removable for cleaning or replacing lens.
- D. Exposed Hardware: Stainless steel latches, fasteners, and hinges.
- E. Spill-Light Control Devices: Internal louvers and external baffles furnished by manufacturer and designed for secure attachment to specific luminaire.

2.4 POWER DISTRIBUTION AND CONTROL

A. Electrical Enclosures Exposed to Weather: NEMA 250, Type 3R enclosure constructed from corrosion-resistant material, with hinged doors fitted with padlock hasps or lockable latches.

PART 3 - EXECUTION

- 3.1 INSTALLATION
 - A. Comply with NECA 1.
 - B. Wiring Method: Install cables in raceways, except when cables are installed within boxes and on poles. Conceal raceways and cables.

Community Park Improvements 20235700

Exterior Athletic Lighting 265668 - 5

- Comply with requirements in Section 260519 "Low-Voltage Electrical Power Conductors and Cables" and Section 260533 "Raceways and Boxes for Electrical Systems" for wiring connections and wiring methods.
- C. Protect equipment during installation to prevent corrosion.
- D. Install controls and ballast housings in cabinets mounted on support structure at heights recommended by manufacturer.
- E. Identify system components, wiring, cabling, and terminals. Comply with requirements for identification specified in Section 260553 "Identification for Electrical Systems."
- F. Light Poles (Bid Alternate only):
 - 1. Install direct-bury wooden poles and other structural units level, plumb, and square.
 - 2. Unless otherwise specified in the Contract Documents, the setting depths for poles shall be equal to or greater than 10% of the pole size, plus 2 feet.
 - 3. The holes shall be at least three times larger than the butt diameter of the pole and shall be as large at the bottom of the hole at the top. The pole shall be minimum 6 inches deeper than the required pole burial depth.
 - 4. After setting, poles shall be backfilled with 3/8-inch limestone traffic bond. The backfill shall be mechanically tamped in 6-inch layers to 6 inches above the top of the hole. Continuous backfilling is not allowed while tamping is in progress. Contractor shall dispose of all excess soil.
 - 5. The structures shall be erected plumb. After project completion, Contractor shall be responsible for adjusting pole plumbness as requested by Owner for 1-year after Substantial Completion.

3.2 FIELD QUALITY CONTROL

- A. Perform the following tests and inspections:
 - 1. After installing sports lighting system and after electrical circuits have been energized, perform proof-of-performance field measurements and analysis for compliance with requirements.
 - 2. Playing and Other Designated Areas: Make field measurements at intersections of grids, dimensioned and located as specified in "Performance Requirements" Article and as described below:
 - a. Baseball Fields: Measure at least 16 points of the infield and 57 points of the outfield.

- 3. Make field measurements at established test points in areas of concern for spill light and glare.
- 4. Perform analysis to demonstrate correlation of field measurements with specified illumination quality and quantity values and corresponding computer-generated values that were submitted with engineered design documents. Submit a report of the analysis. For computer-generated values, use manufacturer's lamp lumens that are adjusted to lamp age at time of field testing.
- B. Correction of Illumination Deficiencies for Playing Areas: Make corrections to illumination quality or quantity, measured in field quality-control tests, that varies from specified illumination criteria by plus or minus 10 percent.
 - 1. Add or replace luminaires; change mounting height and aiming; or install louvers, shields, or baffles.
 - 2. If luminaires are added or mounting height is changed, revise aiming and recalculate and modify or replace support structures if indicated.
 - 3. Retest as specified above after repairs, adjustments, or replacements are made.
 - 4. Report results in writing.
- C. Correction of Excessive Illumination in Spill-Light-Critical Areas: If measurements indicate that specified limits for spill light are exceeded, make corrections to illumination quantity, measured in field quality-control tests, that reduce levels to within specified maximum values.
 - 1. Replace luminaires; change mounting heights and revise aiming; or install louvers, shields, or baffles.
 - 2. If mounting height is changed, revise aiming and recalculate and modify or replace support structures if indicated.
 - 3. Retest as specified above after repairs, adjustments, or replacements are made.
 - 4. Report results in writing.
- D. Sports lighting will be considered defective if it does not pass tests and inspections.
- E. Prepare test and inspection reports.

END OF SECTION 265668

PROJECT LOCATION MAP



A. HUGHETTE ROAD

RIDGEWAY COMMUNITY PARK VILLAGE OF RIDGEWAY RIDGEWAY, WISCONSIN PROJECT NUMBER 22.055

T100
C001
C002
C100
C101
C102
C103
C104
C900
C901
C902
C903
E001
E100
E100A
E100D
C701

INDEX TO DRAWINGS
TITLE SHEET
EXISTING CONDITIONS
EXISTING CONDITIONS PHOTOS
SITE LAYOUT PLAN - OVERVIEW
SITE LAYOUT PLAN - DEPOT AREA
SITE LAYOUT PLAN - DIAMOND DRAINAGE AREA
SITE LAYOUT PLAN - NORTH PARKING AREA
SITE GRADING PLAN - DEPOT AREA
CONSTRUCTION DETAILS
CONSTRUCTION DETAILS
CONSTRUCTION DETAILS
CONSTRUCTION DETAILS
ABBREVIATIONS, SYMBOLS, AND NOTES
ELECTRICAL SITE
ELECTRICAL SITE BID ALTERNATE
ELECTRICAL SITE DEMO
ELECTRICAL SCHEDULES AND DIAGRAMS





SITE DEMOLITION NOTES

- CONTACT DIGGER'S HOTLINE 5 WORKING DAYS PRIOR TO THE START OF CONSTRUCTION.
- . WORKING HOURS TO BE APPROVED BY VILLAGE.
- B. CONSTRUCTION TRAFFIC SHALL BE MINIMIZED TO AVOID CONFLICTS WITH PARK USERS.
- COORDINATE SITE ACCESS, STAGING AND SECURITY WITH VILLAGE STAFF.
- 5. CONSTRUCTION STAFF MUST PARK ON THE STREET OR IN PARKING LOTS - NO PERSONAL VEHICLES ON THE CONSTRUCTION SITE.
- 5. DAMAGE TO EXISTING HARDSCAPE, TREES, OR LAWN AT SITE ACCESS MUST BE REPLACED IN KIND.

I EXISTING CONDITIONS NOTES

 SITE SURVEY PROVIDED BY VILLAGE OF RIDGEWAY.
 ACTUAL FIELD CONDITIONS MAY VARY SLIGHTLY. CONTRACTOR TO VERIFY DIMENSIONS AND ELEVATIONS PRIOR TO CONSTRUCTION.

INACTIVE CAPPED LANDFILL

10





















: V:\22.055 Ridgeway Park\CAD\P-SP.dwg Layout: C100 User: Blake Plotted: Sep 11, 2023 - 4:23pm



le: V:\22.055 Ridgeway Park\CAD\P-SP.dwg_Layout: C101_User: Blake_Plotted: Sep 11, 2023 - .



SIT 1. 2. 3. 4. 5. 6.	E LAYOUT NOTES CONTACT DIGGER'S HOTLINE 5 WORKING DAYS PRIOR TO THE START OF CONSTRUCTION. ALL PROPOSED DIMENSIONS ARE REFERENCED PARALLEL OR PERPENDICULAR TO THE PROPOSED FEATURES SHOWN. WRITTEN DIMENSIONS SUPERSEDE ANY SCALED DIMENSIONS. PROPOSED GRADE AND LEVEL SHALL BE PERIODICALLY REVIEWED IN THE FIELD BY THE OWNER OR A/E. CONCRETE JOINTING IS SHOWN FOR DESIGN INTENT. FINAL SCORING PLANS SHALL BE SUBMITTED TO THE A/E FOR REVIEW AS A SHOP SUBMITTAL. ALL CONCRETE POURS SHALL BE DOWELED TOGETHER UNLESS OTHERWISE APPROVED BY THE A/E. SEED, FERTILIZE, AND CRIMP MULCH ALL GENERAL LANDSCAPE WITH THE SPECIFICATIONS EXCEPT THOSE AREAS INDICATING OTHERWISE. THIS SHALL INCLUDE ANY AREAS OUTSIDE OF THE PROJECT LIMITS THAT ARE DISTURBED BY CONTRACTOR ACTIVITY. ANY EXISTING STRUCTURES AND/OR UTILITIES NOT SHOWN ON	+ PLANNING	Parkitecture 608.886.6808
8. LEC	THESE DOCUMENTS WHICH NEED TO BE REMOVED, RELOCATED, AND OR ADJUSTED SHALL BE THE RESPONSIBILITY OF THE SITE CONTRACTOR AND INCLUDED IN THE BASE BID. CONTRACTOR SHALL BE RESPONSIBLE FOR SITE STAKING. DIGITAL PLAN FILES MAY BE AVAILABLE FROM THE A/E.		IREA
	Area 1	PARK IMPROVEMENTS 82	AN - DIAMOND DRAINAGE AREA
		Project Name: Project Name: Projec	
IM	AGE 2	Date: 09/1 Sheet Number	22.055 REVIEW 1/2023





ïle: V:\22.055 Ridgeway Park\CAD\P-SP.dwg Layout: C104 User: Blake Plotted: Sep 11, 2023 – 4:21pm















V:\22.055 Ridgeway Park\CAD\P-D.dwg Layout: C903 User: Blake Plotted: Sep 11, 2023 – 4:26pm

NOTE: PROVIDE FASTENERS AND INSTALLATION OF BENCHES AS SHOWN. IF INSTALLED ON PAVERS, DO NOT ANCHOR.

Project #: Issued For:	IMUNITY PARK IMPROVEMENTS	PARKITECTURE + PLANNING
09/1	KIDGEWAY, WI 33382	+ 901 Deming Way, Suite 102
REV	Sheet Title:	Parkitecture 608.886.6808
	CONSTRUCTION DETAILS	

ELECTRICAL SYMBOLS, ABBREVIATIONS AND NOTES:

=

+XX

SYMBOLS INDICATED HERE AND NOT USED IN THE CONTRACT DOCUMENTS DO NOT APPLY TO THIS PROJECT. ADDITIONAL SYMBOLS MAY BE INDICATED IN

RECEPTACLES

DUPLEX RECEPTACLE MOUNTING LOCATION: WALL MOUNTING HEIGHT: 1'-6"

MOUNTING LOCATION: WALL

DUPLEX RECEPTACLE WEATHERPROOF MOUNTING LOCATION: WALL MOUNTING HEIGHT: 1'-6"

MOUNTING LOCATION: WALL

MOUNTING HEIGHT: 1'-6"

DOUBLE DUPLEX RECEPTACLE MOUNTING LOCATION: WALL MOUNTING HEIGHT: 1'-6"

LIGHTING

+XX = MOUNTING HEIGHT (IN INCHES)

DUPLEX RECEPTACLE

- THE CONTRACT DOCUMENTS. EXISTING TO REMAIN IS SHOWN WITH SOLID LIGHT LINES. NEW WORK IS SHOWN WITH SOLID HEAVY LINES.
- EX5 INVO TO REMAIN S SHOWN WITH SOLD LIGHT LINES. NEW WORK IS SHOWN WITH SOLD HEAVY LINES. SEE PROJECT SPECIFICATIONS AND DRAWINGS FOR DEVICE/EVEDIMENT RATINGS. ALL MOUNTING HEIGHTS LISTED BELOW OR SHOWN ON DETAIL SHEETS ARE FOR STANDARD INSTALLATIONS. DEVIATIONS TO THE STANDARD MOUNTING HEIGHT MAY OCCUR AS REQUIRED BASED ON FIELD CONDITIONS. IN ALL INSTANCES COORDINATE WITH ARCHITECT AND ENGINEER. ALL DIMENSIONS ARE TO THE CENTER OF BOX. COORDINATE WORK SHOWN IN THIS DRAWING SET WITH ALL PROJECT DRAWINGS.

EQUIPMENT AND WIRING

T TRANSFORMER GEN REMOTE GENERATOR ANNUNCIATOR PUSH BUTTON GROUND CONNECTION GROUND BAR EGS ENGINE GENERATOR SET NON FUSED DISCONNECT SWITCH FUSED DISCONNECT SWITCH AUTOMATIC TRANSFER SWITCH ATS STARTER/DISCONNECT SWITCH CIRCUIT BREAKER DISCONNECT MOTOR CONNECTION MANUAL MOTOR STARTER W/OVERLOAD PROTECTION F = FLUSH MOUNT S = SURFACE MOUNT MMS \$ BRANCH CIRCUIT PANEL DISTRIBUTION PANELBOARD (J JUNCTION BOX HIJ JUNCTION BOX - WALL LIGHTING CONTROL DEVICES SEE LIGHTING CONTROL DEVICE SCHEDULE FOR MORE INFORMATION. LIGHTING CONTACTOR

- $\langle c \rangle$ MOUNTING LOCATION: WALL MOUNTING HEIGHT: AS NOTED
- DIMMER MOUNTING LOCATION: WALL \$ D MOUNTING HEIGHT: 4'-0" PILOT LIGHT SWITCH
- \$_ MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0"
- SINGLE POLE SWITCH s MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0"
- TWO POLE SWITCH \$2 MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0"
- THREE WAY SWITCH MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0" \$,
- FOUR WAY SWITCH MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0" \$,
- OS CEILING MOUNTED OCCUPANCY SENSOR
- DS CEILING MOUNTED DAYLIGHT SENSOR

- KEY SWITCH MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0" °к MOMENTARY CONTACT SWITCH MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0" \$ MC \$_{LV} a,b LOW VOLTAGE SWITCH MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0" a,b = SWITCH DESIGNATION(S)
- REMOTE CONTROL FOR MOTORIZED DAMPER MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0" \$ DM
- SINGLE POLE FIRE PLACE CHIMNEY FAN PROVING SWITCH \$_F MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0"
- SINGLE POLE FIRE PLACE GAS SOLENOID SWITCH \$_G MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0"
- \$_{WP} SINGLE POLE WEATHERPROOF SWITCH MOUNTING LOCATION: WAL MOUNTING HEIGHT: 4'-0"
- OCCUPANCY SENSOR SWITCH MOUNTING LOCATION: WALL MOUNTING HEIGHT: 4'-0" \$_{OS}

SHEET SYMBOLS

N	LINE CONTINUE BREAK	÷	ELEVATION DESIGNATION		
#	DETAILS AND SECTIONS DESIGNATI	ON EQ #	EQUIPMENT DESIGNATION TOP DESIGNATES EQUIPMENT ABBREVIATION BOTTOM DESIGNATES EQUIPMENT		
•	NEW WORK CONNECTION TO EXIST WORK DESIGNATION	ING REFER	PLAN CONTINUATION DESIGNATION		
# X0.0	DETAIL REFERENCE TOP DESIGNAT NUMBER BOTTOM DESIGNATES ON SHEET DETAIL APPEARS	ES DETAIL	REVISION NUMBER DESIGNATION		
#>	KEYED NOTE DESIGNATION				
# X0.0		RENCE TOP DESIGNATES SECTION APPEARS	SECTION NUMBER BOTTOM DESIGNATES O		
# X0.0		SECTION REFERENCE TOP DESIGNATES SECTION NUMBER BOTTOM DESIGNATES WHICH SHEET SECTION APPEARS			



ABBREVIATIONS

bbreviati	ABBREVIATIONS		ABBRI
		Abbreviati	
on A	Description AMPRES	on	FIXTURE
	ABOVE COUNTER TOP	FL	AT FLOOR LINE
	ACOUSTICAL CEILING TILE		FLOOR(ING)
	AMERICANS WITH DISABILITIES ACT		FULL LOAD AMPR
	AMPERE FRAME ABOVE FINISHED FLOOR		FLEXIBLE
	ABOVE FINISHED FLOOR ABOVE FINISHED GRADE		FLOURESCENT FLOW SWITCH
	AMP INTERRUPTING CAPACITY		FUSED SAFETY S
ALT	ALTERNATIVE / ALTERNATE		
	ALUMINUM		GAUGE
	ACCESS PANEL		GALVANIZED
	ARCHITECT / ARCHITECTURAL ABOVE SUSPENDED CEILING		GENERAL CONTR GARBAGE DISPOS
	AMPERE TRIP		GROUND FAULT O
	AUTOMATIC	GFI	GROUND FAULT I
			GROUND FAULT F
	BYPASS AUTOMATIC TRANSFER SWITCH		GREASE INTERCE
	BARE COPPER BELOW FINISHED CEILING		GROUND GALLONS PER MI
	BELOW FINISHED FLOOR		GALVANIZED RIGI
BFL	BELOW FLOOR LEVEL		GROUND TERMIN
	BUILDING		
	BLOCKING		HOSE BIBB
	BOTTOM OF BREAKER		HVAC CONTRACT HUB DRAIN
	BATH TUB		HARDWARE
		HH	HANDHOLE
	CIRCUIT BREAKER		HAND OFF AUTON
			HORSE POWER
	CEILING DIFFUSER CONTRACTOR FURNISHED / OWNER INSTALLED		HEIGHT/HEAT TR/ HIGH VOLTAGE
	CONTRACTOR FURNISHED / CONTRACTOR INSTALLED		HOT WATER SUPP
	CENTER LINE		HOT WATER RECI
	CEILING	HWFU	HOT WATER FIXT
	CLEAR(ANCE)		
	CONCRETE MASONRY UNIT		INSIDE DIAMETER
	CONDUIT CLEAN OUT		INVERT ELEVATIO
	CONCRETE		INTERMEDIATE M
	CONTINUOUS / CONTINUE	INCAND	INCANDESCENT
	CONTRACTOR		INSULATION
	CONTROL RELAY	INT	INTERIOR
	CURRENT TRANSFORMER COLD WATER SUPPLY	IB	JUNCTION BOX
	COLD WATER FIXTURE UNITS		JOINCTION BOX
0111 0		KO	KNOCK-OUT
	DIRECT BURIAL		KITCHEN SINK
	DESIGN BUILD CONTACTOR		KILOVOLT
	DIRECT CURRENT		KILOVOLT-AMPRE
	DEDICATED DETAIL		KILOWATT KILOWATT HOURS
	DRINKING FOUNTAIN		1420111110014
	DIAMETER		LAVITORY
	DISCONNECT		LOCAL AREA NET
	DIVISION DOWN		LIGHT EMITTING I LIGHT FIXTURE
	DISTRIBUTION PANEL		LOCAL TEMPERA
	DISCONNECT SWITCH		LIGHTING
	DOOR UNDER CUT	LTS	LIGHTS
DWGS	DRAWINGS	LV	LOW VOLTAGE
(5) (EXISTING		MASONRY
EXIST	EXISTING		MATERIAL
EC	ELECTRICAL CONTRACTOR		MAXIMUM
	ELECTRIC DUCT BANK		
			MECHANICAL CON
EG	EQUIPMENT GROUND	MCB	MAIN CIRCUIT BR
EG EG	EQUIPMENT GROUND EXHAUST GRILLE	MCB MCC	MAIN CIRCUIT BR MOTOR CONTROL
EG EGS	EQUIPMENT GROUND EXHAUST GRILLE ENGINE GENERATOR SET	MCB MCC MCP	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT F
EG EGS EJ	EQUIPMENT GROUND EXHAUST GRILLE	MCB MCC MCP MDF	MAIN CIRCUIT BR MOTOR CONTROL
EG EGS EJ EL ELEC	EQUIPMENT GROUND EXHAUST GRILLE ENGINE GENERATOR SET EXPANSION JOINT ELEVATION ELECTRIC(AL)	MCB MCC MCP MDF MDF	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY
EG EGS EJ EL ELEC ELEC	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELEVATION ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL)	MCB MCC MCP MDF MDP MECH MG	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTIO MECHANICAL METAL GRATE
EG EGS EJ EL ELEC ELEC ELEC	EQUIPMENT GROUND EXHAUST GRILLE ENGINE GENERATOR SET EXPANSION JOINT ELEVATION ELECTRIC(AL) ELECTRIC(AL) EMERGENCY LIFE SAFETY LIGHTING	MCB MCC MCP MDF MDP MECH MG MH	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTIO MECHANICAL METAL GRATE MANHOLE
EG EGS EJ EL ELEC ELEC ELL ELL	EQUIPNENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELEVATION ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EMERGENCY LIFE SAFETY POWER	MCB MCC MCP MDF MDF MECH MG MH	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MANHOLE MINIMUM
EG EGS EJ ELE ELEC ELEC ELL ELP EMER	EQUIPNENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPRISE GENERATOR SET EXPANSION JOINT ELEVATION ELECTRIC(AL)	MCB MCC MCP MDF MECH MG MHCH MIN MIN	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MANHOLE MINIMUM MAIN LUGS ONLY
EG EGS EJ EL ELEC ELEC ELL ELL ELP EMER EMI	EQUIPNENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELEVATION ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EMERGENCY LIFE SAFETY POWER	MCB MCC MCP MDF MECH MG MH MIN MLO	MAIN CIRCUIT BR MOTOR CONTROU MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MANHOLE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE
EG EGS EJ ELE ELEC ELLC ELLP EMER EMER EMIT EMT EQ	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXONINE GENERATOR SET EXPANSION JOINT ELEVATION ELECTRIC/ELECTRIC/AL) ELECTRIC/ELECTRIC/AL) ELECTRIC/ELECTRIC/AL) EMERGENCY LIFE SAFETY DIGHTING EMERGENCY ELECTROMAGNETIC INTERFERENCE ELECTRIC/AL METALLIC TUBING EDUAL EDUAL EDUAL	MCB MCC MCP MDF MCF MCP MCH MIN MLN MLN MLN MLN MTD	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MANHOLE MINIMUM MAIN LUGS ONLY
EG EGS EJ ELE ELEC ELEC ELLP EMER EMER EMI EMI EQUIP	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELECTRIC(AL) ELECTRIC/ELECTRICAL EMERGENCY LIFE SAFETY LIGHTING EMERGENCY LIFE SAFETY LIGHTING EMERGENCY ELECTRIMAGNETIC INTERFERENCE ELECTRIMAGNETIC INTERFERENCE ELECTRIMAGNETIC INTERFERENCE ELECTRIMAGNETIC INTERFERENCE ELECTRIMAGNETIC INTERFERENCE ELECTRIMAGNETIC INTERFERENCE ELECTRIMAGNETIC INTERFERENCE ELECTRIMAGNETIC INTERFERENCE ELECTRIMAGNETIC INTERFERENCE EQUIPMENT	MCB MCC MDP MDP MCH MG MH MID MID MTD MTD MTD	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MANHOLE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTED MOUNTING MOUNTING HEIGH
EG EGS EJ ELEC ELEC ELEC ELP EMER EMI EMT EQUIP EQUIP ES	EQUIPNENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELEVATION ELECTRICAL ELECTRICAL ELECTRICAL ELECTRICAL EMERGENCY LIFE SAFETY LIGHTING ENERGENCY UIFE SAFETY POWER ENERGENCY UIFE SAFETY POWER ELECTRICAL METALLIC TUBING EQUAL EQUIPMENT EXPLOREMENT EXPLORMENT EXPLORMEN	MCB MCC MCP MDP MCP MG MM MIN MLO MTG MTD MTG MTG HGT	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT MEDIUM DENSITY MAIN DISTRIBUTIK MECHANICAL METAL GRATE MANHOLE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTED MOUNTING MOUNTING HEIGH
EG EGS EJ ELC ELEC ELEC ELLC ELP EMER EMER EMI EQUIP EQUIP ES ETR	EQUIPNENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELECTRICAL ELEVATION ELECTRICAL ELECTRICAL ENERGENCY LIFE SAFETY LIGHTING ENERGENCY LIFE SAFETY POWER ENERGENCY LIFE SAFETY POWER ELECTROMAGNETIC INTERFERENCE ELECTRICAL METALLIC TUBING EOUIPNENT EOUIPNENTENTENTENTENTENTENTENTENTENTENTENTENT	MCB MCC MCP MDP MEPH MGP MI MI MI MI MTT MTT MTT MTT MTT	MAIN CIRCUIT BR MOTOR CIRCUIT I MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MANHOLE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTING MOUNTING MOUNTING HEIGH METAL
EG EGS EJ ELE ELEC ELEC ELEC ELEC EMER EMT EMT EQUIP EQUIP ES ETR EWC	EQUIPNENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELEVATION ELECTRICAL ELECTRICAL ELECTRICAL ELECTRICAL EMERGENCY LIFE SAFETY LIGHTING ENERGENCY UIFE SAFETY POWER ENERGENCY UIFE SAFETY POWER ELECTRICAL METALLIC TUBING EQUAL EQUIPMENT EXPLOREMENT EXPLORMENT EXPLORMEN	MCB MCC MCP MCP MCP MECH MG MH MIN MIN MTD MTD MTD MTD MTTD MTTC HGT HGT	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT MEDIUM DENSITY MAIN DISTRIBUTIK MECHANICAL METAL GRATE MANHOLE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTED MOUNTING MOUNTING HEIGH
EG EGS EJ ELE ELEC ELEL EMER EMT EQUIP EQUIP EQUIP EQUIP ESR ETR EWC EWHR EXP	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELEVATION ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EMERGENCY LIFE SAFETY LIGHTING EMERGENCY ELECTRICAL METALLIC TUBING EOUIPMENT EOUIPMENT EOUIPMENT EQUIPMENT ELECTRICAL WATER COOLER ELECTRICAL WATER COOLER ELECTRICAL WATER MEATER EXPOSED EXPOSED EXPOSED EXPOSED EXPLORED EXPLORE EXPOSED EXPLORE EXPOSED EXPLORED EXP	MCB MCC MCP MDP MECH MG MIN MIN MIN MTD MTD MTT MTT MTT MTT MTT	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTIK MECHANICAL METAL GRATE MINILUGS ONLY MANHOLE MINILUGS ONLY MUSIC AND PAGE MOUNTING MOUNTING HEIGH MOUNTING HEIGH METAL MOTOR / METER MEDUIM VOLTAGI
EG EGS EJ ELE ELEC ELEL EMER EMT EQUIP EQUIP EQUIP EQUIP ESR ETR EWC EWHR EXP	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELEVATION ELECTRICAL) ELECTRICAL ELECTRICAL ELECTRICAL EMERGENCY LIFE SAFETY LIGHTING EMERGENCY UFE SAFETY DWER ENERGENCY LIFE SAFETY DWER ELECTRICAL METALLIC TUBING EQUIAL EQUIPMENT EQUIPMENT EQUIPMENT EQUIPMENT ELECTRICAL WATER HEATER EVENTER ELECTRICAL WATER HEATER EVENTER EV	MCB MCC MCP MCP MCP MCM MF MCM MTD MTD MTD MTD MTD MTD MTT MTL MTT MTL MTR	MAIN CIRCUIT BR MOTOR CIRCUIT I MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MANHOLE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTED MOUNTING HEIGH MOTOR / METER MOTOR / METER MOTOR / METER
EG EGS ELJ ELL ELEC ELLC ELLC EMER EMER EMER EQUIP ES EQUIP ES EVHC EWHC EWHC EXT	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELEVATION ELECTRICAL) ELECTRICAL ELECTRICAL ELECTRICAL EMERGENCY LIFE SAFETY LIGHTING EMERGENCY UFE SAFETY DWER ENERGENCY LIFE SAFETY DWER ELECTRICAL METALLIC TUBING EULATIONAGNETIC INTERFERENCE ELECTRICAL METALLIC TUBING EOUIAL EOUIPMENT EQUIPMENT ELECTRICAL WATER HEATER EXPOSED EXTERIOR E	MCB MCC MCP MDP MECH MG MIN MIN MIN MTG MTG HGT MTG MTG MTG MTG MTG MTG MTG MTG MTC MTC MTC MTC MTC MTC MTC MTC MTC MTC	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MANHOLE MINIMUM MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING METAL MOTOR / METER MEDUIM VOLTAGE NOT APPLICABLE NOT APPLICABLE
EGGE EGS EJJ ELEC ELEC ELEC ELLC ELLC ELLC ELLC	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELEVATION ELECTRIC/ELECTRIC(AL EMERGENCY LIFE SAFETY LIGHTING EMERGENCY UFF SAFETY POWER ELECTRINGANETIC INTERFERENCE ELECTRINGANETIC INTERFERENCE ELECTRINGANENTIC INTERFERENCE ELECTRING TO REMAIN ELECTRICAL WATER COOLER ELECTR	MCB MCC MCP MDP MECH MG MH MIN MIN MTD MTD MTD MTG MTT MTT MTT MTT MTT MTT MTT MTT MTT	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTED MOUNTING M
EGGS EGS EJJ ELE ELEC ELEC ELLC ELLC ELLC ELLC E	EQUIPILENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELEVATION ELECTRICAL) ELECTRICAL) ELECTRICAL) ELECTRICAL VER SAFETY POWER EMERGENCY LIFE SAFETY POWER ELECTRICAL METALLIC TUBING EQUAL EQUIPILENT EQUIPILENT EQUIPILENT EQUIPILENT ELECTRICA WATER COOLER ELECTRIC WATER HEATER EXPOSED EXTERIOR FIRE ALARM FIRE ALARM FIRE ALARM FIRE ALARM EMERGENCE ELECTRICA EVENTEL EXTERIOR EXTER VER VER VER VER VER VER VER VER VER V	MCB MCC MCP MDP MECH MG MH MIN MIN MTD MTG MTG MTG MTG MTG MTG MTG MTG MTG MTC MTR MTR MTR MTR MTR MTR MTR MTR MTR MTR	MAIN CIRCUIT BR MOTOR CIRCUIT I MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MANHOLE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTED MOUNTED MOUNTING HEIGH MOTOR / METAL MOTOR / METAL MOTOR / METAL MOTOR / METAL MOTOR / METAL MOTOR / METAL
EG EG EG EJ EL EL EL EC EL EL EL EL EL EL EL EL EL EL EL EL EL	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELEVATION ELECTRIC/ELECTRIC(AL EMERGENCY LIFE SAFETY LIGHTING EMERGENCY UFF SAFETY POWER ELECTRINGANETIC INTERFERENCE ELECTRINGANETIC INTERFERENCE ELECTRINGANENTIC INTERFERENCE ELECTRING TO REMAIN ELECTRICAL WATER COOLER ELECTR	MCB MCC MCP MDP MECH MG MIN MIN MIN MIN MTG MTG MTG MTG MTG MTG MTG MTG MTG MTG	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTED MOUNTING M
EG EG EG EJ EL EL EL EC EL EC EL EL EL EC EL EL EL EC EL EC EL EC EL EC EL EC EL EC EL EC EL EC EL EC EL EC EL EC EL EC EL EC EL EC EL EC EL EL EL EL EL EL EL EL EL EL EL EL EL	EQUIPINENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ENERGENCY LIFE SAFETY LIGHTING ENERGENCY LIFE SAFETY POWER ENERGENCY LIFE SAFETY POWER ELECTRIC(AL METALL)C TUBING EOUIPINENT EOUIPINENT EOUIPINENT EOUIPINENT EOUIPINENT ELECTRIC(AL WATER COOLER ELECTRIC(AL WATER COOLER ELECTRIC(AL WATER COOLER ELECTRIC(AL METALL) EXISTING TO REMAIN ELECTRIC(AL MATER HEATER EXPOSED EXTERIOR FIRE ALARM FIRE ALARM FIRE ALARM FIRE ALARM ENTER FIRE FIRE EXPOSED EXTERIOR EXTERICAL	MCB MCC MOP MECH MG MH MIN MIN MIN MTD MTT MTT MTT MTT MTT MTT MTT MTT MTT	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MANHOLE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTING NOT APPLICABLE NOT APPLICABLE NOT APPLICABLE NOT APPLICABLE NOT MICONTRAC
EG EG EG EJ EJ EL EL EL EL EL EL EL EL EL EL EL EL EL	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELEVATION ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EMERGENCY LIFE SAFETY LIGHTING EMERGENCY ELECTRIC(AL) EQUIPMENT EQUIPMENT EQUIPMENT EQUIPMENT EQUIPMENT EQUIPMENT ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EQUIPMENT EQUIPMENT EQUIPMENT EQUIPMENT ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EQUIPMENT EQUIPMENT EQUIPMENT ELECTRIC(AL) ELECTR	MCB MCC MCP MDP MECH MG MH MIN MTD MTG MTT HTL MTT MTT MTT MTT MTT MTT MTT MTT NTT MTT M	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT MEDIUM DENSITY MAIN DISTRIBUTI METAL GRATE MINIMUM MAIHOLE MINIMUM MAIHOLE MOUNTED MOUNTED MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUTING MOUTING MOUTING MOUTING MOUTAGI MOTOR / METER MEDUIM VOLTAGI NOT APPLICABLE NON-FUSED DISC NON-FUSED DI
EG EG EG EJ EL EL EL EL EL EL EL EL EL EL E E E E	EQUIPILENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EMERGENCY ELECTRIC(AL) EXHERGENCY ELECTRIC(AL) EXHERGENCY ELECTRIC(AL) EXHERGENCY ELECTRIC(AL) EXHERGENCY ELECTRIC(AL) EXHERGENCY ELECTRIC(AL) EXHERGENCY	MCB MCC MOP MOP MECH MG MH MIN MIN MTD MTT MTT MTT MTT MTT MTT MTT MTT MTT	MAIN CIRCUIT BR MOTOR CIRCUIT I MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MAIHOLE MINIMUM MAIHOLE MINIMUM MAIHOLE MINIMUM MUSIC AND PAGE MOUNTED MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOTOR / METER MOTOR / METER MOTOR / METER MOTOR / METER NOT APPLICABLE NOT APPLICABLE NON-FUSED DISC NON-FUSED DISC NON FUSED DISC NON FUSED DISC NON FUSED DISC NON FUSED DISC NON FUSED DISC NOT IN CONTRAC
EG EG EG EG EJ EL EL EL EL EL EL EL EL EL EL EL EL EL	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EXHERGENCY LIFE SAFETY LIGHTING ENERGENCY LIFE SAFETY POWER EXHERGENCY LIFE SAFETY POWER ELECTRIC(AL METALLC TUBING EOUIPMENT EOUIPMENT EOUIPMENT EOUIPMENT SUPPLIER EXISTING TO REMAIN ELECTRIC(AL WATER COOLER ELECTRIC(AL WATER COOLER ELECTRIC(AL WATER COOLER ELECTRIC(AL MATER VELATER EXPOSED EXTERIOR FIRE ALARM CONTROL PANEL FICOOR CLEANOUT FAN COIL UNIT FICOOR PANIN FEEDER FUSED DISCONNECT SWITCH FIRE EXTINGUISHER (CABINET)	MCB MCC MCP MDP MECH MIN MIN MIN MTG MTG HGT MTG MTG HGT MTG MTG NTG NTG NTG NTG NTG NTG NTG NTG NTG N	MAIN CIRCUIT BR MOTOR CONTROL MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTING MONFUSED SAFE NOT IN CONTRAC NIGHT LIGHT NORMALLY OPEN NOT TO SCALE ON CENTER(S)
EG EGS EJJ ELE ELEC ELEC ELEC ELLC ELLC ELLC E	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRICAL, ELEVATION ELECTRICAL, ELEVATION ELECTRICAL, ELEVATION EMERGENCY LIFE SAFETY POWER EMERGENCY LIFE SAFETY POWER ELECTRICAL METALLIC TUBING EQUIPMENT EQUIPMENT SUPPLIER EXISTING TO REMAIN ELECTRICAL WATER COOLER ELECTRICAL MATER COOLER ELECTRICAL WATER COOLER ELECTRICAL MATER COOLER ELECTRICAL WATER COOLER EXTORS FIRE ALARM FIRE ALARM CONTROL PANEL FILCOR DERAUNT FAN CONTROL PANEL FILCOR CLEVATION FIRE ELEVATION FIRE (ELAMIN) FIRE ELECTNINGUISHER (CABINET) FINSH FLOOR (ELEVATION)	MCB MCC MCP MCP MCP MECH MG MTD MTD MTD MTD MTD MTD MTD MTD MTD MTD	MAIN CIRCUIT BR MOTOR CIRCUIT I MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTIRG MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING HEIGH MOUNTING HEIGH MOUNTING HEIGH MOUNTING HEIGH MOUNTING HEIGH MOUNTING HEIGH MOUNTING HEIGH NOT APPLICABLE NOT APPLICABLE NOT APPLICABLE NOT NCONNECTIO NON-FUSED DISC NON-FUSED DISC NON-FUSED DISC NON-FUSED DISC NON-FUSED DISC NON-FUSED DISC NON-FUSED DISC NOT IN CONTRAC NIGHT LIGHT NORMALLY OPEN NOT NO SCALE ON CENTER(S) OUTSIDE DIAMET
EG EG EG EJ EL EL EL EL EL EL EL EL EL EL EL EL EL	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EXHERGENCY LIFE SAFETY LIGHTING ENERGENCY LIFE SAFETY POWER EXHERGENCY LIFE SAFETY POWER ELECTRIC(AL METALLC TUBING EOUIPMENT EOUIPMENT EOUIPMENT EOUIPMENT SUPPLIER EXISTING TO REMAIN ELECTRIC(AL WATER COOLER ELECTRIC(AL WATER COOLER ELECTRIC(AL WATER COOLER ELECTRIC(AL MATER VELATER EXPOSED EXTERIOR FIRE ALARM CONTROL PANEL FICOOR CLEANOUT FAN COIL UNIT FICOOR PANIN FEEDER FUSED DISCONNECT SWITCH FIRE EXTINGUISHER (CABINET)	MCB MCC MCP MCP MCP MECH MIN MIN MIN MTD MTT MTT MTT MTT MTT MTT MTT MTT MTT	MAIN CIRCUIT BR MOTOR CIRCUIT I MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MAIHOLE MINIMUM MAIHOLE MINIMUM MAIHOLE MINIMUM MUSIC AND PAGE MOUNTED MOUNTING HUSIC AND PAGE MOUNTING HUSIC AND PAGE MOUNTING HEIGH MOTOR / METER MOTOR / METER MOTOR / METER MOTOR / METER NOT APPLICABLE NOT APPLICABLE NON-FUSED DISC NON-FUSED DISC NOT SCALE ON CENTER(S) ON CENTER(S)
EG EG EGS EJ ELEC ELEC ELEC ELEC ELEC ELEC ELEC E	EQUIPILENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) ELECTRIC(AL) EXHERGENCY ELECTRIC(AL) EXHERGENCY ELECTRIC(AL) EXHERGENCY ELECTRIC(AL) EXHERGENCY ELECTRIC(AL) EXHERGENCY EXHERGENC	MCB MCC MCP MDP MECH MIN MIN MIN MIN MTG MTG MTG MTG MTG MTG MTG MTG NTG NTG NTG NTG NTG NTG NTG NTG NTG N	MAIN CIRCUIT BR MOTOR CIRCUIT I MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MAIN LUGS ONLY MUSIC AND PAGE MOUNTIRG MOUNTING MOUNTING MOUNTING MOUNTING HEIGH MOUNTING HEIGH METAL NOT APPLICABLE NOT APPLICABLE NOT APPLICABLE NOT APPLICABLE NOT N-FUSED SAFC NON-FUSED DISC NON-FUSED DISC NOT IN CONTRAC NIGHT LIGHT NORMALLY OPEN NOT NO SCALE ON CENTER(S) OUTSIDE DIAMET
EG EG EG EJ EJ EL EL EL EL EL EL EL EL EL EL EL EL EL	EQUIPMENT GROUND EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXHAUST GRILLE EXPANSION JOINT ELECTRIC(AL) ELECTRICAL ELEVATION ELECTRICAL EMERGENCY LIFE SAFETY LIGHTING EMERGENCY EM	MCB MCC MCP MDP MECH MGP MECH MIN MIN MIN MTG MTG HGT MTG MTG MTG MTG NTG NTG NTG NTG NTG NTG NTG NTG NTG N	MAIN CIRCUIT BR MOTOR CIRCUIT I MOTOR CIRCUIT I MEDIUM DENSITY MAIN DISTRIBUTI MECHANICAL METAL GRATE MINIMUM MAIHOLE MINIMUM MAIHOLE MINIMUM MUSIC AND PAGE MOUNTED MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING MOUNTING NOT APPLICABLE NOT APPLICABLE NOT APPLICABLE NOT NCONTRAC NIGHT LIGHT NOT NC SCALE ON CENTER(S) OUTSIDE DIAMET OPEN ENDED DUK



J

WALL MOUNTED LINEAR MOUNTING LOCATION: WALL	0		RECESSED MOUNTED MOUNTING LOCATION: CEILING
SURFACE LINEAR MOUNTING LOCATION: CEILING	-0-	÷	WALL MOUNTED MOUNTING LOCATION: WALL
	-¢-		PENDANT AND/OR SURFACE
SUSPENDED LINEAR MOUNTING LOCATION: CEILING	٠		SHADING DESIGNATES EMERGENCY POWER SOURCE (EITHER GENERATOR OR BATTERY)
RECESSED GRID MOUNTED MOUNTING LOCATION: CEILING			· · · · · · · · · · · · · · · · · · ·
		~	DIRECTIONAL INDICATOR ADDED TO FIXTURE SYMBOL
1x4 RECESSED GRID MOUNTED MOUNTING LOCATION: CEILING			TO DESIGNATE AIMING DIRECTION
MODIVING LOCKTION. CEILING		\uparrow	EXIT DIRECTIONAL ARROW
2x2 RECESSED GRID MOUNTED MOUNTING LOCATION: CEILING		۲	SINGLE FACE EXIT FIXTURE MOUNTING LOCATION: CEILING
SURFACE MOUNTED MOUNTING LOCATION: CEILING		0	DOUBLE FACE EXIT FIXTURE MOUNTING LOCATION: CEILING
		0 L	SINGLE FACE EXIT FIXTURE MOUNTING LOCATION: WALL
SHADING DESIGNATES EMERGENCY POWER SOURCE (EITHER GENERATOR OR BATTERY)		₽	MOUNTING LOCATION: WALL DOUBLE FACE EXIT FIXTURE

+XX (TV)

TRACK MOUNTING LOCATION: AS NOTED (PE) PHOTO EYE MOUNTING LOCATION: WALL

	Θ	MOUNTING LOCATION: CEILING
OWER	Ŷ	SINGLE FACE EXIT FIXTURE MOUNTING LOCATION: WALL
TERY)	Ð	MOUNTING LOCATION: WALL DOUBLE FACE EXIT FIXTURE
)	Ŕ	DOUBLE HEAD EMERGENCY FIXTURE WITH BATTERY PACK

DUPLEX RECEPTACLE - CEILING MOUNTED MOUNTING LOCATION: CEILING

DUPLEX RECEPTACLE, UPPER HALF SWITCHED

TV OUTLET WITH 120V DUPLEX RECEPTACLE IN COMMON 2 GANG BOX WITH LOW VOLTAGE BARRIER AND COMMON PLATE. TV BY OWNER. MOUNTING LOCATION: WALL

FIXTURE TYPE CIRCUIT TAG

SWITCH DESIGNATION

FLOOR BOX (OR POKE THROUGH) MOUNTING LOCATION: FLOOR

LOWER HALF HOT MOUNTING LOCATION: WALL MOUNTING HEIGHT: 1'-6"

+XX = MOUNTING HEIGHT (IN INCHES)

D-

REVIATIONS		ABBREVIATIONS
Description	Abbreviati on	Description
PRES		POLE PULL BOX / PUSHBUTTON
1120	PC	PLUMBING CONTRACTOR / PHOTOCELL
		POWER DISTRIBUTION UNIT POWER FACTOR
SWITCH	PH	PHASE
		PLUMBING PLYWOOD
		PANEL
TRACTOR		POWER OPERATED DAMPER
POSER T CIRCUIT INTERRUPTER		POWER ROOF VENTILATOR POTENTIAL TRANSFORMER
T INTERRUPTER	PTRV	POWER TYPE ROOF VENTILATOR
T PROTECTION CEPTOR	PWR	POWER
	QTY	QUANTITY
MINUTE IGID CONDUIT		RISER
INAL BOX		RADIUS
	RCP	RECIRCULATING PUMP
CTOR		ROOF DRAIN RECESSED
	RECEPT	RECEPTACLE
		REFERENCE / REFER TO RELOCATE
OMATIC		REQUIRED
{		REMOTE GROUND INDICATOR PANEL
TRACE		ROUGH OPENING ROOF TOP UNIT
JPPLY		
ECIRCULATION PUMP		SUPPLY DIFFUSER
XTURE UNITS		SQUARE FOOT/FEET SHOWER
ER	SHT	SHEET
TION		SIGNAL SIMILAR
MEATL CONDUIT		SUMP PUMP
T		SPECIFICATION
		SAFETY SWITCH / STAINLESS STEEL STATION
		STARTER
		SUSPENDED
		SWITCH SWITCHBOARD
	SWGR	SWITCHGEAR
RES	SYST	SYSTEM
NE0	TCC	TEMPERATURE CONTROLS CONTRACTOR
JRS	TCP	TEMPERATURE CONTROL PANEL
		TELEPHONE TO FLOOR ABOVE
ETWORK	TFB	TO FLOOR BELOW
G DIODE		TRANSFER GRILLE TOP OF BEAM
RATURE CONTROL PANEL		TOP OF CONCRETE
		TOP OF DECK
		TOP OF DUCTBANK TOP OF FOOTING
	TOMH	TOPP OF MANHOLE
		TOP OF PARAPET TOP OF STEEL
		TOP OF WALL
CONTRACTOR		TOILET PARTITION
BREAKER ROL CENTER		TAMPER SWITCH / TIME SWITCH TELEVISION VOLTAGE SURGE SUPRESSION
IT PROTECTOR		TYPICAL
TY FIBERBOARD		
ITION PANEL		UNDER COUNTER UNDER FLOOR
		UNDERGROUND
		UNIT HEATER UNLESS NOTED OTHERWISE
LY		UNINTERRUPTED POWER SUPPLY
GE TERMINAL BOX	V	
		VOLTAGE / VENT VARIABLE FREQUENCY DRIVE
GHT	VIF	VERIFY IN FIELD
		VAPOR PROOF VENT STACK
R		VENT THROUGH ROOF
AGE		
LE		WIRE
FION/NORMALLY CLOSED	WC	WATER CLOSET
SCONNECT FETY SWITCH	WCO	WALL CLEANOUT
ACT		WALL HYDRANT WATER HEATER
	WP	WEATHERPROOF
EN		WATER SOFTENER WATER TIGHT
·	J L WT	WATER IIGHT
		TRANSFORMER
ETER	XP	EXPLOSION PROOF
SHED / CONTRACTOR INSTALLED	#	NUMBER / POUND
SHED / OWNER INSTALLED	&	AND
UPPER	@	AT
	1	



ELECTRICAL GENERAL NOTES:

- 3

\bigcirc 2

2



GENERAL NOTES AND KEYED NOTES ARE TYPICAL ACROSS ALL DRAWINGS. KEYED NOTES MAY NOT BE USED OR BE APPLICABLE TO ALL DRAWINGS. REFER TO DRAWING E001 FOR ADDITIONAL NOTES.

HANDHOLES AND PULL BOXES ARE NOT SPECIFICALLY SHOWN. ELECTRICAL CONTRACTOR SHALL PROVIDE HANDHOLES AND PULL BOXES MOUNTED TO POLES, QUANTITIES AND SIZES AS REQUIRED BY THE NEC AND FOR A COMPLETE AND FUNCTIONAL INSTALLATION. IF HANDHOLES ARE PROVIDED THEY SHALL BE LOCATED OUTSIDE OF THE BALL DIAMOND FENCE LINES TO NOT IMPEDE PLAY.

ALL CONDUIT ROUTES ON THIS DRAWING ARE SCHEMATIC ONLY, ELECTRICAL CONTRACTOR SHALL PROVIDE ALL CONDUITS, INCLUDING THOSE NOT SPECIFICALLY SHOWN, FOR A COMPLETE AND FUNCTIONAL INSTALLATION.

ELECTRICAL KEYED NOTES:

SCOREBOARD. PROVIDE NEW LOCKABLE DISCONNECTING MEANS IN ACCORDANCE WITH THE NEC. DISCONNECTING MEANS NOT SPECIFICALLY SHOWN.

ELECTRICAL BUILDING, PROVIDE NEW PANELBOARD AS SCHEDULED ON 6701, NEW NEMA 5-20R RECEPTACLE WITH GFCI PROTECTION, SURFACE-MOUNTED STRIP LIGHT (LITHONIA CSS-148-4000LM-MVOLT-35%-80CR), CR EQUAL), WITH TOGGLE ON-OFF WALL SWITCH, LIGHTING TIMECLOCK (WITERMATIC TOIN, OR EQUAL), TYPE 2 SPD (L-*: 7001), L-1: 1001, X: 1001, X: 1001, AD (AN ANIMAL RATING, EATON, SQUARE D, LEVITON, LIBERT, ABB, OR EQUAL CONFIRM CIRCUIT BREAKER SIZE IN PANELBOARD WITH SPD MANUFACTURER PRIOR TO ORDERING) AND INSTALL MANUFACTURER-FURSHED BALL DIMOND LIGHTING CONTRCL PAREL WITH LIGHTING CONTACTORS. EXISTING EQUIPMENT AND FEEDERS SHALL TE INTO EXISTING CIRCUITS AS SHOWN IN THE PANEL SCHEDULE ON E701.

3 PROVIDE GROUND ROD AND #6 GROUND WIRE CONNECTOR TO GROUND ROD AND POLE TO GROUND LIGHT POLE IN ACCORDANCE WITH THE

(4) BID ALTERNATE: INSTALL OWNER-FURNISHED LIGHT POLES FOR EXTERIOR ATHLETIC LIGHTING.





GENERAL NOTES AND KEYED NOTES ARE TYPICAL ACROSS ALL DRAWINGS. KEYED NOTES MAY NOT BE USED OR BE APPLICABLE TO ALL

HANDHOLES AND PUIL BOXES ARE NOT SPECIFICALLY SHOWN. ELECTRICAL CONTRACTOR SHALL PROVIDE HANDHOLES AND PUIL BOXES MOUNTED TO POLES, QUANTITIES AND SIZES AS REQUIRED BY THE NEC AND FOR A COMPLETE AND FUNCTIONAL INSTALLATION. IF HANDHOLES ARE PROVIDED THEY SHALL BE LOCATED OUTSIDE OF THE BALL DIAMOND FENCE LINES TO NOT IMPEDE PLAY.

ALL CONDUIT ROUTES ON THIS DRAWING ARE SCHEMATIC ONLY. ELECTRICAL CONTRACTOR SHALL PROVIDE ALL CONDUITS, INCLUDING THOSE NOT SPECIFICALLY SHOWN, FOR A COMPLETE AND FUNCTIONAL INSTALLATION.

SCOREBOARD. PROVIDE NEW LOCKABLE DISCONNECTING MEANS IN ACCORDANCE WITH THE NEC. DISCONNECTING MEANS NOT SPECIFICALLY

ELECTRICAL BUILDING: PROVIDE NEW PANELBOARD AS SCHEDULED ON E701, NEW NEMA 5-20R RECEPTACLE WITH GFCI PROTECTION, SURFACE-MOUNTED STRIP LIGHT (LITHONIA CSS-L48-4000LM-MVOLT-36K-80CRI, OR EQUAL) WITH TOGGLE ON-OFF WALL SWITCH, LIGHTING TIMECLOCK (INTERNATIC TIOI, OR EQUAL), TYPE 2 SPD (L-N: 700V, L-G: 1000V, L-L: 1000V, SCCR: 100K, 20 KA NOMINAL RATING, EATON, SQUARE D, LEVITON, LIEBERT, ABB, OR EQUAL CONFIRM (CROUTI BREAKER SZE IN PANELBOARD WITH SPD MANUFACTURER PRIOR TO ORDERING) AND INSTALL MANUFACTURER-FURNISHED BALL DIAMOND LIGHTING CONTROL PANEL WITH LIGHTING CONTACTORS. EXISTING EQUIPMENT AND FEEDERS SHALL TIE INTO EXISTING CIRCUITS AS SHOWN IN THE PANEL SCHEDULE ON E701.

PARKITECTURE	DNII	4 901 Deming Way, Suite 102	Parkitecture 608.886.6808
C		ICONICA	901 DEMING WAY // MADISON, WI 53717 Ph: 608,684,3500 // F.Y. 605,664,3535 Iconicacreates.com
Project Name: MENTY PARK IMPROVEMENTS		2 RIDGEWAY, WI 53582	
Project : Issued F Date: Sheet Num	or:	09/	35700 /08/23

ELECTRICAL DEMOLITION GENERAL NOTES:

- 1.
- 3.
- 5. CONSTRUCTION.

ELECTRICAL DEMOLITION KEYED NOTES:

- $\langle 1 \rangle$



REMOVE ALL EXISITING ELECTRICAL EQUIPMENT AND MATERIALS ASSOCIATED WITH THE ITEMS BEING REMOVED AND SHOWN ON THIS DRAWING, AS WELL AS ALL EXISTING ELECTRICAL DEVICES, MATERIALS, AND EQUIPMENT NOT BEING REUSED.

2. PATCH ALL STRUCTURE HOLES AND DEFORMATIONS RESULTING FROM DEMOLITION WORK TO MATCH EXISTING INSTALLATION.

RECONNECT OR ALTER AND EXISTING EXISTING CONDUIT AND WIRING FOR EQUIPMENT AND CIRCUITS EXISTING TO REMAIN TO MAINTAIN EXISTING FUNCTIONALITY.

4. THE LOCATIONS AND ELEVATIONS OF EQUIPMENT AND COMPONENTS ARE APPROXIMATE. CONTRACTOR SHALL FIELD VERIFY THE LOCATIONS AND ELEVATIONS OF ALL EXISTING EQUIPMENT AND COMPONENTS IN ORDER TO PERFORM THE CONSTRUCTION SHOWN ON THESE DRAWINGS.

NOT ALL EXISTING EQUIPMENT, PIPING, OR CONDUIT SHON. EXISTING EQUIPMENT AND COMPONENTS WERE OBTAINED FROM FIELD MEASUREMENTS, CONTRACTOR IS RESPONSIBLE FOR VERIFYING THE ACCURACY OF EXISTING INFORMATION AS REQUIRED FOR NEW

REMOVE LIGHT FIXTURES FROM POLE AND ALL ASSOCIATED APPURTENANCES BACK TO SOURCE. POLE SHALL REMAIN FOR INSTALLATION OF NEW LIGHT FIXTURES AND GROUNDING SYSTEM. (POLE SHALL REMAIN AS PART OF BASE BID, REMOVAL OF POLE AND ASSOCIATED APPURTENANCES SHALL OCCUR IN THE BID ALTERNATE AS NOTED ON THESE DRAWINGS.)

C ELECTRICAL BUILDING. REMOVE PANELBOARD, BALL DIAMOND LIGHTING CONTROL PANEL, GROUND GRID, WALL SWITCH, RECEPTACLE, LIGHTING TIMECLOCK, EXTERIOR PHOTOCELL, AND ALL ASSOCIATED CONDUIT, WIRING, UTILITY METER AND ASSOCIATED APPURTENANCES. COORDINATE UTILITY EQUIPMENT WITH UTILITY COMPANY. EQUIPMENT SHALL BE REPLACED IN KIND AS SHOWN ON E100 OR E100A IF THE BID ALTERNATE IS ACCEPTED.

3 SCOREBOARD. REMOVE ASSOCIATED DISCONNECT SWITCH AND CONDUIT AND WIRING BACK TO SOURCE. PREPARE FOR INSTALLATION OF NEW CONDUIT, WIRING, AND DISCONNECT SWITCH AS SHOWN ON E 100 OR E 1000 IF THE BID ALTERNATE IS ACCEPTED.

4 BID ALTERNATE: REMOVE LIGHT POLES AND ASSOCIATED APPURTENANACES. PADDING ON LIGHT POLES SHALL BE SAVED FOR REINSTALLATION ON NEW POLES WHERE SHOWN ON E100A.

Project Issued I Date: Sheet Nu		E		PARKITECTURE
For: (+ PLANNING
09/08	RIDGEWAY, WI 53582	ICONICA	-	901 Deming Way, Suite 102
235700 8/2023 845		901 DEMING WAY // MADISON, WI 53717 Ph: 608.664.3500 // Fx: 608.664.353 iconicacreates.com	Parkitecture	Madison.WI.53717 8

			LIGHTIN	IG FIX	TURE	SCH	EDULE	
TYPE.	DESCRIPTION	MANUFACTURER	MODEL NUMBER	VOLTAGE	WATTS	LAMP	MOLINTING	NOTES
A	BALL DIAMOND LIGHTING	WILL - WISCONSIN LIGHTING LAB	WS-K08940-67-MV-6N-BK-APY-WHP15NP w/WS- PDH & WS-RP0E-8W-X-MV-LG-PBS & NP-4-ACR- BTP-4/2-STUD-180-SBF	240V	5.304 KW	LED	POLE	LIGHT FORTURES SHALL BE MOUNTED ON ENSTING POLEUSING MANUFACTURES RECOMMENDED HARDWARE HARDWARE SHALL BE PRIVIDED BY ELECTRICAL CONTRACTOR
8	BALL DIAMOND LIGHTING.	WILL - WISCONSIN LIGHTING LAB	WS-KE8946-57-MV-5N-BK-APY-WHP15NP w/ WS- POH & WS-RPCE-8W-X-4V-LG-PES & NP-A-ACR- BTP-4/2-STLD-180-SBF	2401/	S 302 AVV	LED	POLE	LIGHT FIXTURES SHALL BE MOUNTED ON EXISTING POLEUSING MANUFACTURER- RECOMMENDED HARDWARE. HARDWARE SHALL BE PROVIDED BY ELECTRICAL CONTRACTOR
G	HALL DIAMOND LIGHTING	WILL - WISCONSIN LIGHTING LAB	WS-K08940-57-MV-5N-BK-APY-WHP15NP w/ WS- PDH & WS-RPCE-8W-X-MV-LG-PBS & NP-A-ACR- BTP-4/2-STUD-180-S8F	240V	3 536 NW	LED	POLE	LIGHT FOTURES SHALL BE MOUNTED ON EXISTING POLEUSING MANUFACTURER- RECOMMENDED HARDWARE HARDWARE SHALL BE PROVIDED BY ELECTRICAL CONTRACTOR

				LIG	HTIN	G PA	NEL						
Senice: 120/240V, 16 Main Breaker: 400A MCE Location: Electrical Built	sing				Enclosure						Mounting Main Bus: SCIC	Surf. Cap 10 to	per
Room Number/Description	Amps	Poles	Ckt.#	Phase A	Phase B	Phase A	Phase B	Ckt.#	Amps	Poles	Room Number/Descrip	tion	
Sheter Building *	100	2	1 3	0	0	0	0	2	20	2	Basketball Lights *		
Yard Light."	15	1	6	0	a manual d	0		6	15	1	Contact B-Ball *		_
Temp Power - Stage *	30	2	7	0	Q	0	0	8	100	2	Food Stand *		
GFI n Dugout *	20	1	- 11	1.	0	Concession of	0	12	20	1	Convenience Outlets *		
Unlabeled Circuit Breaker *	100	2	13	0	1	0	-	14	20	1	Convenience Outlets *		
Unabeled Circuit Dreaker	100	1	15	1000	0		0	16	20	1	Convenience Outlets *		_
Ball Diamond Lighting Contactor (A)	30	2	17	2652	2652	1768	1768	18 20	30	2	Ball Diamond Lighting Contactor (C)		
	30		21	2652	2032	1768	1100	22	30		B. 0. D	La provide a la constante a la constante da la constan	
Ball Diamond Lighting Contactor (A)	-30	2	23	and the second	2652	Contraction in the	1763	24	30	2	Ball Diamond Lighting Contactor (C)		
Ball Diamond Lighting Contactor (B)	30	2	25	2652		1768	1	26	30	2	Ball Diamond Lighting Contactor (C)		
Demonstration of Sector (C)		-	27	1000	2652		1768	28		*	Dan Diamona Eighning Contactor (C	-	
Ball Diamond Lighting Contactor (B)	30	2	29	2652	all the	1768	4.765	30	30	2	Ball Diamond Lighting Contactor (C)		
Ball Diamond Lighting Centrol Panel Control Circuit	20		31	1000	2652	0	1768	32	-				
Spike	20	1	35	1000	0		0	36	30	2	Surge Protection Device (SPD) +		
Space		1	37	0	Constantion of the	0	Constant of the local division of the	38		1	Space		_
Space		1	39	Party and	0	-	0	40		1	Space		_
Space	- 4. L	- t	-41	0		0	-	42	1.	 4) 	Space		
Total Load per Phase per Side (VA)		-		11608	10608	7072	7072	-	-	_			-
Total Load Phase A (VA)		18680	VA		circuit. Loa	d not accou	inted for in	Total Co	nnected	Load		152	1
Total Load Phase B (VA)		17680	VA	feeder loa				Total Connected Load + 25%				189	1
Total Connected Load (VA)		36360	VA	+ Confirm	SPD break	er size with	mfr	Spare 25%				47	+
	-		-					Feeder	Load			237	- 1







Iowa County, Wisconsin

To: Park and Recreation Commission/Board of Trustees **RE**: Bike Racks for Park **From:** Hailey Roessler, Clerk/Treasurer

September 11, 2023

The Ridgeway Community Park currently has one bike rack at the trail head (by the Reeson memorial bench and kiosk/picnic table north of Larson's barn, *pictured below*) with 26 holes for bikes. The Vibrant Spaces Grant, and Ridgeway Farmer's Market Grant award from Iowa County Tourism, requested funds for a bike rack.



Utilizing fund requests from Ridgeway Farm Market's Tourism Grant from Iowa County for \$700 the Park and Rec Commission recommended purchasing an additional black double sided bike rack to be placed by the dumpster.

Double-Sided Grid Bike Rack - 18 Bike Capacity, Black



Everybody's biking! Secures bikes, deters theft. For commuter stations, libraries, schools.

- Sturdy 14-gauge construction, attractive powder coating.
- Rust-resistant.
- Mounting hardware included.

A More Images

							SPECIFY	
MODEL	DESCRIPTION	SIZE	BIKE WT. PRICE EACH		COLOR	IN STOCK		
NO.	DESCRIPTION	L x W x H	CAPACITY	(LBS.)	1	3+	COLOR	SHIPS TODAY
H-2541BL	Double-Sided	110 x 36 x 31"	18	112	\$650	\$620	🔳 Black 👻	1 ADD

SHIPS UNASSEMBLED VIA MOTOR FREIGHT

Village of Ridgeway | 208 Jarvis Street | Suite A | Ridgeway, WI 53582 Item 10.



The below blue wave racks were donated to the Village and picked up by Ridgeway Public Works. They do need to be installed in concrete. Trustee Julene Garner is considering this as part of the village green planning.



Possible locations:

Village Green Area:

People do park here to unload bikes and utilize the trail. This was indicated in the Vibrant Spaces Grant as the location. I have seen people using the trees as a resting spot for their bikes until they are ready.





Item 10.

Park Restrooms:

DNR wasn't sure what purpose the bollards continued to serve or who put them in. The Friends didn't know either, but they are the ones that painted them last. My understanding was that the village put them in to limit traffic from driving through there/crossing the bike trail/raised path in front of the depot.







Village of Ridgeway | 208 Jarvis Street | Suite A | Ridgeway, WI 53582











Recycling & Waste Collection



1120 Broadway • Platteville, WI 53818 608-348-9586

September 5, 2023

Village of Ridgeway C/O Hailey Roessler, Clerk 208 Jarvis St Ridgeway, WI 53582 RECEIVED SEP 11 2023 Village of Ridgeway

RE: Recycling and Solid Waste Collection Renewal Proposal

Village Board,

Thank you for the opportunity to provide recycling and waste service for the Village of Ridgeway, we greatly appreciate the Village's continued business.

Our contract with the Village ends on December 31, 2023, at this time I would like to ask the Board to consider a 5 year contract renewal.

Weekly Recycling and Solid Waste Service 5 year contract term January 1st, 2024 – December 31st, 2028

Totals	\$14.47 per unit, per month
Solid Waste	\$8.81
Recycling	\$5.66
	<u>2024</u>

The rate of solid waste and recycling collection will remain at \$14.47 per residential household until the end of the 2024 calendar year. On January 1st, 2025, 2026, 2027, and 2028 a rate increase of a maximum of 3% may be applied.

Items that are not considered solid waste or recyclables for weekly collection are: leaves, grass clippings, garden waste, brush, wood, cement, construction and remodeling material, furniture, fencing, major appliances, waste tires, waste oils, lead acid batteries, and electronics.



Services Included with Proposal:

- Spring and Fall Village designated clean ups each calendar year.
- Tire and appliance collection would be coordinated with clean up days, and the Village Clerk fees apply.
- Complete preparation of all required State Recycling Grant Applications and Program Accomplishment forms.
- Containers will be donated for any designated community events.
- Recycling instruction and information card will be provided each year to the Village for any direct mailing they may choose.

Again, I would like to thank you for this opportunity, and look forward to the possibility of continuing to provide your Community with recycling and solid waste service.

Sincerely In Ed Faherty

Vice President



Iowa County, Wisconsin

RESOLUTION NO. 2023-08

A RESOLUTION OF AUTHORIZATION AND SUPPORT OF AN APPLICATION FOR ROADWAY IMPROVEMENTS OF CTH HHH BETWEEN LEVEL STREET AND USH 18/151 INTERSECTIONS

Relating to the County of Iowa's participation in various grant application opportunities within the Wisconsin Department of Transportation 2024-2029 Surface Transportation Program – Rural (STP-Rural), or other grant opportunities with the Bipartisan Infra-structure Bill or other discretionary programs from the Federal government.

WHEREAS, State and Federal monies are available under the 2024-2029 Surface Transportation Rural Roads Program and other state / federal funding opportunities to be used for the construction, planning, and design County Trunk Highway facilities for pedestrians, bicyclists, and all other forms of transportation; And,

WHEREAS, after due consideration, the Iowa County Highway Department has recommended that an application be submitted to the State of Wisconsin for the following project:

CTH HHH Reconstruction - Level Street to USH 18 / 151 Intersections, Ridgeway, WI; And,

WHEREAS, it is necessary for the lowa County Highway Department to prepare the application to provide federal funding for the cost for design and construction of improvements to CTH HHH in the Village of Ridgeway. The project will include the following types of work: Complete reconstruction including Intersection improvements, parking lanes, unclassified excavation earthwork, concrete removals, base course, asphalt pavement, curb and gutter, storm sewer and appurtenances, concrete sidewalk, street lighting, landscaping items, along with other improvements and incidental items; And,

WHEREAS, the Iowa County Highway Department will be the project sponsor and lead for the project, including acting as the liaison between the parties and the federal/state government for design, construction, planning, oversight, management, and payments for various types of work as it is completed. The share of costs attributed to the state / federal governments will be as identified in an approved agreement with the Wisconsin Department of Transportation. The remainder of the cost will be divided proportionally per facilities owned and maintained by the Village of Ridgeway and Iowa County as identified; And,

WHEREAS, the Iowa County Highway Department maintained facilities will be the on-street highway portions equivalent to the width of the County Trunk Highway at the Village limits (being the centerline 24-feet in width) through the Village, and the Village of Ridgeway maintained facilities will be the on-street portions beyond the centerline 24-feet and the off-street portions within the County owned right of way. The Village will provide maintenance for all parking lanes; sidewalk facilities, including any retaining walls, stairs, or hand rails if deemed necessary; street lighting; crosswalk and curb ramp



components; retaining walls, stairs or hand railings if deemed necessary; incidental landscaping; and other miscellaneous or special non-participating cost share items, if so incorporated into the project upon completion of construction; And,

WHEREAS, the Village for be wholly (100%) responsible for all costs associated with any modifications, construction, improvement, replacement, or adjustment of any municipal owned utility systems such as sanitary sewer, water main, or etc.; And,

WHEREAS, the Village will equally cost share in any public utility relocations required or performed as a portion of the project if deemed to be necessary for the project and determined to be compensable to the utility during the real estate acquisition process; And,

WHEREAS, the project will require land acquisitions for highway easement the costs for said acquisitions will be 100% funded by the Iowa County Highway Department. The acquisition process will follow all federal requirements; And,

WHEREAS, the anticipated project cost estimate share is 80% funded by the state / federal grant, with the remaining 20% of cost estimate share divided between the County and Village based upon the identified work / maintenance areas listed above. The project cost estimate for the parties is as attached hereto, but only represents an estimate for the project; the actual costs will be as determined through the various agreements for the work as it progresses; And,

WHEREAS, it is anticipated that design will take place in the years of 2024 thru 2026 with construction commencing between the years 2026-2029; subject to state / federal grant funds; and,

NOW, THEREFORE, BE IT RESOLVED, that the Village Board of Ridgeway does APPROVE of and SUPPORT the project and AUTHORIZES the Iowa County Highway Commissioner to prepare and file an application for participation in the 2024-2029 Surface Transportation Program – Rural on its' behalf; and

BE IT FURTHER RESOLVED, that the Iowa County Highway Commissioner is hereby authorized to sign all necessary documents on behalf of the County and the Village of Ridgeway; and shall submit a copy of this resolution to the Wisconsin Department of Transportation Regional Office for consideration with the STP-Rural Program grant application for the project.

The above and forego	oing Resolution was duly	adopted by the	Village Board of the \	/illage of Ridgeway
at its meeting held on	, 2023, by a vote of _	in favor,	opposed, and	not voting.

APPROVED:

ATTEST:

Βv

Michele B. Casper, Village President

By _____ Hailey E. Roessler, Village Clerk



Iowa County, Wisconsin

To: Board of Trustees RE: Sludge From: Public Works/Village Office

September 7, 2023

In the fall of 2022, the Village of Monticello was performing a Wastewater Treatment Facility Upgrade and was limited for sludge storage. Previous discussion between mutual engineer Delta3, Monticello PW and Jeff Brindley, was that Ridgway had adequate space to take on the additional sludge. General discussion was that the Village of Monticello would pay for the sludge hauling costs and would pay for disposal costs. Ridgeway WWTF accepted 54,900 gallons of sludge along with South Wayne accepting 12,200 and Darlington accepting 23,300.

Jordan Fure, Delta3 estimated costs at \$0.05 per gallon and suggested Ridgeway invoice for 54,900 gallons x \$0.05/gal for a total of \$2,745. Invoice 2023-02 was sent in accordance with Jordan's email on May 10, 2023. The invoice was paid in full by the Village of Monticello in the amount of \$2,745.

The above was determined as staff was completing the 2022 CMAR Report. On August 30, 2023 current operator Dale Peterson requested that the discussion be revisited considering that 54,900 gallons of sludge from Monticello was processed thru Ridgeway WWTF into our GeoTube, sludge dewatering system. One GeoTube is filled at 30,000 gallons. Our cost to dispose of one GeoTube from this spring was \$5168.35.

The Village of Monticello paid the Village of Ridgeway \$2745 per guidance from our mutual Engineer, Delta 3, at the proposed fee of \$.05/gallon, which is what we charge Septic Haulers to dump at our facility.

It was asked to be revisited as our costs to dispose of the Village of Monticello's sludge was much higher than the \$.05/gallon and we are still in possession of approximately 24,900 gallons of Monticello's sludge. While we certainly want to help neighboring facilities, we cannot operate doing so at our own ratepayer's expense. It was preferred by the current WWTF operator that Monticello just take sludge back. Jordan Fure, Delta3, responded to the revisited discussion that Monticello could accept the sludge volume back, but Monticello would not be responsible for paying the disposal fee to Ridgeway and Ridgeway would be responsible for trucking the sludge back to them.

Monticello paid the trucking costs to deliver the 54,900 gallons of sludge to Ridgeway's WWTF.

It would cost Ridgeway around \$5,347 to return 54,900 gallons of sludge back to Monticello. It roughly costs \$10,000 to dispose of the sludge received from Monticello.

We could invoice to cover the remaining costs for the path chosen by the board but we don't have room for our own sludge without the second GeoTubes removal this year and the DPW prefers that Monticello take back the gallons of sludge but doesn't feel that Ridgeway WWTF customers should pay to haul it.

Item 13.

5



.

GEORGE'S SMALL ENGINES LLC 5923 State Hwy. 80 Highland, WI 53543-9223 (608) 929-7412

CUSTOMER'S ORDER NO.	PHONE	DATE O/
		0/22/22
NAME	wal Red	gewalt
ADDRESS		1

Å	ヨス	DAG	ul	17 01	Mit)			
SOLD BY	CASH	C.O.D.	CHARGE	ON ACCT.	MDSE. RET'D.	PAID OUT		
QTY.			DESCRIPT	ION	and the second	PRICE	AMO	UNT
6325	SOD		a				21	99
	2hi	1.2	al	hol	rket		35	35
		- (ill	0a	2.b		6	12
	V	lat	50		1		40	00
							93	49
05	28 L	2000	J AL	AZ	fila	hle		
					ΔI			
(a)	1 pl	ra	ble	3/1	10			
	1	\$6	39	29				
		110	/					
	1	AX						
- NC)		ITH PER A					
1 4	0	N ALL ACC	C'TS PAST	DUE 30 D	AYS	TAX		
RECEIVED BY						TOTAL		
D PRODUCT 610T		All claims	and returned	goods must	be accompanied	by this bill.		
671	35						THANK	YOU



/ Home Improvement / Tools / Power Tools / Power Tool Combo Kits / Milwaukee 2767-22GR

Milwaukee



M18 Fuel High Torque Impact Wrench with Grease Gun Kit

Blain # 1445951 | Mfr # 2767-22GR **Today's Price**





Buy Now. Pay Over Time.



Get 0% intro APR financing² for 12 billing cycles on purchases of \$500 or more at Blain's with a Blain's Farm & Fleet Rewards Mastercard®

Apply Now >

Special Offer Available

\$7.99 Shipping on Select Orders \$49+

PayPal Pay in 4 interest-free payments of \$174.75. Learn more

Get \$7.99 Shipping on select orders \$49+ when you choose Ship It! (Exclusions Ap More Details



0



4.6 ***** Google **Customer Reviews**

https://www.farmandfleet.com/products/1445951-milwaukee-m18-fuel-high-torque-impact-wrench-with-grease-gun-kit.html



Renue sealcoating statestry in Asphrait RenuSealCoating.com

P.O. Box 44158 Madison, WI. 53744

Phone: 608-576-7897 Email: shelby.renusealcoating@gmail.com

Date	Estimate #	
8/22/2023	1880	

		Ship To		
Name / Address		206 Kirby St.		
Village of Ridgeway 206 Kirby St. Maintenance Shop Ridgeway, WI				
Customer Phone	Customer E-mail	Customer Contact	P.O. No.	Project
608-206-7365	streetsandparks@ridgeway			206 Kirby St.

Item	Description		Total
Infrared Patching Seal 2 coat	Infrared 17 heats to problem areas. Clean and seal blacktop. Will apply two (2) coats of Eco-fi manufactured, custom rubberized asphalt blend pavement deepest long lasting color and protection. 8,300 sf	riendly, locally sealer producing the	2,550.00 1,500.00
Thank You for Choosin Greg at 608-334-5007	ng Renu Sealcoating. Any questions or concerns please call	Total	\$4,050.00

GUARANTEE: Sealing and Striping: One year guarantee against peeling and fading. All material is guaranteed to be specified and done in a workmanlike manner according to standard practices. Gasoline and oil spillage will soften and break down asphalt sealer and striping. These types of damages will be at the customer's expense. 24-hour Rain Guarantee: Return visits for sealer wash off from rain at no charge. Asphalt and Patch Work: RENU Sealcoating is not responsible for damage to private underground utilities or hidden conditions if the customer fails to give an advance notice of their existence and location. Gasoline and oil spillage will soften and break down asphalt. These types of damages will be at the customer directs work with a minimum 1% grade, it is understood that water ponding may occur and is not guaranteed. Expansion cracks and edge cracking where landscaping /backfilling should have been done is not guaranteed.

CARE AND MAINTENANCE: Sealing and Striping: Avoid driving on newly sealed/striped asphalt for 12-24 hours. Asphalt and Patch Work: Avoid driving on new asphalt for 1-2 days, if temperatures are high, 3-6 days; no parking of vehicles over one half ton. New asphalt surface porosity is visible at first, but tire traffic will knead and seal these pores. Over time, weeds may grow through expansion cracks but are easily killed with any commercial weed killer. No guarantee if care and maintenance is not taken.

Terms: Payment is due upon job completion and receipt of invoice. No monthly statements will be mailed. 1.5% monthly late fee will be applied when payment is recieved after due date [18% annually]. If a collection process takes, customer shall be responsible for all costs including any attorney' fees. All guarantees are void if payment is not made.

The estimate is good for 30 days from this date.

Accepted by: ____

Date:
Wastewater Training Solutions S7304 Gardner Road Viroqua, WI 54665

- Scheduled classes from September to November (reverse side)
- Prepare operators for DNR Certification Exams
- Earn 6 DNR continuing education credits per day

Wastewater Training Solutions www.wastewatertrainingsolutions.com Dan Tomaro (608) 770-5144 <u>dtomaro@yahoo.com</u>



Item 17 PERMIT NO.22

All classes are offered virtually, using Zoom meetings.

You can download the free version of Zoom at: www.Zoom.us

S270 C2 ***** ALL FOR AADC 535

RIDGEWAY WASTEWATER TREATMENT FACILITY 113 DOUGHERTY CT RIDGEWAY WI 53582-9778

իվներկենովներիկներունվերվերությիներ

Wastewater Class Registration Form (also available on-line, www.wastewatertrainingsolutions.com)

Item 17.

Send registration and check to: Wastewater Training Solutions S7304 Gardner Road Viroqua, WI 54665	Lab Biosolids/Sludge Biological Treatment - Attached Growth	September 6 & 7, 2023 September 12 & 13, 2023 Sept. 19 & 20, 2023	Note: Starting in 2024, I will	
Name (s)	Basic General Wastewater	October 16 – 20, 2023	only be	
Address	Advanced Wastewater	October 31, Nov. 1 & 2, 2023	teaching the Basic General	
	Collection System	November 8 & 9, 2023		
	Phosphorus Removal	November 14 & 15, 2023	Wastewater class.	
Phone	Septage Haulers Update Septage Haulers Update	January 16, 2024 January 17, 2024		
	Septage Haulers Update	January 18, 2024	If you have never	
e-mail	Septage Haulers Update	January 23, 2024	used Zoom, we can	
	Septage Haulers Update	January 24, 2024	do a practice	
Class Name	Septage Haulers Update	January 25, 2024	meeting before	
Class date(s)	Basic General Wastewater	February 5 – 9, 2024	class. The app is free and easy to	
	Basic General Wastewater	April 8 – 12, 2024	download.	
Tuition: 1 day class = \$25.2 day class =	Basic General Wastewater	August 19 – 23, 2024	download.	
Tuition: 1-day class = \$85, 2-day class = \$165, 3-day class = \$250 5-day Basic General Wastewater class = \$370	Basic General Wastewater	October 28 – November 1, 2024		



🔝 💿 🧖 🐚 🖷 🖉

7:19 AN 08/15/20

5%



August 17, 2023

Attn: Maggie Johnson

RE: Connect Communities Agreement #AL0523 between the Wisconsin Economic Development Corporation ("WEDC") and Village of Ridgeway ("Local Organization").

Dear Maggie Johnson:

Congratulations! WEDC has determined that the Village of Ridgeway, your Local Organization, is eligible to participate in the Connect Communities Program from July 1, 2023, to June 30, 2024. By signing and returning this letter agreement to the WEDC and paying the participation fee outlined below, the Local Organization agrees to the following terms.

As a participant in the Connect Communities Program, the Local Organization shall:

- Pay WEDC the annual participation fee of Two Hundred Dollars (\$200);
- Actively participate in the Connect Communities Program, by attending a minimum of Two (2) training opportunities offered by WEDC during the year; and
- Submit an annual performance report as required by WEDC, in such form as required by WEDC.

WEDC shall provide the following services to the Local Organization:

- Designate staff to liaise with the Local Organization;
- Provide an on-line communication tool for participants in the Connect Communities Program; and
- Plan and implement workshops and training sessions on downtown revitalization topics based on the needs of the Connect Communities and Wisconsin Main Street program participants.

This letter agreement will take effect when WEDC receives both the Two Hundred Dollar (\$200) participation fee and this letter agreement, signed, by the Local Organization.

Sincerely,

Melissa L. Hughes, Secretary and CEO

ACKNOWLEDGED & AGREED TO BY:

By:

Michele Casper, Village President Date

By signing this agreement, the signer attests that he/she is fully authorized to execute and deliver this letter agreement on behalf of the Local Organization.



WISCNNSIN

Wisconsin Economic Development Corporation

201 W. Washington Avenue Madison, WI 53701-1687 finance@wedc.org INVOICE #: **INV-007154** DATE: 08/21/2023 DUE DATE: 09/21/2023

TOTAL AMOUNT: \$200.00 TOTAL DUE: \$200.00

BILL Ridgeway, Village of TO: 208 Jarvis St Ste A Ridgeway,, WI 53582

DESCRIPTION / MEMO	AMOUNT
Connect Communities FY24 Participation Fee	\$200.00
TOTAL AMOUNT:	\$200.00

PAYMENT DETAILS:

Customer ID - Name: C-00001202 - Ridgeway, Village of Invoice #: INV-007154

CHECK REMIT TO: Wisconsin Economic Development Corporation P.O. Box 78229 Milwaukee, WI 53278-0229

ACH INFORMATION:

Routing Number: 075000022 Account Number: 182380519419

TO PAY ONLINE, GO TO

https://app02.us.bill.com/p/wisconsineconomicdevelopmentco rporation



Connect Communities Gold Tier

1 message

Errin Welty <errin.welty@wedc.org>

Wed, Sep 6, 2023 at 3:42 PM

Item 20.

Afternoon!

You may remember hearing that we plan to incorporate a Gold Tier recognition program for our Connect Communities this year. We are using the Gold Tier status as a way of recognizing Connect Communities that are really active and working to take their communities and programs to the next level and provide some additional resources to help them on that journey. You're receiving this email because your community was one of those that would have received this status for FY23 if it had been in place – so essentially if you keep it up, you'll be recognized this year!

We will recognize communities at the end of the fiscal year (next June), that have met the following criteria:

- Representatives at five or more events (webinars included), and AT LEAST one attendee is different than the community point person
- Completed annual reporting, returned contract, pay \$200 fee
- Active participants in network posts on Network Wisconsin group, engage with Wisconsin Main Street facebook

Specifically, you'd be eligible for the following additional services if you continue to meet these criteria:

- Priority for hosting events in your region
- Eligible to host a consultant for a regional visit (we regularly send consultants to regions for merchandising, social media, restaurant consulting, branding, etc.)
- Main Street staff assistance in hosting a community survey, facilitate strategic planning session or community visioning
- 1 design request per year for a rendering of a key building in need of transition
- 1 event impact study report (completed by Main Street staff with data collected and provided by local program)

Errin Welty, CEcD, EDFP

Senior Downtown Development Director

Wisconsin Economic Development Corporation

errin.welty@wedc.org



+1.608.210.6832



August 24, 2023

Hailey Roessler, Clerk/Treasurer Village of Ridgeway, Wisconsin 208 Jarvis Street Suite A Ridgeway, WI 53582

Re: Written Municipal Advisor Client Disclosure with the Village of Ridgeway ("Client") for 2024 Budget Assistance ("Project" Pursuant to MSRB Rule G-42)

Dear Hailey:

As a registered Municipal Advisor, we are required by Municipal Securities Rulemaking Board (MSRB) Rules to provide you with certain written information and disclosures prior to, upon or promptly, after the establishment of a municipal advisory relationship as defined in Securities and Exchange Act Rule 15Ba1-1. To establish our engagement as your Municipal Advisor, we must inform you that:

- 1. When providing advice, we are required to act in a fiduciary capacity, which includes a duty of loyalty and a duty of care. This means we are required to act solely in your best interest.
- 2. We have an obligation to fully and fairly disclose to you in writing all material actual or potential conflicts of interest that might impair our ability to render unbiased and competent advice to you. We are providing these and other required disclosures in **Appendix A** attached hereto.

As your Municipal Advisor, Ehlers shall provide this advice and service at such fees, as described within **Appendix B** attached hereto.

This documentation and all appendices hereto shall be effective as of its date unless otherwise terminated by either party upon 30 days written notice to the other party.

During the term of our municipal advisory relationship, this writing might be amended or supplemented to reflect any material change or additions.

We look forward to working with you on this Project.

Sincerely,

Ehlers & Associates

David Ferris,CPA Senior Municipal Advisor

¹ This document is intended to satisfy the requirements of MSRB Rule G-42(b) and Rule G-42(c).

Appendix A

DISCLOSURE OF CONFLICTS OF INTEREST/OTHER REQUIRED INFORMATION

Actual/Potential Material Conflicts of Interest

Ehlers has no known actual or potential material conflicts of interest that might impair its ability either to render unbiased and competent advice or to fulfill its fiduciary duty to Client.

Other Engagements or Relationships Impairing Ability to Provide Advice

Ehlers is not aware of any other engagement or relationship Ehlers has that might impair Ehlers' ability to either render unbiased and competent advice to or to fulfill its fiduciary duty to Client.

Affiliated Entities

Ehlers offers related services through two affiliates of Ehlers, Bond Trust Service Corporation (BTSC) and Ehlers Investment Partners (EIP). BTSC provides paying agent services while Ehlers Investment Partners (EIP) provides investment related services and bidding agent service. Ehlers and these affiliates do not share fees. If either service is needed in conjunction with an Ehlers municipal advisory engagement, Client will be asked whether or not they wish to retain either affiliate to provide service. If BTSC or EIP are retained to provide service, a separate agreement with that affiliate will be provided for Client's consideration and approval.

Solicitors/Payments Made to Obtain/Retain Client Business

Ehlers does not use solicitors to secure municipal engagements; nor does it make direct or indirect payments to obtain or retain Client business.

Payments from Third Parties

Ehlers does not receive any direct or indirect payments from third parties to enlist Ehlers recommendation to the Client of its services, any municipal securities transaction or any financial product.

Payments/Fee-splitting Arrangements

Ehlers does not share fees with any other parties and any provider of investments or services to the Client. However, within a joint proposal with other professional service providers, Ehlers could be the contracting party or be a subcontractor to the contracting party resulting in a fee splitting arrangement. In such cases, the fee due Ehlers will be identified in a Municipal Advisor writing and no other fees will be paid to Ehlers from any of the other participating professionals in the joint proposal.

Municipal Advisor Registration

Ehlers is registered with the Securities and Exchange Commission (SEC) and Municipal Securities Rulemaking Board (MSRB).

Material Legal or Disciplinary Events

Neither Ehlers nor any of its officers or municipal advisors have been involved in any legal or disciplinary events reported on Form MA or MA-I nor are there any other material legal or disciplinary events to be reported. Ehlers' application for permanent registration as a Municipal Advisor with the (SEC) was granted on July 28, 2014 and contained the information prescribed under Section 15B(a)(2) of the Securities and Exchange Act of 1934 and rules thereunder. It did not list any information on legal or disciplinary disclosures.

Client may access Ehlers' most recent Form MA and each most recent Form MA-I by searching the Securities and Exchange Commission's EDGAR system (currently available at http://www.sec.gov/edgar/searchedgar/companysearch.html) and searching under either our Company Name (Ehlers & Associates, Inc.) or by using the currently available "Fast Search" function and entering our CIK number (0001604197).

Ehlers has not made any material changes to Form MA or Form MA-I since that date.

Conflicts Arising from Compensation Contingent on the Size or Closing of Any Transaction

The forms of compensation for municipal advisors vary according to the nature of the engagement and requirements of the client. Compensation contingent on the size of the transaction presents a conflict of interest because the advisor may have an incentive to advise the client to increase the size of the securities issue for the purpose of increasing the advisor's compensation. Compensation contingent on the closing of the transaction presents a conflict because the advisor may have an incentive to recommend unnecessary financings or recommend financings that are disadvantageous to the client. If the transaction is to be delayed or fail to close, an advisor may have an incentive to discourage a full consideration of such facts and circumstances, or to discourage consideration of alternatives that may result in the cancellation of the financing or other transaction.

Any form of compensation due a Municipal Advisor will likely present specific conflict of interests with the Client. If a Client is concerned about the conflict arising from Municipal Advisor compensation contingent on size and/or closing of their transaction, Ehlers is willing to discuss and provide another form of Municipal Advisor compensation. The Client must notify Ehlers in writing of this request within 10 days of receipt of this Municipal Advisor writing.

MSRB Contact Information

The website address of the MSRB is www.msrb.org. Posted on the MSRB website is a municipal advisory client brochure that describes the protections that may be provided by MSRB rules and how to file a complaint with the financial regulatory authorities.

Appendix B

Scope of Service

Client has requested that Ehlers assist with the update of the 2024 budget model Excel spreadsheet and provide budget assistance as follows:

- 1. Update and review budget model so that historical audit information is reconciled and model is working properly.
- 2. Review staff preliminary budget estimates.
- 3. Attend meetings, if necessary.

Scope of Service Limitations

Notwithstanding the Scope of Services listed above, Ehlers' engagement related to Project is expressly limited as follows:

- 1. The Village will provide documentation to update budget worksheets and provide certain budget calculations.
- 2. Work will be completed remotely. The Village may provide remote access to accounting records.
- 3. Work will be billed at the hourly rate identified below. Work will be approved by the Village prior to completing the work.

Compensation

In return for the services set forth in the "Scope of Service," Client agrees to compensate Ehlers on an hourly basis as follows:

Budget model update and necessary budget assistance: \$200/hr

Hourly Charges

For any service directed by Client and not covered by this, or another applicable Appendix, Ehlers & Associates will bill Client at an hourly rate that is dependent upon the task/staff required to meet Client request at no less than \$125.00/hour and not to exceed \$350.00/hour.

Payment for Services

For all compensation due to Ehlers & Associates, we will invoice Client for the amount due at the completion of the work. Our fees include our normal travel, printing, computer services, and mail/delivery charges. The invoice is due and payable upon receipt by the Client.

Signed	Si	gı	٦e	d
--------	----	----	----	---

Title

Date



August 24, 2023

Hailey Roessler, Clerk/Treasurer Village of Ridgeway, Wisconsin 208 Jarvis Street Suite A Ridgeway, WI 53582

Re: Written Municipal Advisor Client Disclosure with the Village of Ridgeway ("Client") for 2024 Financial Management Plan Update ("Project" Pursuant to MSRB Rule G-42)

Dear Hailey:

As a registered Municipal Advisor, we are required by Municipal Securities Rulemaking Board (MSRB) Rules to provide you with certain written information and disclosures prior to, upon or promptly, after the establishment of a municipal advisory relationship as defined in Securities and Exchange Act Rule 15Ba1-1. To establish our engagement as your Municipal Advisor, we must inform you that:

- 1. When providing advice, we are required to act in a fiduciary capacity, which includes a duty of loyalty and a duty of care. This means we are required to act solely in your best interest.
- 2. We have an obligation to fully and fairly disclose to you in writing all material actual or potential conflicts of interest that might impair our ability to render unbiased and competent advice to you. We are providing these and other required disclosures in **Appendix A** attached hereto.

As your Municipal Advisor, Ehlers shall provide this advice and service at such fees, as described within **Appendix B** attached hereto.

This documentation and all appendices hereto shall be effective as of its date unless otherwise terminated by either party upon 30 days written notice to the other party.

During the term of our municipal advisory relationship, this writing might be amended or supplemented to reflect any material change or additions.

We look forward to working with you on this Project.

Sincerely,

Ehlers & Associates

David Ferris,CPA Senior Municipal Advisor

¹ This document is intended to satisfy the requirements of MSRB Rule G-42(b) and Rule G-42(c).

Appendix A

DISCLOSURE OF CONFLICTS OF INTEREST/OTHER REQUIRED INFORMATION

Actual/Potential Material Conflicts of Interest

Ehlers has no known actual or potential material conflicts of interest that might impair its ability either to render unbiased and competent advice or to fulfill its fiduciary duty to Client.

Other Engagements or Relationships Impairing Ability to Provide Advice

Ehlers is not aware of any other engagement or relationship Ehlers has that might impair Ehlers' ability to either render unbiased and competent advice to or to fulfill its fiduciary duty to Client.

Affiliated Entities

Ehlers offers related services through two affiliates of Ehlers, Bond Trust Service Corporation (BTSC) and Ehlers Investment Partners (EIP). BTSC provides paying agent services while Ehlers Investment Partners (EIP) provides investment related services and bidding agent service. Ehlers and these affiliates do not share fees. If either service is needed in conjunction with an Ehlers municipal advisory engagement, Client will be asked whether or not they wish to retain either affiliate to provide service. If BTSC or EIP are retained to provide service, a separate agreement with that affiliate will be provided for Client's consideration and approval.

Solicitors/Payments Made to Obtain/Retain Client Business

Ehlers does not use solicitors to secure municipal engagements; nor does it make direct or indirect payments to obtain or retain Client business.

Payments from Third Parties

Ehlers does not receive any direct or indirect payments from third parties to enlist Ehlers recommendation to the Client of its services, any municipal securities transaction or any financial product.

Payments/Fee-splitting Arrangements

Ehlers does not share fees with any other parties and any provider of investments or services to the Client. However, within a joint proposal with other professional service providers, Ehlers could be the contracting party or be a subcontractor to the contracting party resulting in a fee splitting arrangement. In such cases, the fee due Ehlers will be identified in a Municipal Advisor writing and no other fees will be paid to Ehlers from any of the other participating professionals in the joint proposal.

Municipal Advisor Registration

Ehlers is registered with the Securities and Exchange Commission (SEC) and Municipal Securities Rulemaking Board (MSRB).

Material Legal or Disciplinary Events

Neither Ehlers nor any of its officers or municipal advisors have been involved in any legal or disciplinary events reported on Form MA or MA-I nor are there any other material legal or disciplinary events to be reported. Ehlers' application for permanent registration as a Municipal Advisor with the (SEC) was granted on July 28, 2014 and contained the information prescribed under Section 15B(a)(2) of the Securities and Exchange Act of 1934 and rules thereunder. It did not list any information on legal or disciplinary disclosures.

Client may access Ehlers' most recent Form MA and each most recent Form MA-I by searching the Securities and Exchange Commission's EDGAR system (currently available at http://www.sec.gov/edgar/searchedgar/companysearch.html) and searching under either our Company Name (Ehlers & Associates, Inc.) or by using the currently available "Fast Search" function and entering our CIK number (0001604197).

Ehlers has not made any material changes to Form MA or Form MA-I since that date.

Conflicts Arising from Compensation Contingent on the Size or Closing of Any Transaction

The forms of compensation for municipal advisors vary according to the nature of the engagement and requirements of the client. Compensation contingent on the size of the transaction presents a conflict of interest because the advisor may have an incentive to advise the client to increase the size of the securities issue for the purpose of increasing the advisor's compensation. Compensation contingent on the closing of the transaction presents a conflict because the advisor may have an incentive to recommend unnecessary financings or recommend financings that are disadvantageous to the client. If the transaction is to be delayed or fail to close, an advisor may have an incentive to discourage a full consideration of such facts and circumstances, or to discourage consideration of alternatives that may result in the cancellation of the financing or other transaction.

Any form of compensation due a Municipal Advisor will likely present specific conflict of interests with the Client. If a Client is concerned about the conflict arising from Municipal Advisor compensation contingent on size and/or closing of their transaction, Ehlers is willing to discuss and provide another form of Municipal Advisor compensation. The Client must notify Ehlers in writing of this request within 10 days of receipt of this Municipal Advisor writing.

MSRB Contact Information

The website address of the MSRB is www.msrb.org. Posted on the MSRB website is a municipal advisory client brochure that describes the protections that may be provided by MSRB rules and how to file a complaint with the financial regulatory authorities.

Appendix B

Scope of Service

Client has requested that Ehlers & Associates update the Financial Management Plan ("Project"). Ehlers & Associates proposes and agrees to provide the following scope of services:

Confirm Planning Objectives

• Prior to commencing the work, we will review our approach with Client staff to ensure that we have a full understanding of the Client's objectives, any areas of concern or focus, and desired process outcomes. If necessary, we can modify our Scope of Services to meet specific Client needs.

Gather Required Information

• To complete our work, we will need to gather certain information which may include prior year audits and budgets (generally five years); current year budget; capital improvement plans; existing debt service schedules and allocations; strategic planning documents; staffing plans; and policies pertaining to fund balance, debt management; post-issuance compliance and financial management (Ehlers & Associates may already have some or all this information on file).

Prepare Financial Model

- Based on the Client's objectives and the information available, we will prepare a Client-specific Excel based financial planning model that includes:
 - Valuation Forecast. We will project growth in equalized value based on historical valuation trends, and anticipated potential for and timing of new development based on Client input. If applicable, "TID IN" and "TID OUT" forecasts will be provided. The impact of TID closure will be considered based on Client direction. One or more potential growth scenarios may be modeled based on Client's objectives.
 - Fund Forecasts. We will forecast revenues and expenditures for the following funds based on prior year budgetary trends. Based on the Client's objectives and the information available to us, one or more alternate fund forecasts may be developed to reflect adjustments to service levels and staffing.
 - Operating Funds (General Fund & Other Tax Levy Funded Funds)
 - Debt Service Fund
 - Capital Improvement Fund
 - Tax Incremental District Funds

- Capital Planning Model. Using Client's capital improvement plans, we will prepare one or more models identifying funding sources for identified projects. Fund balances, tax levy, debt proceeds, and annual revenues will be evaluated as funding sources.
- Debt Model. We will prepare a current debt service schedule including projected debt abatement sources and tax levy requirements. To the extent that debt financing is required for capital improvement projects, the projected impact of that financing will be modeled. The model will also forecast debt limit capacity utilization and the projected impact of future debt obligations on selected debt profile indicators (for General Obligation debt).
- Consolidated Tax Levy and Rate Projection Model. A summary forecast will be provided projecting the future tax levy that would be required to support the General Fund, Capital Projects Fund, Debt Service Fund, and other levy supported funds. Based on the valuation projection model, a forecasted equalized tax rate will be provided. Future levy requirements will be tested against applicable levy and rate limits. A similar summary forecast will be provided for any enterprise funds included in the model and will include a projection of any additional revenue requirements needed to support the forecast.

Conduct Governing Body Workshops

 Workshops, if needed, can be conducted concurrent with the update of the financial model. The purpose of the workshops will be to present key data, observations, findings, alternatives, and recommendations, and to seek input of Governing Body members and key staff at periodic intervals before the model is finalized. Specific workshop dates, if needed, will be scheduled in consultation with the Client based on the objectives of Client, the availability of the Governing Body and key staff, and the availability of information needed to complete the plan. Workshops are generally more effective when held independent of other Governing Body meetings.

Final Report

• We will prepare a PowerPoint presentation that includes and explains all primary elements of the forecast model and present that to the Village Board at a Village Board meeting. The report will include a summary of key observations and recommendations. If applicable, we will recommend modifications to existing Client policies pertaining to fund balance, debt management, post-issuance compliance and financial management.

Compensation

In return for the services set forth in the "Scope of Service," Client agrees to compensate Ehlers & Associates at an hourly rate of \$250 up to \$5,000.

For any service directed by Client and not covered by this, or another applicable Appendix, Ehlers & Associates will bill Client at an hourly rate that is dependent upon the task/staff required to meet Client request at no less than \$125.00/hour and not to exceed \$350.00/hour.

Payment for Services

For all compensation due to Ehlers & Associates, we will invoice Client for the amount due at the completion of the work. Our fees include our normal travel, printing, computer services, and mail/delivery charges. The invoice is due and payable upon receipt by the Client.

The above Proposal is hereby accepted by the Village of Ridgeway, Wisconsin, by its authorized officer:

Signed

Title

Date